The Potential of Egg White Ovotransferrin as a Functional Food Ingredient against Osteoporosis

by

Nan Shang

A thesis submitted in partial fulfillment of the requirements for the degree of

Doctor of Philosophy

in

Food Science and Technology

Department of Agricultural, Food and Nutritional Science

University of Alberta

© Nan Shang, 2019

ABSTRACT

Ovotransferrin, an iron-binding glycoprotein, accounting for $\sim 12\%$ of egg white protein, is a member of the transferrin family. As a component of innate immunity, ovotransferrin has antimicrobial, anti-viral, anti-oxidant, and immunomodulatory activities. The overall objectives of this thesis were to understand the role of ovotransferrin on bone cells (osteoblasts and osteoclasts) and to explore its potential application as a bioactive protein against osteoporosis.

Using mouse preosteoblast MC3T3-E1 cells, ovotransferrin significantly promoted cell proliferation via regulation of the cell cycle at concentrations of 100 and 1000 µg/mL. Ovotransferrin also significantly stimulated osteoblast differentiation evidenced by increased expression of type I collagen and alkaline phosphates (ALP), as well as mineralization (increased bone matrix deposits) in a dose-dependent manner. Furthermore, ovotransferrin inhibited the production of receptor activator of nuclear factor kappa-B (RANKL), while increased osteoprotegerin (OPG), indicating the potential role of ovotransferrin on preventing osteoclastogenesis. The effect of ovotransferrin on transcription factors involved various pathways were studied. The extracellular signal-regulated kinase 1/2 (ERK1/2) antagonist, but not the c-Jun N-terminal kinase (JNK) antagonist, significantly attenuated ovotransferrin-induced increase of alkaline phosphatase (ALP), runt-related transcription factor 2 (Runx2), \beta-catenin, bone morphogenetic protein 2 (BMP-2), type I collagen synthesis and mineralization; p38 antagonist only decreased ovotransferrin-induced increase of ALP and Runx2. This research further showed that ovotransferrin activated the ERK1/2 cascade, including the phosphorylation of upstream proto-oncogene serine/threonine-protein kinase c-Raf and mitogen-activated protein kinase kinase

MEK1/2, as well as the downstream ribosomal s6 kinase p90RSK and mitogen- and stressactivated protein kinase 1 (MSK1). In addition, phosphoinositide-3-kinase (PI3K)-protein kinase C (Akt) pathway was also involved in ovotransferrin induced osteogenesis as adding ovotransferrin increased the expression of PI3K subunits p85 and p110, which further stimulated the phosphorylation of downstream kinase Akt. Low-density lipoprotein receptor-related protein 1 (LRP1) partially regulated ovotransferrin-stimulated ALP expression and mineralization, while independent to ERK1/2 and PI3K-Akt pathways.

Ovotransferrin was also investigated for its regulatory role on osteoclastogenesis. Macrophage RAW 264.7 was used to differentiate into osteoclasts with RANKL stimulation. Ovotransferrin significantly inhibited osteoclastogenesis, due to the attenuation of RANKL-induced activation of nuclear factor kappa B (NF- κ B) and mitogen-activated protein kinase (MAPK) pathways. Also, the expression of osteoclastogenesis-associated proteins TNF receptor associated factor 6 (TRAF6), c-Fos, nuclear factor of activated T-cells cytoplasmic 1 (NFATc1), and cathepsin K (CathK), were significantly decreased in the presence of ovotransferrin. Ovotransferrin significantly preserved the calcium-phosphate (CaP) coating against osteoclastic resorption and induced cell apoptosis of mature osteoclast by regulating Bcl-2 family.

The potential role of ovotransferrin in slowing the progression of osteoporosis was studied using ovariectomized (OVX) Sprague-Dawley rats. Osteoporosis was developed in OVX rats, mimicking to postmenopausal osteoporosis. Oral administration of ovotransferrin did not affect body weight, food intake and organ weight. After 12 weeks of treatment, feeding ovotransferrin at 1% (w/w) in the diet prevented OVX-induced bone loss and maintained relatively high bone

mineral density and integrated bone microarchitecture. Bone marrow cells extracted from OVX rat fed with ovotransferrin produced less osteoclast when stimulated with RANKL *ex vivo*. Ovotransferrin feeding also decreased the production of serum cytokine tumor necrosis factor α (TNF- α) and interleukin 6 (IL-6), two stimulators for osteoclast differentiation. In addition to its direct regulatory role on bone turnover, ovotransferrin supplementation might benefit osteoporosis prevention by inhibiting adipogenesis, regulating immune response and stimulating short-chain fatty acid production.

In summary, this research demonstrated the potential of ovotransferrin as a functional food ingredient for bone health due to its role in promoting osteoblastic activity and inhibiting osteoclastic activity.

PREFACE

This thesis contains original work done by Nan Shang and has been written according to the guidelines for a paper format thesis of the Faculty of Graduate Studies and Research at the University of Alberta. The concept of this thesis originated from my supervisor Dr. Jianping Wu and the research was funded by a Discovery Grant from the Natural Science and Engineering Research Council of Canada (NSERC). The experimental protocol for the animal study was approved by the Animal Care and Use Committee at the University of Alberta (Protocol AUP 00001960) in accordance with the guidelines issued by the Canada Council on Animal Care.

The thesis consists of seven chapters: Chapter 1 provides a general introduction on the context and the objectives of the thesis; Chapter 2 is a literature review on several topics relevant to this thesis, including egg white ovotransferrin, bone physiology and metabolism, osteoporosis and current therapies, and the potential use of bioactive proteins in osteoporosis prevention; Chapter 3 contains the work on *in vitro* osteogenic activity of egg white ovotransferrin, which has been published as Nan Shang and Jianping Wu, "Egg White Ovotransferrin Shows Osteogenic Activity in Osteoblast Cells" in *Journal of Agricultural and Food Chemistry (2018, vol. 86, issue 11, 2775-2782);* Chapter 4 explores the underlying mechanisms of ovotransferrin-stimulated osteogenic activity; Chapter 5 investigates the effects of ovotransferrin on inhibition of osteoclastogenesis and osteoclastic bone resorption; Chapter 6 further validates the *in vivo* effects of ovotransferrin in preventing ovariectomy-induced osteoporosis; and Chapter 7 gives concluding remarks and discusses future research directions.

Dr. Jianping Wu greatly contributed to the experimental design, data interpretation, thesis preparation and edits. Dr. Lynn McMullen and Dr. Jonathan Curtis helped with the thesis review and evaluation. I was responsible for literature search relevant for the above studies, designing and performing laboratory experiments, data collection and analysis, and drafting the manuscript and thesis. Dr. Michael Doschak and Mrs. Ashely Quinton have provided substantial guidance and assistance for the micro-CT scanning, image construction and data analysis. Mrs. Nicole Coursen has provided technical assistance in animal studies including animal husbandry and tissue sampling upon termination. Dr. Aja Rieger and Mrs. Sabina Baghirova provided guidance for flow cytometry and data analysis.

DEDICATION

Dedicated to my beloved parents,

Qingmao Shang and Pinglan Li

ACKNOWLEDGEMENTS

How time flies! It has been four and half years since I started my Ph.D. program. My experience at the University of Alberta has been nothing short of amazing, due to a lot of support, help, encouragement and inspiration I received from a number of wonderful individuals. Here, I would like to show my deepest appreciation to all of them for being part of this journal and making this thesis possible.

First and foremost, I would like to express my sincere gratitude to my supervisor, Dr. Jianping Wu, for offering me this opportunity to pursue graduate study in his research group, and for his guidance, devoted time and efforts, encouragement, inspiration and patience with me throughout the years. I could not have imagined having a better supervisor and mentor for my Ph.D. study. I also want to thank my supervisory committee members, Dr. Lynn McMullen and Dr. Jonathan Curtis, for their insightful comments and encouragement, especially for their thought-provoking questions during the committee meetings. My special thanks go to Dr. Michael Doschak for his kindness to serve as the examiner for both my candidacy examination and thesis defense, and for allowing me to use his laboratory equipment; I learnt immensely from him about bone biology. I like to express my sincere appreciation to Dr. Jillian Cornish for accepting to be my external examiner and for devoting her valuable time to evaluate my thesis.

I like to specially thank Mrs. Nicole Coursen, for teaching me animal husbandry, sample collection and tissue sampling. I would not be able to get the animal work done without her help. I am also grateful for Mrs. Ashley Quinton for teaching me Micro-CT scanning. I am very grateful for the support form Mrs. Jody Forslund and Ms. Kelly Berdahl, the Graduate Program Administrators of the Department of Agricultural, Food and Nutritional Sciences (AFNS) and Mrs. Holly Horvath, the Executive Assistant to the Chair and Office Coordinator of AFNS. My special thanks go to Mrs. Robin Miles and Eduardo Rodriguez, AFNS Store Manager and Purchasing Technologist, who are always efficient and helpful with chemical ordering.

I would like to express my immense gratitude to all my fellow graduate students, friends and colleague in Dr. Wu's group, both past and present, for their support and company, and for the great memories we shared: Dr. Xiaohong Sun, Dr. Wenlin Yu, Dr. Nandika Bandara, Dr. Qingbiao Xu, Dr. Hui Hong, Dr. Yussef Esparza, Dr. Myoungjin Son, Dr. Ping Liu, Dr. Xiaofeng Ren, Dr. Jiandong Ren, Dr. Wang Liao, Hongbing Fan, Liang Chen, Selene Toledo, Xu Jiang, Khushwart Bhullar, Harshita Arora, and among others.

The financial support I received for my doctoral program from Chinese Scholarship Council (CSC), Natural Sciences and Engineering Research Council of Canada (NSERC), Alberta Innovates Technology Futures (AITF), the Faculty of Agricultural, Life and Environmental Science (ALES) and the Faculty of Graduate Studies and Research (FGSR) at the University of Alberta is highly appreciated.

Last but not lease, my deepest appreciation goes to my dear parents Dr. Qingmao Shang and Dr. Pinglan Li. I would not be able to come so far without their support. They have cherished with me every great moment and supported me whenever I needed. I am indebted to my dear parents for their endless love, trust, encouragements and understanding throughout my life.

CONTENTS

С	HAPTER 1 – General Introduction and Objectives	. 1
	1.1 General Introduction	. 1
	1.2 Objectives and Hypothesis	. 2
	1.3 Chapter Format	. 3
	1.4 References	. 5

CHAPTER 2 - Literature Review	8
2.1 Egg White Ovotransferrin	8
2.1.1 Structure and composition of ovotransferrin	8
2.1.2 Bioactives of Egg White Ovotransferrin and Its-derived Peptides	9
2.1.3 Current Applications of Ovotransferrin	. 14
2.2 Bone Physiology and Metabolism	. 15
2.2.1 Bone Composition	. 15
2.2.2 Bone remodeling	. 15
2.2.3 Osteoblast and Bone Formation	. 17
2.2.4 Osteocyte and Bone Formation	. 18
2.2.5 Osteoclast and Bone Resorption	. 19
2.2.6 Interaction Between Osteoblasts and Osteoclasts	. 20
2.3 Osteoporosis and Current Therapies	. 22
2.3.1 Osteoporosis	. 22
2.3.2 Current Osteoporosis Therapies	. 22
2.3.3 Emerging Therapies and Investigational Drugs	. 24
2.4 Bioactive Proteins and Prevention of Osteoporosis	. 26
2.4.1 Dairy Protein	. 26
2.4.2 Soy Protein	. 27
2.4.3 Egg Protein	. 28
2.5 Perspectives and Potential Challenges	. 30
2.5.1 Current Osteoporosis Drug Market	. 30

2.5.2 Market Opportunity of Bioactive Proteins	. 31
2.5.3 Potential Challenges of Bioactive Protein Application	. 32
2.6 General conclusions	. 32
2.7 References	. 34

CHAPTER 3 - Egg White Ovotransferrin Shows Osteogenic Activity in Osteoblast Cells. 84

CHAPTER 4 - Ovotransferrin Stimulated Osteogenesis via activation of ERK1/2 and
PI3K-Akt pathways independent to low-density lipoprotein receptor-related protein 1
(LRP1)
4.1 Introduction
4.2 Material and Methods 117
4.2.1 Reagents and antibodies117
4.2.2 Cell Culture
4.2.3 Western blot
4.2.4 Immunofluorescence
4.2.5 Mineralization Assay
4.2.6 Cell Migration Assay 120
4.2.7 Statistics
4.3 Results
4.3.1 Ovotransferrin stimulates osteogenesis associated transcription factors 121
4.3.2 Ovotransferrin stimulates osteogenesis via activate MAPK pathways 121
4.3.3 ERK pathway is involved in ovotransferrin-stimulated osteogenesis, but no JNK and
p38 pathways
4.3.4 Ovotransferrin stimulates the phosphorylation and activation of the ERK cascade 123
4.3.5 Ovotransferrin stimulates osteogenesis via activate PI3K-Akt pathways 124
4.3.6 LRP1 is not involved in mediating the osteoinductive effect of ovotransferrin 125
4.4 Discussion
4.5 Conclusion
4.6 References

CHAPTER 5 - Egg White Ovotransferrin Attenuates RANKL-induced Osteoclastogenesis

and Bone Resorption	164
5.1 Introduction	164
5.2 Materials and methods	167
5.2.1 Reagents and antibodies	167
5.2.2 Cell culture	167
5.2.3 Osteoclasts generation and TRAP staining	167

5.2.4 Cell apoptosis assay	. 168
5.2.5 Resorption assay	. 168
5.2.6 Western blot analysis	. 169
5.2.7 RNA extraction and quantitative PCR	. 169
5.2.8 Statistical analysis	. 170
5.3 Results	. 170
5.3.1 Ovotransferrin inhibits osteoclast formation from RAW 264.7	. 170
5.3.2 Ovotransferrin inhibited RANKL-induced activation of NF- κ B and MAPK pathw	ay
	. 170
5.3.3 Ovotransferrin inhibited RANKL-induced expression of the factors involved in	
osteoclastogenesis	. 171
5.3.4 Ovotransferrin inhibits resorptive activity of matured osteoclasts	. 171
5.3.5 Ovotransferrin stimulates the apoptosis of osteoclasts	. 172
5.4 Discussion	. 172
5.5 Conclusions	. 178
56 Deferences	170

CHAPTER 6 – Oral Administration of Egg White Ovotransferrin Prevents Osteoporosis in		
Ovariectomized Rat	. 195	
6.1 Introduction	. 195	
6.2 Materials and methods	. 198	
6.2.1 Chemicals	. 198	
6.2.2 Experimental protocol	. 198	
6.2.3 Sample collection	. 199	
6.2.4 Micro-computerized tomography analysis and bone mineral density determination	199	
6.2.5 Bone biochemical markers and cytokine measurement	. 200	
6.2.6 Isolation of bone marrow cells	. 200	
6.2.7 Isolation of spleen cells	. 201	
6.2.8 Bone marrow cells and spleen cells phenotype analysis	. 201	
6.2.9 Osteoclast differentiation and resorptive activity analysis	. 202	
6.2.10 RNA extraction and qRT-PCR	. 202	

6.2.11 Western blot
6.2.12 Short chain fatty acid analysis
6.2.13 Statistical analysis
6.3 Results
6.3.1 Ovotransferrin showed no significant effects on body weight, food intake and tissue
weight
6.3.2 Ovotransferrin administration prevented the decrease of bone mineral density in OVX
rats
6.3.3 Ovotransferrin administration preserved bone micro-architecture in OVX rats 205
6.3.4 Ovotransferrin administration regulated bone formation and bone resorption in OVX
rats
6.3.5 Ovotransferrin inhibited the production of serum cytokines in OVX rats
6.3.6 Osteoclast precursors derived from ovotransferrin-administrated OVX rats showed
less differentiation ability and resorption activity
6.3.7 Ovotransferrin inhibited osteoclastogenesis by inhibiting the NF-κB pathway 208
6.3.8 Ovotransferrin administration exerted a global regulatory effect in OVX rats 209
6.4 Discussion
6.5 Conclusion
6.6 Reference

CHAPTER 7 - General Summary and Discussion	
7.1 Key Findings of the Present Research	
7.2 Significance of This Research	
7.3 Future Research Directions	
7.4 References	

BIBILIOGRAPHY	r ••••••••••••••••••••••••••••••••••••	268
---------------	---	-----

LIST OF TABLES

Table 2.1 Physiological and pharmacological activities of ovotransferrin and its peptides

 Table 2.2 Current FDA-approved osteoporosis drugs

Table 2.3 The emerging therapies and investigational drugs for osteoporosis

Table 2.4 Summary of current researches on diary bioactive proteins/peptides and their role in osteoporosis treatment

Table 2.5 Summary of human trials on the impact of soy consumption and bone health

 Table 2.6 Summary of research on egg protein/peptide and their impact on bone health

LIST OF FIGURES

Figure 2.1 Bone remodeling cycle

Figure 2.2 Osteoblast differentiation

Figure 2.3 Key regulation pathways in osteoblast differentiation

Figure 2.4 Osteoclast differentiation

Figure 2.5 Key regulation pathways in osteoclast differentiation

Figure 2.6 Interaction between osteoblast and osteoclast

Figure 3.1 Effects of egg white ovotransferrin on cell proliferation in osteoblast cell MC3T3-E1

Figure 3.2 Effects of egg white ovotransferrin on cell cycle in osteoblast cell MC3T3-E1

Figure 3.3 Effects of egg white ovotransferrin on type I collagen expression in osteoblast cell MC3T3-E1

Figure 3.4 Effects of egg white ovotransferrin on ALP expression in osteoblast cell MC3T3-E1

Figure 3.5 Effects of egg white ovotransferrin on mineralization in osteoblast cell MC3T3-E1

Figure 3.6 Effects of egg white ovotransferrin on RANKL expression in osteoblast cell MC3T3-

E1

Figure 3.7 Effects of egg white ovotransferrin on OPG expression in osteoblast cell MC3T3-E1

Figure 4.1 Effects of ovotransferrin on the expression of master factors involved in osteoblast differentiation

Figure 4.2 Effects of ovotransferrin on activation of MAPK pathways

Figure 4.3 Effects of ERK inhibitor on ovotransferrin-induced osteoblast differentiation

Figure 4.4 Effects of JNK inhibitor on ovotransferrin-induced osteoblast differentiation

Figure 4.5 Effects of p38 inhibitor on ovotransferrin-induced osteoblast differentiation

Figure 4.6 Effects of ovotransferrin on activation of ERK MAPK pathway

Figure 4.7 Effects of ovotransferrin on activation of PI3K-Akt pathways

Figure 4.8 Effects of Akt inhibitor on ovotransferrin-induced osteoblast differentiation

Figure 4.9 The effects of ovotransferrin on the receptor expression

Figure 4.10 Effects of LRP1 inhibitor on ovotransferrin-induced osteoblast differentiation

Figure 4.11 Scheme summary of ovotransferrin activated signaling pathways

Figure 5.1 Effect of ovotransferrin on RANKL-induced osteoclastogenesis

Figure 5.2 Effect of ovotransferrin on RANKL-induced NF-κB activation

Figure 5.3 Effect of ovotransferrin on RANKL-induced MAPK activation

Figure 5.4 Effect of ovotransferrin on RANKL-induced expression of proteins involved in osteoclastogenesis

Figure 5.5 Effect of ovotransferrin on RANKL-induced bone resorption in vitro

Figure 5.6 Effect of ovotransferrin on RANKL-induced osteoclasts apoptosis.

Figure 6.1 The experimental design

Figure 6.2 Effects of ovotransferrin diet on body weight, food intake, and uterus weight

Figure 6.3 Effects of ovotransferrin diet on tibia trabecular bone mineral density and cortical bone mineral density

Figure 6.4 Effects of ovotransferrin diet on tibia trabecular micro-architecture

Figure 6.5 Effects of ovotransferrin diet on serum cytokine production

Figure 6.6 Effects of ovotransferrin diet on bone marrow cells-stimulated osteoclasts formation and resorption

Figure 6.7 Effects of ovotransferrin diet on osteoclastogenesis-related protein expression in bone marrow

Figure 6.8 Effects of ovotransferrin treatment on osteoclasts differentiation

Figure 6.9 Regulatory role of ovotransferrin in OVX rats

LIST OF ABBREVIATIONS

- α-MEM: Minimum essential eagle medium alpha modification
- ACE: Angiotensin converting enzyme
- Akt (PKB): Protein kinase B
- ALP: Alkaline phosphatase
- AP-1: Activator protein 1
- APP: Acute-phase protein
- BCAP: Bovine colostrum acidic protein
- Bcl-xL: B-cell lymphoma-extra large
- Bim: BH3-only family member
- BMD: bone mineral density
- BMP: Bone morphogenetic protein
- BSA: Bovine serum albumin
- CAGR: Compound annual growth rate
- CathK: Cathepsin K
- Cbfa1: Core binding factor alpha 1
- CEF: Chicken embryo fibroblasts
- **CPP:** Casein phosphopeptides
- DC-STAMP: Dendritic cell-specific transmembrane protein
- DMEM: Dulbecco's modified eagle medium
- DMSO: Dimethyl sulfoxide
- DNA: Deoxyribonucleic acid

DTT: Dithiothreitol

EAA: Estrogen agonist/antagonist

c-Raf: Proto-oncogene serine/threonine-protein kinase

EDTA: Ethylenediamineteraaceitc acid

ELISA: Enzyme-linked immunosorbent assay

ERK: Extracellular signal-regulated kinase

FasL: Factor associated suicide ligand

FBS: Fetal bovine serum

FDA: Food and drug administration

FGF: Fibroblast growth factor

GI: Gastrointestinal

GIO: Glucocorticoid-induced osteoporosis

GSK3: Glycogen synthase kinase 3

HSV-1: Herpes simplex virus

IgE: Immunoglobulin E

IGF-2: Insulin-like growth factor 2

IGFBP-2: Insulin-like growth factor binding protein 2

IOM: Institute of medicine

ITLN: Intelectin

JNK: c-Jun N-terminal kinase

LF: Lactoferrin

LGR4: Leucine-rich repeat-containing G-protein coupled receptor 4

LPA: Lysophosphatidic acid

LPS: Lipopolysaccharide

LRP: Low-density lipoprotein receptor-related protein

LRP: Low-density lipoprotein receptor-related protein

M-CSF: Macrophage colony stimulating factor

MAPK: Mitogen-activated protein kinase

MAPKAPK: MAPK-activated protein kinase

MBP: Milk basic protein

Mcl-1: Myeloid cell leukaemia sequence 1

MCP-1: Monocyte chemoattractant protein 1

MCS: Mesenchymal stem cell

MDV: Marek's disease virus

MMP9: Matrix metalloproteinase 9

mTOR: Mammalian target of rapamycin

NF-κB: Nuclear factor kappa B

NFATc1: Nuclear factor of activated T-cells cytoplasmic 1

Nrp: Neuropilin

OPG: Osteoprotegerin

PAGE: Polyacrylamide gel electrophoresis

PBS: Phosphate-buffered saline

PCR: Polymerase chain reaction

PHA: Phytohemagglutinin

PI3K: phosphoinositide-3-kinase

PKC: Protein kinase C

PMO: Postmenopausal osteoporosis

PST: Proline/serine/threonine-rich

PTH: Parathyroid hormone

PTH: Parathyroid hormone

RANKL: Receptor activator of nuclear factor kappa-B ligand

Ras: Small GTP-binding protein

RNA: Ribonucleic acid

RTK: Receptor tyrosine kinase

Runx2: Runt-related transcription factor 2

SD: Standard deviation

SDS: Sodium dodecyl sulfate

SEM: Standard error of mean

Sema3A: Semaphoring 3A

Sema4D: Semaphoring 4D

SERM: Selective estrogen inhibitors modulator

SOD: Superoxide dismutase

TRAF6: TNF receptor associated factor 6

TFBP: Transferrin binding protein

TFR: Transferrin receptor

TGF- β 1: Transforming growth factor beta 1

TRAP: Tartrate-resistant acid phosphatase

Wnt: Wingless

X/XOD: Xanthine/xanthine oxidase

ICAM-1: Intercellular adhesion molecule 1

VCAM-1: Vascular cell adhesion protein 1

CHAPTER 1 – General Introduction and Objectives

1.1 General Introduction

Osteoporosis is one of the most serious bone disorders, defined as a decreased bone mass and structural deterioration (US Department of Health and Human Services, 2004). It is estimated that ~20% of elderly men and ~30% of elderly women are at elevated risk of osteoporotic fracture. With the rapidly increasing aging population, osteoporosis is becoming a serious health concern. Pharmacological therapies are available for osteoporosis treatment, such as the anti-resorption agents (e.g. bisphosphonates and estrogen) and anabolic agents (e.g. teriparatide) (Salari Sharif, Abdollahi, & Larijani, 2011). However, due the limited efficacy and inevitable side effects, those approved drugs are far from satisfactory.

Functional foods and nutraceuticals are a growing trend in the food and nutrition industry for promoting well-being and helping reduce the risk of disease (AAFC, 2015). Bioactive proteins constitute an important category of health-promoting nutrients that convey physiological benefits, such as promoting bone health (Mine, Li-Chan, & Jiang, 2010). Milk proteins, such as milk basic proteins (Aoyagi et al., 2010), bovine colostrum acidic protein (BCAP) (Du et al., 2011), and lactoferrin (Naot, Grey, Reid, & Cornish, 2005), have been studied for their bone health promoting activity. In comparison, there is limited research on the potential of egg proteins, other than egg yolk water-soluble protein and phosvitin (Ji, Leem, Kim, & Kim, 2007; Kim, Kim, & Leem, 2008; Liu, Li, Geng, Huang, & Ma, 2017).

Ovotransferrin, an iron-binding glycoprotein, accounts for ~12% of egg white protein. As a member of transferrin family (Wu & Acero-Lopez, 2012), ovotransferrin shares ~ 50% sequence similarity with lactoferrin, a well studied milk protein that was reported to stimulate bone formation and prevent bone resorption (Cornish et al., 2004; Naot et al., 2005). Ovotransferrin is also considered as a member of chick innate immunity (Giansanti, Leboffe, Angelucci, & Antonini, 2015), and its expression was increased during chondrogenesis and endochondral bone formation in developing chick embryo (Gentili et al., 1994; Gentili et al., 1994), indicating its involvement in chick bone development. Thus, ovotransferrin can be possibly developed as functional foods and nutraceuticals in promoting bone health.

1.2 Objectives and Hypothesis

Given the above background, we hypothesized that egg white ovotransferrin can promote bone health through regulating the activity of bone cells. The overall objectives of this research are to understand the regulatory roles of ovotransferrin on bone cells and to explore its potential application as a bioactive protein against osteoporosis. The specific objectives of the research are: 1) To investigate the *in vitro* regulatory role of ovotransferrin on osteoblastic activity and osteogenesis

2) To study the mechanisms involved in the ovotransferrin-stimulated osteogenesis

3) To investigate the *in vitro* effects of ovotransferrin on inhibiting osteoclastogenesis and bone resorptive activity; and

4) To study the *in vivo* efficacy of ovotransferrin on osteoporosis treatment in an ovariectomized (OVX) rat model

1.3 Chapter Format

There are seven chapters in the thesis and the brief description of each chapter are given as follows: **Chapter 1** gives a brief introduction on the current status of osteoporosis, and the growing interests in developing functional foods and/or nutraceuticals to promote bone health. Following with the general introduction, the thesis objectives and hypothesis are descried and the chapter format is presented.

Chapter 2 provides a literature review on the current knowledge of egg white ovotransferrin, bone physiology and metabolism, osteoporosis and current therapies, as well as the studies on bioactive proteins and osteoporosis prevention/treatment. In the end, the perspectives on developing bioactive proteins on health management as well as the potential challenges are summarized.

Chapter 3 reports the *in vitro* effect of ovotransferrin on stimulating osteoblastic activity. Murine preosteoblast MC3T3-E1 was used to investigate the stimulatory effects of ovotransferrin on cell proliferation, cell differentiation and mineralization. The effects on cell proliferation were characterized by BrdU incorporation assay, and cell cycle analysis. The effects on cell differentiation were tested by the expression of differentiation biomarkers using western blot and immunofluorescence. The effects on mineralization were evaluated by Alizarin Red assay. Objective 1 is addressed in this chapter.

Chapter 4 elucidates the signaling pathways underlying the ovotransferrin-stimulated osteoblastic activity and the possible receptors in response to ovotransferrin. Western blot was used to study the activation of signaling pathways, include MAPKs and PI3K-Akt pathway. Signaling antagonists were used to validate the involvement of the target pathways by blocking the activation. Biomarkers used in Chapter 3 were used to evaluate the osteogenic activity. qPCR was used to screen the receptors of ovotransferrin. Objective 2 is addressed in this chapter.

Chapter 5 reports the *in vitro* effects of ovotransferrin on inhibiting osteoclastogenesis and resorptive activity. Murine macrophage RAW 264.7 was used to generate osteoclast. The production and formation of osteoclast were measured by TRAP staining. The resorption of osteoclast was tested with calcium-phosphatase (CaP) resorption assay. Western blot, cell apoptosis assay and qPCR were used to elucidate the mechanism underlying the inhibitory effect of ovotransferrin. Objective 3 is addressed in this chapter.

Chapter 6 describes the *in vivo* effects of ovotransferrin on osteoporosis prevention. Ovariectomized Sprague-Dawley rats were used as the disease model representing the postmenopausal osteoporosis. The body weight, food intake and organ characteristics were recorded for a preliminary safety evaluation. The impact of dietary supplement of ovotransferrin on osteoporosis prevention was evaluated by bone mineral density, bone microarchitecture and serum biomarkers expression. The mechanisms underlying the prevention effects were investigated by serum cytokine production, osteoclastogenesis, as well as the adipogenesis, immune response, and short-chain fatty acid production. Objective 4 is addressed in this chapter.

Chapter 7 presents a general summary on the key findings and the significance of this study. The recommended future studies are outlined according to the limitations and challenges of this research.

1.4 References

- AAFC. (2015). Trends and Market Opportunities for the Functional Foods and Natural Health Products Sector. Retrieved January 23, 2019, from http://www.agr.gc.ca/eng/industrymarkets-and-trade/canadian-agri-food-sector-intelligence/functional-foods-and-naturalhealth-products/trends-and-market-opportunities-for-the-functional-foods-and-naturalhealth-products-sector/?id=1170940080209
- Aoyagi, Y., Park, H., Park, S., Yoshiuchi, K., Kikuchi, H., Kawakami, H., Shephard, R. J. (2010). Interactive effects of milk basic protein supplements and habitual physical activity on bone health in older women: A 1-year randomized controlled trial. *International Dairy Journal*, 20(10), 724–730. https://doi.org/10.1016/J.IDAIRYJ.2010.03.007
- Cornish, J., Callon, K. E., Naot, D., Palmano, K. P., Banovic, T., Bava, U., Reid, I. R. (2004). Lactoferrin is a potent regulator of bone cell activity and increases bone formation in vivo. *Endocrinology*, 145(9), 4366–4374. https://doi.org/10.1210/en.2003-1307
- Du, M., Xu, W., Yi, H., Han, X., Wang, C., & Zhang, L. (2011). Protective effects of bovine colostrum acid proteins on bone loss of ovariectomized rats and the ingredients identification. *Molecular Nutrition & Food Research*, 55(2), 220–228. https://doi.org/10.1002/mnfr.200900593
- Gentili, C., Doliana, R., Bet, P., Campanile, G., Colombatti, A., Cancedda, F. D., & Cancedda, R. (1994). Ovotransferrin and ovotransferrin receptor expression during chondrogenesis and endochondral bone formation in developing chick embryo. *The Journal of Cell Biology*, *124*(4), 579–588.
- Gentili, C., Doliana, R., Bet, P., Campanile, G., Colombatti, A., Cancedda, F. D., & Cancedda, R. (1994). Ovotransferrin and ovotransferrin receptor expression during chondrogenesis and

endochondral bone formation in developing chick embryo. *Journal of Cell Biology*, *124*(4), 579–588.

- Giansanti, F., Leboffe, L., Angelucci, F., & Antonini, G. (2015). The nutraceutical properties of ovotransferrin and its potential utilization as a functional food. *Nutrients*, 7(11), 9105–9115. https://doi.org/10.3390/nu7115453
- Ji, M, Leem, K. H., Kim, M., & Kim, H. K. (2007). Egg Yolk Soluble Protein Stimulates the Proliferation and Differentiation of Osteoblastic MC3T3-E1 Cells. *Bioscience, Biotechnology, and Biochemistry*, 71(5), 1327–1329. https://doi.org/10.1271/bbb.60217
- Kim, H. K., Kim, M. G., & Leem, K. H. (2008). Inhibitory Effects of Egg Yolk Soluble Protein on Bone Resorption. *Bioscience, Biotechnology, and Biochemistry*, 72(7), 1929–1931. https://doi.org/10.1271/bbb.70491
- Liu, Q., Li, C., Geng, F., Huang, X., & Ma, M. (2017). Hen egg yolk phosvitin stimulates osteoblast differentiation in the absence of ascorbic acid. *J Sci Food Agric*, 94, 4532–4538. https://doi.org/10.1002/jsfa.8320
- Mine, Y., Li-Chan. E., & Jiang, B. (2010). Bioactive Proteins and Peptides as Functional Foods and Nutraceuticals : Yoshinori Mine : 9780813813110. (Y. Mine, Li-Chan. E., & B. Jiang, Eds.). Arnes, AI: Iowa State University Press.
- Naot, D., Grey, A., Reid, I. R., & Cornish, J. (2005). Lactoferrin-a novel bone growth factor. *Clinical Medicine & Research.*, 3(2), 93–101. https://doi.org/3/2/93 [pii]
- Salari Sharif, P., Abdollahi, M., & Larijani, B. (2011). Current, new and future treatments of osteoporosis. *Rheumatology International*, 31(3), 289–300. https://doi.org/10.1007/s00296-010-1586-z
- US Department of Health and Human Services. (2004). Bone health and osteoporosis: a report of

the Surgeon General. In US Health and Human Services. https://doi.org/10.2165/00002018-200932030-00004

Wu, J., & Acero-Lopez, A. (2012). Ovotransferrin: Structure, bioactivities, and preparation. Food Research International, 46(2), 480–487. https://doi.org/10.1016/j.foodres.2011.07.012

CHAPTER 2 - Literature Review

2.1 Egg White Ovotransferrin

Hen egg has a long history of use as a human food and is one of the few foods that are consumed throughout the world regardless of religion and ethnic group (Miranda et al., 2015). It is considered as the best source of high quality protein, which is only inferior to breast milk (Kuang et al., 2018; Miranda et al., 2015). Egg white accounts for 60% of whole egg and contains 11% of protein (Abeyrathne, Lee, & Ahn, 2013). Ovotransferrin is the second major protein in egg white that has many functional properties and can be separated from egg white using various approaches (Abeyrathne et al., 2013). In this section, the structure and biological properties of ovotransferrin are reviewed to highlight its potential as a nutraceutical and functional food ingredient.

2.1.1 Structure and composition of ovotransferrin

Hen egg white ovotransferrin was initially called conalbumin and then renamed ovotransferrin after being recognized as an iron-binding protein and a member of transferrin family (Williams, 1968). It is a single glycopeptide chain consisting of 686 amino acids and contains 15 disulfide bonds and no free sulfhydryl groups (Abeyrathne et al., 2013). This single chain is folded into two globular lobes linked by an alpha helix structure (Lee, McKnight, & Palmiter, 1980; Mazurier et al., 1983). Each lobe has the capability to reversibly bind one Fe^{3+} ion concomitantly with one bicarbonate anion (Vliegenthart et al., 1979; Williams, 1968; Williams, 1982). Studies confirmed that egg white ovotransferrin shows ~50% homology with human transferrin and lactoferrin, but differs from its isoelectric point (pI=6.0 for ovotransferrin and pI=8.8 for lactoferrin), glycosylation pattern (galactose and sialic acid are absent in ovotransferrin), and disulfide bonds

(Lee, McKnight, & Palmiter, 1980; Mazurier et al., 1983; Thibodeau, Lee, & Palmiter, 1978; Vliegenthart et al., 1979; Williams, 1968; Williams, 1982). Similarly to other transferrin proteins, ovotransferrin is responsible for iron scavenging and iron delivery due to the capability of reversibly binding two Fe³⁺ ions per molecular (Abeyrathne et al., 2013; Superti, Ammendolia, Berlutti, & Valenti, 2007). The ovotransferrin molecule comprises two homologous globular lobes and each lobe contains one iron-binding site. Based on the iron-binding situation, ovotransferrin can be defined as apo- (iron free) form and holo- (iron bound) form. Different physical and chemical properties have been found in each form. Basically, the holo-form is more resistant to chemical and physical conditions than the apo-form (Abeyrathne et al., 2013).

2.1.2 Bioactives of Egg White Ovotransferrin and Its-derived Peptides

Although egg white ovotransferrin shares great similarity to lactoferrin, research on the bioactivity of ovotransferrin is limited compared to the extensive studies on lactoferrin. The currently recognized biological properties of egg white ovotransferrin and its-derived peptides are summarized in Table 2.1.

2.1.2.1 Antimicrobial activity

Antimicrobial activity was the first recognized bioactivity of ovotransferrin. Ovotransferrin exhibits a strong antimicrobial activity against a broad spectrum of bacteria (Gram-positive and Gram-negative), fungi, and yeasts (Valenti et al., 1983; Valenti & Visca, 1985); thus it has been used as an antimicrobial agent in food products (Del Giacco, Leone, & Ferlazzo, 1985; Valenti et al., 1983). This antimicrobial property is believed due to the iron-binding ability of ovotransferrin that deprive the iron necessary for the growth of microorganisms (Alderton, Ward, & Fevold, 1946; Bullen, Rogers, & Griffiths, 1978). The species most sensitive to iron deprivation effect of

ovotransferrin are *Pseudomonas* spp., *Escherichia coli*, and *Streptococcus mutants*, while the *Streptococcus aureus*, *Proteus* spp., and *Klebsiella* spp. are more resistant (Valenti et al., 1981, 1982, 1983). Furthermore, this bacteriostatic activity of ovotransferrin can be enhanced by adding carbonate ion (MacGillivray et al., 1998; Valenti et al., 1981), increasing the pH (Baron et al., 2014) and immobilizing ovotransferrin by covalent linkage to Sepharose 4B (Valenti et al., 1982). The antimicrobial activity of ovotransferrin is also reported in metal ion saturated ovotransferrin, suggesting a more complex mechanism might be involved in the antimicrobial activity independent to iron deprivation (Tranter & Board, 1982; Valenti et al., 1983b). Both holo- and apo-ovotransferrin showed bactericidal activities against *S. aureus*, *Bacillus cereus* and *E. coli* (Ibrahim, 1996). The antimicrobial ability of holo-ovotransferrin was thought to be a result of the permeable ability of ovotransferrin that allows it to access the inner membrane and leads to both ion leakage from bacteria and the uncoupling of the respiration-dependent energy production (Aguilera, Quiros, & Fierro, 2003).

Additionally, isolation of a bactericidal domain in ovotransferrin further indicated the mechanisms independent of iron-binding ability. OTAP-92, a 9.9kDa peptide, showed strong sequence similarity with defensins and exhibited antibacterial activity (Ehret-Sabatier et al., 1996; Strahilevitz, Mor, Nicolas, & Shai, 1994; Zhou & Smith, 1990). This 92-resudue peptide contains six cysteines engaged in the form of three intrachain disulfide bonds to maintain the tertiary structure, which is essential to the antimicrobial activity. Its antibacterial action is due to the relatively high alkalinity and cysteine array that involves the blocking of the voltage-dependent K^+ channels (Ehret-Sabatier et al., 1996; Galvez et al., 1990; Ibrahim, Iwamori, Sugimoto, & Aoki, 1998; Ibrahim, Sugimoto, & Aoki, 2000; Strahilevitz et al., 1994).

2.1.2.2 Antioxidant activity

Ovotransferrin is a superoxide dismutase (SOD)-mimicking protein exhibiting antioxidant activity through scavenging superoxide radical (O_2^{\square}) under the natural xanthine/xanthine oxidase (X/XOD) coupling system (Ibrahim, Hoq, & Aoki, 2007; Wu & Acero-Lopez, 2012). The overall antioxidant activity could be promoted by increasing the proximity of amino acid residues that scavenge free radicals and chelating pro-oxidative metal (Elias, Kellerby, & Decker, 2008). Thus, compared to intact ovotransferrin, self-cleaved ovotransferrin showed greater O_2^{\square} - scavenging capacity and the N-terminal lobe was more efficient (Ibrahim et al., 2007; Ibrahim, Haraguchi, & Aoki, 2006). Accordingly, ovotransferrin digested with proteolytic enzymes, such as thermolysin and pepsin, resulted a dramatically increased oxygen radical absorption capacity compared to natural ovotransferrin (Huang et al., 2010; Shen, Chahal, Majumder, You, & Wu, 2010). Two tetrapeptides (Trp-Asn-Ile-Pro and Gly-Trp-Asn-Ile) have the highest antioxidant activity after digestion, indicating the important role of the motif Trp-Asn-Ile in response to the high antioxidant activity (Jahandideh, Chakrabarti, Davidge, & Wu, 2016; Shen et al., 2010). Furthermore, conjugation of ovotransferrin with polyphenols, such as catechin, also improved the oxygen radical absorbance capacity (You, Luo, & Wu, 2014).

2.1.2.3 Anti-inflammatory activity

Ovotransferrin is known to be an acute-phase protein (APP) in avian species, which is upregulated during inflammation (Chamanza et al., 1999; Gabay & Kushner, 1999; Hallquist & Klasing, 1994; Tohio et al., 1995; Xie, Huff, Huff, Balog, & Rath, 2002). Thus, the blood level of ovotransferrin is considered as an infection and inflammation marker in chicken (Rath et al., 2009).

The anti-inflammatory activity has been found in several ovotransferrin-derived peptides (Majumder, Chakrabarti, Davidge, & Wu, 2013; Wang et al., 2017). Dipeptide CR (Cys-Arg), FL (Phe-Leu), HC (His-Cys), LL (Leu-Leu) and MK (Met-Lys) significantly inhibited TNF-α-

induced inflammation in Caco-2 cells (Wang et al., 2017). Similarly, tripeptide IRW (Ile-Arg-Trp) and IQW (Ile-Gln-Trp) attenuated TNF-α-induced inflammatory response and oxidative stress in vascular endothelial cells (Majumder, Chakrabarti, Davidge, et al., 2013). Furthermore, the anti-inflammatory activity of IRW was evaluated *in vivo* using spontaneously hypertensive rats. IRW treatment significantly reduced the inflammatory marker, such as the expression of intercellular adhesion molecule 1 (ICAM-1) and vascular cell adhesion protein 1 (VCAM-1) (Majumder et al., 2015; Majumder, Chakrabarti, Morton, et al., 2013).

2.1.2.4 Immunomodulatory activity

Similar to other transferrin family members, ovotransferrin is a component of innate immunity (Giansanti et al., 2002; Giansanti, Giardi, Massucci, Botti, & Antonini, 2007). The immunomodulatory effect of ovotransferrin was found on macrophages and heterophils (Hang Xie et al., 2002). Also, ovotransferrin inhibited the lipopolysaccharide (LPS)- and phytohemagglutinin (PHA)-induced proliferation of mouse spleen lymphocytes (Otani & Odashima, 1997).

2.1.2.5 Other activities

The antiviral activity of ovotransferrin has been reported to the avian herpesvirus Marek's disease virus (MDV) (Giansanti et al., 2002). Although the mechanism is not clear, the antiviral efficacy was not correlated to iron saturation of ovotransferrin (Giansanti et al., 2002). It has been hypothesized that the antiviral activity to MDV is associated with two ovotransferrin fragments DQKDEYELL and KDLLFK, which can block the MDV infection in chicken embryo fibroblasts (CEF) (Giansanti et al., 2005). These fragments share the sequence homology with two lactoferrinderived antiviral peptides that are effective against Herpes simplex Virus (HSV-1) (Siciliano et al., 1999).

Additionally, partially hydrolyzed ovotransferrin produced by thiol-linked auto-cleavage after reduction showed strong anti-cancer activity by remarkably inhibiting the proliferation of human colon (HCT-116) and breast (MCF-7) cancer cells in a dose-dependent manner (Ibrahim & Kiyono, 2009).

Ovotransferrin also showed promotion effect on chick embryo development. A transiently high expression of ovotransferrin and its receptor was found during the differentiation of hypertrophic chondrocytes into osteoblast-like cells in the chick embryo (Gentili et al., 1993; Gentili et al., 1994). This finding suggested the existence of autocrine and paracrine loops of ovotransferrin and its receptor during chondrogenesis and endochondral bone formation (Cancedda et al., 2000; Gentili et al., 1993; 1994). Furthermore, ovotransferrin showed significant improvement of mitochondrial activity and differentiation in embryonic chick brain and neural retina cells (Bruinink, Sidler, & Birchler, 1996).

Ovotransferrin has been developed as a drug carrier to improve the solubility of water-insoluble antibiotics and to facilitate the specific delivery into microbial or infected cells (Ibrahim et al., 2015).

Lastly, several antihypertensive peptides have been isolated from ovotransferrin hydrolysate via the inhibition to angiotensin converting enzyme (ACE), a key enzyme responsible for regulating hypertension (Hartmann & Meisel, 2007; B. Murray & FitzGerald, 2007). Three ovotransferrinderived tripeptides IRW (Ile-Arg-Trp), LKP (Leu-Lys-Pro) and IQW (Ile-Gln-Trp) have been characterized with ACE inhibitory activity and potent in preventing hypertension (Majumder et al., 2015; Majumder, Chakrabarti, Morton, et al., 2013; Majumder, Chakrabarti, Davidge, et al., 2013).

13
2.1.3 Current Applications of Ovotransferrin

The functional properties and biological activities of ovotransferrin have been widely investigated; however, the practical use of ovotransferrin by industry is highly limited. Similar to bovine lactoferrin, ovotransferrin is suggested for use as an iron-supplementing agent in infant formula (Abeyrathne et al., 2013; Huang et al., 2006). Meanwhile, ovotransferrin is also used as an antibacterial agent in infant formula to protect against pathogen infection (Del Giacco et al., 1985; Valenti et al., 1983b). A study confirmed the safety of supplement ovotransferrin into infant formula by feeding 15 babies with ovotransferrin-treated milk. Ovotransferrin did not sensitize ovotransferrin-fed infants and the total immunoglobulin E (IgE) remained within the normal range (Del Giacco et al., 1985). In addition to incorporation into infant formula, ovotransferrin has been suggested to be combined into κ -carrageenan-based edible film to protect food products and to extend shelf life (Seol, Lim, Jang, Jo, & Lee, 2009).

In addition to promoting human health, ovotransferrin has been used in turkey farms for its antimicrobial activity to prevent respiratory disease. Respiratory pathogens are difficult to control in large-scale turkey production and cost a lot for antibiotic uses. Studies reported that use of a ovotransferrin aerosol on a large Belgian turkey farm significantly reduced infection with the respiratory pathogen *Chlamydia psittaci* and resulted in a prevention of respiratory disease, mortality and lowered antibiotic use (Van Droogenbroeck & Vanrompay, 2013; Van Droogenbroeck et al., 2011).

Ovotransferrin and its-derived hydrolysates/peptides have high potential for use in the pharmaceutical, nutraceutical, and food industries; however, the industrial applications are in the start-up stage even though a great number of researches on the use of ovotransferrin have been published in recent years. Future research on egg white ovotransferrin focusing on characterization

and validation of ovotransferrin functions by animal and clinical trials, as well as the development of ovotransferrin application based on food systems will benefit the development of ovotransferrin-based products and improving the sustainability of the egg industry.

2.2 Bone Physiology and Metabolism

2.2.1 Bone Composition

Bone mainly consists of two compounds: the inorganic mineral and the organic components (Morgan, Barnes, & Einhorn, 2008; Rosen, 2013). The inorganic is in the form of crystalline hydroxyapatite [Ca₃(PO₄)₂]₃Ca(OH)₂, which mainly provide strength and rigidity (Heaney, 2009; Morgan et al., 2008). It also contains impurities in addition to calcium and phosphate, such as carbonate, potassium, magnesium, chloride, fluoride and etc. (Brommage & Neuman, 1979; McConnell, 1964; McConnell, Foreman, Drew, Perkins, & Daly, 1971; Ou-Yang, Paschalis, Mayo, Boskey, & Mendelsohn, 2001). The organic component comprises more than 30 proteins with type I collagen being the most abundant (~90%) (Heaney, 2009; Morgan et al., 2008). While the type I collagen represent the major structural component of the bone matrix to offer flexibility, other proteins present in small quantities significantly contribute to the biological functions (Heaney, 2009; Morgan et al., 2008).

2.2.2 Bone remodeling

Throughout life, bone constantly undergoes a cycle of remodeling (Kini, Usha, Nandeesh, 2012). The remodeling process plays a fundamental role in bone metabolism, which is a prerequisite for a proper execution of bone functions (Parfitt, 2008). In a homeostatic equilibrium bone remodeling, the old and micro-damaged bone is continuously replaced by new and mechanically sound bone, so that to preserve bone strength and adapt to mechanical load (Kini, Usha, Nandeesh, 2012). To manage the remodeling process, three types of bone cells are involved, osteoblast, osteocyte, and osteoclast cells. The osteoblast, known as the bone formation cell, is responsible for the production of new bone (Crockett, Rogers, Coxon, Hocking, & Helfrich, 2011; Feng, 2009; Kini, Usha; Nandeesh, 2012; Lian et al., 2012). After the matrix secretion, ~15% of mature osteoblasts are entrapped in the new bone matrix and differentiated into osteocytes, while other cells remain on the bone surface becoming flat lining cells (Crockett et al., 2011; Feng, 2009; Kini, Usha; Nandeesh, 2012). The osteoclast is a giant multinucleated cell lining on the bone surface and is responsible for bone resorption (Crockett et al., 2011; Feng, 2009; Kini, Usha; Nandeesh, 2012). Osteoclasts and osteoblasts closely collaborate in the remodeling process to replace old bone with new bone (Figure 2.1). Briefly, the remodeling cycle consists of three consecutive phases: resorption, reversal, and formation. First, partially differentiated preosteoclasts migrate to the bone surface and are further differentiated into multinucleated osteoclasts. Once the osteoclastic resorption is complete, the reversal phase starts. The main responsibility of the reversal phase is to prepare the bone surface for new bone formation and provide signals for osteoblast differentiation and migration. Following the completion of reversal phase, osteoblasts start producing new bone until the resorbed bone is completely replaced (Crockett et al., 2011; Feng, 2009; Kini, Usha; Nandeesh, 2012; Lian et al., 2012). The remodeling may be in balance when the amount of new-formed bone equals to the amount of resorbed, or maybe be imbalanced when bone formation is either more or less than the bone resorption. Once the remodeling appears to be imbalanced, different kinds of bone disorders will develop, such as osteoporosis.

2.2.3 Osteoblast and Bone Formation

Bone formation by osteoblasts occurs in three successive phases: the production and maturation of the osteoid matrix, followed by mineralization of the matrix. Initially, osteoblasts produce osteoid by rapidly depositing collagen. Then, collagen synthesis decreases and osteoids become fully mineralized. Bone formation by osteoblasts is controlled both locally and systemically. Studies have demonstrated the crucial importance of local bone formation control by bone morphogenetic protein (BMP) (Cao & Chen, 2005) and Wingless (Wnt) (Day, Guo, Garrett-Beal, & Yang, 2005) signaling pathways for osteoblast differentiation and function. BMP2 can act as a potent stimulator of osteoblast differentiation from mesenchymal stem cell (MSC), osteoprogenitor cell expansion, and osteoblastic bone formation activity, which is also used clinically to enhance bone formation, such as fracture repair (Chau, Leong, & Li, 2009). BMPs bind to the cell surface BMP receptors and elicit the canonical BMP-Smad pathway and/or the non-canonical mitogen-activated protein kinase (MAPK) pathway and phosphoinositide-3-kinase - protein kinase B (PI3K-Akt) pathway (Chau et al., 2009; Derynck & Zhang, 2003; Dijke & Hill, 2004). The activation of these pathways regulates the expression of Runt-related transcription factor 2 (Runx2) (Celil & Campbell, 2005; Fujita et al., 2004; Sharff et al., 2009), also known as core binding factor alpha 1, which is an important transcription factor indispensable for osteoblast differentiation (Javed, Chen, & Ghori, 2010; Kerstetter et al., 1997; Lin & Hankenson, 2011; Yavropoulou & Yovos, 2007).

In addition to BMP signaling, Wnt signaling is a key pathway in bone formation. Wnt-protein ligand binds to cell surface receptor Frizzled and co-receptors low-density lipoprotein receptor-related proteins 5 and 6 (LRP5, LRP6) to activate the downstream glycogen synthase kinase 3 (GSK3)/β-catenin pathway (Chau et al., 2009; He, Semenov, Tamai, & Zeng, 2004). The

accumulation of β -catenin in nucleus subsequently turns on Wnt target genes that might play role in osteoblast turnover, such as Myc, cyclin D, c-Jun, Fra-1, BMPs, fibroblast growth factor (FGF) 18 and Runx2 (Chau et al., 2009). Also, Wnt can also activate non-canonical pathways, such as c-Jun N-terminal kinase (JNK) (Qiu, Chen, & Kassem, 2011) and protein kinase C (PKC) (Luna-Ulloa, Hernández-Maqueda, Castañeda-Patlán, & Robles-Flores, 2011). The schematic presentation indicating the osteoblast differentiation and key regulation pathways are shown in Figure 2.2 and Figure 2.3.

2.2.4 Osteocyte and Bone Formation

Osteocytes are the most abundant cell type in bone and are distributed throughout the mineralized bone matrix (Goldring, 2015). They have the function to sense and respond to local biomechanical and systemic stimuli to regulate bone remodeling (Goldring, 2015). Meanwhile, osteocytes mediate the cell-cell interactions with osteoblast and osteoclast to regulate the production of RANKL and OPG (Kramer et al., 2010). Osteocytes also regulate bone remodeling via the release of soluble mediators, such as prostanoids, nitric oxide, and nucleotides (Schaffler et al., 2014; Cheng et al., 2001; Caballero-Alias et al., 2005). They are also the source of a broad spectrum of cytokines and growth factors such as insulin-like growth factor-1 (IGF-1) and vascular endothelial cell growth factor (VEGF) (Sheng, Lau, & Baylink, 2004; Heino, Hentunen, Vaananen, 2002). The recognition that osteocytes possess such diverse capacities in regulating bone remodeling suggests a novel approach to treat skeletal disorders.

2.2.5 Osteoclast and Bone Resorption

Bone resorption by osteoclast is a multistep process initiated by the proliferation of osteoclast precursors, the differentiation of precursors to mature osteoclasts, and finally degradation of the bone matrix (Teitelbaum, 2000). Osteoclast differentiation initially depends on the receptor for macrophage colony stimulating factor (M-CSF) regulated expression of receptor activator of nuclear factor kappa-B (RANK) (Crockett, Mellis, Scott, & Helfrich, 2011) and its ligand RANKL produced by osteoblasts and stromal cells (Leibbrandt & Penninger, 2008). Signaling through M-CSF and RANK leads to activation of the transcription factors nuclear factor kappa B (NF- κ B), activator protein 1 (AP-1) and nuclear factor of activated T-cells cytoplasmic 1 (NFATc1) (Humphrey, Lanier, & Nakamura, 2005). This following regulates the expression of essential osteoclast genes, such as dendritic cell-specific transmembrane protein (DC-STAMP), tartrateresistant acid phosphatase (TRAP), cathepsin K (CathK), matrix metalloproteinase 9 (MMP9) and β 3 integrin, which allow the final differentiation and fusion of the precursors and function of the resulting multinucleated osteoclast. In addition to the essential role of M-CSF and RNAKL in osteoclastogenesis, osteoclast formation is closely associated with immune system and the osteoclastogenesis can be stimulated in inflammatory conditions, such as rheumatoid arthritis. The stimulation effect is through pro-inflammatory cytokines, such as tumor necrosis factor α (TNF- α) and interleukin 6 (IL-6) (Teitelbaum & Ross, 2003). The schematic presentation indicating the osteoclast differentiation and key regulation pathway are shown in Figure 2.4 and Figure 2.5. Osteoclast numbers in bone are not only controlled by osteoclastogenesis process, but also through the regulation of their lifespan. Decreased osteoclast apoptosis lead to increased bone loss, such as in ovariectomy-induced osteoporosis (Hughes et al., 1996; Hughes & Boyce, 1997). Generally, osteoclast apoptosis is regulated by apoptotic factors, such as B-cell lymphoma-extra large (BclxL) (Woo, Kim, & Ko, 2002; Zhang et al., 2005), the BH3-only family member Bim (Akiyama et al., 2003) and myeloid cell leukaemia sequence 1 (Mcl-1) (Masuda et al., 2014). The expression of these apoptotic factors is regulated by signaling pathways including extracellular signal regulated kinase (ERK) (Ramos, 2008), the Akt (Kawamura et al., 2007) and mammalian target of rapamycin (mTOR) (Hadji, Coleman, & Gnant, 2013).

2.2.6 Interaction Between Osteoblasts and Osteoclasts

Bone remodeling is a tightly coupled process between osteoblast and osteoclast activity. Osteoblasts and osteoclasts can communicate and affect each other through direct cell-cell contact, cytokines and extracellular matrix interaction. The schematic presentation indicating the interaction between osteoblast and osteoclast is shown in Figure 2.6.

2.2.6.1 The role of osteoblasts in bone resorption

In general, osteoblasts can affect osteoclast formation, differentiation and apoptosis through several pathways. Osteoblasts can communicate with osteoclast via direct contact and forming gap junctions to allow small water-soluble molecules passage between two cells (Everts et al., 2002; Matsuo & Irie, 2008). However, the most common interaction between these two cells is through cytokines. Among all, the RANK/RANKL/OPG axis is the most well studied. RANKL, produced by osteoblasts, plays an essential role in initiating the osteoclastogenesis. It can bind with its receptor RANK, expressed on osteoclast progenitors, and actives the down signaling pathways related to osteoclast differentiation (Boyle, Simonet, & Lacey, 2003). Osteoblasts also produce an osteoclastogenesis inhibitory factor named osteoprotegerin (OPG). OPG serves as a decoy receptor that can bind with RANKL and block its ability to bind to and activate RANK (Boyle et al., 2003). So far, several other cytokines have been investigated to play important role in the osteoblast

affected osteoclastogenesis, such as Ephrin2/ephb4 (Zhao et al., 2006), M-CSF/MCP-1 (monocyte chemoattractant protein 1) (Kim & Kim, 2016; Li et al., 2007; Sambandam et al., 2010; Wiktor-Jedrzejczak et al., 1990), LGR4 (leucine-rich repeat-containing G-protein coupled receptor 4)/RANKL/RANK (Luo et al., 2016), Sema3A (semaphoring 3A)/Nrp (neuropilin) (Hayashi et al., 2012; Negishi-Koga et al., 2011) and lysophosphatidic acid (LPA) (David et al., 2010; Lapierre et al., 2010; McMichael, Meyer, & Lee, 2010; Panupinthu et al., 2008; Sims, Panupinthu, Lapierre, Pereverzev, & Dixon, 2013). In addition to affecting osteoclast formation, osteoclast apoptosis could also be induced by osteoblasts. Estrogen affects osteoclast survival through the upregulation of factor associated suicide ligand (FasL) in osteoblast leading to the apoptosis of pre-osteoclast (Garcia et al., 2013; Krum et al., 2008; Wang et al., 2015). Osteoporosis drugs, tamoxifen and reloxifene, can also induce pre-osteoclast apoptosis by the same osteoblast-dependent mechanism (Krum et al., 2008).

2.2.6.2 The role of osteoclasts in bone formation

Recently, studies also described the possible influence of osteoclasts in regulating osteoblastic bone formation. Although it is still not well understood, studies reported that osteoclasts might secrete some molecules, such as semaphoring 4D (Sema4D) (Negishi-Koga et al., 2011), sclerostin (Kusu et al., 2003) and MicroRNA/exsomes, to affect osteoblastic bone formation. Sema4D is highly expressed in osteoclasts and showed potential inhibition of bone formation through its receptor plexin-B1 expressed by osteoblasts (Negishi-Koga et al., 2011). For sclerostin, it was thought only expressed in osteocyte and has anti-anabolic effects on bone formation (Winkler et al., 2003); however, recently a study found that sclerostin is also expressed in osteoclast and could inhibit the activity of BMPs (Kusu et al., 2003). It has been shown that microRNAs also regulate osteogenic activity and osteoblastic bone formation (Lian et al., 2012). A recent study reported

that exosomal miR-214-3p produced by osteoclasts might negatively affect the osteoblastic bone formation (Li et al., 2016).

2.3 Osteoporosis and Current Therapies

2.3.1 Osteoporosis

Osteoporosis is defined as low bone mass, structural deterioration, and porous bone, leading to an increased fracture risk. Osteoporosis is by far the most common bone disease all over the world. Osteoporotic fracture can result in disability, diminished function, loss of independence, and premature death (US Department of Health and Human Services, 2004). Hip fracture, the leading cause of osteoporotic-related disability and death, occur predominantly in the aging populations and the incidence increases exponentially with age (Zeng et al., 2013). With the rapid increase in aging people and life expectancy world-wide, the estimated number of hip fractures is predicted to rise from 1.7 million in 1990 to 6.3 million in 2025 (Zeng et al., 2013). Not surprisingly, bone diseases will take a significant financial toll on both society and individuals who suffer from it. The direct care expenditures for osteoporotic fracture alone range from \$12.2 to \$17.9 billion USD each year (US Department of Health and Human Services, 2004). Recognizing that osteoporosis can have a significant impact on the individual health and the whole society, there is an urgent need to extend knowledge of bone pathology, improve diagnosis method, as well as to develop effective and safer treatment approaches.

2.3.2 Current Osteoporosis Therapies

As one of the most serious bone disease worldwide, several approaches have been developed for osteoporosis treatment and prevention. Non-pharmacological management includes adequate calcium and vitamin D intake, weight-bearing exercising, smoking cessation, limitation of alcohol/caffeine consumption, and fall-prevention techniques (Buckley et al., 2017; Camacho et al., 2016; Cosman et al., 2014; Crockett & Das, 2013; NIH, 2001; Tosteson, Burge, Marshall, & Lindsay, 2008; US Department of Health and Human Services, 2004; Watts et al., 2012). The Institute of Medicine (IOM) recommends that dietary calcium intake should be limited to 1000 mg daily for men 50 to 70 years of age and to 1200 mg daily for women 51 years of age and older and for men 71 years of age and older (Taylor, Yaktine, & Del Balle, 2011). The relationship between calcium intake and several diseases, such as cardiovascular risk and kidney stone, has been debated (Moe & Chen, 2008). Therefore, dietary calcium intake is recommended to meet the requirements, rather than using calcium supplements (Cosman et al., 2014). Vitamin D is a key component in calcium absorption and bone health. Although some evidence supports using vitamin D supplementation to reduce fracture risk, recent studies have shown that higher monthly doses of vitamin D are associated with an increased risk of falls (Bischoff-Ferrari et al., 2016; Chung et al., 2011).

The cause of osteoporosis is due to the imbalance of bone remodeling which finally result in more bone resorption than formation. The goal of pharmacological therapies is categorized as either antiresorptive agents (i.e., bisphosphonates, estrogen agonist/antagonists [EAAs], estrogens, calcitonin, and denosumab) or anabolic agents (i.e., teriparatide) (Salari Sharif, Abdollahi, & Larijani, 2011). Antiresorptive medications primarily decrease the rate of bone resorption, while anabolic medications increase bone formation. Although several medications have overlapping indications, it is important to note that not all osteoporosis medications are approved by the Food and Drug Administration (FDA) to treat postmenopausal osteoporosis (PMO), osteoporosis in men, and/or glucocorticoid-induced osteoporosis (GIO) (Table 2.2). Recommendation for treatment options are based on different characteristics, such as gender, degree of fracture risk, and additional risk factors, such as comorbid disease or medications (Buckley et al., 2017; Camacho et al., 2016; Cosman et al., 2014; Watts et al., 2012).

2.3.3 Emerging Therapies and Investigational Drugs

Due to the limitations and side effects of approved drugs, new therapies and investigational drugs are under investigations and looking for approval.

Sclerostin, a protein produced by osteoclasts, interferes with the proliferation and function of osteoblasts. Several humanized monoclonal antibodies that inhibit sclerostin have been developed, named Romosozumab (Evenity, Amgen/UCB) (MacNabb, Patton, & Hayes, 2016; Medscapte, 2017), Blosozumab and BPS 804 (MacNabb et al., 2016). As blosozumab and BPS 804 are still under the process of development and clinical testing, human trials confirmed the osteoporosis prevention effects of romosozumab. However, the FDA rejected approval of romosozumab due to a higher rate of serious adverse cardiovascular events compared with other current drugs, such as alendronate (MacNabb et al., 2016; Medscapte, 2017). Additionally, cathepsin K (CatK), a protease released by osteoclasts, can promote the degradation of bone matrix, especially type I collagen; thus, inhibiting CatK is assumed to prevent bone resorption. Merck & Co. developed a selective CatK inhibitor Odanacatib showed anti-osteoporosis effects, but will significantly increase the risk of stroke (Mullard, 2016). Therefore, Merck & Co. decided to not continue the development of Odanacatib in 2016 (Mullard, 2016).

In addition to antiresorptive agents, novel anabolic agents have been investigated. Ronacaleret, a calcium-sensing receptor antagonist, was suggested to stimulate transient endogenous parathyroid hormone (PTH) release and increase bone formation. However, the efficacy of Ronacaleret was

not sufficient to be pursued (Balan et al., 2009; Fitzpatrick et al., 2011, 2012). New calcilytic compounds have been developed subsequently and additional studies are continuing because of a lack of an orally active anabolic agent (Balan et al., 2009; Widler, 2011). Another approach is to develop glycogen synthase kinase 3 beta (GSK-3 β) inhibitors. GSK-3 β , a multifunctional serine/threonine kinase, is central to Wnt signaling, which is one of the most important regulation pathways of bone formation (Baron & Rawadi, 2007). Treatment of ovariectomized rats with a nonspecific GSK-3 α and -3 β inhibitors lithium chloride or 6-bromo-indirubin-3'-oxime resulted in enhanced bone formation and increased bone mass (Clement-Lacroix et al., 2005). However, GSK-3 β is thought to be related to pathogenesis of Alzheimer's disease and noninsulin-dependent diabetes (Baron & Rawadi, 2007), so, the clinical use of selective GSK-3 β inhibitors in osteoporosis may be limited (Wilting et al., 2007).

The best approach for osteoporosis prevention is to develop agents with dual actions on bone resorption and formation. Strontium ranelate is currently used in Europe (except UK) for the treatment of osteoporosis, although it is not licensed in the US (Lewiecki et al., 2007; Rodrigues et al., 2018). Strontium ranelate is reported to inhibit osteoclast activity, while stimulating osteoblast activity at the same time (Reginster et al., 2005). However, strontium ranelate has been found to be associated with increased risk of venous thromboembolism (Stevenson et al., 2007). More importantly, the change in bone mineral density should be monitored as strontium could be incorporated into bone (Blake, Lewiecki, Kendler, & Fogelman, 2007; Fogelman & Blake, 2005). Table 2.3 summarizes the emerging therapies and drug.

2.4 Bioactive Proteins and Prevention of Osteoporosis

Due to the low efficacy and associated side effects of pharmacological drugs, there is an increasing interest in developing dietary intervention and food-derived bioactive compounds on osteoporosis prevention. Studies have indicated the importance between protein intake and bone health management, especially in the aging population. Recent reports also pointed out the potential use of several food-derived bioactive proteins and peptides in prevention of osteoporosis. In this section, the major findings with regard to using food-derived bioactive proteins in prevention of osteoporosis are reviewed.

2.4.1 Dairy Protein

Dairy foods are considered as a vital component for improving bone health, as well as a great source of bioactive protein/peptides for investigation. Although the beneficial role of dairy products on bone health management was considered to be a result of the rich calcium content, the effects of bioactive proteins and peptides have been proposed and studied (Table 2.4). Milk whey protein, especially the fraction with alkaline isoelectric point (milk basic protein, MBP), suppressed bone resorption and prevented bone loss in ovariectomized aged rats (Toba et al., 2000). In addition, similar protection effects were also observed in young women (Aoe et al., 2001; Itabashi, 2006; Uenishi et al., 2007; Yamamura et al., 2002), menopausal women (Aoe, Koyama, Toba, Itabashi, & Takada, 2005) and healthy older women (>65 years old) (Aoyagi et al., 2010) who supplemented with MBP. Supplementation with MBP significantly increased bone mineral density and suppressed osteoclast-mediated bone resorption. Bovine colostrum acidic proteins (BCAPs) are able to prevent osteoporosis in ovariectomized rats by stimulating bone mineralization and growth (Du et al., 2011). The effective components in the bovine colostrum

acidic proteins were mainly osteopontin, lactoferrin, epidermal growth factor and insulin-like growth factor-2 (IGF-2) (Du et al., 2011). With further study, lactoferrin has been suggested as an anabolic factor for bone formation. Lactoferrin injected over the right hemicalvarium of adult male mice stimulated the osteoid deposition and mineralization (Cornish et al., 2004). Meanwhile, orally administration of lactoferrin also showed prevention of ovariectomy-induce bone loss and deteriorate of bone microarchitecture (Blais, Malet, Mikogami, Martin-Rouas, & Tomé, 2009; Guo et al., 2009; Malet et al., 2011). Among bioactive peptides, which can be derived from milk proteins, casein phosphopeptides (CPPs) are carriers for minerals, especially calcium (FitzGerald, 1998). CPPs showed to enhance calcium solubility and calcium uptake as well as to stimulate differentiation of human osteoblast cells (Donida et al., 2009; Tulipano et al., 2010).

2.4.2 Soy Protein

Soybeans and their constituents have been extensively investigated for their role in preventing chronic disease, such as osteoporosis and cardiovascular disease (Alekel, Weaver, Ronis, & Ward, 2013). Observational studies such as the low rates of hip fractures in Asians (Greendale et al., 2002; Zheng, Lee, & Chun, 2016) suggested that soybeans may contribute to bone health. Results of many human studies have indicated the protective role of soy protein, its isoflavones, or the combination on ovariectomy-induced osteoporosis prevention (Table 2.5). However, it is still not clear whether the protection effect of soy protein is due to its amino acid composition (Omi, Aoi, Murata, & Ezawa, 1994), non-protein constituents like isoflavones (Arjmandi, Birnbaum, et al., 1998; Arjmandi, Getlinger, et al., 1998), or a combination of these factors (Arjmandi, Getlinger, et al., 1998). While the importance of isoflavones in preserving bone has been demonstrated by animal studies (Anderson, Ambrose, & Garner, 1998; Fanti et al., 1998; Picherit et al., 2001),

whether isoflavones can exert similar bone protective effects independently of soy protein still remain controversial (Picherit et al., 2001; Picherit et al., 2001). Thus, more studies need to be conducted to investigate the functional components in soy protein and mechanisms of action.

2.4.3 Egg Protein

Egg, another good protein source, is reported to be of benefit for pediatric bone health (Coheley et al., 2018). Several bioactive proteins have been found in both egg white and yolk, and are suggested to be able to prevent osteoporosis (Table 2.6). Although most experiments were conducted *in vitro*, studies have claimed a positive relation between bone health and several egg white proteins. Ovalbumin has been suggested to stimulate osteoblast proliferation and differentiation, and further promote bone formation (Farrar, Barone, & Morgan, 2010). Ovotransferrin was also reported to play important role in chondrogenesis and endochondral bone formation in chick embryos (Gentili et al., 1993; Gentili et al., 1994). Meanwhile, egg white protein cystatin was found to exhibit inhibition activity on osteoclastic resorption, which can be used to prevent bone loss (Brand et al., 2004). Moreover, in addition to the direct stimulation on bone cells, egg white proteins can also be developed as protein-based scaffold for bone generation, or protein-based delivery system for osteoporosis drugs. Ovalbumin-, ovomucin- and lysozymeincorporated scaffolds promoted bone formation by supporting cell adhesion and regulating osteogenic activity (Carpena, Abueva, Padalhin, & Lee, 2017; Farrar et al., 2010; Martins et al., 2009). Ovomucoid was developed for calcitonin delivery system which could protect calcitonin against breakdown for oral drug delivery (Shah & Khan, 2004).

Compared to bioactive proteins from egg white, several experiments investigated the bioactivities of egg yolk proteins on bone health with animal studies. Leem et al. (2004) studied the effects of

egg yolk proteins on longitudinal bone growth of adolescent male rats. Egg yolk proteins were extracted with ethanol and fractionated into yolk water-soluble protein and yolk water-insoluble protein. Adolescent male rats were fed with whole egg yolk proteins, yolk water-soluble protein and yolk water-insoluble protein all promoted longitudinal bone growth and the height of growth plates (Leem et al., 2004). The protective effect of these proteins was further reported due to the stimulation on bone formation as well as the inhibition on osteoclastogenesis (Ji, Leem, Kim, & Kim, 2007; Kim, Kim, & Leem, 2008). Continuously, the effects of egg yolk water-soluble protein-derived peptide was prepared and showed stronger effects on preventing bone resorption than egg yolk water-soluble protein (Kim, Lee, & Leem, 2011). Similarly, Kobayashi et al. investigated the effects of food enriched with egg yolk hydrolysate on bone metabolism in orchidectomized dogs (Kobayashi et al., 2015) and developed products Bonepep® (Pharma Foods International Co., Ltd., Kyoto, Japan), a commercial bone peptide produced by enzymatic treatment of yolk water-soluble component of defatted chicken eggs. Orchidetomized dogs supplemented with Bonepep® showed significantly promotion on bone formation immediately after orchiectomy (Kobayashi et al., 2015). Among all egg yolk protein, phosvitin is the most promising candidate to maintain bone health because of its strong ability to bind calcium (Hatta, Kapoor, & Juneja, 2008). Phosvitin has been reported to play important role in bone formation of chicken embryos through dephosphorylation (Li, Geng, Huang, Ma, & Zhang, 2014). Liu et al. found that phosvitin stimulated osteoblasts differentiation in the absence of ascorbic acid (Liu, Li, Geng, Huang, & Ma, 2016), which is known to be essential for bone formation. Although the underlying mechanisms are still under study, it might because phosvitin has similar function as ascorbic acid to induce type I collagen and osteocalcin expression during osteoblasts differentiation (Liu et al., 2016).

2.5 Perspectives and Potential Challenges

2.5.1 Current Osteoporosis Drug Market

In the US, an estimated 19% of older men (≥ 65 years old) and 30% of older women (≥ 65 years old) are at elevated risk of osteoporotic fracture and considered to be eligible for treatment. The burden of osteoporosis is similar in Europe and is projected to raise worldwide, with increasing aging populations and urbanization. Thus, rising prevalence of osteoporosis is a serious public health concern. The global osteoporosis treatment market was valued at 11.5 billion USD in 2015, and is expected to expand at a CAGR (compound annual growth rate) of 3.3% from 2017 to 2025 (Grand View Research, 2017). The global osteoporosis treatment market is segmented by drug type into bisphosphonates, parathyroid hormone therapy, calcitonin, selective estrogen inhibitors modulator (SERM) and RANKL inhibitors. Bisphosphonates are extensively used for prevention and treatment of osteoporosis, thus dominating the osteoporosis treatment market. However, a 50% decrease of using bisphosphonates in postmenopausal women from 2008 to 2012 was documented in US (Jha, Wang, Laucis, & Bhattacharyya, 2015; Khosla & Shane, 2016). The reason is probably multifactorial, but the public awareness about the devastating side effects is the most important one. Also, FDA highlighted a statement claiming the anti-fracture efficacy of bisphosphonates after 5 years of use is scarce and perhaps conflicting (Khosla et al., 2007; Shane et al., 2014). Anabolic agents are likely to further revolutionize the market with the arrival of a novel sclerostin inhibitor, romosozumab, and teriparatide biosimilars. Although this diversification of the anabolic offering will help to bolster the treatment paradigm and increase the completion in the market, the associated side effects, insufficient efficacy and high price tag still remains an unmet need in the osteoporosis market.

2.5.2 Market Opportunity of Bioactive Proteins

In the past few decades, growing consciousness among consumers regarding their health and proper diet is expected to increase the research interests in the promising use of foods to promote well-being and to help with the disease prevention. The global market of functional food and natural health products reached approximately 129.4 billion USD in 2015 and is expected to continue growing (AAFC, 2015; Grand View Research, 2016).

Bone health is closely related to diet and nutrition. Changing dietary patterns and/or providing supplemented nutrients have been suggested to play outstanding role in preventing osteoporosis. Currently, calcium and vitamin D are still the predominant supplementations in bone health management market. However, with the growing concern on the increasing risk of cardiovascular disease and kidney disease by supplementing calcium-vitamin D, there are emerging opportunities for other nutraceuticals, such as bioactive proteins. Bioactive proteins and peptides are an important category of nutraceuticals food sector, which is reported to valued at 75 billion USD per year currently (Mine, Li-Chan, & Jiang, 2010). The use of bioactive proteins in disease intervention is suitable for long-term use, and can provide additional nutritional value compare to pharmaceutical drugs. Currently, a number of bioactive-based functional foods or nutraceuticals are commercially available carrying health claims, such as the colostrum powder, whey protein powder, and collagen products. However, a further investigation and development of bioactive proteins application in osteoporosis is still needed to meet the growing increased market requirements.

2.5.3 Potential Challenges of Bioactive Protein Application

To further broaden the application of bioactive proteins in health promoting, a major challenge is to ensure the efficacy and convenient delivery. In general, bioactive proteins are susceptible to gastrointestinal (GI) digestion, in which the proteins are hydrolyzed into small peptides, and further digested to produce amino acid (Matthews & Laster, 1965). Thus, development of bioactive proteins poses high challenges for stability against GI digestion to ensure the intact proteins can be delivery to proper cellular site of action. Additionally, the mechanisms underlying the effects of bioactive proteins in disease prevention is difficult to elucidate. Even though bioactive proteins successfully go through the GI system and remain intact, the protein absorption and transportation to the target site can be also challenging. Although research reported that the intact lactoferrin has been found in the circulation (Kitagawa et al., 2003; Takeuchi, Kitagawa, & Harada, 2004), proteins with high molecular weight and size are hardly absorbed in small intestine. So, mechanism research is needed to provide the theoretical basis to support the beneficial effects of bioactive proteins. Finally, most of the current studies on bioactive proteins were conducted with only cells or animals. Therefore, further studies to establish the efficacy of bioactive proteins by clinical trials is inevitable.

2.6 General conclusions

There is growing interest in using bioactive compounds for health management and disease prevention. Osteoporosis is one of the most serious chronic bone disorders worldwide, while the therapies are far from satisfactory. Bone health is greatly affected by remodeling; thus, opportunities exist to identify bioactive compounds that can modulate the remodeling process. Currently, food-derived bioactive proteins have been suggested to play important roles in prevention of osteoporosis. Although a number of research reports provided exciting results using bioactive proteins to promote bone health, much of the work is still in its infancy. The degradation of proteins through the gastrointestinal (GI) tract is one of the most important reasons that has limited the application of bioactive proteins. GI digestion may significantly affect the activity of bioactive proteins; therefor, the bioavailability and efficacy of orally administration of bioactive proteins should be further evaluated with animal and human study. Meanwhile, the mechanisms should be further studied to elucidate the principles underlying the beneficial action, which may provide more possible targets to exert the function of bioactive proteins. It may also provide hints to develop novel delivery methods in addition to orally administration, which can minimized the negative influence of GI digestion. Although there are still difficulties need get over, the great potential of food-derived bioactive proteins are expected to be a promising approach to benefit for human health as well as to add value to food industry.

2.7 References

- AAFC. (2015). Trends and Market Opportunities for the Functional Foods and Natural Health Products Sector. Retrieved January 23, 2019, from http://www.agr.gc.ca/eng/industrymarkets-and-trade/canadian-agri-food-sector-intelligence/functional-foods-and-naturalhealth-products/trends-and-market-opportunities-for-the-functional-foods-and-naturalhealth-products-sector/?id=1170940080209
- Abeyrathne, E. D. N. S., Lee, H. Y., & Ahn, D. U. (2013). Egg white proteins and their potential use in food processing or as nutraceutical and pharmaceutical agents - a review. *Poultry Science*, 92, 3292–3299.
- Aguilera, O., Quiros, L. M., & Fierro, J. F. (2003). Transferrins selectively cause ion efflux through bacterial and artificial membranes. *FEBS Letters*, *548*(1–3), 5–10.
- Akiyama, T., Bouillet, P., Miyazaki, T., Kadono, Y., Chikuda, H., Chung, U. I., Tanaka, S.
 (2003). Regulation of osteoclast apoptosis by ubiquitylation of proapoptotic BH3-only Bcl-2 family member Bim. *The EMBO Journal*, *22*(24), 6653–6664.
- Alderton, G., Ward, W. H., & Fevold, H. L. (1946). Identification of the bacteria-inhibiting ironbinding protein of egg white as conalbumin. *Archives of Biochemistry*, *11*, 9–13.
- Alekel, D. L., Germain, A. S., Peterson, C. T., Hanson, K. B., Stewart, J. W., & Toda, T. (2000).
 Isoflavone-rich soy protein isolate attenuates bone loss in the lumbar spine of
 perimenopausal women. *The American Journal of Clinical Nutrition*, 72(3), 844–852.
- Alekel, D. L., Weaver, C. M., Ronis, M. J. J., & Ward, W. E. (2013). Nutritional Influences on Bone Health and Overview of Methods. *Bioactive Food as Dietary Interventions for the Aging Population*, 357–370.

Anderson, J. J., Ambrose, W. W., & Garner, S. C. (1998). Biphasic effects of genistein on bone

tissue in the ovariectomized, lactating rat model. *Proceedings of the Society for Experimental Biology and Medicine. Society for Experimental Biology and Medicine (New York, N.Y.), 217*(3), 345–350.

- Aoe, S., Koyama, T., Toba, Y., Itabashi, A., & Takada, Y. (2005). A controlled trial of the effect of milk basic protein (MBP) supplementation on bone metabolism in healthy menopausal women. *Osteoporosis International*, 16(12), 2123–2128.
- Aoe, S., Toba, Y., Yamamura, J., Kawakami, H., Yahiro, M., Kumegawa, M., Takada, Y.
 (2001). Controlled Trial of the Effects of Milk Basic Protein (MBP) Supplementation on Bone Metabolism in Healthy Adult Women. *Bioscience, Biotechnology, and Biochemistry*, 65(4), 913–918.
- Aoyagi, Y., Park, H., Park, S., Yoshiuchi, K., Kikuchi, H., Kawakami, H., Shephard, R. J. (2010). Interactive effects of milk basic protein supplements and habitual physical activity on bone health in older women: A 1-year randomized controlled trial. *International Dairy Journal*, 20(10), 724–730.
- Arboleya, L., & Castañeda, S. (2013). Osteoimmunology: the study of the relationship between the immune system and bone tissue. *Reumatología Clínica*, *9*(5), 303–315.
- Arjmandi, B. H., Birnbaum, R., Goyal, N. V, Getlinger, M. J., Juma, S., Alekel, L., Kukreja, S.
 C. (1998). Bone-sparing effect of soy protein in ovarian hormone-deficient rats is related to its isoflavone content. *The American Journal of Clinical Nutrition*, 68(6), 1364S–1368S.
- Arjmandi, B. H., Getlinger, M. J., Goyal, N. V, Alekel, L., Hasler, C. M., Juma, S., Kukreja, S. C. (1998). Role of soy protein with normal or reduced isoflavone content in reversing bone loss induced by ovarian hormone deficiency in rats. *The American Journal of Clinical Nutrition*, 68(6), 1358S–1363S.

- Arjmandi, B. H., Lucas, E. A., Khalil, D. A., Devareddy, L., Smith, B. J., McDonald, J., Mason,
 C. (2005). One year soy protein supplementation has positive effects on bone formation
 markers but not bone density in postmenopausal women. *Nutrition Journal*, 4(1), 8.
- Balan, G., Bauman, J., Bhattacharya, S., Castrodad, M., Healy, D. R., Herr, M., Paralkar, V. M. (2009). The discovery of novel calcium sensing receptor negative allosteric modulators. *Bioorganic & Medicinal Chemistry Letters*, 19(12), 3328–3332.
- Baron, F., Jan, S., Gonnet, F., Pasco, M., Jardin, J., Giudici, B., Nau, F. (2014). Ovotransferrin Plays a Major Role in the Strong Bactericidal Effect of Egg White against the Bacillus cereus Group. *Journal of Food Protection*, 77(6), 955–962.
- Baron, R., & Rawadi, G. (2007). Targeting the Wnt/β-Catenin Pathway to Regulate Bone Formation in the Adult Skeleton. *Endocrinology*, 148(6), 2635–2643.
- Bischoff-Ferrari, H. A., Dawson-Hughes, B., Orav, E. J., Staehelin, H. B., Meyer, O. W., Theiler, R., Egli, A. (2016). Monthly High-Dose Vitamin D Treatment for the Prevention of Functional Decline. *JAMA Internal Medicine*, 176(2), 175.
- Blais, A., Malet, A., Mikogami, T., Martin-Rouas, C., & Tomé, D. (2009). Oral bovine lactoferrin improves bone status of ovariectomized mice. *American Journal of Physiology-Endocrinology and Metabolism*, 296(6), E1281–E1288.
- Blake, G. M., Lewiecki, E. M., Kendler, D. L., & Fogelman, I. (2007). A Review of Strontium Ranelate and Its Effect on DXA Scans. *Journal of Clinical Densitometry*, *10*(2), 113–119.
- Bolander, M. E., Youngf, M. F., Fishert, L. W., Yamada, Y., & Terminet, J. D. (1988).
 Osteonectin cDNA sequence reveals potential binding regions for calcium and hydroxyapatite and shows homologies with both a basement membrane protein (SPARC) and a serine proteinase inhibitor (ovomucoid) (bone/calcium-binding proteins).

Biochemistry, 85, 2919–2923.

- Boyle, W. J., Simonet, W. S., & Lacey, D. L. (2003). Osteoclast differentiation and activation. *Nature*, *423*(6937), 337–342.
- Brand, H. S., Lerner, U. H., Grubb, A., Beertsen, W., Nieuw Amerongen, A. V., & Everts, V. (2004). Family 2 cystatins inhibit osteoclast-mediated bone resorption in calvarial bone explants. *Bone*, 35(3), 689–696.
- Brommage, R., & Neuman, W. F. (1979). Passive accumulation of magnesium, sodium, and potassium by chick calvaria. *Calcified Tissue International*, *28*(1), 57–63.
- Bruinink, A., Sidler, C., & Birchler, F. (1996). Neurotrophic effects of transferrin on embryonic chick brain and neural retinal cell cultures. *International Journal of Developmental Neuroscience : The Official Journal of the International Society for Developmental Neuroscience*, 14(6), 785–795.
- Buckley, L., Guyatt, G., Fink, H. A., Cannon, M., Grossman, J., Hansen, K. E., McAlindon, T. (2017). 2017 American College of Rheumatology Guideline for the Prevention and Treatment of Glucocorticoid-Induced Osteoporosis. *Arthritis & Rheumatology*, 69(8), 1521–1537.
- Bullen, J. J., Rogers, H. J., & Griffiths, E. (1978). Role of iron in bacterial infection. Current Topics in Microbiology and Immunology, 80, 1–35. Retrieved from
- Caballero-Alias, A. M., Loverldge, N., Pitsillides, A., Parker, M., Kaptoge, S., Lyon, A., Reeve, J. (2005). Osteocytic expression of constitutive NO synthase isoforms in the femoral neck cortex: a case-control study of intracapsular hip fracture. *J. Bone Miner. Res.*, 20(2), 268-273.
- Camacho, P. M., Petak, S. M., Binkley, N., Clarke, B. L., Harris, S. T., Hurley, D. L., Watts, N.

B. (2016). American Association of Clinical Endocrinologists and American Collage of Endocrinology Clinical Practice Guidelines for the Diagnosis and Treatment of Postmenopausal Osteoporosis - 2016. *Endocrine Practice*, *22*(9), 1111–1118.

- Cancedda, R., Castagnola, P., Cancedda, F. D., Dozin, B., & Quarto, R. (2000). Developmental control of chondrogenesis and osteogenesis. *The International Journal of Developmental Biology*, 44(6), 707–714.
- Cao, X., & Chen, D. (2005). The BMP signaling and in vivo bone formation. *Gene*, 357(1), 1–8.
- Carpena, N. T., Abueva, C. D. G., Padalhin, A. R., & Lee, B. T. (2017). Evaluation of egg white ovomucin-based porous scaffold as an implantable biomaterial for tissue engineering. *Journal of Biomedical Materials Research Part B: Applied Biomaterials*, 105(7), 2107–2117.
- Celil, A. B., & Campbell, P. G. (2005). BMP-2 and Insulin-like Growth Factor-I Mediate Osterix (Osx) Expression in Human Mesenchymal Stem Cells via the MAPK and Protein Kinase D Signaling Pathways. *Journal of Biological Chemistry*, 280(36), 31353–31359.
- Chamanza, R., Toussaint, M. J. M., van Ederen, A. M., van Veen, L., Hulskamp-Koch, C., & Fabri, T. H. F. (1999). Serum amyloid a and transferrin in chicken. A preliminary investigation of using acute - phase variables to assess diseases in chickens. *Veterinary*

Quarterly, 21(4), 158–162.

- Chau, J. F. L., Leong, W. F., & Li, B. (2009). Signaling pathways governing osteoblast proliferation, differentiation and function. *Histology and Histopathology*, 24(12), 1593– 1606.
- Cheng, B., Kato, Y., Zhao, S., Luo, J., Sprague, E., Bonewald, L. F., Jiang, J. X. (2001). PGE(2) is essential for gap junction-mediated intercellular communication between osteocyte-like

MLO-Y4 cells in response to mechanical strain. *Endocrinology*, 142(8), 3464-3473.

- Chung, M., Lee, J., Terasawa, T., Lau, J., & Trikalinos, T. A. (2011). Vitamin D With or
 Without Calcium Supplementation for Prevention of Cancer and Fractures: An Updated
 Meta-analysis for the U.S. Preventive Services Task Force. *Annals of Internal Medicine*, *155*(12), 827.
- Clement-Lacroix, P., Ai, M., Morvan, F., Roman-Roman, S., Vayssiere, B., Belleville, C., Rawadi, G. (2005). Lrp5-independent activation of Wnt signaling by lithium chloride increases bone formation and bone mass in mice. *Proceedings of the National Academy of Sciences*, *102*(48), 17406–17411.
- Coheley, L. M., Kindler, J. M., Laing, E. M., Oshri, A., Hill Gallant, K. M., Warden, S. J., Lewis, R. D. (2018). Whole egg consumption and cortical bone in healthy children. *Osteoporosis International*, 1–9.
- Cornish, J., Callon, K. E., Naot, D., Palmano, K. P., Banovic, T., Bava, U., Reid, I. R. (2004).
 Lactoferrin is a potent regulator of bone cell activity and increases bone formation in vivo.
 Endocrinology, 145(9), 4366–4374.
- Cosman, F., de Beur, S. J., LeBoff, M. S., Lewiecki, E. M., Tanner, B., Randall, S., National Osteoporosis Foundation. (2014). Clinician's Guide to Prevention and Treatment of Osteoporosis. Osteoporosis International : A Journal Established as Result of Cooperation between the European Foundation for Osteoporosis and the National Osteoporosis Foundation of the USA, 25(10), 2359–2381.
- Crockett, J. C., Mellis, D. J., Scott, D. I., & Helfrich, M. H. (2011). New knowledge on critical osteoclast formation and activation pathways from study of rare genetic diseases of osteoclasts: focus on the RANK/RANKL axis. *Osteoporosis International*, *22*(1), 1–20.

- Crockett, J. C., Rogers, M. J., Coxon, F. P., Hocking, L. J., & Helfrich, M. H. (2011). Bone remodelling at a glance. *Journal of Cell Science*, *124*(Pt 7), 991–998.
- Crockett, J., & Das, J. C. (2013). Osteoporosis: a current view of pharmacological prevention and treatment. *Drug Design, Development and Therapy*, 7, 435.
- David, M., Wannecq, E., Descotes, F., Jansen, S., Deux, B., Ribeiro, J., Peyruchaud, O. (2010).
 Cancer Cell Expression of Autotaxin Controls Bone Metastasis Formation in Mouse through Lysophosphatidic Acid-Dependent Activation of Osteoclasts. *PLoS ONE*, 5(3), e9741.
- Day, T. F., Guo, X., Garrett-Beal, L., & Yang, Y. (2005). Wnt/β-Catenin Signaling in Mesenchymal Progenitors Controls Osteoblast and Chondrocyte Differentiation during Vertebrate Skeletogenesis. *Developmental Cell*, 8(5), 739–750.
- Del Giacco, G. S., Leone, A. L., & Ferlazzo, A. (1985). Total IgE in newborns treated prophylactically with conalbumin. *International Journal of Tissue Reactions*, 7(6), 535– 537.
- Derynck, R., & Zhang, Y. E. (2003). Smad-dependent and Smad-independent pathways in TGFβ family signalling. *Nature*, *425*(6958), 577–584.
- Dijke, P. ten, & Hill, C. S. (2004). New insights into TGF-β–Smad signalling. *Trends in Biochemical Sciences*, *29*(5), 265–273.
- Donida, B. M., Mrak, E., Gravaghi, C., Villa, I., Cosentino, S., Zacchi, E., Ferraretto, A. (2009). Casein phosphopeptides promote calcium uptake and modulate the differentiation pathway in human primary osteoblast-like cells. *Peptides*, *30*(12), 2233–2241.
- Droogenbroeck, C. Van, & Vanrompay, D. (2013). Use of ovotransferrin on a turkey farm to reduce respiratory disease. *Veterinary Record*, *172*(3), 71.1-71.4

- Du, M., Wang, K., Wu, C., & Zhang, L. (2009). Effects of bovine colostrum acid protein on bone loss and hemobiochemistry indexes in rats. *Dairy Science and Technology*, 89(5), 449–461.
- Du, M., Xu, W., Yi, H., Han, X., Wang, C., & Zhang, L. (2011). Protective effects of bovine colostrum acid proteins on bone loss of ovariectomized rats and the ingredients identification. *Molecular Nutrition & Food Research*, 55(2), 220–228.
- Ehret-Sabatier, L., Loew, D., Goyffon, M., Fehlbaum, P., Hoffmann, J. A., van Dorsselaer, A., & Bulet, P. (1996). Characterization of novel cysteine-rich antimicrobial peptides from scorpion blood. *The Journal of Biological Chemistry*, 271(47), 29537–29544.
- Elias, R. J., Kellerby, S. S., & Decker, E. A. (2008). Antioxidant Activity of Proteins and Peptides. *Critical Reviews in Food Science and Nutrition*, *48*(5), 430–441.
- Evans, E. M., Racette, S. B., Van Pelt, R. E., Peterson, L. R., & Villareal, D. T. (2007). Effects of soy protein isolate and moderate exercise on bone turnover and bone mineral density in postmenopausal women. *Menopause*, *14*(3), 481–488.
- Everts, V., Delaissé, J. M., Korper, W., Jansen, D. C., Tigchelaar-Gutter, W., Saftig, P., &
 Beertsen, W. (2002). The Bone Lining Cell: Its Role in Cleaning Howship's Lacunae and
 Initiating Bone Formation. *Journal of Bone and Mineral Research*, 17(1), 77–90.
- Fanti, P., Monier-Faugere, M. C., Geng, Z., Schmidt, J., Morris, P. E., Cohen, D., & Malluche,
 H. H. (1998). The Phytoestrogen Genistein Reduces Bone Loss in Short-Term
 Ovariectomized Rats. *Osteoporosis International*, 8(3), 274–281.
- Farrar, G., Barone, J., & Morgan, A. (2010). Ovalbumin-based porous scaffolds for bone tissue regeneration. *Journal of Tissue Engineering*, 2010, 209860.
- Feng, X. (2009). Chemical and Biochemical Basis of Cell-Bone Matrix Interaction in Health and

Disease. Current Chemical Biology, 3(2), 189–196.

- FitzGerald, R. J. (1998). Potential Uses of Caseinophosphopeptides. *International Dairy Journal*, 8(5–6), 451–457.
- Fitzpatrick, L. A., Dabrowski, C. E., Cicconetti, G., Gordon, D. N., Fuerst, T., Engelke, K., & Genant, H. K. (2012). Ronacaleret, a calcium-sensing receptor antagonist, increases trabecular but not cortical bone in postmenopausal women. *Journal of Bone and Mineral Research*, 27(2), 255–262.
- Fitzpatrick, L. A., Dabrowski, C. E., Cicconetti, G., Gordon, D. N., Papapoulos, S., Bone, H. G., & Bilezikian, J. P. (2011). The Effects of Ronacaleret, a Calcium-Sensing Receptor
 Antagonist, on Bone Mineral Density and Biochemical Markers of Bone Turnover in
 Postmenopausal Women with Low Bone Mineral Density. *The Journal of Clinical Endocrinology & Metabolism*, 96(8), 2441–2449.
- Fogelman, I., & Blake, G. M. (2005). Strontium ranelate for the treatment of osteoporosis. *BMJ* (*Clinical Research Ed.*), 330(7505), 1400–1401.
- Fujita, T., Azuma, Y., Fukuyama, R., Hattori, Y., Yoshida, C., Koida, M., Komori, T. (2004).
 Runx2 induces osteoblast and chondrocyte differentiation and enhances their migration by coupling with PI3K-Akt signaling. *The Journal of Cell Biology*, *166*(1), 85–95.
- Gabay, C., & Kushner, I. (1999). Acute-Phase Proteins and Other Systemic Responses to Inflammation. *New England Journal of Medicine*, *340*(6), 448–454.
- Galvez, A., Gimenez-Gallego, G., Reuben, J. P., Roy-Contancin, L., Feigenbaum, P.,
 Kaczorowski, G. J., & Garcia, M. L. (1990). Purification and characterization of a unique,
 potent, peptidyl probe for the high conductance calcium-activated potassium channel from
 venom of the scorpion Buthus tamulus. *The Journal of Biological Chemistry*, 265(19),

11083–11090.

- Garcia, A. J., Tom, C., Guemes, M., Polanco, G., Mayorga, M. E., Wend, K., Krum, S. A. (2013). ERα signaling regulates MMP3 expression to induce FasL cleavage and osteoclast apoptosis. *Journal of Bone and Mineral Research*, 28(2), 283–290.
- Gentili, C., Bianco, P., Neri, M., Malpeli, M., Campanile, G., Castagnola, P., Cancedda, F. D. (1993). Cell proliferation, extracellular matrix mineralization, and ovotransferrin transient expression during in vitro differentiation of chick hypertrophic chondrocytes into osteoblast-like cells. *Journal of Cell Biology*, *122*(3), 703–712.
- Gentili, C., Doliana, R., Bet, P., Campanile, G., Colombatti, A., Cancedda, F. D., & Cancedda, R. (1994). Ovotransferrin and ovotransferrin receptor expression during chondrogenesis and endochondral bone formation in developing chick embryo. *The Journal of Cell Biology*, *124*(4), 579–588.
- Gentili, C., Doliana, R., Bet, P., Campanile, G., Colombatti, A., Cancedda, F. D., & Cancedda, R. (1994). Ovotransferrin and ovotransferrin receptor expression during chondrogenesis and endochondral bone formation in developing chick embryo. *Journal of Cell Biology*, *124*(4), 579–588.
- Giansanti, F., Giardi, M. F., Massucci, M. T., Botti, D., & Antonini, G. (2007). Ovotransferrin expression and release by chicken cell lines infected with Marek's disease virus. *Biochemistry and Cell Biology*, 85(1), 150–155.
- Giansanti, F., Massucci, M. T., Giardi, M. F., Nozza, F., Pulsinelli, E., Nicolini, C., Antonini, G.
 (2005). Antiviral activity of ovotransferrin derived peptides. *Biochemical and Biophysical Research Communications*, 331(1), 69–73.

Giansanti, F., Rossi, P., Massucci, M. T., Botti, D., Antonini, G., Valenti, P., & Seganti, L.

(2002). Antiviral activity of ovotransferrin discloses an evolutionary strategy for the defensive activities of lactoferrin. *Biochemistry and Cell Biology*, *80*(1), 125–130.

Giansanti, F., Rossi, P., Massucci, M. T., Botti, D., Antonini, G., Valenti, P., & Seganti, L. (2002). Antiviral activity of ovotransferrin discloses an evolutionary strategy for the defensive activities of lactoferrin. *Biochem Cell Biol*, 80(1), 125–130.

Goldring, S. R. (2015). The osteocyte: key player in reguating bone turnover. *RMD Open*, 15, 1.
Grand view research. (2017). *Osteoporosis drug market analysis by product (branded, bishosphonates, pharathyroid hormone therapy, calcitonin, selective estrogen inhibitors modulator (SERM), Rank ligand inhibitos, generics), and segment forecasts, 2018-2025.*

- Grand View Research. (2016). Functional foods market analysis by product (carotenoids, dietary fibers, fitty acids, minerals, prebiotics, probiotics, vitamins), by application, by End-Use (Sports nutrition, weight managment, immunity, digestive health) and segment forecasts, 2018 to 2024.
- Greendale, G. A., FitzGerald, G., Huang, M. H., Sternfeld, B., Gold, E., Seeman, T., Sowers, M. (2002). Dietary Soy Isoflavones and Bone Mineral Density: Results from the Study of Women's Health Across the Nation. *American Journal of Epidemiology*, *155*(8), 746–754.
- Guo, H. Y., Jiang, L., Ibrahim, S. A., Zhang, L., Zhang, H., Zhang, M., & Ren, F. Z. (2009). Orally administered lactoferrin preserves bone mass and microarchitecture in ovariectomized rats. *J Nutr*, *139*(C), 958–964.
- Hadji, P., Coleman, R., & Gnant, M. (2013). Bone effects of mammalian target of rapamycin (mTOR) inhibition with everolimus. *Critical Reviews in Oncology/Hematology*, 87(2), 101–111.

Hallquist, N. A., & Klasing, K. C. (1994). Serotransferrin, ovotransferrin and metallothionein

levels during an immune response in chickens. *Comparative Biochemistry and Physiology*. *Biochemistry and Molecular Biology*, *108*(3), 375–384.

- Hartmann, R., & Meisel, H. (2007). Food-derived peptides with biological activity: from research to food applications. *Current Opinion in Biotechnology*, *18*(2), 163–169.
- Hatta, H., Kapoor, M. P., & Juneja, L. R. (2007). Bioactive Components in Egg Yolk. In Egg Bioscience and Biotechnology (pp. 185–237). Hoboken, NJ, USA: John Wiley & Sons, Inc.
- Hayashi, M., Nakashima, T., Taniguchi, M., Kodama, T., Kumanogoh, A., & Takayanagi, H.(2012). Osteoprotection by semaphorin 3A. *Nature*, 485(7396), 69–74.
- He, X., Semenov, M., Tamai, K., & Zeng, X. (2004). LDL receptor-related proteins 5 and 6 in Wnt/-catenin signaling: Arrows point the way. *Development*, 131(8), 1663–1677.
- Heaney, R. P. (2009). Dairy and bone health. *Journal of the American College of Nutrition*, 28 Suppl 1, 82S–90S.
- Heino, T. J., Hentunen, T. A., Vaananen, H. K. (2002). Osteocytes inhibit osteoclastic bone resorption through transforming growth factor-beta: enhancement by estrogen. J. Cell Biochem., 85(1), 185-197.
- Huang, D., Zhao, L., Zong, Y., Yang, S., Zhang, L., Song, Q., Lu, C. (2006, May 26). Ovotransferrin iron supplement agent, and its preparing method and use.
- Huang, W. Y., Majumder, K., & Wu, J. (2010). Oxygen radical absorbance capacity of peptides from egg white protein ovotransferrin and their interaction with phytochemicals. *Food Chemistry*, 123(3), 635–641.
- Huang, W., Chakrabarti, S., Majumder, K., Jiang, Y., Davidge, S. T., & Wu, J. (2010). Eggderived peptide IRW inhibits TNF-α-induced inflammatory response and oxidative stress in endothelial cells. *Journal of Agricultural and Food Chemistry*, 58(20), 10840–10846.

- Hughes, D. E., & Boyce, B. F. (1997). Apoptosis in bone physiology and disease. *Molecular Pathology : MP*, 50(3), 132–137.
- Hughes, D. E., Dai, A., Tiffee, J. C., Li, H. H., Mundy, G. R., & Boyce, B. F. (1996). Estrogen promotes apoptosis of murine osteoclasts mediated by TGF-beta. *Nature Medicine*, 2(10), 1132–1136.
- Humphrey, M. B., Lanier, L. L., & Nakamura, M. C. (2005). Role of ITAM-containing adapter proteins and their receptors in the immune system and bone. *Immunological Reviews*, 208(1), 50–65.
- Ibrahim, H. R. (1996). Insights into the structure-function relationships of ovalbumin, ovotransferrin, and lysozyme. In *Hen Eggs: Basic and Applied Science*.
- Ibrahim, H. R., Haraguchi, T., & Aoki, T. (2006). Ovotransferrin is a redox-dependent autoprocessing protein incorporating four consensus self-cleaving motifs flanking the two kringles. *Biochimica et Biophysica Acta (BBA) - General Subjects*, 1760(3), 347–355.
- Ibrahim, H. R., Hoq, M. I., & Aoki, T. (2007). Ovotransferrin possesses SOD-like superoxide anion scavenging activity that is promoted by copper and manganese binding. *International Journal of Biological Macromolecules*, 41(5), 631–640.
- Ibrahim, H. R., Iwamori, E., Sugimoto, Y., & Aoki, T. (1998). Identification of a distinct antibacterial domain within the N-lobe of ovotransferrin. *Biochimica et Biophysica Acta*, *1401*(3), 289–303.
- Ibrahim, H. R., & Kiyono, T. (2009). Novel Anticancer Activity of the Autocleaved Ovotransferrin against Human Colon and Breast Cancer Cells. *Journal of Agricultural and Food Chemistry*, 57(23), 11383–11390.

Ibrahim, H. R., Sugimoto, Y., & Aoki, T. (2000). Ovotransferrin antimicrobial peptide (OTAP-

92) kills bacteria through a membrane damage mechanism. *Biochimica et Biophysica Acta*, *1523*(2–3), 196–205.

- Ibrahim, H. R., Tatsumoto, S., Ono, H., Van Immerseel, F., Raspoet, R., & Miyata, T. (2015). A novel antibiotic-delivery system by using ovotransferrin as targeting molecule. *European Journal of Pharmaceutical Sciences*, 66, 59–69.
- Itabashi, A. (2006). [Prevention of osteoporosis by foods and dietary supplements. Milk basic protein (MBP) increases bone mineral density in young adult women and perimenopausal women]. *Clinical Calcium*, 16(10), 1632–1638.
- Jahandideh, F., Chakrabarti, S., Davidge, S. T., & Wu, J. (2016). Antioxidant Peptides Identified from Ovotransferrin by the ORAC Method Did Not Show Anti-Inflammatory and Antioxidant Activities in Endothelial Cells. *Journal of Agricultural and Food Chemistry*, 64(1), 113–119.
- Javed, A., Chen, H., & Ghori, F. Y. (2010). Genetic and Transcriptional Control of Bone Formation. Oral and Maxillofacial Surgery Clinics of North America, 22(3), 283–293.
- Jha, S., Wang, Z., Laucis, N., & Bhattacharyya, T. (2015). Trends in Media Reports, Oral Bisphosphonate Prescriptions, and Hip Fractures 1996-2012: An Ecological Analysis. *Journal of Bone and Mineral Research*, 30(12), 2179–2187.
- Ji, M., Leem, K. H., Kim, M., & Kim, H. K. (2007). Egg Yolk Soluble Protein Stimulates the Proliferation and Differentiation of Osteoblastic MC3T3-E1 Cells. *Bioscience, Biotechnology, and Biochemistry*, 71(5), 1327–1329.
- Kapinas, K., & Delany, A. M. (2011). MicroRNA biogenesis and regulation of bone remodeling. *Arthritis Research & Therapy*, *13*(3), 220.

Kawamura, N., Kugimiya, F., Oshima, Y., Ohba, S., Ikeda, T., Saito, T., & Shinoda, Y. (2007).

Akt1 in Osteoblasts and Osteoclasts Controls Bone Remodeling, (10).

- Kenny, A. M., Mangano, K. M., Abourizk, R. H., Bruno, R. S., Anamani, D. E., Kleppinger, A., Kerstetter, J. E. (2009). Soy proteins and isoflavones affect bone mineral density in older women: a randomized controlled trial. *The American Journal of Clinical Nutrition*, 90(1), 234–242.
- Kerstetter, J. E., Caseria, D. M., Mitnick, M. E., Ellison, A. F., Gay, L. F., Liskov, T. A. P., Insogna, K. L. (1997). Increased circulating concentrations of parathyroid hormonein healthy, youngwomen consuminga, 1188–1196.
- Khosla, S., Burr, D., Cauley, J., Dempster, D. W., Ebeling, P. R., Felsenberg, D., American Society for Bone and Mineral Research. (2007). Bisphosphonate-Associated Osteonecrosis of the Jaw: Report of a Task Force of the American Society for Bone and Mineral Research. *Journal of Bone and Mineral Research*, 22(10), 1479–1491.
- Khosla, S., & Shane, E. (2016). A Crisis in the Treatment of Osteoporosis. *Journal of Bone and Mineral Research*, *31*(8), 1485–1487.
- Kim, H. K., Kim, M. G., & Leem, K. H. (2008). Inhibitory Effects of Egg Yolk Soluble Protein on Bone Resorption. *Bioscience, Biotechnology, and Biochemistry*, 72(7), 1929–1931.
- Kim, H. K., Lee, S., & Leem, K.-H. (2011). Protective effect of egg yolk peptide on bone metabolism. *Menopause*, 18(3), 307–313.
- Kim, J. H., Kim, M. S., Oh, H. G., Lee, H. Y., Park, J. W., Lee, B. G., Chae, H. J. (2013). Treatment of eggshell with casein phosphopeptide reduces the severity of ovariectomyinduced bone loss. *Laboratory Animal Research*, 29(2), 70.
- Kim, J. H., & Kim, N. (2016). Signaling Pathways in Osteoclast Differentiation. Chonnam Medical Journal, 52(1), 12–17.

- Kim, M. C., Hong, M. H., Lee, B. H., Choi, H. J., Ko, Y. M., & Lee, Y. K. (2015). Bone Tissue Engineering by Using Calcium Phosphate Glass Scaffolds and the Avidin–Biotin Binding System. *Annals of Biomedical Engineering*, 43(12), 3004–3014.
- Kini, Usha; Nandeesh, B. N. (2012). *Physiology of Bone Formation, Remodeling, and Metabolism. Radionuclide and Hybrid Bone Imaging* (Vol. 9783642024).
- Kitagawa, H., Yoshizawa, Y., Yokoyama, T., Takeuchi, T., Talukder, M. J. R., Shimizu, H., Harada, E. (2003). Persorption of bovine lactoferrin from the intestinal lumen into the systemic circulation via the portal vein and the mesenteric lymphatics in growing pigs. *The Journal of Veterinary Medical Science*, 65(5), 567–572.
- Ko, K. Y., Mendoncam, A. F., Ismail, H., & Ahn, D. U. (2009). Ethylenediaminetetraacetate and lysozyme improves antimicrobial activities of ovotransferrin against Escherichia coli
 O157:H7. *Poultry Science*, 88(2), 406–414.
- Kobayashi, T., Koie, H., Watanabe, A., Ino, A., Watabe, K., Kim, M., Otsuji, K. (2015). Effects of food enriched with egg yolk hydrolysate (bone peptide) on bone metabolism in orchidectomized dogs. *J. Vet. Med. Sci*, 77(4), 503–506.
- Kramer, I., Halleux, C., Keller, H., Pegurri, M., gooi, J. H., Weber, P. B., Feng, J. Q., Bonewald,
 L. F., Knessel M. (2010). Osteocyte Wnt/beta-catenin signaling is required for normal bone
 homeostasis. *Mol. Cell Biol.*, 30, 3071-3085
- Kreijkamp-Kaspers, S., Kok, L., Grobbee, D. E., de Haan, E. H. F., Aleman, A., Lampe, J. W., & van der Schouw, Y. T. (2004). Effect of Soy Protein Containing Isoflavones on Cognitive Function, Bone Mineral Density, and Plasma Lipids in Postmenopausal Women. *JAMA*, 292(1), 65–74.
- Kruger, M. C., Plimmer, G. G., Schollum, L. M., Haggarty, N., Ram, S., & Palmano, K. (2017).
The effect of whey acidic protein fractions on bone loss in the ovariectomised rat. *British Journal of Nutrition*, *93*, 244–252.

- Krum, S. A., Miranda-Carboni, G. A., Hauschka, P. V, Carroll, J. S., Lane, T. F., Freedman, L.
 P., & Brown, M. (2008). Estrogen protects bone by inducing Fas ligand in osteoblasts to regulate osteoclast survival. *The EMBO Journal*, *27*(3), 535–545.
- Kuang, H., Yang, F., Zhang, Y., Wang, T., & Chen, G. (2018). The Impact of Egg Nutrient Composition and Its Consumption on Cholesterol Homeostasis. *Cholesterol*, 2018, 6303810.
- Kulkarni, N. H., Onyia, J. E., Zeng, Q., Tian, X., Liu, M., Halladay, D. L., Ma, Y. L. (2006).
 Orally Bioavailable GSK-3α/β Dual Inhibitor Increases Markers of Cellular Differentiation
 In Vitro and Bone Mass In Vivo. *Journal of Bone and Mineral Research*, *21*(6), 910–920.
- Kusu, N., Laurikkala, J., Imanishi, M., Usui, H., Konishi, M., Miyake, A., Itoh, N. (2003).
 Sclerostin Is a Novel Secreted Osteoclast-derived Bone Morphogenetic Protein Antagonist with Unique Ligand Specificity. *Journal of Biological Chemistry*, 278(26), 24113–24117.
- Lapierre, D. M., Tanabe, N., Pereverzev, A., Spencer, M., Shugg, R. P. P., Dixon, S. J., & Sims,
 S. M. (2010). Lysophosphatidic Acid Signals through Multiple Receptors in Osteoclasts to
 Elevate Cytosolic Calcium Concentration, Evoke Retraction, and Promote Cell Survival. *Journal of Biological Chemistry*, 285(33), 25792–25801.
- Lau, K. H., Baylink, D. J., Zhou, X. D., Rodriguez, D., Bonewald, L. F., Li, Z., Ruffoni, D.,
 Muller, R., Kesavan, C., Sheng, M. H. (2013). Osteocyte-derived insulin-like growth factor
 I is essential for determining bone mechanosensitivity. *Am. J. Physiol. Endocrinol. Metab.*, 305(2), E271-281.

Lee, D. C., McKnight, G. S., & Palmiter, R. D. (1980). The chicken transferrin gene. Restriction

endonuclease analysis of gene sequences in liver and oviduct DNA. *The Journal of Biological Chemistry*, 255(4), 1442–1450.

- Lee, N., Cheng, J., enomoto, T., & Nakano, Y. (2006). One peptide derived from hen ovotransferrin as pro-drug to inhibit angiotensin converting enzyme. *Journal of Food and Drug Analysis*, 14(1), 31–35.
- Leem, K. H., Kim, M. G., Kim, H. M., Kim, M., Lee, Y. J., & Kim, H. K. (2004). Effects of Egg Yolk Proteins on the Longitudinal Bone Growth of Adolescent Male Rats. *Bioscience, Biotechnology, and Biochemistry*, 68(11), 2388–2390.
- Leibbrandt, A., & Penninger, J. M. (2008). RANK/RANKL: Regulators of Immune Responses and Bone Physiology. *Annals of the New York Academy of Sciences*, *1143*(1), 123–150.
- Lewiecki, E. M., Miller, P. D., McClung, M. R., Cohen, S. B., Bolognese, M. A., Liu, Y., AMG 162 Bone Loss Study Group. (2007). Two-Year Treatment With Denosumab (AMG 162) in a Randomized Phase 2 Study of Postmenopausal Women With Low BMD. *Journal of Bone* and Mineral Research, 22(12), 1832–1841.
- Li, C., Geng, F., Huang, X., Ma, M., & Zhang, X. (2014). Phosvitin phosphorus is involved in chicken embryo bone formation through dephosphorylation. *Poultry Science*, 93(12), 3065– 3072.
- Li, D., Liu, J., Guo, B., Liang, C., Dang, L., Lu, C., Zhang, G. (2016). Osteoclast-derived exosomal miR-214-3p inhibits osteoblastic bone formation. *Nature Communications*, 7(1), 10872.
- Li, X., Qin, L., Bergenstock, M., Bevelock, L. M., Novack, D. V., & Partridge, N. C. (2007).
 Parathyroid Hormone Stimulates Osteoblastic Expression of MCP-1 to Recruit and Increase the Fusion of Pre/Osteoclasts. *Journal of Biological Chemistry*, 282(45), 33098–33106.

- Lian, J. B., Stein, G. S., van Wijnen, A. J., Stein, J. L., Hassan, M. Q., Gaur, T., & Zhang, Y.
 (2012). MicroRNA control of bone formation and homeostasis. *Nature Reviews Endocrinology*, 8(4), 212–227.
- Lin, G. L., & Hankenson, K. D. (2011). Integration of BMP, Wnt, and notch signaling pathways in osteoblast differentiation. *Journal of Cellular Biochemistry*, *112*(12), 3491–3501.
- Liu, Q., Li, C., Geng, F., Huang, X., & Ma, M. (2017). Hen egg yolk phosvitin stimulates osteoblast differentiation in the absence of ascorbic acid. *J Sci Food Agric*, *94*, 4532–4538.
- Luna-Ulloa, L. B., Hernández-Maqueda, J. G., Castañeda-Patlán, M. C., & Robles-Flores, M.
 (2011). Protein kinase C in Wnt signaling: Implications in cancer initiation and progression. *IUBMB Life*, 63(10), 915–921.
- Luo, J., Yang, Z., Ma, Y., Yue, Z., Lin, H., Qu, G., Liu, M. (2016). LGR4 is a receptor for RANKL and negatively regulates osteoclast differentiation and bone resorption. *Nature Medicine*, 22(5), 539–546.
- MacGillivray, R. T. A., Moore, S. A., Chen, J., Anderson, B. F., Baker, H., Luo, Y., Baker, E. N. (1998). Two High-Resolution Crystal Structures of the Recombinant N-Lobe of Human Transferrin Reveal a Structural Change Implicated in Iron Release [†], [‡]. *Biochemistry*, *37*(22), 7919–7928.
- MacNabb, C., Patton, D., & Hayes, J. S. (2016). Sclerostin Antibody Therapy for the Treatment of Osteoporosis: Clinical Prospects and Challenges. *Journal of Osteoporosis*, 2016, 1–22.
- Majumder, K., Chakrabarti, S., Davidge, S. T., & Wu, J. (2013). Structure and Activity Study of Egg Protein Ovotransferrin Derived Peptides (IRW and IQW) on Endothelial Inflammatory Response and Oxidative Stress. *Journal of Agricultural and Food Chemistry*, *61*(9), 2120–2129.

- Majumder, K., Chakrabarti, S., Morton, J. S., Panahi, S., Kaufman, S., Davidge, S. T., & Wu, J. (2013). Egg-derived tri-Peptide IRW exerts antihypertensive effects in spontaneously hypertensive rats. *PLoS ONE*, 8(11), e82829.
- Majumder, K., Liang, G., Chen, Y., Guan, L., Davidge, S. T., & Wu, J. (2015). Egg ovotransferrin-derived ACE inhibitory peptide IRW increases ACE2 but decreases proinflammatory genes expression in mesenteric artery of spontaneously hypertensive rats. *Molecular Nutrition & Food Research*, 59(9), 1735–1744.
- Malet, A., Bournaud, E., Lan, A., Mikogami, T., Tomé, D., & Blais, A. (2011). Bovine lactoferrin improves bone status of ovariectomized mice via immune function modulation. *Bone*, 48(5), 1028–1035.
- Martins, A. M., Pham, Q. P., Malafaya, P. B., Raphael, R. M., Kasper, F. K., Reis, R. L., & Mikos, A. G. (2009). Natural Stimulus Responsive Scaffolds/Cells for Bone Tissue Engineering: Influence of Lysozyme upon Scaffold Degradation and Osteogenic Differentiation of Cultured Marrow Stromal Cells Induced by CaP Coatings. *Tissue Engineering Part A*, *15*(8), 1953–1963.
- Masuda, H., Hirose, J., Omata, Y., Tokuyama, N., Yasui, T., Kadono, Y., Tanaka, S. (2014). Anti-apoptotic Bcl-2 family member Mcl-1 regulates cell viability and bone-resorbing activity of osteoclasts. *Bone*, *58*, 1–10.
- Matsuo, K., & Irie, N. (2008). Osteoclast–osteoblast communication. *Archives of Biochemistry* and Biophysics, 473(2), 201–209.
- Matthews, D. M., & Laster, L. (1965). Absorption of protein digestion products: a review. *Gut*, 6(5), 411–426.
- Mazurier, J., Metz-Boutigue, M. H., Jollès, J., Spik, G., Montreuil, J., & Jollès, P. (1983).

Human lactotransferrin: molecular, functional and evolutionary comparisons with human serum transferrin and hen ovotransferrin. *Experientia*, *39*(2), 135–141.

McConnell, D. (1964). Bone Mineral. Science, 145(3638), 1336–1336.

- McConnell, D., Foreman, D. W., Drew, I., Perkins, D., & Daly, P. (1971). Texture and composition of bone. *Science (New York, N.Y.)*, 172(3986), 971–973.
- McMichael, B. K., Meyer, S. M., & Lee, B. S. (2010). c-Src-mediated Phosphorylation of Thyroid Hormone Receptor-interacting Protein 6 (TRIP6) Promotes Osteoclast Sealing Zone Formation. *Journal of Biological Chemistry*, 285(34), 26641–26651.
- Medscapte. (2017). FDA rejects Amgen's osteoporosis hope romosozumab. Retrieved September 19, 2018, from www.medscapte.com/view-artical/882966
- Mine, Y., Li-Chan. E., & Jiang, B. (2010). Bioactive Proteins and Peptides as Functional Foods and Nutraceuticals : Yoshinori Mine : 9780813813110. (Y. Mine, Li-Chan. E., & B. Jiang, Eds.). Arnes, AI: Iowa State University Press.
- Miranda, J. M., Anton, X., Redondo-Valbuena, C., Roca-Saavedra, P., Rodriguez, J. A., Lamas,
 A., Cepeda, A. (2015). Egg and egg-derived foods: effects on human health and use as
 functional foods. *Nutrients*, 7(1), 706–729.
- Moe, S. M., & Chen, N. X. (2008). Mechanisms of vascular calcification in chronic kidney disease. *Journal of the American Society of Nephrology : JASN*, *19*(2), 213–216.
- Morgan, E. F., Barnes, G. L., & Einhorn, T. A. (2008). The bone organ system: form and function. In R. Marcus, D. Feldman, D. W. Dempster, M. Luckey, & J. A. Cauley (Eds.), *Osteoporosis* (3rd ed., pp. 3–26). San Diego: Elsevier Academic Press.
- Mullard, A. (2016). Merck & amp; Co. drops osteoporosis drug odanacatib. *Nature Reviews Drug Discovery*, *15*(10), 669–669.

- Murray, B., & FitzGerald, R. (2007). Angiotensin Converting Enzyme Inhibitory Peptides Derived from Food Proteins: Biochemistry, Bioactivity and Production. *Current Pharmaceutical Design*, 13(8), 773–791.
- Murray, M. J., Meyer, W. R., Lessey, B. A., Oi, R. H., DeWire, R. E., & Fritz, M. A. (2003). Soy protein isolate with isoflavones does not prevent estradiol-induced endometrial hyperplasia in postmenopausal women: a pilot trial. *Menopause*, *10*(5), 456–464.
- Negishi-Koga, T., Shinohara, M., Komatsu, N., Bito, H., Kodama, T., Friedel, R. H., & Takayanagi, H. (2011). Suppression of bone formation by osteoclastic expression of semaphorin 4D. *Nature Medicine*, 17(11), 1473–1480.
- NIH Consensus Development Panel on Osteoporosis Prevention, Diagnosis, and Therapy. (2001). Osteoporosis prevention, diagnosis, and therapy. *JAMA*, *285*(6), 785–795.
- Omi, N., Aoi, S., Murata, K., & Ezawa, I. (1994). Evaluation of the effect of soybean milk and soybean milk peptide on bone metabolism in the rat model with ovariectomized osteoporosis. *Journal of Nutritional Science and Vitaminology*, 40(2), 201–211.
- Otani, H., & Odashima, M. (1997). Inhibition of proliferative responses of mouse spleen lymphocytes by lacto - and ovotransferrins. *Food and Agricultural Immunology*, 9(3),

193–201.

- Ou-Yang, H., Paschalis, E. P., Mayo, W. E., Boskey, A. L., & Mendelsohn, R. (2001). Infrared Microscopic Imaging of Bone: Spatial Distribution of CO32–. *Journal of Bone and Mineral Research*, 16(5), 893–900.
- Panupinthu, N., Rogers, J. T., Zhao, L., Solano-Flores, L. P., Possmayer, F., Sims, S. M., & Dixon, S. J. (2008). P2X7 receptors on osteoblasts couple to production of lysophosphatidic acid: a signaling axis promoting osteogenesis. *The Journal of Cell Biology*, 181(5), 859–

871.

- Parfitt, A. M. (2008). Osteoporosis. In R. Marcus, D. Feldman, D. A. Nelson, & C. J. Rosen (Eds.), Osteoporosis (pp. 71–92). San Diego: Academic Press.
- Picherit, C., Bennetau-Pelissero, C., Chanteranne, B., Lebecque, P., Davicco, M.-J., Barlet, J.-P., & Coxam, V. (2001). Soybean Isoflavones Dose-Dependently Reduce Bone Turnover but
 Do Not Reverse Established Osteopenia in Adult Ovariectomized Rats. *The Journal of Nutrition*, 131(3), 723–728.
- Picherit, C., Chanteranne, B., Bennetau-Pelissero, C., Davicco, M. J., Lebecque, P., Barlet, J. P.,
 & Coxam, V. (2001). Dose-dependent bone-sparing effects of dietary isoflavones in the ovariectomised rat. *The British Journal of Nutrition*, 85(3), 307–316.
- Qiu, W., Chen, L., & Kassem, M. (2011). Activation of non-canonical Wnt/JNK pathway by
 Wnt3a is associated with differentiation fate determination of human bone marrow stromal (mesenchymal) stem cells. *Biochemical and Biophysical Research Communications*, *413*(1), 98–104.
- Ramos, J. W. (2008). The regulation of extracellular signal-regulated kinase (ERK) in mammalian cells. *The International Journal of Biochemistry & Cell Biology*, 40(12), 2707– 2719.
- Rath, N. C., Anthony, N. B., Kannan, L., Huff, W. E., Huff, G. R., Chapman, H. D., Wakenell,
 P. (2009). Serum ovotransferrin as a biomarker of inflammatory diseases in chickens. *Poultry Science*, 88(10), 2069–2074.
- Reginster, J. Y., Sarlet, N., Lejeune, E., & Leonori, L. (2005). Strontium ranelate: a new treatment for postmenopausal osteoporosis with a dual mode of action. *Current Osteoporosis Reports*, 3(1), 30–34.

- Rodrigues, T. A., Freire, A. O., Bonfim, B. F., Cartágenes, M. S. S., Garcia, J. B. S., Rodrigues, T. A., Garcia, J. B. S. (2018). Strontium ranelate as a possible disease-modifying osteoarthritis drug: a systematic review. *Brazilian Journal of Medical and Biological Research*, 51(8).
- Rosen, C. J. (2013). Primer on the metabolic bone diseases and disorders of mineral metabolism. (C. J. Rosen, R. Bouillon, J. Compston, & V. Rosen, Eds.) (8th ed.). Ames, Iowa: Wiley-Blackwell.
- Sakamoto, S., Sakamoto, M., Goldhaber, P., & Glimcher, M. J. (1974). The inhibition of mouse bone collagenase by lysozyme. *Calcified Tissue Research*, *14*(1), 291–299.
- Salari Sharif, P., Abdollahi, M., & Larijani, B. (2011). Current, new and future treatments of osteoporosis. *Rheumatology International*, 31(3), 289–300.
- Sambandam, Y., Blanchard, J. J., Daughtridge, G., Kolb, R. J., Shanmugarajan, S., Pandruvada,
 S. N. M., Reddy, S. V. (2010). Microarray Profile of Gene Expression During Osteoclast
 Differentiation in Modelled Microgravity. *Journal of Cellular Biochemistry*, *111*(5), 1179–1187.
- Schaffler, M. B., Cheung, W. Y., Majeska, R., Kennedy, O. (2014). Osteocytes: master orchestrators of bone. *Calcif. Tissue. Int.*, 94(1): 5-24.
- Seol, K., Lim. D., Jang, A., Jo, C., & Lee, M. (2009). Antimicrobial effect of j-carrageenanbased edible film containing ovotransferrin in fresh chicken breast stored at 5 C. *Meat Science*, 83, 479–483.
- Shah, R. B., & Khan, M. A. (2004). Protection of salmon calcitonin breakdown with serine proteases by various ovomucoid species for oral drug delivery. *Journal of Pharmaceutical Sciences*, 93(2), 392–406.

- Shane, E., Burr, D., Abrahamsen, B., Adler, R. A., Brown, T. D., Cheung, A. M., Whyte, M. P. (2014). Atypical Subtrochanteric and Diaphyseal Femoral Fractures: Second Report of a Task Force of the American Society for Bone and Mineral Research. *Journal of Bone and Mineral Research*, 29(1), 1–23.
- Shang, N., & Wu, J. (2018). Egg White Ovotransferrin Shows Osteogenic Activity in Osteoblast Cells. *Journal of Agricutrual Food and Chemistry*, 66, 2775-2782.
- Sharff, K. A., Song, W. X., Luo, X., Tang, N., Luo, J., Chen, J., He, T. C. (2009). Heyl Basic Helix-Loop-Helix Protein Plays an Important Role in Mediating BMP9-induced Osteogenic Differentiation of Mesenchymal Progenitor Cells. *Journal of Biological Chemistry*, 284(1), 649–659.
- Shen, S., Chahal, B., Majumder, K., You, S. J., & Wu, J. (2010). Identification of Novel Antioxidative Peptides Derived from a Thermolytic Hydrolysate of Ovotransferrin by LC-MS/MS. *Journal of Agricultural and Food Chemistry*, 58(13), 7664–7672.
- Sheng, M. H., Lau, K. H., Baylink, D. J. (2014). Role of osteocyte-derived insulin-like growth factor I in developmental growth, modeling, remodeling, and regeneration of the bone. J. Bone Metab. 21(1), 41-54.
- Shimo-Oka, T., Hagiwara, Y., & Ozawa, E. (1986). Class specificity of transferrin as a muscle trophic factor. *Journal of Cellular Physiology*, *126*(3), 341–351.
- Siciliano, R., Rega, B., Marchetti, M., Seganti, L., Antonini, G., & Valenti, P. (1999). Bovine
 Lactoferrin Peptidic Fragments Involved in Inhibition of Herpes Simplex Virus Type 1
 Infection. *Biochemical and Biophysical Research Communications*, 264(1), 19–23.
- Sims, S. M., Panupinthu, N., Lapierre, D. M., Pereverzev, A., & Dixon, S. J. (2013). Lysophosphatidic acid: A potential mediator of osteoblast–osteoclast signaling in bone.

Biochimica et Biophysica Acta (BBA) - Molecular and Cell Biology of Lipids, *1831*(1), 109–116.

- Stevenson, M., Davis, S., Lloyd-Jones, M., & Beverley, C. (2007). The clinical effectiveness and cost-effectiveness of strontium ranelate for the prevention of osteoporotic fragility fractures in postmenopausal women. *Health Technology Assessment (Winchester, England)*, 11(4), 1–134.
- Strahilevitz, J., Mor, A., Nicolas, P., & Shai, Y. (1994). Spectrum of antimicrobial activity and assembly of dermaseptin-b and its precursor form in phospholipid membranes. *Biochemistry*, 33(36), 10951–10960.
- Superti, F., Ammendolia, M. G., Berlutti, F., & Valenti, P. (2007). Chapter 7 Ovotransferrin. In
 R. Huopalahti, R. López-Fandiño, M. Anton, & R. Schade (Eds.), *Bioactive Egg Compounds* (pp. 43–48). Berlin Heidelberg: Springer.
- Takeuchi, T., Kitagawa, H., & Harada, E. (2004). Evidence of lactoferrin transportation into blood circulation from intestine via lymphatic pathway in adult rats. *Experimental Physiology*, 89(3), 263–270.
- Taylor, C. L., Yaktine, A. L., & Del Balle, H. B. (2011). Dietary Reference Intakes for Calcium and Vitamin D. Washington, D.C.: National Academies Press.
- Teitelbaum, S. L. (2000). Bone resorption by osteoclasts. *Science (New York, N.Y.)*, 289(5484), 1504–1508.
- Teitelbaum, S. L., & Ross, F. P. (2003). Genetic regulation of osteoclast development and function. *Nature Reviews Genetics*, 4(8), 638–649.
- Thibodeau, S. N., Lee, D. C., & Palmiter, R. D. (1978). Identical precursors for serum transferrin and egg white conalbumin. *The Journal of Biological Chemistry*, *253*(11), 3771–3774.

- Toba, Y., Takada, Y., Yamamura, J., Tanaka, M., Matsuoka, Y., Kawakami, H., Kumegawa, M. (2000). Milk basic protein: a novel protective function of milk against osteoporosis. *Bone*, 27(3), 403–408.
- Tohjo, H., Miyoshi, F., Uchida, E., Niiyama, M., Syuto, B., Moritsu, Y., Takeuchi, M. (1995).
 Polyacrylamide Gel Electrophoretic Patterns of Chicken Serum in Acute Inflammation
 Induced by Intramuscular Injection of Turpentine. *Poultry Science*, 74(4), 648–655.
- Tosteson, A. N. A., Burge, R. T., Marshall, D. A., & Lindsay, R. (2008). Therapies for treatment of osteoporosis in US women: cost-effectiveness and budget impact considerations. *The American Journal of Managed Care*, 14(9), 605–615.
- Tranter, H. S., & Board, R. G. (1982). The antimicrobial defense of avian eggs: biological perspective and chemical basis [Poultry]. *Journal of Applied Biochemistry*.
- Tu, K. N., Lie, J. D., Wan, C. K. V., Cameron, M., Austel, A. G., Nguyen, J. K., Hyun, D.
 (2018). Osteoporosis: A Review of Treatment Options. *P & T : A Peer-Reviewed Journal* for Formulary Management, 43(2), 92–104.
- Tulipano, G., Bulgari, O., Chessa, S., Nardone, A., Cocchi, D., & Caroli, A. (2010). Direct effects of casein phosphopeptides on growth and differentiation of in vitro cultured osteoblastic cells (MC3T3-E1). *Regulatory Peptides*, 160(1–3), 168–174.
- Uenishi, K., Ishida, H., Toba, Y., Aoe, S., Itabashi, A., & Takada, Y. (2007). Milk basic protein increases bone mineral density and improves bone metabolism in healthy young women. *Osteoporosis International*, 18(3), 385–390.
- US Department of Health and Human Services. (2004). Bone health and osteoporosis: a report of the Surgeon General. In *US Health and Human Services*.

Valenti, P., Antonini, G., Fanelli, M. R., Orsi, N., & Antonini, E. (1982). Antibacterial activity of

matrix-bound ovotransferrin. Antimicrobial Agents and Chemotherapy, 21(5), 840-841.

- Valenti, P., Antonini, G., Von Hunolstein, C., Visca, P., Orsi, N., & Antonini, E. (1983). Studies of the antimicrobial activity of ovotransferrin. *International Journal of Tissue Reactions*, 5(1), 97–105.
- Valenti, P., Stasio, A., Mastromerino, P., Seganti, L., Sinibaldi, L., & Orsi, N. (1981). Influence of bicarbonate and citrate on the bacteriostatic action of ovotransferrin towards staphylococci. *FEMS Microbiology Letters*, 10(1), 77–79.
- Valenti, P., Visca, P., Antonini, G., Mycopathologia, N. O.-, & 1985, undefined. (1985).
 Antifungal activity of ovotransferrin towards genus Candida. *Mycopathologia*, 89(3), 169–175.
- Valenti, P., Visca, P., Antonini, G., Orsi, N., & Antonini, E. (1987). The effect of saturation with Zn2+ and other metal ions on the antibacterial activity of ovotransferrin. *Medical Microbiology and Immunology*, 176(3), 123–130.
- Van Droogenbroeck, C., Dossche, L., Wauman, T., Van Lent, S., Phan, T. T. T., Beeckman, D. S. A., & Vanrompay, D. (2011). Use of ovotransferrin as an antimicrobial in turkeys naturally infected with Chlamydia psittaci, avian metapneumovirus and Ornithobacterium rhinotracheale. *Veterinary Microbiology*, 153(3–4), 257–263.
- Vliegenthart, J. F. G., Dorland, L., Haverkamp, J., Spik, G., Fournet, B., & Montreuil, J. (1979). Investigation by 360-MHz 1H-nuclear-magnetic-resonance spectroscopy and methylation analysis of the single glycan chain of chicken ovotransferrin. *European Journal of Biochemistry*, 100(2), 569–574.
- Vupadhyayula, P. M., Gallagher, J. C., Templin, T., Logsdon, S. M., & Smith, L. M. (2009). Effects of soy protein isolate on bone mineral density and physical performance indices in

postmenopausal women-a 2-year randomized, double-blind, placebo-controlled trial. *Menopause*, *16*(2), 320–328.

- Waalen, J. (2010). Current and emerging therapies for the treatment of osteoporosis. *Journal of Experimental Pharmacology*, 121.
- Wang, L., Liu, S., Zhao, Y., Liu, D., Liu, Y., Chen, C., Jin, Y. (2015). Osteoblast-induced osteoclast apoptosis by fas ligand/FAS pathway is required for maintenance of bone mass. *Cell Death & Differentiation*, 22(10), 1654–1664.
- Wang, X., Zhao, Y., Yao, Y., Xu, M., Du, H., Zhang, M., & Tu, Y. (2017). Anti-inflammatory activity of di-peptides derived from ovotransferrin by simulated peptide-cut in TNF-αinduced Caco-2 cells. *Journal of Functional Foods*, 37, 424–432.
- Watts, N. B., Adler, R. A., Bilezikian, J. P., Drake, M. T., Eastell, R., Orwoll, E. S., Endocrine Society. (2012). Osteoporosis in Men: An Endocrine Society Clinical Practice Guideline. *The Journal of Clinical Endocrinology & Metabolism*, 97(6), 1802–1822.
- Wellman-Labadie, O., Picman, J., & Hincke, M. T. (2008). Comparative antibacterial activity of avian egg white protein extracts. *British Poultry Science*, 49(2), 125–132.
- Widler, L. (2011). Calcilytics: antagonists of the calcium-sensing receptor for the treatment of osteoporosis. *Future Medicinal Chemistry*, *3*(5), 535–547.
- Wiktor-Jedrzejczak, W., Bartocci, A., Ferrante, A. W., Ahmed-Ansari, A., Sell, K. W., Pollard,
 J. W., & Stanley, E. R. (1990). Total absence of colony-stimulating factor 1 in the
 macrophage-deficient osteopetrotic (op/op) mouse. *Proceedings of the National Academy of Sciences of the United States of America*, 87(12), 4828–4832.
- Williams. J. (1968). A comparison of glycopeptides from the ovotransferrin and serum transferrin of the hen. *Biochemical Journal*, *108*(1), 57–67.

- Williams, J. (1982). The evolution of transferrin. *Trends in Biochemical Sciences*, 7(11), 394–397.
- Wilting, I., de Vries, F., Thio, B. M. K. S., Cooper, C., Heerdink, E. R., Leufkens, H. G. M., van Staa, T. P. (2007). Lithium use and the risk of fractures. *Bone*, 40(5), 1252–1258.
- Winkler, D. G., Sutherland, M. K., Geoghegan, J. C., Yu, C., Hayes, T., Skonier, J. E., Latham,
 J. A. (2003). Osteocyte control of bone formation via sclerostin, a novel BMP antagonist. *The EMBO Journal*, 22(23), 6267–6276.
- Woo, K. H., Kim, H. M., & Ko, J. S. (2002). Macrophage colony-stimulating factor promortes the survival of osteoclast precursors by upregulating Bcl-X(L). *Experimental & Molecular Medicine*, 34(5), 340–346.
- Wu, J., & Acero-Lopez, A. (2012). Ovotransferrin: Structure, bioactivities, and preparation. Food Research International, 46(2), 480–487.
- Xie, H., Huff, G. R., Huff, W. E., Balog, J. M., Holt, P., & Rath, N. C. (2002). Identification of Ovotransferrin as an Acute Phase Protein in Chickens. *Poultry Science*, 81(1), 112–120.
- Xie, H., Huff, G. R., Huff, W. E., Balog, J. M., & Rath, N. C. (2002). Effects of ovotransferrin on chicken macrophages and heterophil-granulocytes. *Developmental and Comparative Immunology*, 26(9), 805–815.
- Xie, H., Newberry, L., Clark, F. D., Huff, W. E., Huff, G. R., Balog, J. M., & Rath, N. C. (2002).
 Changes in Serum Ovotransferrin Levels in Chickens with Experimentally Induced
 Inflammation and Diseases. *Avian Diseases*, 46(1), 122–132.
- Yamaguchi, M., Tezuka, M., Yoshihiro TUKADA, S., & Kish, M. (1998). Casein
 Phosphopeptides in Dietary Calcium Tofu Ovariectomized Rats: Prevention of Bone Loss
 Enhance Calcium Availability in. *Food Sci. Technol. Int.*, 4(3), 209–212.

- Yamamura, J., Aoe, S., Toba, Y., Motouri, M., Kawakami, H., Kumegawa, M., Takada, Y.
 (2002). Milk Basic Protein (MBP) Increases Radial Bone Mineral Density in Healthy Adult
 Women. *Bioscience, Biotechnology, and Biochemistry*, 66(3), 702–704.
- Yavropoulou, M. P., & Yovos, J. G. (n.d.). The role of the Wnt signaling pathway in osteoblast commitment and differentiation. *Hormones (Athens, Greece)*, *6*(4), 279–294.
- You, J., Luo, Y., & Wu, J. (2014). Conjugation of ovotransferrin with catechin shows improved antioxidant activity. *Journal of Agricultural and Food Chemistry*, 62(12), 2581–2587.
- Zeng, F., Fan, F., Xue, W., Xie, H., Wu, B., Tu, S., Chen, Y. (2013). The association of red meat, poultry, and egg consumption with risk of hip fractures in elderly Chinese: a casecontrol study. *Bone*, 56(2), 242–248.
- Zerega, B., Camardella, L., Cermelli, S., Sala, R., Cancedda, R., & Cancedda, F. D. (2001). Avidin expression during chick chondrocyte and myoblast development in vitro and in vivo: regulation of cell proliferation. *Journal of Cell Science*, *114*, 1473–1482.
- Zhang, Q., Badell, I. R., Schwarz, E. M., Boulukos, K. E., Yao, Z., Boyce, B. F., & Xing, L.
 (2005). Tumor necrosis factor prevents alendronate-induced osteoclast apoptosis in vivo by stimulating Bcl-xL expression through Ets-2. *Arthritis & Rheumatism*, 52(9), 2708–2718.
- Zhao, C., Irie, N., Takada, Y., Shimoda, K., Miyamoto, T., Nishiwaki, T., Matsuo, K. (2006).
 Bidirectional ephrinB2-EphB4 signaling controls bone homeostasis. *Cell Metabolism*, 4(2), 111–121.
- Zheng, X., Lee, S.-K., & Chun, O. K. (2016). Soy Isoflavones and Osteoporotic Bone Loss: A Review with an Emphasis on Modulation of Bone Remodeling. *Journal of Medicinal Food*, 19(1), 1–14.

Zhou, Z. R., & Smith, D. L. (1990). Assignment of disulfide bonds in proteins by partial acid

hydrolysis and mass spectrometry. Journal of Protein Chemistry, 9(5), 523-532.

Figure 2.1 Bone remodeling cycle

Bone remodeling cycle mainly occurs in three phases: resorption, reversal, and formation. In the first phase, osteoclast precursors are stimulated to differentiation and adherence to bone surface to initiate the resorption. During the resorption, the bone matrix (old bone) is digested. Reversal phase is considered as a preparation stage for osteoblast to form new bone at the resorption site. Osteoblasts appear at the resorption site to synthesize new bone matrix and become either lining cells resting on the bone surface or osteocytes embedding into the new bone matrix. This figure was made according to Kapinas & Delany, (2011)



Figure 2.2 Osteoblast differentiation

Signalings by Runx2, BMP, and Wnt direct the mesenchymal stem cell differentiated to the osteoblast lineage. Fully differentiated osteoblasts have potential to express alkaline phosphatase, type I collagen, osteocalcin, osteopontin, and bone sialoprotein as well as to incorporate the hydroxyapatite into the newly deposited osteoid. At the end of bone formation, a subset of osteoblast further differentiated into osteocytes (embedded into bone matrix) or become inactive bone lining cells. The remaining osteoblasts undergo apoptosis. In addition to osteoblasts, mesenchymal stem cells can also differentiate into adipocytes, myoblasts, fibroblasts and chondrocytes. This figure was made according to Arboleya & Castañeda, (2013)



Figure 2.3 Key regulation pathways in osteoblast differentiation

Three key signaling pathways regulate the osteoblast differentiation: the BMP signaling, Wnt signaling and FGF signaling. (a) Smad-dependent-BMP signaling activated by binding to BMP receptor and then transducing signaling to Samds (Smad 1/5/8). Activated Smads form a complex with Smad4 then translocate into nucleus and interact with other transcription factors to trigger target gene expression. (b) Wnt signaling activated by binding to a receptor complex including a frizzled receptor coupled with LRP5/6 as co-receptor. The Wnt signaling induces a cascade of intracellular events that stabilize β -catenin and facilitate the transportation into nucleus, where it binds transcription factors and modulate target gene expression. (c) FGF signaling diverse roles in osteoblast differentiation; however, the intracellular signaling cascades responsible to FGF-regulated osteoblast differentiation remain unclear. So far, FGF signaling is known to active MERKs, Akt and PKC pathways to affect osteoblast differentiation.



Figure 2.4 Osteoclast differentiation

In response to M-CSF, hematopoietic stem cells differentiated into macrophage colony-forming units (CFU-M), which are the common precursor cells of macrophages and osteoclasts. RANKL is essential for the differentiation step from CFU-M to multinucleated osteoclasts by cell-cell fusion. Finally, multinucleated osteoclasts undergo maturation and acquire bone-resorbing activity.



Figure 2.5 Key regulation pathways in osteoclast differentiation

M-CSF and RANKL are essential for osteoclast differentiation. M-CSF binds to its receptor c-Fms and activates ERK and Akt pathways to induce cell proliferation and survival of osteoclast precursors. Meanwhile, RANKL recruits TRAF6 to activate MAPKs, Akt and NF- κ B pathways to promote differentiation of osteoclast precursors to osteoclasts. Additionally, various proinflammatory cytokines, such as TNF- α , IL-6 and IL-1, cooperatively orchestrate enhanced osteoclastogenesis. For example, TNF- α stimulates osteoclastogenesis via activate NF- κ B and MAPKs pathways.



Figure 2.6 Interaction between osteoblast and osteoclast

Osteoblasts stimulate proliferation, differentiation, survival and apoptosis of osteoclasts by M-CSF/c-Fms, RANK/RANKL, LPA/LPA receptor, and Fas/FasL. Osteoblast also inhibits the proliferation, differentiation of osteoclasts by OPG, RANKL/LGR4 and Sema3A/Nrp. Meanwhile, osteoclasts produce sclerostin, Sema4D and MicroRNAs or exsomes and inhibit the differentiation or activity of osteoblast. This figure was made according to Matsuo & Irie, (2008)



Activity	Protein/peptide	Mechanisms	Reference
Antimicrobial Activit	ty		
	Ovotransferrin	Iron deprivation necessary for the growth of microorganisms	Alderton, Ward, & Fevold (1946); Valenti et al. (1983); Valenti, Antonini, Fanelli, Orsi, & Antonini (1982)
		Bicarbonate addition increases antibacterial activity by forming a "protein-ferric-bicarbonate" complex	Valenti et al. (1981)
	Ovotransferrin peptide OTAP-92 (Leu109-	Zn ²⁺ -ovotransferrin complex enhances antibacterial activity EDTA and/or lysozyme prevents <i>E. coli</i> O157:H7 of <i>L. monocytogenes</i> proliferation Ion leakage and uncoupling of the respiration- dependent energy production (holo-form) Block the voltage-dependent K ⁺ channel	Valenti, Visca, Antonini, Orsi, & Antonini (1987) r Ko, Mendoncam, Ismail, & Ahn (2009) Alderton et al. (1946); Valenti et al. (1983); Wellman-Labadie, Picman, & Hincke (2008) Ibrahim, Iwamori, Sugimoto, & Aoki (1998); Ibrahim,
	Asp200)		Sugimoto, & Aoki (2000)
Antiviral Activity			
	Ovotransferrin	Antiviral activity against Marek's disease virus (MDV)	Giansanti et al. (2005)
		Viral adsorption inhibition	Giansanti et al. (2002)
	Ovotransferrin peptide DQKDEYELL and KDLLFK	Inhibit viral infection	Giansanti et al. (2002); Giansanti et al. (2005)

Table 2.1 Physiological and pharmacological activities of ovotransferrin and its peptides

Antioxidant activity

	Ovotransferrin	Superoxide dismutase (SOD)-like activity O_2^- scavenge and O_2^- dismutation capacity	Ibrahim, Hoq, & Aoki (2007) Ibrahim et al. (2007)
	Ovotransferrin peptide DLLFKDSAIMLK and FFSASCVPGATIE	Oxygen radical absorbance capacity	(Huang, Majumder, & Wu, 2010)
	Ovotransferrin peptide WNIP and GWNI	Motif WNI is responsible for antioxidant activity	Shen, Chahal, Majumder, You, & Wu (2010)
Anti-cancer activity			
	Reduced autoclaved ovotransferrin	Inhibit the cancer cell proliferation	Ibrahim & Kiyono (2009)
Anti-inflammation ad	ctivity		
	Ovotransferrin	Acute-phase protein (APP) in chicken	Rath et al. (2009); Tohjo et al. (1995); Xie et al. (2002); Xie et al. (2002); Xie, Huff, Huff, Balog, & Rath (2002)
	Ovotransferrin peptide IRW and IQW	Attenuate TNF- α -induced inflammation	Majumder, Chakrabarti, Davidge, & Wu (2013)
Anti-hypertensive ac	tivity		
	Ovotransferrin peptide IRW, IQW and LKP	ACE inhibition	Majumder et al. (2015); Majumder et al. (2013)
	Ovotransferrin peptide KVREGT	ACE inhibition	Lee, Cheng, Enomoto, & Nakano (2006)
Immunomodulating	activity		

	Ovotransferrin	Component of innate immunity	Giansanti et al. (2002); Giansanti, Giardi, Massucci, Botti, & Antonini (2007)
		Modulation of macrophages and heterophils functions	Xie et al. (2002)
		Inhibit LPS and PHA-induced lymphocyte proliferation	Otani & Odashima, (1997)
Growth Promotion			
	Ovotransferrin	Involve in chondrogenesis and endochondral bone formation in chicken	Cancedda, Castagnola, Cancedda, Dozin, & Quarto (2000); Gentili et al. (1993, 1994)
		Improve mitochondrial activity and differentiation in chick brain and neural retina cells Myotrophic activity	Bruinink, Sidler, & Birchler (1996) Shimo-Oka, Hagiwara, & Ozawa (1986)
Drug delivery			
	Ovotransferrin	- Binding and delivery of water insoluble antibiotics (sulphantibiotics)	s Ibrahim et al. (2015)

Table 2.2 Current FDA-approved osteoporosis drugs

Drug	Brand, Manufacturer	Treatment		Prevention		
		Postmenopausal osteoporosis	Osteoporosis in men	Glucocorticoid -induced osteoporosis	Postmenopausal osteoporosis	Glucocorticoid- induced osteoporosis
Alendronate	Fosamax, Merck	Yes	Yes	Yes	Yes	Yes
Alendronate/ch olecalciferol	Fosamax Plus D, Merck	Yes	Yes	No	No	No
Alendronate effervescent	Binosto, Mission Pharmacal	Yes	Yes	No	No	No
Risedronate IR	Actonel, Warner Chilcott	Yes	Yes	Yes	Yes	Yes
Risedronate DR	Atelvia, Warner Chilcott	Yes	No	No	No	No
Ibandronate	Boniva, Genentech	Yes	No	No	No	No
Ibandronate	Boniva, Genentech	Yes	No	No	Yes	No
Zoledronic acid	Reclast, Novartis	Yes	Yes	Yes	Yes	Yes
Denosumab	Prolia, Amgen	Yes	Yes	No	No	No
Raloxifene	Evista, Lilly USA	Yes	No	No	Yes	No

Conjugated estrogens/bazed oxifene	Duavee, Pfizer	No	No	No	Yes	No
Teriparatide	Forteo, Lilly USA	Yes	No	Yes	No	No
Abaloparatide	Tymlos, Radius Health	Yes	No	No	No	No
Calcitonin- salmon	Miacalcin injection, Novartis Swiss	Yes (When alternative treatments are not suitable)	No	No	No	No
This table was ma	de according to Tu et a	l., (2018) and Waalen,	(2010)			

Drug	Manufacturer	Mechanism	Investigation process	Side effects	Reference
Antiresorptive Ag	ent				
Romosozumab	Evenity, Amgen/UBC	Inhibit sclerostin	Rejected by FDA, but will submit application again with updated data	Increase risk of cardiovascular diseases	Medscapte (2017)
Blosozumab	Eli Lilly	Inhibit sclerostin	Complete phase II clinical trials	No serious adverse effects reported	MacNabb, Patton, & Hayes (2016)
BPS 804	Novartis	Inhibit sclerostin	Complete phase II clinical trials	Increase risk of chronic kidney disease	MacNabb et al. (2016)
Odanacatib	Merck & Co.	Inhibit cathepsin K	Discontinued	Increase risk of stroke	Mullard (2016)
Lasofoxifene	Sermonix	SERM	Approved in Europe, but not in U.S. yet	Increase risk of breast cancer, coronary heart disease and stroke	
Anabolic Agents					
Ronacaleret	N/A	Stimulate PTH	Clinical trial is in progress	N/A	Balan et al. (2009); Fitzpatrick et al. (2011, 2012)
GSK-3β inhibitor	N/A	Inhibit GSK-3β	Less clinical data reported	Associate with Alzheimer's disease and diabetes	Baron & Rawadi (2007); Clement- Lacroix et al. (2005); Kulkarni et al. (2006); Wilting et al. (2007)

Table 2.3 The emerging therapies and investigational drugs for osteoporosis

Dual Action Ag	gents				
Strontium ranelate	N/A	Inhibit osteoclast activity, while stimulating osteoblast activity	Approved in Europe (except UK), but not in U.S. yet	Increase risk of venous thromboembolism	Blake, Lewiecki, Kendler, & Fogelman (2007); Reginster, Sarlet, Lejeune, & Leonori (2005); Stevenson, Davis, Lloyd- Jones, & Beverley (2007)

	Experimental design	Outcomes	Reference
Milk basic protein (MBP)			
Animal study			
21 51 week old OVX	0.01% or $0.1%$ MDD for 17 weaks	Increased PMD	Tobe at al
	0.01/0 01 0.1/0 WIDI 101 1/ WCCKS	- Increased DND	(2000)
75.6 month old OVX	3 g/kg MBP for 1 months	Pecovered OVX induce hope loss	(2000) Kruger et al
	5 g/kg MBF 101 4 months	- Recovered OVA-induce bolic loss	(2017)
Tats		collagen	(2017)
		- Preserved hone architecture	
		- Increased organic matter in rat femurs	
34 Healthy adult women	40 mg BMP per day for 6 months	- Increased BMD	A oe et al
5 Theating adult women	to hig bith per day for o months	- Decreased urinary cross-linked N-teleonentides	(2001)
		of type I collagen and deoxypyridinoline	(2001)
33 Healthy adult women	50 mL experimental beverage	- Significantly increase BMD	Yamamura et
<u> </u>	containing 40 mg MBP for 6 month		al. (2002)
35 healthy young	40 mg BMP per day for 6 months	- Increased BMD	Uenishi et al.
women		- Decreased urinary cross-linked N-telopeptides	(2007)
		of type I collagen	
		- Increased serum osteocalcin	
100 healthy menopausal	50 mL experimental beverage	- 1.5% increased BMD	Aoyagi et al.
women	containing 40 mg MBP for 12 month	- Decreased urinary cross-linked N-teleopeptides	(2010)
		of type I collagen	
32 healthy menopausal	40 mg MBP per day for 6 months	- Increased BMD	Aoe,
women		- Decreased urinary cross-linked N-teleopeptides	Koyama,
		of type I collagen	Toba,
			Itabashi, &
			Takada
			(2005)

Table 2.4 Summary of current research on dairy bioactive proteins/peptides and their role in osteoporosis treatment

Bovine colostrum acid proteins (BCAPs)

Animal study

48 3-month OVX SD rats 48 OVX rats	50 mg/day, 10 mg/day or 2 mg/day BCAP for 12 weeks 0.4 g/day, 0.2 g/day or 0.04 g/day BCAP for 12 weeks	 Increased BMC and BMD in a dose-depended manner Increased femur weight and length Increased BMD 	Du et al. (2011) Du, Wang, Wu, & Zhang (2009)
Lactoferrin (LF)			
Animal study			
45 sexually mature male mice	4 mg, 0.4 mg, or 0.4 mg LF injected into right hemicalvaria for 5 days	- Increased bone formation	Cornish et al. (2004)
42 6-week-old OVX mice	20 g/kg, 10 g/kg, 5 g/kg or 1 g/kg LF for 27 weeks	- Increased BMD and femoral failure load	Blais, Malet, Mikogami, Martin- Rouas, & Tomé (2009)
100 12-week-old OVX mice	10 g/kg LF for 1, 2, and 4 months	- Prevented OVX-induced bone loss	Malet et al. (2011)
60 3-month-old OVX rat	85 mg/kg, 8.5 mg/kg or 0.85 mg/kg LF for 3 months	 Increased BMD Increased mechanical strength Decreased serum TNF-αand IL-6 	Guo et al. (2009)
Casein phosphopeptides (CPPs)		
Animal study			
20 OVX rat	100 mg/kg eggshell-CPPs for 4 weeks	 Increased calcium absorption Increased femoral bone weight Rescued serum osteocalcin and BMD 	Kim et al. (2013)
16 5-week-old OVX rat	28 mg/100g CPP tofu or 28 mg/100 dephospho-CPP tofu for 4 weeks	- Prevented OVX-induced bone loss	Yamaguchi, Tezuka, Yoshihiro Tukada, & Kish (1998)

BMD: Bone mineral density

OVX: Ovariectomy

Participants	Dietary Intervention	Outcomes	Reference
Healthy, peri- menopausal women	40g soy protein for 6 months	 Bone-specific alkaline phosphatase N-terminal telopeptide of type I collagen 	Alekel et al. (2000)
Healthy, post- menopausal women	25g soy protein for 12 months	 Bone mineral density Total body Bone mineral content Deoxypyridinoline Bone-specific alkaline phosphatase 	Arjmandi et al. (2005)
Healthy, post- menopausal woman	25.6g soy protein for 9 months	 Bone-specific alkaline phosphatase C-terminal telopeptide of type I collagen 	Evans, Racette, Van Pelt, Peterson, & Villareal (2007)
Healthy, post- menopausal woman	18g soy protein for 12 months	 Bone mineral density N-terminal telopeptide of type I collagen Bone-specific alkaline phosphatase 	Kenny et al. (2009)
Healthy, post- menopausal woman	25.6g soy protein for 12 months	 Bone mineral density Bone-specific alkaline phosphatase 	Kreijkamp-Kaspers et al. (2004)
Healthy, post- menopausal woman	38g soy protein for 6 months	- N-terminal telopeptide of type I collagen	Murray et al. (2003)
Healthy, post- menopausal woman	25g soy protein isolate for 24 months	 Bone mineral density N-terminal telopeptide of type I collagen 	Vupadhyayula, Gallagher, Templin, Logsdon, & Smith (2009)

Table 2.5 Summary of human trials on the impact of soy consumption on bone health

Protein/Peptide	Research Model	Outcomes	Reference
Egg white protein/P	eptide		
Ovalbumin	 Ovalbumin-based scaffolds In vitro study with MC3T3-E1 cells 	 Improved proliferation of MC3T3-E1 Increased expression of alkaline phosphatase (ALP) and osteocalcin 	Farrar, Barone, and Morgan (2010)
Ovotransferrin	- In vitro study with MC3T3-E1	 Stimulation of cell proliferation Increased expression of type I collagen and ALP Promoted cell differentiation 	Shang and Wu (2018)
Ovomucoid	 Ovomucoid-based delivery system N/A 	 Protection of calcitonin breakdown for oral drug delivery Homology between ovomucoid and osteonectin The potential on binding collagen, calcium and hydroxyapatite 	Shah and Khan (2004) Bolander et al. (1988)
Ovomucin	 Ovomucin-based scaffolds In vitro study with bone marrow-derived mesenchymal stem cell 	- Supported cell adhesion and growth	Carpena et al. (2017)
Lysozyme	 N/A Lysozyme incorporated scaffolds In vitro study with rat marrow stromal cells 	 The regulation of collagen degradation during bone remodeling Improved osteogenic activity Increased production of mineralized extracellular matrix 	Sakamoto et al. (1974) Martins et al. (2009)
Cystatin	- N/A	 Inhibited bone matrix degradation Inhibited osteoclastic bone resorption	Brand et al. (2004)

Table 2.6 Summary of research on egg protein/peptides and their impact on bone health

Avidin	 N/A Avidin-biotin binding system In vitro study with human osteoblast cell line MG-63 	 The expression of avidin is related to growth plate hypertrophic cartilage develop in chick embryos Improved osteoblast-like cell adhesion and proliferation Increased the expression of type I collagen, osteopontin, ALP, and osteocalcin 	Zerega et al. (2001) Kim et al. (2015)
Egg Yolk Protein/Pep	tide		
Egg yolk proteins	- In vivo study with adolescent male rats	- Promoted longitudinal bone growth and height of growth plates	Leem et al. (2004)
Egg yolk water- soluble protein	- In vitro study with MC3T3-E1 cells	- Promoted cell proliferation and differentiation	Ji et al. (2007)
-	- In vitro study with bone marrow cells from male ICR mice	- Inhibited TNFα-induced osteoclastogenesis	Kim, Kim, and Leem (2008)
Egg yolk water- soluble peptide	- In vivo study with ovariectomized rats	 Increased bone mineral density and cortical thickness in tibia Prevented ovariectomy-induced cancellous bone loss 	Kim, Lee, and Leem (2011)
Egg yolk water- soluble peptide product Bonepep®	- In vivo study with orchiectomized dogs	- Promoted bone formation after orchiectomy immediately	Kobayashi et al. (2015)
Phosvitin	- In vitro study with live calvarial bone organ culture models	 Stimulated osteoblast differentiation in the absence of ascorbic acid Inhibited parathyroid hormone-induced osteoclastogenesis 	Liu et al. (2016)

CHAPTER 3 - Egg White Ovotransferrin Shows Osteogenic Activity in Osteoblast Cells

3.1 Introduction

Osteoporosis, defined as a skeletal disorder, is characterized by an impairment of bone mass, strength, and microarchitecture, and is caused by an imbalanced bone formation and bone resorption, which is regulated respectively by osteoblast and osteoclast (WHO, 2003). As a serious public health concern, osteoporosis affects over 200 million people worldwide, leading to severe and localized pain, increased bone fractures, and reduced height (US Department of Health and Human Services, 2004). Moreover, osteoporosis is also a major cause of morbidity and health expenditure in ageing populations (Naot, Grey, Reid, & Cornish, 2005). Currently, antiresorptive medications such as bisphosphonates, selective estrogen receptor modulators and hormones are the main pharmacological treatment options for osteoporosis (Chen & Sambrook, 2012). Teriparatide, a recombinant fragment of parathyroid hormone (PTH) has been used as an anabolic agent in osteoporosis treatment by stimulating the bone formation (Rosen, 2004). Despite their great therapeutic value, these agents are limited in their ability to restore bone mass, and are associated with side effects such as nausea, irregular heartbeat, loose bowel movement, and even increased risk of cancer (Sharif, Abdollahi, & Larijani, 2011). Therefore, there is an increasing interest in using natural components as the alternatives for the prevention and treatment of osteoporosis (Jillian Cornish et al., 2004).

Ovotransferrin, a 78 kDa iron-binding glycoprotein found in avian egg white and in avian serum, is a member of transferrin family (Giansanti, Leboffe, Pitari, Ippoliti, & Antonini, 2012). Egg ovotransferrin displays several bioactivities, including antimicrobial, antioxidant, and

immunomodulating activities (Giansanti et al., 2012). Likewise, ovotransferrin is a component of innate immunity (Xie, Huff, Huff, Balog, & Rath, 2002). In chicken macrophages, ovotransferrin was reported to stimulate the production of IL-6, nitrite and Matrix Metallo-proteinase, suggesting that ovotransferrin can modulate macrophage and heterophil function (Xie et al., 2002). More recently, oral administration of egg ovotransferrin reduced the levels of inflammatory cytokines (Andersen, 2015). Since inflammation is known to play an important role in bone metabolism (Redlich & Smolen, 2012), ovotransferrin may have beneficial effects on bone health. Previous study also suggested that the expression levels of ovotransferrin and its receptor were increased during initial stages of bone formation in developing chick embryo (Gentili et al., 1994), further indicating its significant role on cartilage and bone growth. Lactoferrin, another member of transferrin family, was reported to increase osteoblast activity while decreasing osteoclast activity in vitro (Jillian Cornish et al., 2004). Injection of lactoferrin increases bone growth in adult male mice (Jillian Cornish et al., 2004; Naot et al., 2005). Guo et al. (2009) further proved that orally administrated lactoferrin increases bone mineral density and mechanical strength on ovariectomized-induced osteoporosis (Guo et al., 2009). In this research, we investigated the effects of ovotransferrin on cell proliferation and differentiation of bone osteoblast cell MC3T3-E1, a widely used cell model in osteoblastic activity study.

3.2 Materials and methods

3.2.1 Reagents

Ovotransferrin (Conalbumin from chicken egg white, at a purity of 98%) and Alizarin-S red stain were bought from Sigma Chemicals Co (St Louis, MO, USA). α-minimum essential medium (α-MEM) and Fetal Bovine Serum (FBS) were obtained from Gibco/Invitrogen (Carlsbad, CA, USA).
AlamarBlue cell viability reagent and BrdU labeling reagent were bought from ThermoFisher Scientific Inc. (Waltham, MA, USA). Propidium iodide flow cytometry kit was bought from Abcam plc. (Toronto, ON, Canada). Collagen I alpha antibody was bought from Novus Biological Canada ULC (Oakville, ON, Canada). Hoechst 33342 (Trihydrochloride), rabbit anti-mouse IgG (H+L) secondary antibody, and goat anti-rabbit IgG (H+L) secondary antibody were bought from Molecular Probes (Waltham, MA, USA). BrdU mouse antibody was bought from Cell Signaling Technology Inc. (Danvers, MA, USA). Triton-X-100 was from VWR International (West Chester, PA, USA). Goat anti-rabbit and donkey anti-mouse fluorochrome-conjugated secondary antibodies were purchased from Licor Biosciences (Lincoln, NB, USA).

3.2.2 Cell Culture

The murine osteoblastic cell line MC3T3-E1 (subclone 4, ATCC CRL-2593) was purchased from ATCC (Manassas, VA, USA). Cells were cultured in α -MEM supplemented with 10% FBS and penicillin-streptomycin in an incubator under 95% air and 5% CO₂ at 37 °C. Cells were subcultured using 0.25% trypsin every 2 to 3 days. All experiments were performed on 80-90% confluent cells grown in tissue culture grade plastic 96 or 48 well plates. To examine the activity of egg white ovotransferrin, the cells were incubated with different concentrations of ovotransferrin or 1000 µg/mL lactoferrin (the positive control) for different time periods prior to BrDU incorporation, western blotting, and immunofluorescence.

3.2.3 BrDU incorporation assay

The cells were seeded on 48 well tissue culture plates at a concentration of 1×10^4 cell/well, and incubated in α -MEM with 10% FBS. After 4 h of incubation, the medium was changed and different concentrations of ovotransferrin were added. Lactoferrin was also used as a positive control. Following 24 h of incubation, the cells were washed with PBS and placed in fresh α -MEM

with 1% FBS, containing 1% bromodeoxyuridine (BrDU) for 1 h. The cells were then fixed in 70% ethanol for 20 min, treated with 1N hydrochloric acid (HCl) for 20 min to antigen exposure, then permeabilized with 0.1% Triton-X-100 in phosphate buffered saline for 5 min and blocked in 1% bovine serum albumin (BSA) in phosphate buffered saline for 60 min, and finally incubated with mouse monoclonal antibody against BrDU (1:1000) at 4 °C. All the steps except addition of primary antibody were performed at room temperature. Following overnight incubation with the primary antibody, the cells were treated with anti-mouse secondary antibody for 30 min in the dark. Nuclei were stained with the Hoechst33342 nuclear dye. Cells were visualized under an Olympus IX81 fluorescent microscope. For each data point, 3 random fields were chosen. The percentage of nuclei positive for BrDU staining was noted in each field and the mean calculated.

3.2.4 Cell cycle analysis

Cell cycle distribution was analyzed by measuring DNA content using flow cytometer (FACS Canto II, BD Bioscience, CA). The cells were seeded on 6 well tissue culture plates at a concentration of 1×10^4 cells/well, and incubated in α -MEM with 10% FBS until confluence. After being treated with different concentrations of ovotransferrin or 1000 µg/mL lactoferrin for 24 h the adherent cells were washed once with PBS, then trypsinized, and centrifuged at 500 g for 5 min. The cell pellets collected were gently suspended in 400 µL PBS, slowly added 800 µL ice-cold 100% ethanol at 4 °C and mixed for at least 2 h. After centrifugation at 500 g for 5 min, the cell pellets were incubated with 200 µL staining solution (9.45 mL PBS containing 500 µL 20-fold diluted propidium iodide stock solution and 50 µL 200×RNase) at 37 °C in the dark for 30 min prior to the measurement.

3.2.5 Immunofluorescence

The immunofluorescence studies were performed similar to our previous studies (Liao, Chakrabarti, Davidge, & Wu, 2016). Briefly, cells were fixed in 4% formalin, permeabilized with 0.1% Triton-X-100 in PBS and immunostained overnight with a rabbit polyclonal antibody against type I collagen. Cells were treated with Alexa Fluor546 (red) conjugated goat anti-rabbit secondary antibody for 30 min in the dark. Nuclei were stained with the Hoechst33342 nuclear dye (1:10000) for 10 min. After washing to remove unbound antibody/dye, the immunostained cells were observed under an Olympus IX81 fluorescence microscope (Olympus, Tokyo, Japan). Images were obtained using Metamorph imaging software (Molecular Devices, Sunnyvale, CA, USA). Mean fluorescence intensity was calculated from the intensity of the red fluorescent signal (determined by Adobe Photoshop Elements 2.0 software; Adobe Systems Inc., San Jose, CA, USA) from 3 randomly selected fields per group.

3.2.6 Western blot

The cells were seeded on 48 well tissue culture plates at a concentration of 1×10^4 cells/well, and incubated in α -MEM with 10% FBS. The cells were treated with different concentrations of ovotransferrin or 1000 µg/mL lactoferrin for 72 h. After incubation, the culture medium was removed and the cells lysed in boiling hot Laemmle's buffer containing 50 µM dithiothreitol (DTT) and 0.2% Triton-X-100 to prepare samples for western blot as described previously (Liao et al., 2016). These cell lysates were run in SDS-PAGE, blotted to nitrocellulose membranes and immunoblotted with antibodies against ALP, RANKL, OPG and the loading control α -tubulin. Anti-tubulin was used at 0.4 µg/mL, while the others were used at 1 µg/mL. The protein bands were detected by a Licor Odyssey BioImager and quantified by densitometry using corresponding software (Licor Biosciences, Lincoln, NB, US). Each band of ALP, RANKL and OPG was normalized to its corresponding band of loading control. Cell lysates from untreated cells were loaded onto every gel. The results were expressed as percentage of the corresponding untreated control.

3.2.7 Mineralization

The mineralization studies were performed similar to previous studies (Reinholz et al., 2000). The degree of mineralization was determined in the 12-well plates using Alizarin Red staining. Cells were incubated with different concentrations of ovotransferrin for 10 days. The medium was removed and cells were rinsed twice with PBS. The cells were fixed with ice-cold 70% (v/v) ethanol for 1 h. The ethanol was removed by aspiration and cells were washed twice with Milli-Q water. The cells were then stained with 1% (w/v) Alizarin-S Red in Milli-Q (pH 4.2) for 10 min at room temperature. After washing with Milli-Q water, the samples were observed under light and pictures were photographed.

3.2.8 Statistics

All data are presented as mean \pm SEM (standard error of mean) of between 4 and 7 independent experiments. Data were analyzed using One Way analysis of variance (ANOVA) with Dunnett's post-hoc test for comparisons to control. The PRISM 6 statistical software (GraphPad Software, San Diego, CA) was used for the analyses. P< 0.05 was considered significant.

3.3 Results

3.3.1 Ovotransferrin promotes cell proliferation of osteoblasts

Ovotransferrin did not show cytotoxicity to osteoblasts at concentrations up to 1000 μ g/mL. To investigate the effect of ovotransferrin on cell proliferation, osteoblasts were incubated with thymidine analog BrDU, and the incorporation of BrDU into newly synthesized DNA was detected by fluorescence microscopy. As shown in Figure 3.1, the percentage of BrDU positive cells

significantly increased after adding lactoferrin (positive control, 1000 μ g/mL) or ovotransferrin at concentrations of 100 μ g/mL and 1000 μ g/mL. At the concentration of ovotransferrin (1000 μ g/mL), the percentage of BrDU-incorporated cells increased to 18.3±0.7%, which was twofold increment of the untreated group (10.4±1.2%). These results indicated that ovotransferrin could promote osteoblast proliferation, suggesting the potential role of ovotransferrin as an osteoinductive factor in bone formation.

3.3.2 Ovotransferrin inhibits the G0/G1 arrest but promotes S and G2/M arrest

To determine whether the promotion of cell proliferation can be attributed to cell cycle arrest, cell nuclei were stained with propidium iodide and cell cycle was analyzed by flow cytometry. As shown in Figure 3.2, the percentage of cells in G0/G1 phase decreased dose-dependently in the ovotransferrin groups, while increased in the lactoferrin group. At the concentration of 1000 μ g/mL ovotransferrin, the percentage of cells in the G0/G1 phase decreased significantly to 74.3±0.6%, as compared to 86.4±0.4% in the untreated group and 87.6±3.0% in the lactoferrin group. On the contrary to G0/G1 phase, the percentage of cells in S and G2/M phases increased in the ovotransferrin groups while decreased in lactoferrin group; the highest percentage was shown at 100 μ g/mL ovotransferrin treatment instead of 1000 μ g/mL ovotransferrin group. The percentage of cells in the S phase was increased significantly to 15.9±0.3% in 100 μ g/mL ovotransferrin group. And the percentage of cells in G2/M phase was not affected by ovotransferrin addition.

3.3.3 Ovotransferrin increases expression of type I collagen in osteoblasts

Type I collagen expressed by osteoblasts play an important role as the structural component in bone formation. A high level of type I collagen expression, biosynthesis, and secretion is a typical indicator of osteoblasts differentiation. Thus, the expression of type I collagen was also examined in this study to investigate the effect of ovotransferrin on cell differentiation. As shown in Figure 3.3, the expression of type I collagen increased after treatment with lactoferrin (positive control) and ovotransferrin; at a concentration of 100 μ g/mL ovotransferrin, type I collagen showed 1.5-fold increase compared with the untreated group. These results suggested that ovotransferrin could stimulate the expression of bone matrix component type I collagen, supporting further a beneficial role of ovotransferrin on promoting cell differentiation.

3.3.4 Ovotransferrin promotes expression of ALP in osteoblasts

Differentiation of osteoblast precursor to postmitotic matrix-producing osteoblasts is one of the most important stages in bone formation. Determination of alkaline phosphatase (ALP) expression is a widely used indicator for both osteoblast differentiation and bone formation. As shown in Figure 3.4, the expression of ALP was significantly increased after treatment with lactoferrin (positive control) and ovotransferrin (100 μ g/mL and 1000 μ g/mL) for 72 h. Interestingly, the maximal stimulation was observed at 100 μ g/mL, not at the concentration of 1000 μ g/mL. The expression level of ALP increased to 149.3±21.5% when treated with 100 μ g/mL ovotransferrin, compared to 138.3±13.7% for 1000 μ g/mL treatment. Our results suggested ovotransferrin could promote ALP expression when treated with ovotransferrin, further supporting the ability of ovotransferrin on stimulating osteoblast differentiation.

3.3.5 Ovotransferrin promotes mineralization of osteoblasts

Mineralization is the most important step of osteoblast differentiation and is frequently used as a marker to characterize bone formation. To examine the effect of ovotransferrin on mineralization in osteoblasts, MC3T3-E1 cells were incubated with ovotransferrin for 5 days, 10 days, 15 days and 20 days, and mineralization assay was carried out using Alizarin red S staining. The magnitude of the absorbance for the developed color is proportional to the quantity of calcium deposit, which is a direct indication of cell mineralization. As shown in Figure 3.5, at increasing incubation time, the calcium deposits were increased in MC3T3-E1 in all groups. In the first 5 days, there was no significant difference between untreated group and ovotransferrin treatment. At 10 days, a significantly increase of absorbance was observed at 1000 μ g/mL ovotransferrin group, indicating ovotransferrin stimulated the calcium nodules and deposits in osteoblasts. At 20 days, the absorbance was significantly increased at both 100 μ g/mL and 1000 μ g/mL ovotransferrin could promote the calcium deposit and calcium nodules production, suggesting that ovotransferrin could enhance mineralization.

3.3.6 Ovotransferrin inhibits expression of RANKL in osteoblasts

Besides the effect on bone formation, we also investigated the effect of ovotransferrin on bone resorption. RANKL, produced by osteoblasts, acts as a chemoattractant to initiate the osteoclastogenesis and contributes to osteoclast differentiation; therefore expression of RANKL is critical in regulating osteoclasts formation and bone resorption (Rucci, 2008). As shown in Figure 3.6, expression of RANKL decreased after treated with lactoferrin (positive control) and ovotransferrin in a dose-dependent manner. The expression of RANKL was significantly

decreased to $60.3\pm1.9\%$ (100 µg/mL ovotransferrin), or to $55.4\pm8.6\%$ (1000 µg/mL ovotransferrin), or to $58.8\pm2.1\%$ (lactoferrin). These results indicated that ovotransferrin could inhibit the expression of RANKL, suggesting its potential role on preventing osteoclastogenesis.

3.3.7 Ovotransferrin promotes expression of OPG in osteoblasts

In addition to RANKL, osteoblasts also release OPG to counteract the function of RANKL and to prevent osteoclast-mediated bone resorption (Rucci, 2008). Thus, to confirm the effect of ovotransferrin on preventing bone resorption, expression of OPG was also determined by western blot. As shown in Figure 3.7, the expression of OPG was increasing in a dose-dependent manner, showing a 1.5-fold increase at 1000 μ g/mL ovotransferrin treatment, compared with the untreated group. These results suggested that ovotransferrin could also promote the expression of OPG, which further supported the protective role of ovotransferrin on preventing osteoclast-induced bone resorption.

3.4 Discussion

Although a potential positive role of high-quality food protein intake on bone health has been suggested (Shams-White et al., 2017), effect of egg proteins has not been studied. In addition to their excellent essential amino acid profile and high digestibility, egg proteins are known to impart a wide range of physiological benefits including antimicrobial, anticancer, antioxidant and immunomodulation activities (Mine, Li-Chan. E., & Jiang, 2010). In this study, we showed for the first time that ovotransferrin could directly stimulate osteoblasts proliferation and differentiation as an osteogenic agent. In addition, ovotransferrin also contributes to inhibit osteoclastogenesis. The effects of ovotransferrin on both proliferation and differentiation of osteoblasts are profound,

being comparable to lactoferrin, a well-accepted nutraceutical for bone formation. Lactoferrin, extracted from milk belonging to transferrin family, is one of the most well studied bioactive proteins that acts as a promoter of osteoblasts growth and increases calvarial formation in vivo (Naot et al., 2005). Some reports claimed that the effects of lactoferrin on proliferation and survival of osteoblasts even exceed other established anabolic factors such as transforming growth factorβ, parathyroid hormone (PTH), amylin, and insulin (Cornish et al., 1999). In this study, we showed that ovotransferrin promoted osteoblast proliferation in a dose-depend manner, and reached significance at concentrations of 100 and 1000 μ g/mL; in comparison, lactoferrin showed the same trend and the same magnitude (Cornish et al., 2004). In general, transferrin family is recognized to have stimulatory effects on proliferation. In addition to the stimulation effects on osteoblasts, lactoferrin was also reported to induce the growth and proliferation of human enterocytes (Buccigrossi et al., 2007), human endometrial stromal cells and human lymphoblastic T cells (Bi, Lefebvre, Duś, Spik, & Mazurier, 1997). Laskey et al. (1988) gave the evidence that transferrin supported cell proliferation by supplying iron for DNA synthesis (Laskey, Webb, Schulman, & Ponka, 1988). And then transferrin has been studied in different cell lines and showed simulative effects on lymphoblast cells (Laskey et al., 1988), osteoblast cells (Tsunoi et al., 1984), and oligodendrocytes progenitor cells (Silvestroff, Franco, & Pasquini, 2013) with different magnitude, while showed suppression effects on human hepatoma cells (A. W. M. Lee, Oates, & Trinder, 2003). Richie et al. (1987) reported that human milk-derived lactoferrin inhibited the mitogen and alloantigen induced human lymphocytes proliferation (Richie, Hilliard, Gilmore, & Gillespie, 1987). Zhang, Lima, & Rodrigues (2015) also described the potential anticancer activity of lactoferrin by inhibiting the growth of breast cancer cells (Zhang, Lima, & Rodrigues, 2015). Similarly, ovotransferrin has been successfully used in inhibiting the proliferative response of mouse spleen lymphocytes stimulated by lipopolysaccharide and phytohemagglutinin (Otani & Odashima, 1997). Mizunoya et al. (2015) studied the growth promoting activity of several egg white proteins in C2C12 myoblast cell line and observed that ovalbumin and ovomucoid could stimulate the proliferation of myoblasts while ovotransferrin decreased cell proliferation (Mizunoya, Tashima, Sato, Tatsumi, & Ikeuchi, 2015). Therefore, it seems that the effect of ovotransferrin on cell proliferation as well as the sensitivity of the cells to respond to ovotransferrin vary depending on the cell types.

Cell growth to high density often reflects loss of cell cycle control (Smith et al., 2000). Thus, cell cycle analysis was performed to investigate the effect of ovotransferrin on cell cycle control. Treatment of osteoblasts with ovotransferrin showed controlled cell cycle, stimulated more cells at the S phase but less at G0/G1 phase, indicating active DNA synthesis and cell proliferation, which is consistent with BrDU incorporation results. In addition to osteoblasts, controlled cell cycle also occurs in adipocytes (Richon, Lyle, & McGehee, 1997; Yeh, Bierert, & Mcknightt, 1995), as well as mesenchymal cells (Hall & Miyake, 2000). Cell cycle progression has been described as an important mechanism contributing to bone cell activity. Glucocorticoids, a widely used immunosuppressive and anti-inflammatory drug, has been reported to inhibit bone formation by inhibiting cell cycle progression in osteoblasts and finally lead to osteoporosis and increased fracture risk in chronic glucocorticoid-treated patients (Smith et al., 2000; Smith, Coetzee, & Frenkel, 2002). San Martin et al. (2009) proposed that deregulation of cell cycle may contribute to the pathogenesis of osteosarcoma. Several osteogenic compounds have also been reported to stimulate cell proliferation through affecting cell cycle (San Martin et al., 2009). Cao et al. (2012) reported that icarii, a prenyl flavonoid from Epimedium considered as a strong candidate used in bone tissue engineering, increased the percentage of S phase cell and decreased G1 phase in MC3T3-E1 cells (Cao et al., 2012). Polygonum amplexicaule var. sinense, a traditional herb, also showed simulative effects on osteoblasts by affecting cell cycle progression (Xiang, Su, Hu, & Yan, 2011).

The osteoinductive activity of ovotransferrin is also showed by its capacity to substantially promote osteoblast differentiation. The progress of bone formation was classified into three stages: preosteoblast recruitment characterized by cell proliferation and type I collagen secretion, followed by osteoblast precursor differentiation into postmitotic matrix-producing osteoblasts accompanied with downregulated replication, and osteoid mineralization by terminally differentiated osteoblasts associated with increased extracellular matrix accumulation and ALP expression and activity (Lee et al., 2004; Owen et al., 1990). As the most abundant protein among all bone matrixes, type I collagen is conductive to mineral deposition and also binds to noncollagenous matrix proteins to initiate and regulate mineralization (Ehara et al., 2003). Also, ALP is involved in bone matrix formation before crystallization of the calcium and phosphate ions (Lee et al., 2004) and may play a role in transport of phosphate (Millán & Whyte, 2016). Therefore, type I collagen and ALP are considered as the most important biomarkers to study osteoblastic activity. Lactoferrin was reported to enhance osteoblast differentiation by increasing the expression and activity of type I collagen and ALP in both mouse osteoblasts MC3T3-E1 and human osteoblastlike cells SaOS-2 (Cornish et al., 2004; Fritsche et al., 2001). Calcium phosphates such as αtricalcium phosphate (α -TCP) and tera-calcium phosphate (TetCP) have been clinically applied as components of bone-substitutes materials. Ehara et al. (2003) reported that α -TCP and TetCP stimulated MC3T3-E1 differentiation through increasing ALP activity and type I collagen expression (Ehara et al., 2003). In this study, ovotransferrin promoted both type I collagen and ALP expression, respectively, in the early and late phases, suggesting that, in addition to promoting matrixes production and osteoblasts differentiation in the early stage, it also affected late phase of calcification of bone matrix.

Proteins from transferrin family are suggested to play an important role in cell differentiation. Lactoferrin was reported to stimulate cell differentiation not only in osteoblasts, but also other cell lines such as stromal cells ST2 (Cornish et al., 2004), mesenchymal cells (Yagi et al., 2009), and human adipose stem cells (Ying et al., 2012). Macedo et al. (2004) studied the possible involvement of transferrin in T lymphocyte differentiation with hypotransferrinaemic mice and found transferrin itself or a pathway triggered by the interaction of transferrin with its receptor is crucial for normal early T-cell differentiation (Macedo et al., 2004). Paez et al. (2005) overexpressed human transferrin in two oligodendroglial cell lines and found transferrin is required for a more complete maturation of oligodendroglia (Paez, García, Campagnoni, Soto, & Pasquini, 2005).

As the most direct evidence of calcium deposit capability, mineralization was also investigated to evaluate the effect of ovotransferrin on bone formation. Ovotransferrin was found to increase bone-like tissue formation by MC3T3-E1 cells in a dose-dependent manner. This might be, in part, due to the increased cell activity and increased production of extra-cellular matrix like type I collagen or other non-collagenous proteins. The other possible reason for the enhanced formation of calcified tissue is that ovotransferrin may function as suppliers of original materials (e.g. amino acids) for bone formation. Ehara et al. (2003) also implied the possible function of α -TCP and

TetCP as calcium and phosphate materials for bone formation (Ehara et al., 2003). Above all, ovotransferrin acts to expand the pool of osteoblasts by exerting mitogenic effects as well as driving differentiation of precursors to produce more mature osteoblastic phenotypes capable of promoting bone matrix deposition and mineralization. This anabolic potency suggests that ovotransferrin or its analogs should be explored as therapies for osteoporosis that can restore skeletal strength because most current interventions merely arrest further structural decline.

Bone mass is regulated by continuous remodeling, which is based on a delicate balance between osteoblastic bone formation and osteoclastic bone resorption. The maturation and activation of osteoclasts is dependent on the RANK/RANKL/OPG system. Osteoblasts are the primary source of receptor activator of nuclear factor kB ligand (RANKL), which stimulates osteoclast differentiation and activates bone resorption after binding to RANK on osteoprogenitor cells (Rucci, 2008). Thus, the osteoblastic bone resorption depends on a delicate balance between the expression of RANKL and OPG. Ovotransferrin inhibited RANKL expression while promoted OPG expression in osteoblasts, indicating its inhibitory effect on bone resorption. Bone resorption prevention was also reported for milk protein (Lorget et al., 2002), soybean protein and soybean isoflavone (Messina & Messina, 2000). Lactoferrin reduced bone-resorbing activity was due to inhibited RANKL expression while promoted OPG expression (Lorget et al., 2002). Some drugs, like denosumab, showed prevention of bone resorption by inhibiting RANKL expression as well (Hamdy, 2008). Besides the regulation of RANK/RANKL/OPG system, ovotransferrin was shown anti-inflammatory activity (Xie et al., 2002). Given the inflammatory nature of osteoporotic condition, ovotransferrin may play a role in counterbalancing the inflammation-induced bone

resorption. The anti-inflammatory property of ovotransferrin may complement with its osteogenic effect on bone cells.

3.5 Conclusion

In conclusion, this study demonstrated for the first time that the egg white protein ovotransferrin could function as an osteogenic agent by stimulated proliferation and differentiation of osteoblasts as well as inhibited osteoclastogenesis. These findings suggested an important role of ovotransferrin in mediating bone remodeling and rebuilding the balance between bone formation and resorption. Although further research is warranted to test its *in vivo* efficacy, ovotransferrin shows the potential as a complementary therapeutic target for alleviation of bone disorders such as osteoporosis as well as an anabolic agent for promotion of bone repair.

3.6 References

- Andersen, C. J. (2015). Bioactive egg components and inflammation. *Nutrients*, 7(9), 7889–7913.
- Bi, B. Y., Lefebvre, A. M., Duś, D., Spik, G., & Mazurier, J. (1997). Effect of lactoferrin on proliferation and differentiation of the Jurkat human lymphoblastic T cell line. *Archivum Immunologiae et Therapiae Experimentalis*, 45(4), 315–320.
- Buccigrossi, V., De Marco, G., Bruzzese, E., Ombrato, L., Bracale, I., Polito, G., & Guarino, A. (2007). Lactoferrin induces concentration-dependent functional modulation of intestinal proliferation and differentiation. *Pediatric Research*, 61(4), 410–414.
- Cao, H., Ke, Y., Zhang, Y., Zhang, C. J., Qian, W., & Zhang, G. L. (2012). Icariin stimulates MC3T3-E1 cell proliferation and differentiation through up-regulation of bone morphogenetic protein-2. *International Journal of Molecular Medicine*, 29(3), 435–439.
- Chen, J. S., & Sambrook, P. N. (2012). Antiresorptive therapies for osteoporosis: a clinical overview. *Nature Reviews Endocrinology*, 8(2), 81–91.
- Cornish, J., Callon, K. E., Lin, C. Q., Xiao, C. L., Gamble, G. D., Cooper, G. J. S., & Reid, I. R. (1999). Comparison of the Effects of Calcitonin Gene-Related Peptide and Amylin on Osteoblasts. *Journal of Bone and Mineral Research*, 14(8), 1302–1309.
- Cornish, J., Callon, K. E., Naot, D., Palmano, K. P., Banovic, T., Bava, U., Reid, I. R. (2004). Lactoferrin is a potent regulator of bone cell activity and increases bone formation in vivo. *Endocrinology*, 145(9), 4366–4374.
- Ehara, A., Ogata, K., Imazato, S., Ebisu, S., Nakano, T., & Umakoshi, Y. (2003). Effects of αTCP and TetCP on MC3T3-E1 proliferation, differentiation and mineralization. *Biomaterials*, 24(5), 831–836.

- Fritsche, K. L., Setchell, K. D. R., Conneely, O. M., Beck, M. A., Prasad, K. N., Keen, C. L., & Klurfeld, D. M. (2001). Anti-inflammatory activities of lactoferrin. *Journal of the American College of Nutrition*, 20(5 SUPPL.), 396S–397S.
- Gentili, C., Doliana, R., Bet, P., Campanile, G., Colombatti, A., Cancedda, F. D., & Cancedda, R. (1994). Ovotransferrin and ovotransferrin receptor expression during chondrogenesis and endochondral bone formation in developing chick embryo. *The Journal of Cell Biology*, *124*(4), 579–588.
- Giansanti, F., Leboffe, L., Pitari, G., Ippoliti, R., & Antonini, G. (2012). Physiological roles of ovotransferrin. *Biochimica et Biophysica Acta (BBA) General Subjects*, *1820*(3), 218–225.
- Guo, H. Y., Jiang, L., Ibrahim, S. A., Zhang, L., Zhang, H., Zhang, M., & Ren, F. Z. (2009). Orally administered lactoferrin preserves bone mass and microarchitecture in ovariectomized rats. *J Nutr*, 139(C), 958–964.
- Hall, B. K., & Miyake, T. (2000). All for one and one for all: Condensations and the initiation of skeletal development. *BioEssays*, 22(2), 138–147.
- Hamdy, N. A. T. (2008). Denosumab: RANKL inhibition in the management of bone loss. *Drugs* of Today, 44(1), 7.
- Laskey, J., Webb, I., Schulman, H. M., & Ponka, P. (1988). Evidence that transferrin supports cell proliferation by supplying iron for DNA synthesis. *Experimental Cell Research*, 176(1), 87–95.
- Lee, A. W. M., Oates, P. S., & Trinder, D. (2003). Effects of cell proliferation on the uptake of transferrin-bound iron by human hepatoma cells. *Hepatology*, 38(4), 967–977.
- Lee, Y. K., Song, J., Lee, S. B., Kim, K. M., Choi, S. H., Kim, C. K., Kim, K. N. (2004). Proliferation, differentiation, and calcification of preosteoblast-like MC3T3-E1 cells

cultured onto noncryrstalline calcium phosphate glass. *Journal of Biomedical Materials Research - Part A*, 69(1).

- Liao, W., Chakrabarti, S., Davidge, S. T., & Wu, J. (2016). Modulatory effects of egg white ovotransferrin-derived tripeptide IRW (Ile-Arg-Trp) on vascular smooth muscle cells against Angiotensin II stimulation. *Journal of Agricultural and Food Chemistry*, 64(39), 7342–7347.
- Lorget, F., Clough, J., Oliveira, M., Daury, M. C., Sabokbar, A., & Offord, E. (2002). Lactoferrin reduces in vitro osteoclast differentiation and resorbing activity. *Biochemical and Biophysical Research Communications*, 296(2), 261–266.
- Macedo, M. F., De Sousa, M., Ned, R. M., Mascarenhas, C., Andrews, N. C., & Correia-Neves,
 M. (2004). Transferrin is required for early T-cell differentiation. *Immunology*, *112*(4), 543–549.
- Messina, M., & Messina, V. (2000). Soyfoods, soybean isoflavones, and bone health: a brief overview. Journal of Renal Nutrition : The Official Journal of the Council on Renal Nutrition of the National Kidney Foundation, 10(2), 63–68.
- Millán, J. L., & Whyte, M. P. (2016). Alkaline Phosphatase and Hypophosphatasia. *Calcified Tissue International*, 98(4), 398–416.
- Mine, Y., Li-Chan. E., & Jiang, B. (2010). Bioactive Proteins and Peptides as Functional Foods and Nutraceuticals : Yoshinori Mine : 9780813813110. (Y. Mine, Li-Chan. E., & B. Jiang, Eds.). Arnes, AI: Iowa State University Press.
- Mizunoya, W., Tashima, A., Sato, Y., Tatsumi, R., & Ikeuchi, Y. (2015). The growth-promoting activity of egg white proteins in the C2C12 myoblast cell line. *Animal Science Journal*, 86(2), 194–199.

- Naot, D., Grey, A., Reid, I. R., & Cornish, J. (2005). Lactoferrin--a novel bone growth factor. *Clinical Medicine & Research.*, *3*(2), 93–101.
- Otani, H., & Odashima, M. (1997). Inhibition of proliferative responses of mouse spleen lymphocytes by lacto - and ovotransferrins. *Food and Agricultural Immunology*, 9(3),

193–201.

- Owen, T. A., Aronow, M., Shalhoub, V., Barone, L. M., Wilming, L., Tassinari, M. S., Stein, G. S. (1990). Progressive development of the rat osteoblast phenotype in vitro reciprocal relationships in expression of genes associated with osteoblast proliferation and differentiation during formation of the bone extracellular-matrix. *Journal of Cellular Physiology*, *143*(3), 420–430.
- Paez, P. M., García, C. I., Campagnoni, A. T., Soto, E. F., & Pasquini, J. M. (2005).
 Overexpression of human transferrin in two oligodendroglial cell lines enhances their differentiation. *Glia*, 52(1), 1–15.
- Redlich, K., & Smolen, J. S. (2012). Inflammatory bone loss: pathogenesis and therapeutic intervention. *Nature Reviews Drug Discovery*, *11*(3), 234–250.
- Reinholz, G. G., Getz, B., Pederson, L., Sanders, E. S., Subramaniam, M., Ingle, J. N., & Spelsberg, T. C. (2000). Bisphosphonates directly regulate cell proliferation, differentiation, and gene expression in human osteoblasts. *Cancer Research*, 60(21), 6001–6007.
- Richie, E. R., Hilliard, J. K., Gilmore, R., & Gillespie, D. J. (1987). Human milk-derived lactoferrin inhibits mitogen and alloantigen induced human lymphocyte proliferation. *Journal of Reproductive Immunology*, 12(2), 137–148.
- Richon, V. M., Lyle, R. E., & McGehee, R. E. (1997). Regulation and expression of retinoblastoma proteins p107 and p130 during 3T3-L1 adipocyte differentiation. *Journal of*

Biological Chemistry, 272(15), 10117–10124.

- Rosen, C. J. (2004). What's new with PTH in osteoporosis: where are we and where are we headed? *Trends in Endocrinology & Metabolism*, 15(5), 229–233.
- Rucci, N. (2008). Molecular biology of bone remodelling. Clinical Cases in Mineral and Bone Metabolism : The Official Journal of the Italian Society of Osteoporosis, Mineral Metabolism, and Skeletal Diseases, 5, 49–56.
- Salari Sharif, P., Abdollahi, M., & Larijani, B. (2011). Current, new and future treatments of osteoporosis. *Rheumatology International*, *31*(3), 289–300.
- San Martin, I. A., Varela, N., Gaete, M., Villegas, K., Osorio, M., Tapia, J. C., Galindo, M.
 (2009). Impaired cell cycle regulation of the osteoblast-related heterodimeric transcription
 factor Runx2-Cbfβ in osteosarcoma cells. *Journal of Cellular Physiology*, 221(3), 560–571.
- Shams-White, M. M., Chung, M., Du, M., Fu, Z., Insogna, K. L., Karlsen, M. C., Weaver, C. M. (2017). Dietary protein and bone health: a systematic review and meta-analysis from the National Osteoporosis Foundation. *The American Journal of Clinical Nutrition*, 105(6), ajcn145110.
- Silvestroff, L., Franco, P. G., & Pasquini, J. M. (2013). Neural and Oligodendrocyte Progenitor Cells: Transferrin Effects on Cell Proliferation. *ASN Neuro*, *5*(1), AN20120075.
- Smith, E., Coetzee, G. A., & Frenkel, B. (2002). Glucocorticoids inhibit cell cycle progression in differentiating osteoblasts via glycogen synthase kinase-3β. *Journal of Biological Chemistry*, 277(20), 18191–18197.
- Smith, E., Redman, R. A., Logg, C. R., Coetzee, G. A., Kasahara, N., & Frenkel, B. (2000). Glucocorticoids inhibit developmental stage-specific osteoblast cell cycle: Dissociation of cyclin A-cyclin-dependent kinase 2 from E2F4-p130 complexes. *Journal of Biological*

Chemistry, 275(26), 19992–20001.

- Tsunoi, M., Hakeda, Y., Kurihara, N., Maeda, N., Utsum, N., & Kumegawa, M. (1984). Effect of transferrin on alkaline phosphatase activity and collagen synthesis in osteoblastic cell derived from newborn mouse calvaria. *Experimental Cell Research*, 153, 240–244.
- US Department of Health and Human Services. (2004). Bone health and osteoporosis: a report of the Surgeon General. In *US Health and Human Services*.
- WHO. (2003). Prevention and management of osteoporosis. World Health Organization technical report series (Vol. 921). Retrieved from http://www.ncbi.nlm.nih.gov/pubmed/15293701
- Xiang, M. X., Su, H. W., Hu, J., & Yan, Y. J. (2011). Stimulative effects of Polygonum amplexicaule var. sinense on osteoblastic MC3T3-E1 cells. *Pharmaceutical Biology*, 49(10), 1091–1096.
- Xie, H., Huff, G. R., Huff, W. E., Balog, J. M., & Rath, N. C. (2002). Effects of ovotransferrin on chicken macrophages and heterophil-granulocytes. *Developmental & Comparative Immunology*, 26(9), 805–815.
- Yagi, M., Suzuki, N., Takayama, T., Arisue, M., Kodama, T., Yoda, Y., Ito, K. (2009). Effects of lactoferrin on the differentiation of pluripotent mesenchymal cells. *Cell Biology International*, 33(3), 283–289.
- Yeh, W. C., Bierert, B. E., & Mcknightt, S. L. (1995). Rapamycin inhibits clonal expansion and adipogenic differentiation of 3T3-L1 cells. *Biochemistry*, 92(November), 11086–11090.
- Ying, X., Cheng, S., Wang, W., Lin, Z., Chen, Q., Zhang, W., Zhu Lu, C. (2012). Effect of lactoferrin on osteogenic differentiation of human adipose stem cells. *International Orthopaedics*, 36(3), 647–653.

Zhang, Y., Lima, C. F., & Rodrigues, L. R. (2015). In vitro evaluation of bovine lactoferrin potential as an anticancer agent. *International Dairy Journal*, 40, 6–15.

Figure 3.1 Effects of egg white ovotransferrin on cell proliferation in osteoblast cell MC3T3-

E1

MC3T3-E1 cells were seeded into 48-well plate for 4 h and treated with 1000 μ g/mL lactoferrin or different concentrations (1000, 100, 10, and 1 μ g/mL) of ovotransferrin for 24 h prior to adding BrDU reagent, fixed, permeabilized and immunostained. Cell nuclei were counter-stained with Hoechst33342 dye. Percentage of BrDU(+) nuclei were counted in 3 random fields per group and their mean value determined. Data were expressed as mean±SEM. *, ** and *** indicate p<0.05, p<0.01 and p<0.001 respectively; as compared to the untreated control group.



Figure 3.2 Effects of egg white ovotransferrin on cell cycle in osteoblast cell MC3T3-E1

Confluent monolayers of MC3T3-E1 cells were treated with different concentrations (1000, 100, 10, and 1 μ g/mL) of ovotransferrin for 24 h prior to being trypsinized, collected, fixed and subjected to PI flow cytometry. Representative panels are shown. Percentage of each cell cycle phase was measured. Data were expressed as mean±SEM. * indicates p<0.05 as compared to the untreated control group.



G0/G1

(B)

ŝ

G2/M

Figure 3.3 Effects of egg white ovotransferrin on type I collagen expression in osteoblast cell

MC3T3-E1

MC3T3-E1 cells were seeded into 48-well plate for 4 h and treated with 1000 μ g/mL lactoferrin or different concentrations (1000, 100, 10, and 1 μ g/mL) of ovotransferrin for 72 h prior to being fixed, permeabilized and immunostained for type I collagen. Cell nuclei were counter-stained with Hoechst33342 dye. Mean fluorescence intensity was measured as the average of total fluorescent intensity from 3 randomly selected fields per group and expressed as percentage of the untreated control. Data were expressed as mean±SEM. * indicates p<0.05 as compared to the untreated control group.





Figure 3.4 Effects of egg white ovotransferrin on ALP expression in osteoblast cell MC3T3-

E1

Confluent monolayers of MC3T3-E1 cells were treated with 1000 μ g/mL lactoferrin or different concentration (1000, 100, 10, and 1 μ g/mL) of ovotransferrin for 72 h prior to being lysed and immunoblotted for ALP. A representative set of immunoblots is shown. Fluorescence intensity of the immunoblots was measured. Data were expressed as mean±SEM. * and ** indicate p<0.05 and p<0.01 respectively; as compared to the untreated control group.



Figure 3.5 Effects of egg white ovotransferrin on mineralization in osteoblast cell MC3T3-

E1

Confluent monolayers of MC3T3-E1 cells were treated with different concentrations (1000, 100, 10, and 1 μ g/mL) of ovotransferrin with medium contain ascorbic acid and β -glycerophosphate for different days (5, 10, 15 20 days) prior to being stained with Alizarin Red S. Representative images of mineralization are shown. Absorbance at 562 nm was measured after cell distained with cetylpyridinium chloride in sodium phosphate. Data were expressed as mean±SEM. * indicates p<0.05 as compared to the untreated control group.



Figure 3.6 Effects of egg white ovotransferrin on RANKL expression in osteoblast cell MC3T3-E1

Confluent monolayers of MC3T3-E1 cells were treated with 1000 μ g/mL lactoferrin or different concentrations (1000, 100, 10, and 1 μ g/mL) of ovotransferrin for 72 h prior to being lysed and immunoblotted for RANKL. A representative set of immunoblots is shown. Data were expressed as mean±SEM. * indicates p<0.05 as compared to the untreated control group.



Figure 3.7 Effects of egg white ovotransferrin on OPG expression in osteoblast cell MC3T3-

E1

Confluent monolayers of MC3T3-E1 cells were treated with 1000 μ g/mL lactoferrin or different concentrations (1000, 100, 10, and 1 μ g/mL) of ovotransferrin for 72 h prior to being lysed and immunoblotted for OPG. A representative set of immunoblots is shown. Data were expressed as mean±SEM. * indicates p<0.05 as compared to the untreated control group.



CHAPTER 4 - Ovotransferrin Stimulated Osteogenesis via activation of ERK1/2 and PI3K-Akt pathways independent to low-density lipoprotein receptor-related protein 1 (LRP1)

4.1 Introduction

Bone is a highly dynamic organ whose homeostasis is maintained through continuously remodeling (Kini, Usha; Nandeesh, 2012; Wozney et al., 1988). Bone remodeling is a delicate process controlling the balance between bone formation and resorption to keep bone integrity and quality (Kini, Usha; Nandeesh, 2012; Shahi, Peymani, & Sahmani, 2017). The bone-forming osteoblasts and the bone resorbing osteoclasts are two major components involving in and mediating this process (Kini, Usha; Nandeesh, 2012; Shahi et al., 2017). Osteoporosis is the most serious bone disorder worldwide, which leads to a systemic impairment of bone mass, strength, micro-architecture and an increased risk of osteoporotic fractures (Bates, 2007). The development of osteoporosis is mainly due to an imbalance between osteoblast and osteoclast activity, as well as an uncoupling of bone resorption and formation, which finally results in a greater bone resorption than formation (Bates, 2007). Two strategies have been introduced to osteoporosis prevention by either inhibiting bone resorption or promoting bone formation (Crockett & Das, 2013). The antiresorptive agents have been widely used in osteoporosis treatment, but their low efficacy, associated side effects and concurrent comorbidities challenged the use of these drugs (Crockett & Das, 2013). Compare to various antiresorptive agents, parathyroid hormone (PTH) is the only available drug to stimulate the osteoblast-controlled bone formation currently (Crockett & Das, 2013; Morley, Whitfield, & Willick, 2001). However, the high cost and short-term efficacy limits the application of this anabolic agent (Morley et al., 2001; Ponnapakkam, Katikaneni,

Sakon, Stratford, & Gensure, 2014). Thus, it is desirable to identify and develop natural anabolic agents to enhance bone formation by stimulating osteoblast activity.

Several lines of evidence have shown that food proteins, especially bioactive proteins, may contribute to bone health management (Bharadwaj, Naidu, Betageri, Prasadarao, & Naidu, 2009; Cornish et al., 2004; Eaton-Evans, 1994; Marshall, 2004; Messina & Messina, 2000). Among all, milk-derived lactoferrin is one of the most well studied bioactive proteins that has been reported to be a potent bone growth factor both *in vitro* and *in vivo* (Cornish et al., 2004; Guo et al., 2009; Hou, Xue, & Lin, 2012). Our previous work reported a similar effect of egg-derived ovotransferrin as lactoferrin in stimulating osteoblastic activity (Shang & Wu, 2018). Ovotransferrin, a 78-kDa iron-binding glycoprotein, belonging to the same transferrin family as lactoferrin, shares similar structures and biological activity. Egg white ovotransferrin was demonstrated to promote osteoblast proliferation and to inhibit osteoclastogenesis *in vitro*. In particular, ovotransferrin exerted beneficial effect on osteoblast differentiation and extracellular matrix calcification. Osteoblast differentiation is a prerequisite of bone formation *in vivo*. However, the molecular mechanisms underlying the effect of ovotransferrin on osteoblast differentiation remain poorly understood.

Mitogen-activated protein kinases (MAPKs) are serine/threonine kinases that are well-known mediators converting extracellular stimuli into cellular responses, including many of the extracellular ligands relevant to osteoblasts (Cargnello & Roux, 2011). The mammalian MAPK family includes extracellular signal-regulated kinase (ERK), c-Jun NH2-terminal kinase (JNK) and p38 (Cargnello & Roux, 2011). ERK MAPK pathway was reported to be responsible for the

lactoferrin induced mitogenic effects (Grey et al., 2004), as well as the collagen peptide induced osteogenic activity in osteoblast (Kim et al., 2013). Also, ERK1/2 was described to be implicated in bone morphogenic protein 2 (BMP-2) (Jaiswal et al., 2000) and 1,25(OH)₂D₃-induced osteoblast differentiation (Chae et al., 2002). In skeletal biology, alternation in JNK activity is responsible for altered osteoblast activity (Yamashita et al., 2005). Treatment with a JNK inhibitor was reported to reduce mineralization and expression of mature osteoblast markers, while osteoblast differentiation was enhanced by overexpression of JNK2 (Matsuguchi et al., 2009a). Compare to ERK and JNK pathway, more evidence indicated the critical role of p38 MAPK pathway in osteoblast differentiation. The p38 MAPK was required for both natural (Greenblatt et al., 2010) and BMP-2 - induced differentiation in osteoblasts (Hu, Chan, Wang, & Li, 2003). Moreover, an increasing amount of research has shown that p38 MAPK is indispensable for osteoblast differentiation induced by various osteotropic factors, such as epinephrine (Suzuki, Palmer, Bonjour, & Caverzasio, 1999) and PTH (Rey, Manen, Rizzoli, Ferrari, & Caverzasio, 2007). In addition to MAPK signaling, the PI3K-Akt pathway plays a central role in the control of cell survival, growth, and proliferation (McGonnell et al., 2012). Both in vitro and in vivo studies also indicated that the phosphoinositide 3-kinase (PI3K)-protein kinase B (Akt) pathway is involved in regulation of osteoporosis through promoting osteoblast proliferation, differentiation and bone formation (Xi et al., 2015). More importantly, study reported that Akt is a molecular switch for transforming growth β -1 (TGF- β 1)-induced osteoblastic differentiation. TGF- β 1 could be either an inducer or an inhibitor of osteoblastic differentiation, which is determined by the state of Akt phosphorylation in MC3T3-E1 cells (Suzuki et al., 2014).

Critical cellular decisions such as proliferation, migration and differentiation, are regulated by stimulatory cues from the extracellular environment and convert to a cellular response through binding to membrane receptors that can trigger the signal transduction pathway inside the cells (Groves & Kuriyan, 2010). Both MAPK pathway and PI3K-Akt pathway can be activated by several receptors, such as RTKs (receptors tyrosine kinases) and G-protein-coupled receptors (Katz, Amit, & Yarden, 2007; Naor, Benard, & Seger, 2000). It is believe that most of the bioactive macromolecules exert effects though binding specific receptors, rather than directly transport into cell plasma. Lactoferrin receptors are thought to have pivotal roles in mediating multiple functions of lactoferrin. Studies indicated that lactoferrin promoted osteoblast proliferation through low-density lipoprotein receptor-related protein 1 (LRP1) - mediated ERK1/2 MAPK activation (Grey et al., 2004), whereas lactoferrin-induced osteoblast survival and differentiation is LRP1-independent (Grey, Zhu, Watson, Callon, & Cornish, 2006; W. Zhang et al., 2014).

In this chapter, the signaling pathways that transduce the stimulating signal of ovotransferrin during osteoblast differentiation were examined. Meanwhile, the involvement of possible ovotransferrin receptors, especially LRP1, in ovotransferrin-treated osteoblast was evaluated.

4.2 Material and Methods

4.2.1 Reagents and antibodies

Ovotransferrin (Conalbumin from chicken egg white, at a purity of 98%) and Alizarin-S red stain were bought from Sigma Chemicals Co. (St Louis, MO, USA). α-minimum essential medium (α-MEM) and Fetal Bovine Serum (FBS) were obtained from Gibco/Invitrogen (Carlsbad, CA, USA). Triton-X-100 was bought from VWR International (West Chester, PA, USA). CytoSelect wound healing assay kit was purchased from Cell Biolabs Inc., (San Diego, CA, USA). Collagen I alpha antibody was bought from Novus Biological Canada ULC (Oakville, ON, Canada). Mouse monoclonal primary antibody against ALP, β-catenin, BMP-2/4, PI3-kinase p85α, PI3-kinase p110a, phosphory-CREB-1 and CREB-1 were purchased from Santa Cruz Biotechnology (Mississauga, ON, Canada). Mouse monoclonal primary antibody against Runx2 and transferrin receptor, and rabbit monoclonal primary antibody against TFR1, ITLN1/2, LRP1, α-Tubulin and GAPDH were purchased from Abcam (Toronto, ON, Canada). Phospho-ERK1/2 pathway sampler kit and rabbit monoclonal primary antibody against phosphor-JNK (ser73), JNK, phosphor-p38 MAPK (Thr180/Tyr182) and p38 were purchased from Cell Signaling Technology (Whitby, ON, Canada). Rabbit monoclonal primary antibody against phosphor-Akt (Ser473) and Akt were purchased from Cell Signaling Technology (Danvers, MA, USA). EKR1/2 antagonist U0126, JNK antagonist SP600125, p38 antagonist SB203580, Akt antagonist API-1 and LRP1 antagonist Rap were purchased from Tocris (Oakville, ON, Canada). Hoechst 33342 (Trihydrochloride), rabbit anti-mouse IgG (H+L) secondary antibody, and goat anti-rabbit IgG (H+L) secondary antibody were bought from Molecular Probes (Waltham, MA, USA). Goat anti-rabbit and donkey antimouse fluorochrome-conjugated secondary antibodies were purchased from Licor Biosciences (Lincoln, NB, USA).

4.2.2 Cell Culture

The murine pre-osteoblastic cell line MC3T3-E1 (subclone 4, ATCC CRL-2593) was purchased from ATCC (Manassas, VA, USA). Cells were cultured under the same condition described in Chapter 3. Briefly, cells were cultured in α -MEM supplemented with 10% FBS and penicillinstreptomycin in an incubator under 95% air and 5% CO₂ at 37°C. All experiments were performed on 80-90% confluent cells grown in tissue culture grade plastic 48 or 24 well plates. To investigate the involvement of the signaling pathways in ovotransferrin-induced osteoblastic activity, cells were incubated with ovotransferrin in different conditions prior to western blotting, immunofluorescence and mineralization assay as described below.

4.2.3 Western blot

The cells were seeded on 48 well tissue culture plates at a concentration of 1×10^4 cells/well counted on a haemocytometer, and incubated in α -MEM with 10% FBS. After treatment, the culture medium was removed and the cells lysed in boiling hot Laemmle's buffer containing 50 μ M dithiothreitol (DTT) and 0.2% Triton-X-100 to prepare samples for western blot as described previously (Shang & Wu, 2018). These cell lysates were run in SDS-PAGE, blotted to nitrocellulose membranes and immunoblotted with primary antibodies. The protein bands were detected by a Licor Odyssey BioImager and quantified by densitometry using the corresponding software (Licor Biosciences, Lincoln, NB, US). Each target band was normalized to the corresponding band of loading control. Cell lysates from untreated cells were loaded onto every gel. The results were expressed as percentage of the corresponding untreated control.

4.2.4 Immunofluorescence

The immunofluorescence studies were performed similar to our previous studies (Shang & Wu, 2018). Briefly, cells were fixed in 4% paraformaldehyde, permeabilized with 0.1% Triton-X-100 in PBS and immunostained overnight with a rabbit polyclonal antibody against type I collagen. Cells were treated with Alexa Fluor546 (red) conjugated goat anti-rabbit secondary antibody for 30 min in the dark. Nuclei were stained with the Hoechst33342 nuclear dye (1:10000) for 10 min. After washing with PBS to remove unbound antibody/dye, the immunostained cells were observed under an Olympus IX81 fluorescence microscope (Olympus, Tokyo, Japan). Images were obtained using Metamorph imaging software (Molecular Devices, Sunnyvale, CA, USA). Mean

fluorescence intensity was calculated from the intensity of the red fluorescent signal (determined by Adobe Photoshop Elements 2.0 software; Adobe Systems Inc., San Jose, CA, USA) from 3 randomly selected fields per group.

4.2.5 Mineralization Assay

The mineralization studies were performed similar to previous studies (Shang & Wu, 2018). The degree of mineralization was determined in the 12-well plates using Alizarin Red staining. Cells were incubated with different concentrations of ovotransferrin for 15 days. The medium was removed and cells were rinsed twice with PBS. The cells were fixed with ice-cold 70% (v/v) ethanol for 1 h. The ethanol was removed by aspiration and cells were washed twice with Milli-Q water. The cells were stained with 1% (w/v) Alizarin-S Red in Milli-Q (pH 4.2) for 10 min at room temperature. After washing with Milli-Q water, the samples were observed under light microscopy and photographed.

4.2.6 Cell Migration Assay

Migration of osteoblast MC3T3-E1 cells were evaluated by a CytoSelect wound healing assay kit (Cell Biolabs, Inc., San Diego, CA, USA) according to the manual. Briefly, cells were seeded into a 24-well plate with a wound healing insert in the middle of each well. Cells were cultured overnight until the monolayer formed. The wound healing insert was removed to create a 0.9 mm wound healing area. Cells were pre-treated with 50 µM Akt inhibitor for 1h and then treated with 1000 µg/mL ovotransferrin and Akt inhibitor together for 12 h. Afterward, cells were fixed with 4% of paraformaldehyde solution and cell nuclei were stained with Hoechst33342 nuclear dye (1:10000) for 10 min. Cells were visualized by an Olympus IX83 fluorescence microscope (Olympus, Shinjuku, Tokyo, Japan). The wound healing area was measured by Image J software.

A smaller area indicated a higher migration ability of the cells. Results of each treatment group were normalized to the corresponding untreated group.

4.2.7 Statistics

All data are presented as mean±SEM (standard error of mean) of 6 independent experiments. Data were analyzed using One Way analysis of variance (ANOVA) with Dunnett's post-hoc test for comparisons to control. The PRISM 6 statistical software (GraphPad Software, San Diego, CA) was used for the analyses. P<0.05 was considered significant.

4.3 Results

4.3.1 Ovotransferrin stimulates osteogenesis associated transcription factors

To investigate the mechanism underlying the ovotransferrin stimulated osteogenesis, we first evaluated several transcription factors that play vital role in regulating osteoblastic activity including osteoblast differentiation and mineralization. In general, ovotransferrin does-dependently affected the expression of Runx2, BMP-2, β -catenin and phosphorylation of CREB on osteoblast MC3T3-E1 cells (Figure 4.1A, B, C, D, E). The strongest stimulation on the expression of BMP-2 (~1.5 folds) and β -catenin (~2.5 folds) was found at the concentration of 1000 µg/mL, while the expression of Runx2 (~2 folds) and phosphorylation of CREB (~2.5 folds) increased more at the concentration of 100 µg/mL.

4.3.2 Ovotransferrin stimulates osteogenesis via activate MAPK pathways

Previous studies demonstrated that the MAPK pathways are involved in the regulation of osteoblast differentiation and osteogenesis. Thus, we investigated the effect of ovotransferrin on three MAPK members, including ERK, JNK and p38. Our results showed that ovotransferrin
treatment significantly activated all three MAPK pathways (Figure 4.2). ERK1/2 phosphorylation was significantly increased during the time treated but the highest expression was observed at 20-30 min and then started to decline (though remained in a high level of phosphorylation); once the old medium was replaced by new medium (containing ovotransferrin), the phosphorylation of ERK1/2 was dramatically increased again, which suggesting that ovotransferrin could continuously induce the activation of ERK1/2 pathway (Figure 4.2A). Compared with ERK1/2, ovotransferrin only increased the phosphorylation of JNK at the initial 20 to 30 min period and then declined to basal level; replenishment with new medium (containing ovotransferrin) after 12 h did not change the level of phosphorylation of JNK (Figure 4.2B). The stimulation of p38 phosphorylation was time-response, increased at prolonged incubation and peaked during 30 min to 1 h and then stated to decrease afterwards.(Figure 4.2C); replenishment with new medium (containing ovotransferrin) at 12 h did not increase further phosphorylation of p38. These results suggest that ovotransferrin may activate JNK and p38 pathways in an acute manner, while can continuously activate ERK1/2 pathway thus exhibiting long-term effect in regulating osteogenesis.

4.3.3 ERK pathway is involved in ovotransferrin-stimulated osteogenesis, but no JNK and p38 pathways

To further validate the involvement of MAPK pathways in ovotransferrin-stimulated osteogenesis, different antagonists were used to inhibit the activation of its corresponding pathways. Three different antagonists were used to inhibit ERK, JNK and p38 pathways, at the concentrations not toxic to cells (Figure 4.7.1). Several biomarkers were used to indicate the effects of ovotransferrin-stimulated osteogenesis, including the expression of ALP, Runx2, β -catenin, BMP-2, type I collagen, and the mineralization level. ERK1/2 antagonist alone showed no significant influence

on the expression of the biomarkers (Figure 4.3), while it significantly attenuated the increase of ovotransferrin-induced expression of ALP (Figure 4.3A), Runx2 (Figure 4.3B), β -catenin (Figure 4.3C), BMP-2 (Figure 4.3D), and type I collagen (Figure 4.3E). More importantly, ERK1/2 antagonist obviously attenuated the stimulation of mineralization induced by ovotransferrin treatment (Figure 4.3F), which provided direct evidence that the ERK pathway plays an important role in ovotransferrin-regulated osteogenesis.

However, JNK antagonist did not show such significant effects (Figure 4.4). Pre-treatment with antagonist of JNK did not attenuate the high expression of neither ALP (Figure 4.4A) nor Runx2 (Figure 4.4B). For p38 pathway, p38 antagonist significantly decreased ovotransferrin-induced expression of ALP (Figure 4.5A) and Runx2 (Figure 4.5B); however, this decrease might be due to the inhibition of basal expression level as the treatment with p38 antagonist alone showed a significant inhibition in ALP and Runx2 expression.

4.3.4 Ovotransferrin stimulates the phosphorylation and activation of the ERK cascade

ERK1/2 cascade acts downstream of Ras (small GTP-binding protein) and usually involves sequential phosphorylation and activation of Raf, MEK1/2 (MAPK/ERK kinases), ERK1/2 and MAPKAPKs (MAPK-activated protein kinases) (Wortzel & Seger, 2011). Although ERK1/2 are considered as the core components, MAPKAPKs components (such as p90RSK and MSK1/2) phosphorylate a large number of substrates that eventually regulate most cellular processes, including proliferation, differentiation, survival, apoptosis and more (Wortzel & Seger, 2011). Thus, the phosphorylation of several ERK1/2 upstream and downstream kinases was investigated to evaluate the role of ovotransferrin in activating the ERK cascade. Ovotransferrin at concentrations of 100 and 1000 μ g/mL significantly stimulated the phosphorylation of ERK

upstream kinase c-Raf (Figure 4.6A) and MEK1/2 (Figure 4.6B). Meanwhile, the phosphorylation of downstream MAPKAPK p90RSK (Figure 4.6C) and MSK1 (Figure 4.6D) were also increased by ovotransferrin. The activation of p90RSK and MSK1 may further regulate the transcription factors like Runx2, β-catenin, and CREB, therefore stimulating osteogenesis.

4.3.5 Ovotransferrin stimulates osteogenesis via activate PI3K-Akt pathways

In addition to MAPK pathways, activation of PI3K-Akt pathway by many molecules is known to play a vital role in osteogenesis was studied. Here, the effects of ovotransferrin in activating PI3K-Akt pathway. The phosphorylation of Akt was dramatically increased within the initial 6 h, and then declined to basal level until new medium was replaced at 24 h (Figure 4.7A). As the downstream signaling of PI3K, the increased phosphorylation of Akt might be due to the activation of PI3K. The current study found that 1000 μ g/mL ovotransferrin stimulated the expression of two PI3K subunits, p85 and p110 (Figure 4.7B). These results showed that ovotransferrin could activate PI3K-Akt pathway in osteoblast cells and contribute to osteogenesis.

To further confirm the involvement of the PI3K-Akt pathway in ovotransferrin stimulated osteoblastic activity, Akt antagonist was used to evaluate the expression of Runx2 and BMP-2 while the Akt pathway was inhibited. This study showed that Akt antagonist significantly attenuated ovotransferrin induced expression of Runx2 (Figure 4.8A) and BMP-2 (Figure 4.8B). The PI3K-Akt pathway has been found to play an important role in regulating osteoblast migration via mediating Runx2 and BMP-2 expression; thus, the cell migration assay was performed to evaluate the influence of Akt inhibition on ovotransferrin promoted cell migration. Ovotransferrin treatment significantly reduced the area of migration as well as the distance of the migration area, while pre-treatment with Akt antagonist before ovotransferrin significant diminished the positive

effects (Figure 4.8C). The area of migration and the distance of the migration area were significantly larger in Akt antagonist added group than ovotransferrin treatment alone. Finally, the role of PI3K-Akt pathway on ovotransferrin-induced osteogenesis was confirmed directly by mineralization assay. Reduced mineralization was observed when the cells treated with both Akt antagonist and ovotransferrin, compared with cell treated only with ovotransferrin (Figure 4.8E).

4.3.6 LRP1 is not involved in mediating the osteoinductive effect of ovotransferrin

Although our previous results revealed the involvement of ERK MAPK and PI3K-Akt pathway in ovotransferrin-induced osteogenesis, the receptors that respond to ovotransferrin and activate the corresponding signaling still remain unclear. Thus, in this study, we investigated the receptors that may act in response to ovotransferrin. According to the literatures, 7 different receptors that respond to ovotransferrin, transferrin family and lactoferrin, were selected for study (Table 4.7.1). PCR results showed that only TFBP (transferrin binding protein), TFR1 (transferrin receptor 1), ITLN1 (intelectin-1) and LRP1 (low-density lipoprotein receptor-related protein 1) were expressed in osteoblast MC3T3-E1 (Figure 4.7.2). The expression of TFR1, ITLN1 and LRP1 was further confirmed by western blot (Figure 4.9). Cells treated with ovotransferrin significant increase in the expression of ITLN1 and LRP1 (Figure 4.9), suggesting the response of these 2 receptors to ovotransferrin.

To further validate the participation of LRP1 in the osteoinductive effects of ovotransferrin, a natural LRP1 antagonist Rap was used to block the binding of all known ligand to receptor (Herz & Strickland, 2001; Zhang et al., 2014). Rap was added before ovotransferrin treatment to block the interaction between ovotransferrin and LRP1. With Rap pretreatment, ovotransferrin-induced high expression of ALP (Figure 4.10A) was inhibited, while the expression of Runx2 (Figure

4.10B) and phosphorylation of ERK1/2 (Figure 4.10C) and Akt (Figure 4.10D) were not affected. Interestingly, the mineralization assay showed that Rap pre-treatment inhibited ovotransferrininduced mineralization, at least partly (Figure 4.10E). These results suggested the involvement of LRP1 in response to ovotransferrin stimulated osteoblastic activity, but not via the activation of ERK and Akt pathways.

4.4 Discussion

Ovotransferrin has been recognized as a new anabolic factor that potently promotes bone formation and inhibits bone resorption (Shang & Wu, 2018). In this chapter, we showed that ovotransferrin stimulated osteoblast differentiation mainly through LRP1-independent ERK1/2 MAPK and PI3K-Akt pathways. This work, to the best of our knowledge, is the first evidence of the signaling mechanism underlying the beneficial effect of ovotransferrin on osteoblast differentiation.

Osteoblast differentiation is tightly controlled by several master regulators, such as Runx2, Wnt/βcatenin and BMPs (Qi et al., 2003). In this study, ovotransferrin significantly increased the expression of Runx2, BMP-2, β-catenin and the phosphorylation of CREB, suggesting that the stimulatory effect of ovotransferrin on osteoblast differentiation is through these master regulators (Figure 4.1). Runx2 is the most extensively studied master regulator in osteogenesis (Rutkovskiy, Stensløkken, & Vaage, 2016). It activates the genetic pathways controlling osteoblast differentiation as well as the expression of osteopontin, bone sialoprotein, osteocalcin, osteoprotegerin, RANKL, and many others (Li et al., 2012; Rutkovskiy et al., 2016). BMPs are another key molecule in the regulation of osteoblast differentiation as well as act as strong inducers of bone formation (Bae, Kim, Won, Min, & Hwang, 2017; Chen, Deng, & Li, 2012). In particular, BMPs activate the phosphorylation and translocation of specific proteins to increase the transcription of Runx2, osteocalcin, type I collagen, and ALP (Bae et al., 2017; Yamaguchi, Komori, & Suda, 2000). BMP signaling also activates β -catenin, a signal transducer of Wnt/Notch system, to induce osteoblast differentiation (Bain, Müller, Wang, & Papkoff, 2003). Normally, cytoplasm β -catenin translocates to the nucleus and activates by binding to its cofactors and results in a subsequent regulation of Runx2, as well as other pro-osteogenic genes (Takada, Kouzmenko, & Kato, 2009). Upon BMP stimulation, more β -catenin accumulates and translocates to the nucleus, and accelerate osteoblast differentiation (Bae et al., 2017). Furthermore, BMPs promote Runx2 phosphorylation and bind to coactivator CREB to induce the osteogenic differentiation (Lee, Hong, & Bae, 2002), at least three CREB proteins (CREB1, CREB 3, and CREB 5) have been found Runx2-responsive (Teplyuk et al., 2008).

Mitogen-activated protein kinase (MAPK) pathways provide a major link between the external environment and nucleus to control cell proliferation and differentiation in different cell types, including osteoblasts. The critical role of ERK MAPK pathway in osteoblast differentiation and mineralization has been well established *in vivo*. Mice with a deletion of *Erk1* and *Erk2* in osteoblasts display substantially reduced bone mineralization, demonstrating the importance of ERK for osteoblast mineralization (Matsushita et al., 2009). More importantly, the activation of ERK pathway has been suggested to be involved in several potent bone formation anabolites, such as drug PTH (Chen et al., 2004; Swarthout, Doggett, Lemker, & Partridge, 2001), and natural functional components lactoferrin (Grey et al., 2004; Liu et al., 2018) and isoflavone (Bhargavan et al., 2009). In this study, ovotransferrin treatment significant increased the phosphorylation of ERK and other important components involved in ERK signaling pathway, such as c-Raf,

MEK1/2, p90RSK and MSK1 (Figure 4.2 and Figure 4.6). Meanwhile, blocking ERK1/2 with its antagonist U0126 inhibited ovotransferrin-induced increase of bone formation biomarkers (ALP, Runx2, BMP-2, β -catenin and type I collagen) and mineralization. These findings are consistent with other studies using different cell types. A decrease of ALP activity was reported when ERK activity was inhibited in human osteoblasts HOB and bone marrow stromal cells (Lai et al., 2001). Meanwhile, the inhibition of ERK also decreased the matrix mineralization by regulating the bone matrix proteins, such as type I collagen, osteopontin, and bone sialoprotein (Lai et al., 2001). ERK inhibition impaired expression and activities of matrix proteins due to the regulation of transcription factors, especially Runx2 (Kanno, Takahashi, Tsujisawa, Ariyoshi, & Nishihara, 2007; G Xiao et al., 2000). Recently, MEK, the up-stream effector of MAPK, has been shown to regulate the activity of Runx2 (Xiao et al., 2000). Stimulation of MAPK by constitutively active MEK enhances, whereas the MEK inhibitor PD98059 inhibits, the expression of osteocalcin (Shimo et al., 2007). Notably, studies also suggested that active ERK may bind to the C-terminal proline/serine/threonine-rich (PST) domain of Runx2 and contribute to Runx2 transcriptional activity (Ge et al., 2012; Guozhi Xiao, Jiang, Gopalakrishnan, & Franceschi, 2002). Moreover, ERK has been found to bind to the promoter of osteocalcin and bone sialoprotein in osteoblast via the association between ERK and Runx2, intriguing the possibility that ERK activity may be differentially regulated in the context of specific osteoblast-relevant promoters (Ge et al., 2012; Guozhi Xiao et al., 2002).

In addition to ERK signaling, our study found that the phosphorylation of JNK was also increased by ovotransferrin treatment; however, blocking JNK with inhibitor SP600125 did not affect ovotransferrin-induced expression of ALP and Runx2 (Figure 4.4). Currently, the regulatory role of JNK pathway in osteoblast differentiation remains unclear. Treatment with JNK inhibitor was reported to reduce the expression of osteocalcin and bone sialoprotein, two biomarkers of laststage of differentiation (Matsuguchi et al., 2009b). ALP, however, is considered as an early-stage biomarker; thus, adding JNK inhibitor did not affect the expression of ALP as JNK only functions at late-stage. Furthermore, Runx2 may not be the target in response to JNK pathway, although Runx2 is considered as the most important factor required in osteoblast differentiation and bone formation. Several other transcription factors also play indispensable role in osteoblast differentiation, such as AP-1 family members (JunB and Fra-1) (Eferl et al., 2004; Kenner et al., 2004) and CREB/ATF family member ATF4 (Yang et al., 2004). For example, inactivation of JNK did not inhibit the expression of Runx2, but significantly inhibited the induction of ATF4, (Matsuguchi et al., 2009a). Additionally, activation of JNK kinase cascade was also found to mobilize the AP-1/ATF transcription factors and regulate osteoblastic activity (Yamashita et al., 2005). Based on these, some late-stage biomarkers and transcription factors in addition to ALP and Runx2 should be tested to better understand the involvement of JNK pathway in response to ovotransferrin stimulation.

p38 MAPKs have been shown to play an important role in osteoblast commitment and maturation as well as proliferation (Greenblatt et al., 2010; Rodríguez-Carballo et al., 2014; Thouverey & Caverzasio, 2012). Studies have found that p38 MAPK was activated as the calvarial osteoblast differentiates along with ERKs (Hu et al., 2003). When p38 was inhibited, the expression of ALP and the mineral deposition were significantly reduced (Hu et al., 2003). Meanwhile, osteoblast MC3T3-E1 expressing dominant-negative p38 MAPK also displayed sign of delay in mineralization (Hu et al., 2003). Similar results were observed in our study. The p38 inhibitor SB203580 significantly inhibited the basal expression of ALP and Runx2, indicating the importance of p38 in osteoblast differentiation (Figure 4.5). Interestingly, in this study, we found that ovotransferrin significantly increased the phosphorylation of p38 within 24 h, suggesting the stimulatory role of ovotransferrin in activating p38 signaling. Adding p38 inhibitor together with ovotransferrin reduced the expression of ALP and Runx2 compared with ovotransferrin alone, but remained significantly higher than control (Figure 4.5). These results suggested that ovotransferrin might affect the expression of ALP and Runx2 via pathways other than p38 MAPK pathway, such as the BMP pathway.

The PI3K-Akt signaling pathway plays a central role in the control of cell survival, growth, and proliferation throughout the body. With regard to bone, there is an increasing amount of evidence suggesting the potential osteoblast-specific effect of PI3K-Akt pathway. The Akt knockout animals showed impairment of skeletal development and functional integrity, indicating the essential role of Akt activation in bone health (Easton et al., 2005; Garofalo et al., 2003; Kawamura et al., 2007; Peng et al., 2003). Recent evidence suggests that Runx2 functions are not only regulated by phosphorylation of ERK and p38 MAPK, but also affected by PI3K-Akt signaling. The master factor Runx2 was reduced in Akt-1 knockout mice, indicating that the Runx2 might be a downstream target of Akt pathway (Kawamura et al., 2007). However, studies further informed that Runx2-Akt interactions are unlikely to occur via direct phosphorylation because Runx2 lacks an Akt consensus phosphorylation site (Kawamura et al., 2007). Thus, it is believed that Akt may regulate Runx2 indirectly via affecting the activity of GSK3β, an inhibitor of Runx2 DNA binding ability (Kugimiya et al., 2007), or FoxO1, a repressor of Runx2-dependent osteocalcin transcription (Yang et al., 2011; Zhang et al., 2011). In our study, ovotransferrin significantly

stimulated the phosphorylation of Akt and resulted in an increase of Runx2 expression, BMP-2 expression, osteoblast migration activity and mineralization (Figure 4.7). More importantly, these enhancements induced by ovotransferrin were attenuated after blocking with Akt inhibitor API-1, indicating a key role of Akt activation in ovotransferrin-induced osteogenesis (Figure 4.8). As activation of Akt is always considered via PI3K, our study further investigated the effect of ovotransferrin on PI3K expression. PI3K itself consist of a regulatory p85 subunit and a catalytic p110 subunit (McGonnell et al., 2012). The recruitment and phosphorylation of p85 and p110 subunit is required for activation of Akt (McGonnell et al., 2012). In our study, ovotransferrin significantly increased the expression of both p85 subunit and p110 subunit of PI3K, indicating a key role of activation of PI3K-Akt signaling in response to ovotransferrin treatment.

There are studies reported the ability of ovotransferrin in regulating different signaling pathways, such as MAPK pathways (Lee, Ahn, & Paik, 2018) and NF-κB pathway (Giansanti et al., 2002), to exert the immune-enhancing activity and anti-inflammatory activity. However, to our best knowledge, there is no study published to elucidate the possible receptors involved in response to ovotransferrin. As ovotransferrin is a macromolecule, we believed that the most likely way to trigger signal transduction by ovotransferrin is through binding to membrane receptors. Thus, based on our findings that ovotransferrin exert osteogenesis activity by activating the MAPK and PI3K-Akt pathways, we tentatively investigated the possible receptors that may be involved in eliciting the signal transduction. In this study, 7 possible receptors were selected based on their ability to bind overall transferrin family members and lactoferrin-specific binding receptors, including ovotransferrin receptor TFBP (Gentili et al., 1993; 1994), transferrin family receptors TFR1 (Kawabata, 2018), TFR2 (Kawabata, 2018), Culilin/LRP2, IGFBP-2 (Miljuš et al., 2015;

Weinzimer et al., 2001) and lactoferrin receptors ITLN1 (Akiyama et al., 2013; Shin et al., 2008) and LRP1 (Grey et al., 2004, 2006). Currently, TFBP is the only ovotransferrin receptor reported. Our study showed expression of TFBP was increased by ovotransferrin; a previous study also found that the expression of ovotransferrin receptor was increased during chondrogenesis and endochondral bone formation in developing chick embryo (Gentili et al., 1994). Unfortunately, there is no commercial TFBP antibody or inhibitor available currently (Figure 4.9); thus it is difficult to investigate relation the role of TFBP on ovotransferrin-induced osteoblast differentiation. Our study found that the expression of ITLN1 and LRP1 was significantly increased, while the expression of TFR1 did not change (Figure 4.9). Intelectin, also called small intestine lactoferrin receptor, is the well documented lactoferrin receptor that specifically recognizes lactoferrin in human small intestine (Suzuki, Lopez, & Lönnerdal, 2005). However, there is no evidence available for the functions of intelectin in osteoblast activity to date. LRP1 is the only known lactoferrin receptor that mediates the action of lactoferrin in osteoblasts (Naot et al., 2011). Studies have shown that lactoferrin stimulated osteoblast proliferation through LRP1meidated activation of ERK pathway (Grey et al., 2004). Meanwhile, LRP1 not only acts as a membrane receptor of lactoferrin, but also regulates the endocytosis to transport lactoferrin into cytoplasm (Grey et al., 2004). Thus, LRP1 could be the most promising receptor for ovotransferrin to mediate different physiological processes of osteoblasts. Blocking LRP1 with its inhibitor Rap did not affect the expression of Runx2, ERK and Akt pathways, while ALP and mineralization was partially attenuated (Figure 4.10). These results suggested that LRP1 was independent to ovotransferrin-induced expression of Runx2 as well as the activation of ERK and Akt pathway; but LRP1 might play roles in regulating ovotransferrin induced ALP expression and

mineralization. Therefore, further studies are needed to identify the role of LRP1 in ovotransferrin promoted osteoblast differentiation.

4.5 Conclusion

In conclusion, this study found that ovotransferrin stimulated osteoblast differentiation mainly through LRP1-independent c-Raf-MEK-ERK and PI3K-Akt signaling pathways. Ovotransferrin activated ERK and PI3K-Akt signaling transduction and regulated several bone-specific factors, such as Runx2, β-catenin and BMP-2, leading to an enhanced osteoblast differentiation. Our study also indicated that LRP1 could partially regulate the function of ovotransferrin via ERK- and AKT-independent signaling pathways.

4.6 References

- Akiyama, Y., Oshima, K., Kuhara, T., Shin, K., Abe, F., Iwatsuki, K., Matsuda, T. (2013). A lactoferrin-receptor, intelectin 1, affects uptake, sub-cellular localization and release of immunochemically detectable lactoferrin by intestinal epithelial Caco-2 cells. *Journal of Biochemistry*, 154(5), 437–448.
- Bae, S. J., Kim, H. J., Won, H. Y., Min, Y. K., & Hwang, E. S. (2017). Acceleration of osteoblast differentiation by a novel osteogenic compound, DMP-PYT, through activation of both the BMP and Wnt pathways. *Scientific Reports*, 7(1), 8455.
- Bain, G., Müller, T., Wang, X., & Papkoff, J. (2003). Activated beta-catenin induces osteoblast differentiation of C3H10T1/2 cells and participates in BMP2 mediated signal transduction. *Biochemical and Biophysical Research Communications*, 301(1), 84–91.
- Bates, P. (2007). Metabolic bone disease. Orthopaedics and Trauma, 31(5), 306–311.
- Bharadwaj, S., Naidu, A. G. T., Betageri, G. V., Prasadarao, N. V., & Naidu, A. S. (2009). Milk ribonuclease-enriched lactoferrin induces positive effects on bone turnover markers in postmenopausal women. *Osteoporosis International*, 20(9), 1603–1611.
- Bhargavan, B., Gautam, A. K., Singh, D., Kumar, A., Chaurasia, S., Tyagi, A. M., ... Chattopadhyay, N. (2009). Methoxylated isoflavones, cajanin and isoformononetin, have non-estrogenic bone forming effect via differential mitogen activated protein kinase (MAPK) signaling. *Journal of Cellular Biochemistry*, 108(2), 388–399.
- Cargnello, M., & Roux, P. P. (2011). Activation and function of the MAPKs and their substrates, the MAPK-activated protein kinases. *Microbiology and Molecular Biology Reviews* : *MMBR*, 75(1), 50–83.

Chae, H. J., Jeong, B. J., Ha, M. S., Lee, J. K., Byun, J. O., Jung, W. Y., Kim, H. R. (2002). Erk

Map Kinase is required in 1,25(OH)₂D₃-induced differentiatio in human osteobalsts. *Immunopharmacology and Immunotoxicology*, *24*(1), 31–41.

- Chen, C., Koh, A. J., Datta, N. S., Zhang, J., Keller, E. T., Xiao, G., McCauley, L. K. (2004).
 Impact of the Mitogen-activated Protein Kinase Pathway on Parathyroid Hormone-related
 Protein Actions in Osteoblasts. *Journal of Biological Chemistry*, 279(28), 29121–29129.
- Chen, G., Deng, C., & Li, Y. P. (2012). TGF-β and BMP signaling in osteoblast differentiation and bone formation. *International Journal of Biological Sciences*, 8(2), 272–288.
- Cornish, J., Callon, K. E., Naot, D., Palmano, K. P., Banovic, T., Bava, U., Reid, I. R. (2004).
 Lactoferrin is a potent regulator of bone cell activity and increases bone formation in vivo.
 Endocrinology, 145(9), 4366–4374.
- Crockett, J., & Das, J. C. (2013). Osteoporosis & amp;ndash; a current view of pharmacological prevention and treatment. *Drug Design, Development and Therapy*, 7, 435.
- Easton, R. M., Cho, H., Roovers, K., Shineman, D. W., Mizrahi, M., Forman, M. S., Birnbaum, M. J. (2005). Role for Akt3/Protein Kinase B in Attainment of Normal Brain Size. *Molecular and Cellular Biology*, 25(5), 1869–1878.
- Eaton-Evans, J. (1994). Osteoporosis and the role of diet. *British Journal of Biomedical Science*, *51*(4), 358–370.
- Eferl, R., Hoebertz, A., Schilling, A. F., Rath, M., Karreth, F., Kenner, L., Wagner, E. F. (2004). The Fos-related antigen Fra-1 is an activator of bone matrix formation. *The EMBO Journal*, *23*(14), 2789–2799.
- Garofalo, R. S., Orena, S. J., Rafidi, K., Torchia, A. J., Stock, J. L., Hildebrandt, A. L., ... Coleman, K. G. (2003). Severe diabetes, age-dependent loss of adipose tissue, and mild growth deficiency in mice lacking Akt2/PKB beta. *The Journal of Clinical Investigation*,

112(2), 197–208.

- Ge, C., Yang, Q., Zhao, G., Yu, H., Kirkwood, K. L., & Franceschi, R. T. (2012). Interactions between extracellular signal-regulated kinase 1/2 and P38 Map kinase pathways in the control of RUNX2 phosphorylation and transcriptional activity. *Journal of Bone and Mineral Research*, 27(3), 538–551.
- Gentili, C., Bianco, P., Neri, M., Malpeli, M., Campanile, G., Castagnola, P., Cancedda, F. D. (1993). Cell proliferation, extracellular matrix mineralization, and ovotransferrin transient expression during in vitro differentiation of chick hypertrophic chondrocytes into osteoblast-like cells. *Journal of Cell Biology*, *122*(3), 703–712.
- Gentili, C., Doliana, R., Bet, P., Campanile, G., Colombatti, A., Cancedda, F. D., & Cancedda, R. (1994). Ovotransferrin and ovotransferrin receptor expression during chondrogenesis and endochondral bone formation in developing chick embryo. *Journal of Cell Biology*, *124*(4), 579–588.
- Giansanti, F., Rossi, P., Massucci, M. T., Botti, D., Antonini, G., Valenti, P., & Seganti, L. (2002). Antiviral activity of ovotransferrin discloses an evolutionary strategy for the defensive activities of lactoferrin. *Biochem Cell Biol*, 80(1), 125–130.
- Greenblatt, M. B., Shim, J.-H., Zou, W., Sitara, D., Schweitzer, M., Hu, D., Glimcher, L. H. (2010). The p38 MAPK pathway is essential for skeletogenesis and bone homeostasis in mice. *The Journal of Clinical Investigation*, 120(7), 2457–2473.
- Grey, A., Banovic, T., Zhu, Q., Watson, M., Callon, K., Palmano, K., Cornish, J. (2004). The low-density lipoprotein receptor-related protein 1 is a mitogenic receptor for lactoferrin in osteoblastic cells. *Molecular Endocrinology (Baltimore, Md.)*, 18(9), 2268–2278.

Grey, A., Zhu, Q., Watson, M., Callon, K., & Cornish, J. (2006). Lactoferrin potently inhibits

osteoblast apoptosis, via an LRP1-independent pathway. *Molecular and Cellular Endocrinology*, *251*(1–2), 96–102.

- Groves, J. T., & Kuriyan, J. (2010). Molecular mechanisms in signal transduction at the membrane. *Nature Structural & Molecular Biology*, *17*(6), 659–665.
- Guo, H. Y., Jiang, L., Ibrahim, S. A., Zhang, L., Zhang, H., Zhang, M., & Ren, F. Z. (2009). Orally administered lactoferrin preserves bone mass and microarchitecture in ovariectomized rats. *J Nutr*, 139(C), 958–964.
- Herz, J., & Strickland, D. K. (2001). LRP: a multifunctional scavenger and signaling receptor. Journal of Clinical Investigation, 108(6), 779–784.
- Hou, J.-M., Xue, Y., & Lin, Q.-M. (2012). Bovine lactoferrin improves bone mass and microstructure in ovariectomized rats via OPG/RANKL/RANK pathway. *Acta Pharmacologica Sinica*, 33(10), 1277–1284.
- Hu, Y., Chan, E., Wang, S. X., & Li, B. (2003). Activation of p38 Mitogen-Activated Protein Kinase Is Required for Osteoblast Differentiation. *Endocrinology*, 144(5), 2068–2074.
- Jaiswal, R. K., Jaiswal, N., Bruder, S. P., Mbalaviele, G., Marshak, D. R., & Pittenger, M. F. (2000). Adult human mesenchymal stem cell differentiation to the osteogenic or adipogenic lineage is regulated by mitogen-activated protein kinase. *The Journal of Biological Chemistry*, 275(13), 9645–9652.
- Kanno, T., Takahashi, T., Tsujisawa, T., Ariyoshi, W., & Nishihara, T. (2007). Mechanical stress-mediated Runx2 activation is dependent on Ras/ERK1/2 MAPK signaling in osteoblasts. *Journal of Cellular Biochemistry*, 101(5), 1266–1277.
- Katz, M., Amit, I., & Yarden, Y. (2007). Regulation of MAPKs by growth factors and receptor tyrosine kinases. *Biochimica et Biophysica Acta*, *1773*(8), 1161–1176.

- Kawabata, H. (2018). Transferrin and transferrin receptors update. *Free Radical Biology and Medicine*.
- Kawamura, N., Kugimiya, F., Oshima, Y., Ohba, S., Ikeda, T., Saito, T., & Shinoda, Y. (2007).Akt1 in Osteoblasts and Osteoclasts Controls Bone Remodeling, (10).
- Kenner, L., Hoebertz, A., Beil, F. T., Keon, N., Karreth, F., Eferl, R., Wagner, E. F. (2004). Mice lacking JunB are osteopenic due to cell-autonomous osteoblast and osteoclast defects. *The Journal of Cell Biology*, *164*(4), 613–623.
- Kim, H., Kim, M. G., Leem, K. H., Kim, H. K., Kim, M. G., & Leem, K. H. (2013). Osteogenic Activity of Collagen Peptide via ERK/MAPK Pathway Mediated Boosting of Collagen Synthesis and Its Therapeutic Efficacy in Osteoporotic Bone by Back-Scattered Electron Imaging and Microarchitecture Analysis. *Molecules*, 18(12), 15474–15489.
- Kini, Usha; Nandeesh, B. N. (2012). *Physiology of Bone Formation, Remodeling, and Metabolism. Radionuclide and Hybrid Bone Imaging* (Vol. 9783642024).
- Kugimiya, F., Kawaguchi, H., Ohba, S., Kawamura, N., Hirata, M., Chikuda, H., Chung, U.
 (2007). GSK-3β Controls Osteogenesis through Regulating Runx2 Activity. *PLoS ONE*, 2(9), e837.
- Lai, C. F., Chaudhary, L., Fausto, A., Halstead, L. R., Ory, D. S., Avioli, L. V, & Cheng, S. L.
 (2001). Erk is essential for growth, differentiation, integrin expression, and cell function in human osteoblastic cells. *The Journal of Biological Chemistry*, 276(17), 14443–14450.
- Lee, J. H., Ahn, D. U., & Paik, H. D. (2018). In vitro Immune-Enhancing Activity of
 Ovotransferrin from Egg White via MAPK Signaling Pathways in RAW 264.7
 Macrophages. *Korean Journal for Food Science of Animal Resources*.

Lee, K. S., Hong, S. H., & Bae, S. C. (2002). Both the Smad and p38 MAPK pathways play a

crucial role in Runx2 expression following induction by transforming growth factor- β and bone morphogenetic protein. *Oncogene*, *21*(47), 7156–7163.

- Li, Y., Ge, C., Long, J. P., Begun, D. L., Rodriguez, J. A., Goldstein, S. A., & Franceschi, R. T. (2012). Biomechanical stimulation of osteoblast gene expression requires phosphorylation of the RUNX2 transcription factor. *Journal of Bone and Mineral Research*, 27(6), 1263– 1274.
- Liu, M., Fan, F., Shi, P., Tu, M., Yu, C., Yu, C., & Du, M. (2018). Lactoferrin promotes MC3T3-E1 osteoblast cells proliferation via MAPK signaling pathways. *International Journal of Biological Macromolecules*, 107(Pt A), 137–143.
- Marshall, K. (2004). Therapeutic applications of whey protein. *Alternative Medicine Review : A Journal of Clinical Therapeutic*, 9(2), 136–156.
- Matsuguchi, T., Chiba, N., Bandow, K., Kakimoto, K., Masuda, A., & Ohnishi, T. (2009a). JNK Activity Is Essential for *Atf4* Expression and Late-Stage Osteoblast Differentiation. *Journal of Bone and Mineral Research*, *24*(3), 398–410.
- Matsuguchi, T., Chiba, N., Bandow, K., Kakimoto, K., Masuda, A., & Ohnishi, T. (2009b). JNK Activity Is Essential for *Atf4* Expression and Late-Stage Osteoblast Differentiation. *Journal of Bone and Mineral Research*, *24*(3), 398–410.
- Matsushita, T., Chan, Y. Y., Kawanami, A., Balmes, G., Landreth, G. E., & Murakami, S. (2009). Extracellular signal-regulated kinase 1 (ERK1) and ERK2 play essential roles in osteoblast differentiation and in supporting osteoclastogenesis. *Molecular and Cellular Biology*, 29(21), 5843–5857.
- McGonnell, I. M., Grigoriadis, A. E., W-F Lam, E., Price, J. S., Sunters, A., & Gartland, A. (2012). A specific role for phosphoinositide 3-kinase and AKT in osteoblasts?, *719*(19), 28–

1.

- Messina, M., & Messina, V. (2000). Soyfoods, soybean isoflavones, and bone health: a brief overview. Journal of Renal Nutrition : The Official Journal of the Council on Renal Nutrition of the National Kidney Foundation.
- Miljuš, G., Malenković, V., Đukanović, B., Kolundžić, N., & Nedić, O. (2015). IGFBP-3/transferrin/transferrin receptor 1 complexes as principal mediators of IGFBP-3 delivery to colon cells in non-cancer and cancer tissues. *Experimental and Molecular Pathology*, 98(3), 431–438.
- Morley, P., Whitfield, J. F., & Willick, G. E. (2001). Parathyroid hormone: an anabolic treatment for osteoporosis. *Current Pharmaceutical Design*, *7*(8), 671–687.
- Naor, Z., Benard, O., & Seger, R. (2000). Activation of MAPK cascades by G-protein-coupled receptors: the case of gonadotropin-releasing hormone receptor. *Trends in Endocrinology* and Metabolism: TEM, 11(3), 91–99.
- Naot, D., Chhana, A., Matthews, B. G., Callon, K. E., Tong, P. C., Lin, J. M., Cornish, J. (2011).
 Molecular mechanisms involved in the mitogenic effect of lactoferrin in osteoblasts. *Bone*, 49(2), 217–224.
- Peng, X. -d., Xu, P.-Z., Chen, M.-L., Hahn-Windgassen, A., Skeen, J., Jacobs, J., Hay, N. (2003). Dwarfism, impaired skin development, skeletal muscle atrophy, delayed bone development, and impeded adipogenesis in mice lacking Akt1 and Akt2. *Genes & Development*, 17(11), 1352–1365.
- Ponnapakkam, T., Katikaneni, R., Sakon, J., Stratford, R., & Gensure, R. C. (2014). Treating osteoporosis by targeting parathyroid hormone to bone. *Drug Discovery Today*, 19(3), 204– 208.

- Qi, H., Aguiar, D. J., Williams, S. M., Pean, A. La, Pan, W., & Verfaillie, C. M. (n.d.). Identification of genes responsible for osteoblast differentiation from human mesodermal progenitor cells.
- Rey, A., Manen, D., Rizzoli, R., Ferrari, S. L., & Caverzasio, J. (2007). Evidences for a role of p38 MAP kinase in the stimulation of alkaline phosphatase and matrix mineralization induced by parathyroid hormone in osteoblastic cells. *Bone*, 41(1), 59–67.
- Rodríguez-Carballo, E., Gámez, B., Sedó-Cabezón, L., Sánchez-Feutrie, M., Zorzano, A.,
 Manzanares-Céspedes, C., Ventura, F. (2014). The p38α MAPK Function in
 Osteoprecursors Is Required for Bone Formation and Bone Homeostasis in Adult Mice.
 PLoS ONE, 9(7), e102032.
- Rutkovskiy, A., Stensløkken, K.-O., & Vaage, I. J. (2016). Osteoblast Differentiation at a Glance. *Medical Science Monitor Basic Research*, *22*, 95–106.
- Shahi, M., Peymani, A., & Sahmani, M. (2017). Regulation of Bone Metabolism. Reports of Biochemistry & Molecular Biology, 5(2), 73–82.
- Shang, N., & Wu, J. (2018). Egg White Ovotransferrin Shows Osteogenic Activity in Osteoblast Cells.
- Shang, N., & Wu, J. (2018). Egg White Ovotransferrin Shows Osteogenic Activity in Osteoblast Cells. Journal of Agricultural and Food Chemistry, 66(11), 2775–2782.
- Shimo, T., Matsumura, S., Ibaragi, S., Isowa, S., Kishimoto, K., Mese, H., Sasaki, A. (2007).
 Specific inhibitor of MEK-mediated cross-talk between ERK and p38 MAPK during differentiation of human osteosarcoma cells. *Journal of Cell Communication and Signaling*, *1*(2), 103–111.
- Shin, K., Wakabayashi, H., Yamauchi, K., Yaeshima, T., & Iwatsuki, K. (2008). Recombinant

human intelectin binds bovine lactoferrin and its peptides. *Biological & Pharmaceutical Bulletin*, *31*(8), 1605–1608. Retrieved from

- Suzuki, A., Palmer, G., Bonjour, J. P., & Caverzasio, J. (1999). Regulation of Alkaline
 Phosphatase Activity by p38 MAP Kinase in Response to Activation of Gi Protein-Coupled
 Receptors by Epinephrine in Osteoblast-Like Cells. *Endocrinology*, 140(7), 3177–3182.
- Suzuki, E., Ochiai-Shino, H., Aoki, H., Onodera, S., Saito, A., Saito, A., & Azuma, T. (2014).
 Akt Activation is Required for TGF-β1-Induced Osteoblast Differentiation of MC3T3-E1
 Pre-Osteoblasts. *PLoS ONE*, 9(12), e112566.
- Suzuki, Y. A., Lopez, V., & Lönnerdal, B. (2005). Mammalian lactoferrin receptors: structure and function. *Cellular and Molecular Life Sciences*, *62*(22), 2560–2575.
- Swarthout, J. T., Doggett, T. A., Lemker, J. L., & Partridge, N. C. (2001). Stimulation of extracellular signal-regulated kinases and proliferation in rat osteoblastic cells by parathyroid hormone is protein kinase C-dependent. *The Journal of Biological Chemistry*, 276(10), 7586–7592.
- Takada, I., Kouzmenko, A. P., & Kato, S. (2009). Wnt and PPARγ signaling in osteoblastogenesis and adipogenesis. *Nature Reviews Rheumatology*, *5*(8), 442–447.
- Teplyuk, N. M., Galindo, M., Teplyuk, V. I., Pratap, J., Young, D. W., Lapointe, D., van Wijnen,
 A. J. (2008). Runx2 regulates G protein-coupled signaling pathways to control growth of
 osteoblast progenitors. *The Journal of Biological Chemistry*, 283(41), 27585–27597.
- Thouverey, C., & Caverzasio, J. (2012). The p38α MAPK positively regulates osteoblast function and postnatal bone acquisition. *Cellular and Molecular Life Sciences*, 69(18), 3115–3125.

Weinzimer, S. A., Gibson, T. B., Collett-Solberg, P. F., Khare, A., Liu, B., & Cohen, P. (2001).

Transferrin Is an Insulin-Like Growth Factor-Binding Protein-3 Binding Protein¹. *The Journal of Clinical Endocrinology & Metabolism*, *86*(4), 1806–1813.

- Wortzel, I., & Seger, R. (2011). The ERK Cascade: Distinct Functions within Various Subcellular Organelles. *Genes & Cancer*, 2(3), 195–209.
- Wozney, J. M., Rosen, V., Celeste, A. J., Mitsock, L. M., Whitters, M. J., Kriz, R. W., ... Wang,
 E. A. (1988). Novel regulators of bone formation: molecular clones and activities. *Science* (*New York, N.Y.*), 242(4885), 1528–1534.
- Xi, J. C., Zang, H. Y., Guo, L. X., Xue, H. B., Liu, X. D., Bai, Y. B., & Ma, Y. Z. (2015). The PI3K/AKT cell signaling pathway is involved in regulation of osteoporosis. *Journal of Receptors and Signal Transduction*, 35(6), 640–645.
- Xiao, G., Jiang, D., Gopalakrishnan, R., & Franceschi, R. T. (2002). Fibroblast Growth Factor 2 Induction of the Osteocalcin Gene Requires MAPK Activity and Phosphorylation of the Osteoblast Transcription Factor, Cbfa1/Runx2. *Journal of Biological Chemistry*, 277(39), 36181–36187.
- Xiao, G., Jiang, D., Thomas, P., Benson, M. D., Guan, K., Karsenty, G., & Franceschi, R. T.
 (2000). MAPK pathways activate and phosphorylate the osteoblast-specific transcription factor, Cbfa1. *The Journal of Biological Chemistry*, 275(6), 4453–4459.
- Yamaguchi, A., Komori, T., & Suda, T. (2000). Regulation of Osteoblast Differentiation
 Mediated by Bone Morphogenetic Proteins, Hedgehogs, and Cbfa1. *Endocrine Reviews*, 21(4), 393–411.
- Yamashita, M., Ying, S.-X., Zhang, G., Li, C., Cheng, S. Y., Deng, C., & Zhang, Y. E. (2005).
 Ubiquitin Ligase Smurf1 Controls Osteoblast Activity and Bone Homeostasis by Targeting MEKK2 for Degradation. *Cell*, *121*(1), 101–113.

- Yang, S., Xu, H., Yu, S., Cao, H., Fan, J., Ge, C., Xiao, G. (2011). Foxo1 Mediates Insulin-like Growth Factor 1 (IGF1)/Insulin Regulation of Osteocalcin Expression by Antagonizing Runx2 in Osteoblasts. *Journal of Biological Chemistry*, 286(21), 19149–19158.
- Yang, X., Matsuda, K., Bialek, P., Jacquot, S., Masuoka, H. C., Schinke, T., Karsenty, G. (2004). ATF4 is a substrate of RSK2 and an essential regulator of osteoblast biology; implication for Coffin-Lowry Syndrome. *Cell*, 117(3), 387–398.
- Zhang, H., Pan, Y., Zheng, L., Choe, C., Lindgren, B., Jensen, E. D., ... Huang, H. (2011). FOXO1 Inhibits Runx2 Transcriptional Activity and Prostate Cancer Cell Migration and Invasion. *Cancer Research*, 71(9), 3257–3267.
- Zhang, W., Guo, H., Jing, H., Li, Y., Wang, X., Zhang, H., Ren, F. (2014). Lactoferrin stimulates osteoblast differentiation through PKA and p38 pathways independent of Lactoferrin's receptor LRP1. *Journal of Bone and Mineral Research*, 29(5), 1232–1243.

Figure 4.1 Effects of ovotransferrin on the expression of master factors involved in osteoblast differentiation.

MC3T3-E1 cells treated with different concentration of ovotransferrin for 24 h and proteins were extracted for western blot analysis. (A) The expression of Runx2; (B) The expression of BMP-2; (C) The expression of β -catenin; (D) The phosphorylation of CREB; (E) The western blot for each of the biomarkers. Data were expressed as Mean±SEM (n=6). Means without a common letter indicated p<0.05.

В

А

p-CREB t-CREB α-Tubulin









Figure 4.2 Effects of ovotransferrin on activation of MAPK pathways.

MC3T3-E1 cells treated with 1000 μ g/mL of ovotransferrin for different time (indicated in figures) and proteins were extracted for western blot analysis. (A) The time course of ERK1/2 phosphorylation; (B) The time course of JNK phosphorylation; (C) The time course of p38 phosphorylation.



А

146

Figure 4.3 Effects of ERK inhibitor on ovotransferrin-induced osteoblast differentiation.

MC3T3-E1 cells were pre-treated with ERK inhibitor for 1 h and then treated with 1000 μ g/mL of ovotransferrin for western blot analysis (24 h treatment), immunofluorescence analysis (36 h) and mineralization (15 d). (A) The expression of ALP; (B) The expression of Runx2; (C) The expression of β -catenin; (D) The expression of BMP-2; (E) The expression of type I collagen; (F) The osteoblast mineralization. Data were expressed as Mean±SEM (n=6). Means without a common letter indicated p<0.05.

В

D

А











F



OVT + / ERKi +

Figure 4.4 Effects of JNK inhibitor on ovotransferrin-induced osteoblast differentiation.

MC3T3-E1 cells were pre-treated with JNK inhibitor for 1 h and then treated with 1000 μ g/mL of ovotransferrin for 24 h and proteins were extracted for western blot analysis. (A) The expression of ALP; (B) The expression of Runx2. Data were expressed as Mean±SEM (n=6). Means without a common letter indicated p<0.05.

В

А

200b b ALP over α-Tubulin (% OVT- / JNKi-) 150а 100-50-0 OVT ÷ + JNKi + _ + ALP α-Tubulin



Figure 4.5 Effects of p38 inhibitor on ovotransferrin-induced osteoblast differentiation.

MC3T3-E1 cells were pre-treated with p38 inhibitor for 1 h and then treated with 1000 μ g/mL of ovotransferrin together for 24 h and proteins were extracted for western blot analysis. (A) The expression of ALP; (B) The expression of Runx2. Data were expressed as Mean±SEM (n=6). Means without a common letter indicated p<0.05.

В

A

200 d ALP over α-Tubulin (% OVT-/p38i-) С 150b 100-50· n OVT ÷ ÷ p38i ÷ ÷ ALP α-Tubulin



Figure 4.6 Effects of ovotransferrin on activation of the ERK MAPK pathway.

MC3T3-E1 cells were treated with 100 μ g/mL and 1000 μ g/mL of ovotransferrin together for 24 h and proteins were extracted for western blot analysis. (A) The phosphorylation of c-Ras; (B) The phosphorylation of MEK1/2. (C) The phosphorylation of p90RSK; (D) The phosphorylation of MSK1. Data were expressed as Mean±SEM (n=6). Means without a common letter indicated p<0.05. Untr: cells without ovotransferrin treatment.



Figure 4.7 Effects of ovotransferrin on activation of PI3K-Akt pathways.

MC3T3-E1 cells treated with ovotransferrin and proteins were extracted for western blot analysis. (A) The time course of Akt phosphorylation (MC3T3-E1 cells treated with 1000 μ g/mL ovotransferrin for different times). (B) The expression of PI3K p85 and p110 subunit (MC3T3-E1 cells treated with 100 or 1000 μ g/mL ovotransferrin for 24 h). Data were expressed as Mean±SEM (n=6). Means without a common letter indicated p<0.05. Untr: cells without ovotransferrin treatment.



В



Figure 4.8 Effects of Akt inhibitor on ovotransferrin-induced osteoblast differentiation.

MC3T3-E1 cells were pre-treated with Akt inhibitor for 1 h and then treated with 1000 μ g/mL of ovotransferrin together for western blot analysis (24 h treatment), cell migration (12 h) and mineralization (15 d). (A) The expression of Runx2; (B) The expression of BMP-2; (C) (i) The area of migration, (ii) The distance between the migration area; (D) The osteoblast mineralization. Data were expressed as Mean±SEM (n=6). Means without a common letter indicated p<0.05.

А

C(i)

В





C(ii)





OVT + / Akti -



OVT - / Akti -

OVT - / Akti +

OVT + / Akti -

OVT + / Akti +

Figure 4.9 The effects of ovotransferrin on the receptor expression.

MC3T3-E1 cells were treated with 1000 μ g/mL of ovotransferrin and proteins were extracted for western blot analysis. The expression of TFBP, TFR1 and LRP1. Data were expressed as Mean±SEM (n=6). ** indicated p<0.01 and *** p<0.001.



Figure 4.10 Effects of LRP1 inhibitor on ovotransferrin-induced osteoblast differentiation.

MC3T3-E1 cells were pre-treated with LRP1 inhibitor for 2 h and then treated with 1000 μ g/mL of ovotransferrin for western blot analysis (24 h treatment) and mineralization (15 d). (A) The expression of ALP; (B) The expression of Runx2; (C) The phosphorylation of ERK1/2; (D) The phosphorylation of Akt; (E) The osteoblast mineralization. Data were expressed as Mean±SEM (n=6). Means without a common letter indicated p<0.05.

А

В





OVT - / LRP1i -

OVT - / LRP1i +

OVT + / LRP1i -

OVT +/LRP1i +
Figure 4.11 Scheme summary of ovotransferrin activated signaling pathways.

Ovotransferrin stimulation activated the MAPK pathways, including ERK1/2, JNK and p38. The activation of ERK1/2 cascade further regulated the expression of Runx2, β -catenin, and phosphorylation of CREB and stimulated osteoblast differentiation and mineralization. Ovotransferrin also activated PI3K-Akt pathway and regulated osteoblast differentiation via Runx2. The activation of PI3K-Akt pathway also regulated the migration of osteoblast.



4.7 Supplementary data and information

Figure 4.7.1 Toxicity of antagonists

Ovotransferrin treated with different signaling antagonist for 24h and Alarma Blue assay was used to evaluate the cell toxicity of each antagonist.



Figure 4.7.2 The PCR amplify of the selected receptors

The mRNA was extracted from MC3T3-E1 and PCR amplify was sued to investigate the expression of selected receptors.

ITLN1

LRP1



Name	Abbreviation	Functions	Reference
Transferrin binding protein	TFBP	Ovotransferrin receptor	Gentili et al., 1993; 1994
Transferrin receptor 1	TFR1	Primary target of transferrin in the iron transport system	Kawabata, 2018
Transferrin receptor 2	TFR2	Target for transferrin family members, while the expression is restricted in cell type	Kawabata, 2018
Cubilin/LRP2 complex	N/A	Function as a receptor for transferrin and also acts as endocytic receptors for a variety of proteins including hemoglobin, albumin, vitamin- carrier and lipoprotein	Miljuš et al., 2015; Weinzimer et al., 2001
Insulin-like growth factor (IGF)-binding protein-3	IGFBP-3	Has been reported to bind transferrin	Miljuš et al., 2015; Weinzimer et al., 2001
Intelectin-1	ITLN1	Intestinal lactoferrin receptor	Akiyama et al., 2013; Shin et al., 2008
Low-density lipoprotein receptor-related protein 1	LRP1	Lactoferrin receptor that expressed on osteoblast, chondrocyte and cartilage tissue that involved in various biological processes	Grey et al., 2004; 2006

Table 4.7.1 List of the selected receptors

Gene	Primer Sequences (5'-3')	
TFBP-1	Forward	GGAAGGAGTTCGGCACGAATA
	Reverse	TCCACACGGGATTCATAGCG
TFBP-2	Forward	GGAAGGAGTTCGGCACGAATA
	Reverse	TCCTTCTCTGTTGCTTCCCG
TFBP-3	Forward	CGCCACAACAGCCAACTATT
	Reverse	TGGTCCCTGGTCCAGATGAT
TFBP-4	Forward	GAACAGCGCCAGTTCAACAA
	Reverse	GGCTCTGGAAGAGTGTTGCT
TFR1-1	Forward	TTGCCATGGGGTCAGTTCAA
	Reverse	GGTTGGCAAGTCCAAACCAAA
TFR1-2	Forward	AGCATCTGCTAATGAGACCCA
	Reverse	ACCCTGAGAGCCTAGTGACAT
TFR1-3	Forward	TCTAGCATGAACCAGGGGGA
	Reverse	GTAGTGCGTGTCAGTCCACA
TFR2-1	Forward	GTTGCTGAGCTGAACTTGGC
	Reverse	TAGTGCGTGTCAGTCCACAC
TFR2-2	Forward	AGATCTTTCTGGACCTGGCTGC
	Reverse	GGGATCCTGGGAGAAGTCTGA
TFR2-3	Forward	ATTGTTTTGCAGTCTGCCCG
	Reverse	TAGTGCGTGTCAGTCCACAC
Culilin-1	Forward	GAACAGCGCCAGTTCAACAA
	Reverse	GGCTCTGGAAGAGTGTTGCT

 Table 4.7.2 The primers designed in this chapter

Culilin-2	Forward	CGCCACAACAGCCAACTATT
	Reverse	TGGTCCCTGGTCCAGATGAT
LRP2	Forward	TAACCCCAGCTGTGACCCT
	Reverse	AGCAACTGCTCAGAAGAGCC
IGFBP-3-1	Forward	AGATGCGAGCTTAGAGCGG
	Reverse	GCGCGCACTGGGACA
IGFBP-3-2	Forward	CAGTGCGCGCCTCCG
	Reverse	ATTATGTGGCACGGAGCATCT
IGFBP-3-3	Forward	GCTGTGCATGTCCAACACAA
	Reverse	ACGGCTCCCTCATACTCCTT
ITLN1-1	Forward	GGAGCACACAAAGGCACAAG
	Reverse	GAGAAGTCAGGGCCAATCCC
ITLN1-2	Forward	TGATTGCCACGAGAGTGTG
	Reverse	CTGGGCCATTGTCAGTCCAA
LRP1-1	Forward	GAGTGTTCCGTGTATGGCAC
	Reverse	GATGCCTTGGATGATGGTC
LRP1-2	Forward	TATGAAGGTGGAGAGCCCGA
	Reverse	CAGCTTCCAGGGGTATGCTC
LRP1-3	Forward	CGTGCCTACCTTCCCGAC
	Reverse	CGGACTTGAGTGAGCCAGG

CHAPTER 5 - Egg White Ovotransferrin Attenuates RANKL-induced Osteoclastogenesis and Bone Resorption

5.1 Introduction

Bone is a dynamic organ, which undergoes continuously remodeling to changes its mass and form (Kini, Usha; Nandeesh, 2012). Normal bone metabolism relies on a balance between bone resorption and formation that regulated by osteoclasts and osteoblasts, respectively (Takatsuna et al., 2004). When this balance turns to favour of bone resorption, it results in osteolytic conditions and leads to bone diseases such as osteoporosis (Takatsuna et al., 2004). The reduction in bone mass and its architectural integrity are considered to result from the massive recruitment of osteoclasts and the excessive osteoclastic activity (Takatsuna et al., 2004; Wei et al., 2018). Thus, inhibition of osteoclastic activity and the subsequent bone resorption has profound effect on regulating abnormal remodeling process and is therapeutically important for osteoporosis prevention.

Osteoclasts are giant, multinucleated cells derived from monocytic/macrophage lineage, which are responsible for bone resorption (Tanaka et al., 2006). The essential step of osteoclasts generation is the binding of receptor-activator of NF-κB ligand (RANKL) to its receptor RANK in osteoclast precursors. This binding triggers the trimerization of TNF receptor-associated factor 6 (TRAF6), leading to the activation of downstream signaling pathways, such as NF-κB, Scr and mitogen-activated protein kinases (MAPKs) pathways. The MAPKs include p38 MAPK, c-jun-N-terminal kinase (JNK), and extracellular signal-regulated kinase (ERK) (Boyle, Simonet, & Lacey, 2003; Thummuri et al., 2015). Activation of NF-κB and MAPKs subsequently results in the expression

of transcription factors such as proto-oncogene c-Fos (c-Fos) and nuclear factor of activated Tcells, cytoplasmic 1 (NFATc1), which are crucial for the differentiation of mature osteoclasts from the precursors (Thummuri et al., 2015). Hence, targeting NF- κ B and MAPKs/AP-1 signaling to decrease osteoclastogenesis has been a promising strategy for treating osteolytic disorders.

In addition to impairing osteoclast formation, the induction of apoptosis of matured osteoclasts is another strategy for osteoporosis treatment. The significance of osteoclast apoptosis in bone disorders has been identified in several genetically modified animal models (Xing & Boyce, 2005). Decreased osteoclast apoptosis in general leads to increased bone loss, which is seen after ovariectomy due to estrogen deficiency (Hughes et al., 1996; Hughes & Boyce, 1997). Decreased osteoclast apoptosis in inflamed joints may contribute in part to reduced efficacy of bisphosphonate therapy to prevent local bone loss in patients with rheumatoid arthritis, compared to those patients with osteoporosis (Zhang et al., 2005). The apoptosis process is triggered by two different pathways: the intrinsic pathway through mitochondrial and the extrinsic pathway by ligand activation via death receptors (Elmore, 2007; Roux et al., 2005; Xing & Boyce, 2005). Although the extrinsic pathway in osteoclast apoptosis has several features differing from those in other cell types (for example, TNF promotes survival in osteoclasts rather than death in other cell types (Glantschnig, Fisher, Wesolowski, Rodan, & Reszka, 2003; Lee et al., 2001; Zhang et al., 2005)), osteoclasts contain the general apoptosis machinery of the intrinsic pathway (Xing & Boyce, 2005). Caspases are one of the main adhesion molecules involved in bone cell apoptosis (Mollazadeh, Fazly Bazzaz, & Kerachian, 2015). The release of protein-activated caspase through the mitochondria is governed by Bcl-2 family (Hung & Chow, 1997). Bad, Bax, and Bid are the apoptotic members while Bcl-2 and Bcl-xL are considered as anti-apoptotic members (Elmore,

2007). Among Bcl-2 family members, Bcl-xL has been mostly studied in osteoclasts. M-CSF, RANKL and TNF all increase Bcl-xL protein level in mature osteoclasts (Woo, Kim, & Ko, 2002; Xing & Boyce, 2005). Over-expression of Bcl-xL in osteoclasts *in vitro* prolongs their life span and protects them from bisphosphonate-induced cell death (Zhang, Badell, Schwarz, Boyce, & Xing, 2004). Bcl-2 expression and Bcl-2: Bax ratio are significantly increased in pagetic osteoclasts, which have more nuclei than normal osteoclasts. Abnormally high levels of Bcl-2 protein delayed the apoptosis of matured osteoclasts in pagetic patients (Brandwood et al., 2003; Mee, 1999).

To date, most FDA-approved drugs against osteoporosis are anti-resorptive agents such as bisphosphonates. These anti-resorptive agents prevent bone loss by inhibiting osteoclasts formation and/or inducing apoptosis in matured osteoclasts, but are associated with severe side effects such as renal impairment and osteonecrosis of the jaw (Drake, Clarke, & Khosla, 2008). Food-derived natural products are gaining the momentum towards the development of alternative against osteoporosis. Ovotransferrin is an iron-binding glycoprotein, which is rich in egg white. Egg ovotransferrin is found to have profound pharmacological activities, such as antimicrobial, antioxidant, and immunomodulating activities (Wu & Acero-Lopez, 2012). Our previous study reported that ovotransferrin stimulated osteogenic activity (Shang & Wu, 2018) via down regulation of RANKL/OPG ratio, suggesting its possible additional effects on mediating osteoclastogenesis. In this study, we hypothesized that ovotransferrin could help prevent RANKL-induced osteoclastogenesis and induce apoptosis in matured osteoclasts. Hence, we tested the activity of ovotransferrin in cell model of osteoclastogenesis and illustrated the underlying molecular mechanisms. We found that ovotransferrin significantly inhibits RANKL-induced

osteoclastogenesis accompanied with the inhibition of NF-κB and MAPK signaling. Furthermore, ovotransferrin induced cell apoptosis in matured osteoclasts via intrinsic apoptosis pathway by regulating Bcl-2 family members.

5.2 Materials and methods

5.2.1 Reagents and antibodies

Recombinant mouse soluble RANKL was purchased from R&D Systems (Oakville, ON, Canada). Dulbecco modified Eagles medium (DMEM), fetal bovine serum (FBS), penicillin-streptomycin (1000 U/mL) and TRIzol Reagent was purchased from Thermo Fisher Scientific (Burlington, ON, Canada). Ovotransferrin and TRAP-staining kit was purchased from Sigma-Aldrich (St. Louis, MO, USA). Annexin V-FITC Apoptosis Staining/Detection Kit was purchased from Abcam (Toronto, ON, Canada). NF-κB pathway sampler kit was purchased from Cell Signaling Techology (Whitby, ON, Canada).

5.2.2 Cell culture

The murine monocyte cell line RAW 264.7 was purchased from ATCC (Manassas, VA, USA). RAW 264.7 cells were cultured at 37°C in 5% CO₂ atmosphere in Dulbecco modified Eagles medium (DMEM) supplemented with 10% fetal bovine serum (FBS) and 1% pen-strep. All experiments were conducted using the cells between passage number 18 to 25.

5.2.3 Osteoclasts generation and TRAP staining

Osteoclast cells were generated from culture of murine monocyte cell line RAW 264.7. RAW 264.7 cells were seeded in 24-well plate at density of 1×10^4 cells/well and incubated for 6 h to allow the cells to attach to the surface. After 6h, the culture medium was replaced with medium contains 100 ng/ml RANKL. Osteoclasts were successfully induced by RANKL added every 48 h

over the course of 4 days. RAW 264.7 cells were treated with different concentration of ovotransferrin (1 μ g/mL to 1000 μ g/mL) for 4 days for its effects on osteoclastogenesis. To confirm and count the generation of multinucleated osteoclast-like cells, the cultured cells were fixed in 4% paraformaldehyde for 1 h at 4 °C and then stained for the enzyme tartrate-resistant acid phosphatase (TRAP) using the TRAP-staining kit according to the manufacturer instructions (0.1mg/mL naphthol AS-MX phosphate and 0.3 mg/mL Fast Red Violet LB staining). The cells were observed under a light microscope with the magnification of 10X (Olympus IX83) and the images were captured by Metamorphy (Olympus). TRAP-positive cells containing three or more nuclei were counted. The experiment was repeated 5 times.

5.2.4 Cell apoptosis assay

Osteoclasts were generated from RAW 264.7 cells stimulated with RANKL (100 ng/mL) for 4-6 days as described in 2.3 and identified by microscopy. Cells were treated with 1-1000 μ g/mL ovotransferrin for 12 h and collected by centrifugation. The apoptosis detection was conducted using the apoptosis detection kit according to manufacturer's instruction. Briefly, the cells were re-suspended in 500 μ L binding buffer, followed by the addition of 5 μ L of Annexin V-FITC and 5 μ L of propidium iodide, and then incubation for 5 min in the darkness at room temperature. The Annexin V-FITC/PI binding was analyzed by flow cytometry (FACSCanto II, BD Biosciences, Mississauga, ON, Canada).

5.2.5 Resorption assay

RAW 264.7 cells were seeded in a calcium phosphate (CaP)-coated 48-well plate (Cosmo Bio Co., Ltd, Tokyo, Japan) and cultured with 100 ng/mL RANKL for 6 days to generated multinucleated osteoclasts, and treated with 1-1000 μ g/mL ovotransferrin for 5 days. An aliquot of 150 μ L new medium was added into the well every 2 days without removing the old medium. The generated

multinucleated osteoclasts were identified with microscopy. An aliquot of $100 \ \mu$ L culture medium was used to identify the resorption by measuring the fluorescence intensity every 2 days. After 6 days, the adherent cells were removed by the treatment of 5% hypochlorite for 5 min and the plate was washed with water and then dried., Microscopic images of each well were captured using a microscope and the pit areas are measured by Image J.

5.2.6 Western blot analysis

The RAW 264.7 cells were seeded in 48 well tissue culture plates at a density of 1×10^4 cells/well, and incubated in α -MEM with 10% FBS. The cells were treated with different concentrations of ovotransferrin and RANKL. At the end of incubation, the culture medium was removed and the cells lysed in boiling Laemmle's buffer containing 50 µM dithiothreitol (DTT) and 0.2% Triton-X-100 to prepare samples for western blot as described previously. These cell lysates were run in SDS-PAGE, blotted to nitrocellulose membranes and immunoblotted with antibodies. The concentration of each antibody was according to the manufacturer's instruction. The protein bands were detected by a Licor Odyssey BioImager and quantified by densitometry using corresponding software (Licor Biosciences, Lincoln, NB, USA). Each band was normalized to its corresponding band of loading control. Cell lysates from untreated cells were loaded onto every gel. The results were expressed as percentage of the corresponding untreated control.

5.2.7 RNA extraction and quantitative PCR

Total RNA was extracted from the cells using TRIzol[®] reagent by following the manufacturer's instruction, and cDNA was synthesized using SuperScript II Reverse Transcriptase (Invitrogen, Burlington, ON, Canada) according to the manufacturer's instruction. Real-time PCR was performed using Fast SYBR Green PCR Master Mix (Applied Biosystem, Burlington, ON, Canada) and an ABI 7300 Sequencing Detection System (Applied Biosystem, Burlington, ON, Canada).

The amplification conditions were as follow: initial denaturation at 95°C for 10 min, followed by 35 cycles of 10 s at 95°C, 15 s at 60°C, and 10 s at 72°C. Real-time PCR primers used in this study are listed in Table 5.7.1.

5.2.8 Statistical analysis

All data are presented as mean \pm SEM (standard error of mean) from 4 and 8 independent experiments. Data were analyzed using One Way analysis of variance (ANOVA) with Dunnett's post-hoc test for comparisons to control. The PRISM 6 statistical software (GraphPad Software, San Diego, CA) was used for the analyses. P< 0.05 was considered significant.

5.3 Results

5.3.1 Ovotransferrin inhibits osteoclast formation from RAW 264.7

To examine the effect of ovotransferrin on osteoclast formation from mouse osteoclast precursors, RAW 264.7 were cultured in osteoclastogenic medium containing 100 ng/mL RANKL with various concentrations of ovotransferrin (1-1000 μ g/mL) for 6 days. RANKL addition significantly increased osteoclast formation, whereas ovotransferrin addition significantly suppressed RANKL-induced osteoclast formation (Figure 5.1A). Further, count of TRAP-positive (TRAP⁺) osteoclasts confirmed that ovotransferrin dose dependently inhibited RANKL-induced osteoclast differentiation (Fig. 5.1B).

5.3.2 Ovotransferrin inhibited RANKL-induced activation of NF-KB and MAPK pathway

RANKL-induced osteoclast differentiation involves activation of NF- κ B and MAPK pathways. Phosphorylation of NF- κ B enhances its transcriptional potential. We found that RANKL significantly increased the phosphorylation of IKK α/β (Ser 176/178), I κ B α (Ser 32), and NF- κ B p65 (Ser 536), whereas ovotransferrin pre-treatment suppressed their phosphorylation (Fig. 5.2). During osteoclastogenesis, RANKL also increases the activation of MAPK signaling. So, we further investigated the effect ovotransferrin on MAPKs under RANKL stimulation in osteoclast precursors. Ovotransferrin treatment significantly inhibited RANKL-induced phosphorylation of MAPKs in a dose-depended manner (Fig. 5.3). These data indicate that ovotransferrin inhibits osteoclast formation and differentiation by attenuating NF-κB and MAPK pathways.

5.3.3 Ovotransferrin inhibited RANKL-induced expression of the factors involved in osteoclastogenesis

We further assessed the expression of the factors involved in osteoclastogenesis including TRAF6, c-Fos, NFATc1 and Cathepsin K. Western blotting showed that RANKL stimulation significantly increased the expression of TRAF6, NFATc1 and cathepsin K, while pre-treatment of ovotransferrin significantly decreased RANKL-induced expression of osteoclast-specific proteins (Fig. 5.4). Interestingly, the RANKL did not increase the expression of c-Fos, but pre-treatment of ovotransferrin significantly decreased the expression in a dose-dependent manner (Fig. 5.4C). These data indicate that ovotransferrin inhibits osteoclast differentiation by inhibiting several osteoclastogenesis-related factors.

5.3.4 Ovotransferrin inhibits resorptive activity of matured osteoclasts

Since ovotransferrin inhibited osteoclast formation and expression of osteoclastic markers, we further investigated whether ovotransferrin could inhibit the osteoclastic resorption *in vitro*. RAW 264.7 cultured in a fluoresceinated calcium phosphate (CaP)-coated plate were treated with 100 ng/mL RANKL and different concentrations of ovotransferrin (1-1000 µg/mL) for 6 days to

generate multinucleated osteoclasts. RANKL increased the fluorescent intensity of culture medium with time (Fig. 5.5A), indicating the generation of osteoclasts and osteoclastic resorption of CaP; whereas ovotransferrin treatment significantly decreased RANKL-induced osteoclastic resorption (Fig. 5.5A and 5.5B). After 6 days, the total areas of resorption pits were measured and the formation of resorption pits were substantially reduced in the presence of ovotransferrin (Fig. 5.5C and 5.5D). These data indicate that ovotransferrin inhibited not only osteoclast differentiation but also the bone-resorbing activity of mature osteoclasts *in vitro*.

5.3.5 Ovotransferrin stimulates the apoptosis of osteoclasts

We further studied ovotransferrin-induced cell apoptosis of osteoclasts using flow cytometry. Matured multinucleated osteoclasts were identified using microscope and then treated with different concentrations of ovotransferrin (1-1000 μ g/mL). Flow cytometry revealed that ovotransferrin induced apoptosis of matured osteoclasts in a dose-depended manner (Fig. 5.6A and 5.6B). Next, we further assessed the expression of *Bcl-2* family, which is one of the major regulators for cell apoptosis. RT-PCR analysis showed that ovotransferrin treatment significantly increased the expression of *Bim and Bad*, while decreased *Bcl-2* and *Bcl-x* expression (Fig. 5.6C). These findings suggested that ovotransferrin regulates *Bcl-2* family, and may further acting on osteoclasts mitochondria and induce apoptosis.

5.4 Discussion

Bone is constantly remodelling by bone formation osteoblast cells and bone resorbing osteoclast cells (Kini, Usha; Nandeesh, 2012). The over-activation of osteoclasts leads to excessive bone resorption and is a characteristic of several osteolytic disorders, such as osteoporosis. Thus,

osteoclasts remain one of the key targets for the development of potential osteoporosis therapies. However, current treatments targeting osteoclasts to prevent bone loss are associated with some side-effects such as renal toxicity and osteonecrosis with bisphosphonates (Perazella & Markowitz, 2008) and endometrial cancer risk with selective estrogen receptor modulators (Brown, 2009). Diet is one of the most important factors that may contribute to a proper bone structure and function. Nutrients like vitamin D and calcium are well studied in respect to their effects on bone matrix production and mineralization (Alekel, Weaver, Ronis, & Ward, 2013). However, other nutritional factors may also be implied in the process, such as protein. It was reported that a proper increase of protein intake is associated with higher bone mineral density (Bonjour, 2005; Kerstetter & Insogna, 2003). Furthermore, natural compounds including bioactive protein have been a source of therapeutic in bone health management. For instance, soy proteins have been suggested to improve bone quality by inhibiting osteoclast activity (Alekel et al., 2000; Messina & Messina, 2000). Lactoferrin, a protein derived from milk was also reported to promote bone formation by stimulating osteoblastic activity (Cornish et al., 2004) while preventing bone resorption by inhibiting osteoclast differentiation and resorptive activity (Lorget et al., 2002).

Although a positive role of diet protein intake on bone health has been suggested, the effect of egg protein has not been well studied. In addition to their essential amino acid profile and high digestibility, egg proteins are known to impart a wide range of physiological benefits including antimicrobial, anticancer, antioxidant and immunomodulation activities (Wu & Acero-Lopez, 2012). In our previous study, we showed for the first time that egg white protein ovotransferrin could directly stimulate osteoblasts proliferation and differentiation as an osteogenic agent (Shang & Wu, 2018) Thus, the present study aims to characterize the cellular and molecular effects of egg

white ovotransferrin on mouse osteoclasts. In this study, we demonstrated that ovotransferrin exerts an anti-osteoclastogenesis effect via suppress RANKL-induced activation of NF- κ B and MAPK signaling pathway. In addition, we also found that ovotransferrin could induce cell apoptosis in matured multinuclear osteoclasts by regulating the expression of Bcl-2 family members.

Osteoclastogenesis is a complex process, which is primarily regulated by RANKL and RANK. Binding of RANKL to RANK is the essential step in osteoclast development. It leads to recruitment of TRAF6 and activation of downstream signaling such as NF-kB and three MAPKs including ERK, JNK, and p38 (Karin, 2005; Matsumoto, Sudo, Saito, Osada, & Tsujimoto, 2000; Thummuri, Naidu, & Chaudhari, 2017). The role of NF-kB in RANKL/RANK pathway for osteoclast differentiation has been confirmed by both in vitro and in vivo studies (Abu-Amer, 2013). NF- κ B signaling comprises a number of activation steps. Under normal conditions, NF- κ B exists as a complex with the cytoplasmic inhibitory protein I κ Ba. With the stimulation of RANKL, I κ B kinase (IKK) is activated and stimulates the phosphorylation of IkB proteins at Ser32 and 36 residues, which leads to the polyubiquitination of IkB and subsequent proteasomal degradation. IkB degradation increases nuclear translocation of NF-kB p65 and leads to the transcription of the genes involved in osteoclast differentiation (Thummuri et al., 2017). In our study, we found that ovotransferrin inhibits RANKL-induced phosphorylation of IKK α/β (Ser 176/180), IkB α (Ser 32), and NF- κ B p65 (Ser 536), suggesting the regulatory role of ovotransferrin in NF- κ B pathway. In addition to NF- κ B pathway, activation of MAPK signaling is another pivotal regulation in RANKL-induced osteoclast activation and differentiation (Tanaka et al., 2006; Thummuri et al., 2017). MAPKs play an essential role in the transduction of extracellular stimuli into intracellular

responses and thereby contribute to the regulation of diverse cellular activities (Lee et al., 2016). Hence, in the current study, we also determined the effect of ovotransferrin on RANKL-mediated phosphorylation of MAPK and found that ovotransferrin markedly inhibits RANKL-induced phosphorylation of three MAPKs. These results indicate that the anti-osteoclastogenic effects of ovotransferrin are associated with the suppression of RANKL-induced NF-κB and MAPK signaling pathways.

The phosphorylation of NF-κB and MAPK signaling pathways is the early stage of osteoclastogenesis. Following the activation of NF-κB and MAPK signaling, the transcription of osteoclast differentiation factors such as c-Fos and NFATc1 are expressed and acts as the regulator for osteoclast differentiation at early or mid stage (Teitelbaum, 2004; Thummuri et al., 2017). Binding of NF-κB and NFATc1 to their promoter regulates the expression of various osteoclast-specific genes, including TRAP, cathepsin K, and DC-STAMP (Takatsuna et al., 2004; Thummuri et al., 2017). In our results, we observed that ovotransferrin decreases the expression of c-Fos, NFATc1 and cathepsin K in RANKL-stimulated osteoclast precursors. These results suggest that ovotransferrin inhibits osteoclastogenesis by decreasing the expression of c-Fos, NFATc1, and its downstream regulators that are crucial for osteoclastogenesis.

In addition to inhibit the osteoclast formation, pharmaceutical agents such as bisphosphonates and tamoxifen also inhibit osteoclastic bone resorption, at least in part, by inducing osteoclast apoptosis via different mechanisms (Drake et al., 2008; Hughes et al., 1996). Bisphosphonates can bind to and inhibit the activity of farnesyl pyrophosphate synthase, a key regulatory enzyme in mevalonic acid pathway, and inhibited the posttranslational modification of several specific proteins (e.g., the

small guanosine triphosphate-binding proteins Rab, Rac, and Rho), which play central roles in the regulation of core osteoclast activation, therefore ultimately leading to osteoclast apoptosis (Luckman et al., 1998). Tamoxifen, acting as estrogen receptor agonist, prevents excessive bone loss by limiting osteoclast life span through the promotion of osteoclast apoptosis by mediating TGF- β (transforming growth factor β) (Hughes et al., 1996). Thus, in this study, we also investigated whether egg white ovotransferrin may affect osteoclast function by inducing apoptosis. Here, we found that ovotransferrin could induce cell apoptosis in matured osteoclasts via affecting the gene expression of *Bcl-2* family members. The expression of anti-apoptotic protein *Bcl-2* and *Bcl-xL* in osteoclasts was decreased, while the pro-apoptotic protein *Bim* and *Bax* was increased in the presence of ovotransferrin.

Generally, apoptotic process is triggered by two pathways. One is established by ligand activation via death receptor pathway (Xing & Boyce, 2005). These death receptors belong to tumor necrosis factor (TNF) receptor superfamily, including Fas, TNF-related apoptosis-inducing ligand (TRAIL), and TNF-R1 (Mollazadeh et al., 2015). However, the regulation of the death receptor pathway in osteoclasts has several features that differ from those in other cell types. For example, TNF promotes osteoclast survival rather than death in other cells (Glantschnig et al., 2003; Lee et al., 2001) The other pathway is regulated by pro- and anti-apoptotic Bcl-2 family members and involves mitochondrial release of cytochrome c, thus which is also known as mitochondrial pathway (Xing & Boyce, 2005). The anti-apoptotic Bcl-2 family members include Bcl-2, Bcl-xL, Bcl-w, Mcl-1, A1 and Boo/Diva, which all contain Bcl-2 homology (BH) domains (Xing & Boyce, 2005). The pro-apoptotic members, depending on the composition of their BH domains, are further divided in to the BH3-only pro-apoptotic members (Bim, Bad, and Bid) and other pro-apoptotic

proteins (Bak, Bax, and Bod) (Xing & Boyce, 2005). Although the regulatory mechanisms of the Bcl-2 family underlying osteoclasts apoptosis have not been fully identified and characterized, studies have demonstrated the significant relationship between the expression of Bcl-2 family and osteoclast apoptosis. Among all Bcl-2 family members, anti-apoptotic protein Bcl-xL and proapoptotic protein Bim have been the ones mostly studied in osteoclasts. Over-expression Bcl-xL in osteoclasts *in vitro* prolongs their life span and protects them from bisphosphonate-induced cell death, suggesting the adverse effects of Bcl-xL on inducing osteoclast apoptosis and prevent bone loss (Zhang et al., 2004). However, interesting results have been found in vivo. Iwasawa et al. found that the anti-apoptotic protein Bcl-xL positively regulates osteoclast survival, but negatively regulate osteoclast activity by down-regulating extracellular matrix (ECM) proteins production and c-Src kinase activity (Iwasawa et al., 2009). However the effects only showed the inhibitor treatment (ABT-737) in the study; thus more studies are warranted to validate the effects of BclxL protein, or even the Bcl-2 family proteins on osteoclasts survival and their ultimate resorbing activity. Compared to Bcl-xL, the effect of pro-apoptotic protein Bim are consistent in different studies. The increased number of osteoclasts along with the elongated life span was observed in Bim-/- mice, which suggested the negative effect of Bim in osteoclast survival (Akiyama et al., 2003). Moreover, TGF- β 1, an important functional modulator of osteoclasts, can induce osteoclast apoptosis by affecting the expression of Bim. The expression of Bim was increased in the presence of TGF- β 1, which led to the upregulation of activated caspase 9 (Houde, Chamoux, Bisson, & Roux, 2009). In addition to Bcl-xL and Bim, Bcl-2 also plays role in osteoclast apoptosis and in the pathology of postmenopausal osteoporosis. The mRNA and protein levels of Bcl-2 in osteoclasts were distinctly increased in postmenopausal osteoporosis patients, leading to the inhibition of osteoclast apoptosis and excessive bone loss (Pang, Gong, Han, & Liu, 2018).

5.5 Conclusions

Our study established that egg white ovotransferrin could suppress RANKL-mediated osteoclastogenesis and resorption activity. Moreover, we demonstrated that the prevention effects of ovotransferrin occur through suppression of NF-κB and MAPK activation during the process of osteoclast differentiation along with the induction of cell apoptosis in matured osteoclasts. It is now clear that the balance among osteoclasts, osteoblasts and osteocytes is an important determinant of bone mass and strength. However, some of the drugs used for osteoporosis prevention by regulating osteoclastogenesis and osteoclast life span are affecting the survival of osteoblasts and/or osteocytes, and therefore is harmful to the skeleton's integrity in long term therapy. Our study found that egg white protein ovotransferrin inhibits osteoclasts differentiation and survival, suggesting its potential as a favorable nutritional approach for bone health management.

5.6 References

- Abu-Amer, Y. (2013). NF-κB signaling and bone resorption. Osteoporosis International : A Journal Established as Result of Cooperation between the European Foundation for Osteoporosis and the National Osteoporosis Foundation of the USA, 24(9), 2377–2386.
- Akiyama, T., Bouillet, P., Miyazaki, T., Kadono, Y., Chikuda, H., Chung, U. I., Tanaka, S.
 (2003). Regulation of osteoclast apoptosis by ubiquitylation of proapoptotic BH3-only Bcl-2 family member Bim. *The EMBO Journal*, *22*(24), 6653–6664.
- Alekel, D. L., Germain, A. S., Peterson, C. T., Hanson, K. B., Stewart, J. W., & Toda, T. (2000).
 Isoflavone-rich soy protein isolate attenuates bone loss in the lumbar spine of
 perimenopausal women. *The American Journal of Clinical Nutrition*, 72(3), 844–852.
- Alekel, D. L., Weaver, C. M., Ronis, M. J. J., & Ward, W. E. (2013). Nutritional Influences on Bone Health and Overview of Methods. *Bioactive Food as Dietary Interventions for the Aging Population*, 357–370.
- Bonjour, J. P. (2005). Dietary protein: An essential nutrient for bone health. *Journal of the American College of Nutrition*, 24(6), 526S–536S.
- Boyle, W. J., Simonet, W. S., & Lacey, D. L. (2003). Osteoclast differentiation and activation. *Nature*, *423*(6937), 337–342.
- Brandwood, C. P., Hoyland, J. A., Hillarby, M. C., Berry, J. L., Davies, M., Selby, P. L., & Mee,
 A. P. (2003). Apoptotic gene expression in Paget's disease: a possible role for Bcl-2. *The Journal of Pathology*, 201(3), 504–512.
- Brown, K. (2009). Is tamoxifen a genotoxic carcinogen in women? *Mutagenesis*, 24(5), 391–404.
- Cornish, J., Callon, K. E., Naot, D., Palmano, K. P., Banovic, T., Bava, U., Reid, I. R. (2004).

Lactoferrin is a potent regulator of bone cell activity and increases bone formation in vivo. *Endocrinology*, *145*(9), 4366–4374.

- Drake, M. T., Clarke, B. L., & Khosla, S. (2008). Bisphosphonates: mechanism of action and role in clinical practice. *Mayo Clinic Proceedings*, *83*(9), 1032–1045.
- Elmore, S. (2007). Apoptosis: A Review of Programmed Cell Death. *Toxicologic Pathology*, *35*(4), 495–516.
- Glantschnig, H., Fisher, J. E., Wesolowski, G., Rodan, G. A., & Reszka, A. A. (2003). M-CSF, TNFα and RANK ligand promote osteoclast survival by signaling through mTOR/S6 kinase. *Cell Death & Differentiation*, *10*(10), 1165–1177.
- Houde, N., Chamoux, E., Bisson, M., & Roux, S. (2009). Transforming growth factor-beta1 (TGF-beta1) induces human osteoclast apoptosis by up-regulating Bim. *The Journal of Biological Chemistry*, 284(35), 23397–23404.
- Hughes, D. E., & Boyce, B. F. (1997). Apoptosis in bone physiology and disease. *Molecular Pathology : MP*, 50(3), 132–137. Retrieved from
- Hughes, D. E., Dai, A., Tiffee, J. C., Li, H. H., Mundy, G. R., & Boyce, B. F. (1996). Estrogen promotes apoptosis of murine osteoclasts mediated by TGF-beta. *Nature Medicine*, 2(10), 1132–1136. Retrieved from
- Hung, R. W., & Chow, A. W. (1997). Apoptosis: Molecular mechanisms, regulation and role in pathogenesis. *The Canadian Journal of Infectious Diseases = Journal Canadien Des Maladies Infectieuses*, 8(2), 103–109.
- Iwasawa, M., Miyazaki, T., Nagase, Y., Akiyama, T., Kadono, Y., Nakamura, M., Tanaka, S. (2009). The antiapoptotic protein Bcl-xL negatively regulates the bone-resorbing activity of osteoclasts in mice. *The Journal of Clinical Investigation*, *119*(10), 3149–3159.

- Karin, M. (2005). Inflammation-activated Protein Kinases as Targets for Drug Development. *Proceedings of the American Thoracic Society*, 2(4), 386–390.
- Kerstetter, J. E., & Insogna, K. L. (2003). Dietary protein, calcium metabolism, and skeletal homeostasis. *The American Journal of Clinical Nutrition*, 78, 584–592.
- Kini, Usha; Nandeesh, B. N. (2012). *Physiology of Bone Formation, Remodeling, and Metabolism. Radionuclide and Hybrid Bone Imaging* (Vol. 9783642024).
- Lee, K., Chung, Y. H., Ahn, H., Kim, H., Rho, J., & Jeong, D. (2016). Selective Regulation of MAPK Signaling Mediates RANKL-dependent Osteoclast Differentiation. *International Journal of Biological Sciences*, 12(2), 235–245.
- Lee, S. E., Chung, W. J., Kwak, H. B., Chung, C.-H., Kwack, K., Lee, Z. H., & Kim, H. H. (2001). Tumor Necrosis Factor-α Supports the Survival of Osteoclasts through the Activation of Akt and ERK. *Journal of Biological Chemistry*, *276*(52), 49343–49349.
- Lorget, F., Clough, J., Oliveira, M., Daury, M. C., Sabokbar, A., & Offord, E. (2002). Lactoferrin reduces in vitro osteoclast differentiation and resorbing activity. *Biochemical and Biophysical Research Communications*, 296(2), 261–266.
- Luckman, S. P., Hughes, D. E., Coxon, F. P., Russell, R. G. G., Rogers, M. J., & Rogers, M. J. (1998). Nitrogen-Containing Bisphosphonates Inhibit the Mevalonate Pathway and Prevent Post-Translational Prenylation of GTP-Binding Proteins, Including Ras. *Journal of Bone and Mineral Research*, 13(4), 581–589.
- Matsumoto, M., Sudo, T., Saito, T., Osada, H., & Tsujimoto, M. (2000). Involvement of p38
 Mitogen-activated Protein Kinase Signaling Pathway in Osteoclastogenesis Mediated by
 Receptor Activator of NF-κB Ligand (RANKL). *Journal of Biological Chemistry*, 275(40), 31155–31161.

- Mee, A. P. (1999). Paramyxoviruses and Paget's disease: the affirmative view. *Bone*, *24*(5 Suppl), 19S–21S.
- Messina, M., & Messina, V. (2000). Soyfoods, soybean isoflavones, and bone health: a brief overview. Journal of Renal Nutrition : The Official Journal of the Council on Renal Nutrition of the National Kidney Foundation.
- Mollazadeh, S., Fazly Bazzaz, B., & Kerachian, M. (2015). Role of apoptosis in pathogenesis and treatment of bone-related diseases. *Journal of Orthopaedic Surgery and Research*, *10*(1), 15.
- Pang, T., Gong, M., Han, J., & Liu, D. (2018). Relationship between osteoporosis and expression of Bcl-2 and CXCL12. *Experimental and Therapeutic Medicine*, 15(2), 1293–1297.
- Perazella, M. A., & Markowitz, G. S. (2008). Bisphosphonate nephrotoxicity. *Kidney International*, 74(11), 1385–1393.
- Roux, S., Lambert-Comeau, P., Saint-Pierre, C., Lépine, M., Sawan, B., & Parent, J. L. (2005).
 Death receptors, Fas and TRAIL receptors, are involved in human osteoclast apoptosis.
 Biochemical and Biophysical Research Communications, 333(1), 42–50.
- Shang, N., & Wu, J. (2018). Egg White Ovotransferrin Shows Osteogenic Activity in Osteoblast Cells.
- Takatsuna, H., Asagiri, M., Kubota, T., Oka, K., Osada, T., Sugiyama, C., Umezawa, K. (2004). Inhibition of RANKL-Induced Osteoclastogenesis by (–)-DHMEQ, a Novel NF-κB Inhibitor, Through Downregulation of NFATc1. *Journal of Bone and Mineral Research*, *20*(4), 653–662.
- TANAKA, S., Miyazaki, T., Fukuda, A., Akiyama, T., Kadono, Y., Wakeyama, H., Nakamura,K. (2006). Molecular Mechanism of the Life and Death of the Osteoclast. *Annals of the New*

York Academy of Sciences, 1068(1), 180–186.

- Teitelbaum, S. L. (2004). RANKing c-Jun in osteoclast development. *The Journal of Clinical Investigation*, *114*(4), 463–465.
- Thummuri, D., Jeengar, M. K., Shrivastava, S., Nemani, H., Ramavat, R. N., Chaudhari, P., & Naidu, V. G. M. (2015). Thymoquinone prevents RANKL-induced osteoclastogenesis activation and osteolysis in an in vivo model of inflammation by suppressing NF-KB and MAPK Signalling. *Pharmacological Research*, 99, 63–73.
- Thummuri, D., Naidu, V. G. M., & Chaudhari, P. (2017). Carnosic acid attenuates RANKLinduced oxidative stress and osteoclastogenesis via induction of Nrf2 and suppression of NF-κB and MAPK signalling. *Journal of Molecular Medicine*, *95*(10), 1065–1076.
- Wei, C. M., Liu, Q., Song, F. M., Lin, X. X., Su, Y. J., Xu, J., Zhao, J. M. (2018). Artesunate inhibits RANKL-induced osteoclastogenesis and bone resorption in vitro and prevents LPSinduced bone loss in vivo. *Journal of Cellular Physiology*, 233(1), 476–485.
- Woo, K. H., Kim, H. M., & Ko, J. S. (2002). Macrophage colony-stimulating factor promortes the survival of osteoclast precursors by upregulating Bcl-X(L). *Experimental & Molecular Medicine*, 34(5), 340–346.
- Wu, J., & Acero-Lopez, A. (2012). Ovotransferrin: Structure, bioactivities, and preparation. Food Research International, 46(2), 480–487.
- Xing, L., & Boyce, B. F. (2005). Regulation of apoptosis in osteoclasts and osteoblastic cells. Biochemical and Biophysical Research Communications, 328(3), 709–720.
- Zhang, Q., Badell, I. R., Schwarz, E. M., Boulukos, K. E., Yao, Z., Boyce, B. F., & Xing, L.
 (2005). Tumor necrosis factor prevents alendronate-induced osteoclast apoptosis in vivo by stimulating Bcl-xL expression through Ets-2. *Arthritis & Rheumatism*, 52(9), 2708–2718.

Zhang, Q., Badell, I., Schwarz, E. M., Boyce, B. F., & Xing, L. (2004). TNF Protects Osteoclasts from Alendronate - Induced Apoptosis by Stimulating Bcl - xL Expression through the

Transcription Factor, Ets 2. Journal of Bone and Mineral Research, 19(S1), S171.

Figure 5.1 Effect of ovotransferrin on RANKL-induced osteoclastogenesis.

(A) RAW 264.7 cells were incubated with RANKL (100 ng/mL) and various concentrations of ovotransferrin (OVT) for 4 days and then TRAP staining was performed. (B) TRAP⁺ multinucleated cells with at least three nuclei were counted as osteoclasts. Data were expressed as means \pm SEM (n=8). * and *** indicated p < 0.05 and p < 0.001.

А



В



Figure 5.2 Effect of ovotransferrin on RANKL-induced NF-KB activation.

RAW 264.7 cells were pretreated with ovotransferrin (1-1000 μ g/mL) for 30min prior to stimulation with RANKL (100 ng/mL). Then RAW 264.7 cells were treated with both RANKL and ovotransferrin for 4 h. Whole cell lysates were used for western blotting analysis of p-NF- κ B p65 (ser 536)/NF- κ B p65 (A), p-I κ B (ser 32)/I κ B (B), and p-IKK α/β (ser 176/180)/IKK α +IKK β (C). All bands were shown in (D). Data were expressed as means ± SEM (n=6). Means without a common letter indicated p<0.05.

В

А

150p-NF-кВ p65 (Ser 536) / NF-кВ p65 150p-lkBα (ser 32) / lkBα (% RANKL[†]/ovt⁻) (% RANKL[†]/OVT⁻) d 100b h 100 50· 50 n RANKL + RANKL + + + + + + + + оут OVT 1000 1000 100 1 100 10 1 10 --(µg/mL) (µg/mL) С D RANKL OVT





Figure 5.3 Effect of ovotransferrin on RANKL-induced MAPK activation.

RAW 264.7 cells were pretreated with ovotransferrin (1-1000 μ g/mL) for 30min prior to stimulation with RANKL (100 ng/mL). Then RAW 264.7 cells were treated with both RANKL and ovotransferrin for 4 h Whole cell lysates were used for western blotting analysis of p-p44/42 Erk1/2 (Thr 202/Tyr 204)/p44/42 Erk1/2 (A), p-p38 (Thr 180/Tyr182)/p38 (B), and p-JNK (Thr 183/Tyr 185)/JNK (C). All bands were shown in (D). Data were expressed as means ± SEM (n=6). Means without a common letter indicated p<0.05.



Figure 5.4 Effect of ovotransferrin on RANKL-induced expression of proteins involved in osteoclastogenesis.

RAW 264.7 cells were pretreated with ovotransferrin (1-1000 μ g/mL) for 2 h prior to stimulation with RANKL (100 ng/mL). Then RAW 264.7 cells were treated with both RANKL and ovotransferrin for 12 h. Whole cell lysates were used for western blotting analysis of TRAF6 (A), cFos (B), NFATc1 (C) and cathepsin K (D). All bands were shown in (E). Data were expressed as means ± SEM (n=6). Means without a common letter indicated p<0.05.

А

В





+

1

С







Figure 5.5 Effect of ovotransferrin on RANKL-induced bone resorption in vitro.

RAW 264.7 were cultured onto fluoresceinated calcium phosphate (CaP)-coated plate and stimulated with RANKL (100 ng/mL). After 1day for adhesion, cells were treated with ovotransferrin (1-1000 μ g/mL) and RANKL together for 6 days. (A) The fluorescence intensity of resorbed CaP during 6-day treatment. * indicated p < 0.05. (B) The total resobed CaP after 6 days. (C) The total area of resorption pits after 6 days. (D) The image of resorption area. Data were expressed as means ± SEM (n=4). Means without a common letter indicated p<0.05.







RANKL -OVT -



RANKL + OVT 1000 μg/mL



RANKL + OVT 100 μg/mL

RANKL + OVT 10 μg/mL

RANKL + OVT 1 μg/mL

Figure 5.6 Effect of ovotransferrin on RANKL-induced osteoclasts apoptosis.

Osteoclasts were generated by stimulating RAW 264.7 with RANKL (100 ng/mL) for 4-6 days and multinucleated cells were identified by microscopy. Then cells were treated with ovotransferrin (1-1000 μ g/mL) for 12 h. (A) The osteoclasts apoptosis percentage. (B) The osteoclasts apoptosis measured by flow cytometry. The lower left quadrand: PI-/V-FITC-; upper left quadrand: PI+/V-FITC-; lower right quadrand: PI-/V-FITC+; upper right quadrand: PI+/V-FITC+. (C) The gene expression of Bcl-2 family (*Bcl-2, Bcl-x, Bim, Bid*). Data were expressed as means ± SEM (n=4). Means without a common letter indicated p<0.05.* and ** indicated p<0.05 and p<0.01.





С
5.7 Supplementary data and information

Table 5	.7.1	Primers	designed	for	aPCR
I GOIC C		I I IIII U I J	acoignea	101	91 OI

Gene	Primer Sequ	ences (5'-3')
Bcl-2	Forward	TGAACCGGCATCTGCACAC
	Reverse	CGTCTTCAGAGACAGCCAGGAG
Bcl-x	Forward	GCTGGGACACTTTTGTGGAT
	Reverse	TGTCTGGTCACTTCCGACTG
Bax	Forward	ACCAGCTCTGAACAGATCATG
	Reverse	ACTTTAGTGCACAGGGCCTTG
Bim	Forward	CTTCCATACGACAGTCTC
	Reverse	AACCATTTGAGGGTGGTCTTC
β -actin	Forward	AGATGTGGATCAGCAAGCAG
	Reverse	GCGCAAGTTAGGTTTTGTCA

CHAPTER 6 – Oral Administration of Egg White Ovotransferrin Prevents Osteoporosis in Ovariectomized Rat

6.1 Introduction

Osteoporosis is a multifactorial systemic skeletal disease, characterized by decreased bone mineral density and micro-architectural deterioration of bone tissue, resulting in increased bone fragility and susceptibility to fracture (Tu et al., 2018). As an aging-related disease, osteoporosis is a major health and economic burden with the extension of average life span. Although both men and women will suffer from bone loss during their life, women are more susceptible to osteoporosis and osteoporotic fractures, especially postmenopausal women (Ji & Yu, 2015). Healthy bone requires continuous remodeling which is pivotal for bone density maintenance (Ji & Yu, 2015); however, the normal bone turnover cycle can be impaired and result in bone loss. Several factors are associated with osteoporosis such as decreased level of sex hormones (e.g. estrogen, testosterone) (de Villiers et al., 2013; Rahman, Bhattacharya, Banu, Kang, & Fernandes, 2009) and growth factors (e.g. insulin-like growth factor, fibroblast growth factor) (Fromigué, Modrowski, & Marie, 2004; Giustina, Mazziotti, & Canalis, 2008; Rosen, Donahue, & Hunter, 1994), drug side effects (e.g. glucocorticoid-induced osteoporosis) (Smith et al., 2000), lack of essential nutrients (e.g. mineral, vitamin) (Nieves, 2005), as well as acquired habit, such as smoking and alcohol intake (Wilsgaard et al., 2009).

Current osteoporosis therapies rely on either anti-resorptive agents like bisphosphonates, selective estrogen receptor modulator (SERMs), denosumab, or an anabolic agent like parathyroid hormone (PTH) analogs (Tu et al., 2018). However, these therapies are associated with side effects such as

gastrointestinal toxicity (Drake, Clarke, & Khosla, 2008), and increased risk of cancer, strokes and cardiovascular system diseases (Crockett & Das, 2013). The presence of numerous side effects have prompted further research to explore alternatives, especially food derived bioactive components.

In the recent decades, food-derived bioactive components are gaining the momentum towards the development of new therapies for osteoporosis prevention and treatment. The potential of soy isoflavones has been demonstrated in a number of studies (Arjmandi et al., 2005; Kenny et al., 2009; Messina & Messina, 2000). Evidence from epidemiologic studies also supported that dietary soy isoflavones attenuated menopause-induced osteoporotic bone loss by decreasing bone resorption and stimulating bone formation (Lauderdale et al., 1997; Silverman & Madison, 1988). Lactoferrin, an iron binding protein, has gained attention for its ability to prevent osteoporosis by stimulating bone formation while inhibiting bone resorption (Cornish et al., 2004; Grey, Zhu, Watson, Callon, & Cornish, 2006; Guo et al., 2009). Moreover, as a well-studied bioactive protein, lactoferrin is also able to benefit for bone health by promoting the differentiation of pluripotent mesenchymal cells into osteoblastic and chondroblastic lineage (Yagi et al., 2009) and inhibiting the production of inflammatory cytokines, such as IL-6 and TNF- α (Conneely, 2001), suggesting a multifunction of bioactive proteins in regulating bone health.

Egg white ovotransferrin, a 76 kDa iron-binding glycoprotein belonging to the transferrin family, has been found to offer similar capabilities as lactoferrin on bone health management (Shang & Wu, 2018; Wu & Acero-Lopez, 2012). Ovotransferrin not only potently stimulates the proliferation and differentiation of osteoblasts, but also inhibits osteoclastogenesis and resorption

activity. These results suggested the potential of ovotransferrin in preventing osteoporosis. Moreover, as a major functional protein in egg white, ovotransferrin has been widely studied for its multiple biological functions such as immunomodulatory, anti-inflammatory activity and antioxidative activity (Wu & Acero-Lopez, 2012), which may also play important roles in maintaining bone integrity. Although orally administered proteins are hydrolyzed in the gastrointestinal tract and thus may lose their biological activity, oral uptake of ovotransferrin has been found to exhibit similar physiologic effects *in vivo* as displayed *in vitro*, such as its anti-inflammatory activity (Kobayashi et al., 2015). However, all our previous studies were conducted in experiments that allowed direct contact of ovotransferrin with bone cells. To the best of our knowledge, the efficacy of orally administered ovotransferrin on bone health has yet to be examined.

The *in vivo* effects of egg white ovotransferrin on preventing menopausal-induced osteoporosis were investigated using ovariectomized (OVX) rats. The objectives of this work were 1) to evaluate the safety (based on organ measurements) and effects of ovotransferrin treatment on osteoporosis prevention by measuring bone mineral density and micro-architecture; 2) to investigate the regulatory role of ovotransferrin on bone metabolism by measuring the expression of bone formation/resorption biomarkers (representing the bone remodeling process), and the production of osteoclastogenesis associated cytokines; 3) to investigate the auxiliary functions of ovotransferrin on bone health management including the regulatory role on adipogenesis, phenotypes of immune cells in bone marrow cells (representing local immunity) and spleen (representing systemic immunity), and short chain fatty acid production (representing gut microbiota).

6.2 Materials and methods

6.2.1 Chemicals

Alendronate sodium was purchased from Sigma-Aldrich (Oakville, ON, Canada) and diluted in water for injection. Dulbecco's modified Eagle's medium (DMEM), fetal bovine serum (FBS), bovine serum albumin (BSA) and penicillin-streptomycin were purchased from Sigma-Aldrich (Oakville, ON, Canada). The Krebs-Ringer-HEPES (KRH) buffer and ACK lysis buffer was prepared according to the previous study (Gars et al., 2018). Fluorescent pre-labeled monoclonal antibodies CD3, CD4, CD80, CD28 were purchased from eBioscience (San Diego, CA, USA), antibodies CD8, CD25, CD71, OX62, CD11b/c, CD51 were purchased from Biolegend (Vindeland, ON, Canada), and antibodies RANK, OX12 and OX6 were purchased from BD Biosciences Pharmingen (Mississauga, ON, Canada). All primers were synthesized by Invitrogen (Burlington, ON, Canada) and the sequences are shown in Table 6.7.4. AIN-93M purified animal diets and casein were purchased from Envigo (Madison, WI, USA). The nutritional characteristics of experimental diets is shown in Table 6.7.5.

6.2.2 Experimental protocol

This experiment protocol was approved by the University of Alberta Animal Welfare Committee (Protocol AUP00001960) in accordance with the guidelines issued by the Canadian Council on Animal Care and also adhered to the Guide for the Care and Use of Laboratory Animals published by the United States National Institutes of Health. The experiment was carried out using 12-week-old virgin female Sprague-Dawley rats weighing 292 ± 20 g obtained from Charles River (Senneville, QC, Canada), in which 40 rats were ovariectomized (OVX) and 8 rats were shamoperated (Sham). Upon arrival, the total of 48 rats were housed at University of Alberta animal facility with a 12:12 h cycle of light: dark in a humidity and temperature controlled (60% RH, and

23°C) environment and maintained on diet and water *ad libitum*. In addition to the sham group (n=8), another 40 OVX rats were randomly divided into 5 dietary groups (n=8/group); 1) negative control (NC), 2) positive control (PC, injected alendronate sodium 3 times/week), 3) high ovotransferrin treatment (HOvt, a normal diet supplemented with 1% w/w ovotransferrin), 4) medium ovotransferrin treatment (MOvt, a normal diet supplemented with 0.2% w/w ovotransferrin and 0.8% w/w casein), 5) low ovotransferrin treatment (LOvt, a normal diet supplemented with 0.04% w/w ovotransferrin and 0.96% w/w casein). The doses of ovotransferrin were designed according to the published literatures (Blais, Malet, Mikogami, Martin-Rouas, & Tome, 2009; Guo et al., 2009; Hou, Xue, & Lin, 2012; Malet et al., 2011; Yoshimaki et al., 2013). Sham, NC and PC groups were provided with a normal diet supplemented with 1% w/w casein to exclude the nutritional effect of ovotransferrin as a protein. After 12 weeks, rats were sacrificed and tissues were collected. The diagram of an experimental protocol is shown in Fig. 1.

6.2.3 Sample collection

Rats were weighed every 2 days and food intake was measured 3 times per week. Serum and fecal samples were collected every 4 weeks and stored at -80°C. After 12 weeks of treatment, the rats were placed in metabolic cage for urine collection. During sacrifice, the serum was collected immediately after euthanasia. Tissues were removed, weighted and stored at -80°C until analysis.

6.2.4 Micro-computerized tomography analysis and bone mineral density determination

The right tibia was measured by micro-computerized tomography analysis (micro-CT) using Skyscan 1076 (Kontich BE, Belgium) at 18 μ m resolution with 70kV, 100 μ A with a 1.0mm Aluminum filter. Briefly, inhalant isofluorane was used to induce animal anaesthesia by gradually increasing the concentration of isofluorane from 0.5% to 2% in 100% oxygen. Then animal was placed on the bed of micro-CT scanner face down and the right leg was straightened and

immobilized with tapes. Gently pushed the bed into the scanner and performed a Scout Scan of the animal to get the images of tibia. Projected images were reconstructed using vendor-supplied software Nrecon 1.6.1.5 (SkyScan NV, Kontich BE) with images oriented in axial plane of section. A circular region of interest (ROI) relative to the size of the tibia bone was used to sample a standardized amount of trabecular bone in the vertebral body in order to conduct standard histomorphometric analysis. The tibia growth plate was utilized as anatomical landmarks in order to consistently segment and sample the same region of trabecular bone from all samples. Vendor morphometric software CTAn 1.10.0.1 (SkyScan, Kontich BE) was then used to analyze tibia trabecular microarchitecture parameters (percentage bone volume, trabecular thickness, trabecular numbers and trabecular separation) and tissue mineral density against the vendor-supplied bone phantoms of known mineral density (0.25g/cm³ and 0.75g/cm³) to calibrate BMD values from Hounsfield units.

6.2.5 Bone biochemical markers and cytokine measurement

Serum and urinary calcium were measured in triplicate with a calcium assay kit (Abcam, Toronto, ON, Canada) and urinary creatinine was measured in triplicate with a creatinine assay kit (Abcam, Toronto, ON, Canada). Serum osteocalcin, parathyroid hormone (PTH), and urinary pyridinoline (PYD) were measured in duplicate with ELISA kit (Quidel, San Diego, CA, U.S.A.). Serum bone alkaline phosphatase (BALP) and procollagen I intact N-terminal (PINP) were measured in duplicate with ELISA kit (Abcam, Toronto, TA, U.S.A.). Serum TNFα and IL-6 were measured in triplicate with ELISA kit (Abcam, Toronto, ON, Canada).

6.2.6 Isolation of bone marrow cells

Bone marrow was harvested by inserting a syringe needle (23 gauge) into one end of the femur and flushed with DMEM containing 10% FBS and 5% pen-strep. Marrow cells were centrifuged and treated with ACK lysis buffer for 5 min at room temperature, then washed and centrifuged. Cells were re-suspended in fresh DMEM and the total counting was determined with a haemocytometer. The extracted bone marrow cells were used for immune cells phenotype determination and osteoclast differentiation. Leftover cell pellets were transferred to 1.5 mL microcentrifuge tubes, rinsed with PBS and storage at -80°C for qRT-PCR and western blot analysis.

6.2.7 Isolation of spleen cells

At necropsy, spleens were removed and placed in sterile Krebs-Ringer-HEPES (KRH) buffer (pH 7.4) supplemented with 0.5% (w/v) BSA. Spleen cells were extracted by push spleen through the cell strainers (40 µm) using barrel of sterile syringe. Isolated spleen cells were treated with ACK lysis buffer to remove excess red blood cells for 5 min on ice. Cells were washed and re-suspended in the KRH buffer and counted on a haemocytometer. The extracted spleen cells were used for phenotype determination.

6.2.8 Bone marrow cells and spleen cells phenotype analysis

Freshly isolated immune cells from both bone marrow and spleen were analyzed using a direct four-color labelled immunofluorescence assay. Cells were incubated with pre-labelled monoclonal antibodies to identify and quantify the phenotypes. The 96 well V-bottom plate was pre-coated with 200 μ L of 4% FBS in PBS for 30 min at room temperature. 20-40 μ L (1×10⁵~4×10⁵ cells/well) bone marrow cells or spleen cells were added into V-bottom plate and stained with antibodies according to manufacturer's recommendations. The antibody combinations are shown in Table 6.7.5. After 30 min incubation in dark, cells were fixed in 4% paraformaldehyde, and the proportion of positive cells for each marker was determined according to the relative fluorescence

intensity by flow cytometry Canto II (BD Biosciences, Mississauga, ON, Canada) using the FlowJo software (FlowJo LLC, Ashland, OR, USA).

6.2.9 Osteoclast differentiation and resorptive activity analysis

Bone marrow cells were seeded in 24-well plate at a density of $1 \times 10^4 \sim 5 \times 10^4$ cells/well and cultured with DMEM containing 10% FBS, 5% pen-strep and 10 ng/mL M-CSF for 2 days. Then, the culture medium was replaced with DMEM containing 10 ng/mL M-CSF and 100 ng/ml RANKL to stimulate osteoclastogenesis. Osteoclasts were successfully differentiated by incubation with M-CSF and RANKL after 4 days. TRAP-staining kit was used to confirm and count the multinucleated osteoclast-like cells according to the manufacturer's instruction. Simply, cells were fixed in 4% paraformaldehyde for 1 h in 4°C and then stained with TRAP staining solution (0.1mg/mL naphthol AS-MX phosphate and 0.3 mg/mL Fast Red Violet LB staining). Cells were observed with a light microscope under the 10×lense (Olympus IX83, Richmond Hill, ON, Canada) and images were captured by Metamorphy (Olympus, Richmond Hill, ON, Canada). Any TRAP-positive multinucleated cells containing three or more nuclei were counted.

6.2.10 RNA extraction and qRT-PCR

Total RNA was extracted from the cells using TRIzol[®] reagent following the manufacturer's instruction, and cDNA was synthesized using SuperScript II Reverse Transcriptase (Invitrogen, Burlington, ON, Canada) according to the manufacturer's instruction. Real-time PCR was performed using Fast SYBR Green PCR Master Mix (Applied Biosystem, Burlington, ON, Canada) and an ABI 7300 Sequencing Detection System (Applied Biosystem, Burlington, ON, Canada). The amplification conditions were as follow: initial denaturation at 95°C for 10 min, followed by 35 cycles of 10 s at 95°C, 15 s at 60°C, and 10 s at 72°C.

6.2.11 Western blot

Whole-cell protein extracts were prepared from the bone marrow cells. The cells were lysed in boiling hot Laemmle's buffer containing 50 μ M dithiothreitol (DTT) and 0.2% Triton-X-100 to prepare samples for western blot as described previously. The cell lysates were run in SDS-PAGE, blotted to nitrocellulose membranes and immunoblotted with antibodies. The concentration of each antibody was determined according to the manufacturer's instruction. The protein bands were detected by a Licor Odyssey BioImager and quantified by densitometry using corresponding software (Licor Biosciences, Lincoln, NB, USA). Each band was normalized to its corresponding band of a loading control. The results were expressed as percentage of the corresponding negative control or untreated control.

6.2.12 Short chain fatty acid analysis

Three to four replicates of frozen fecal samples (50 mg) were weighed into a 2 mL polypropylene tube and dissolved in 25% phosphoric acid. After centrifugation, 0.8 mL of supernatant sample was added a 1.8 mL GC vial, mixed 0.2 mL of 25% phosphoric acid, and 0.2 mL of internal standard solution. Standards were prepared by combining 1 mL of standard solution and 0.2 mL of internal standard solution (23.253 μ mol/mL isocaproic acid). For GC analysis, 1 μ L of the sample was injected to 1177 Split/Splitless injector or 0.2 μ L for 1079 PVT injector on Stabliwax-DA meter, 0.53 mm ID, 0.5 μ m df (Restek Corporation, Bellefonte, PA, USA). Injection temperature and detector temperature were 250°C. Helium was used as carrier gas at a constant flow rate of 20 mL/min.

6.2.13 Statistical analysis

Data are presented as mean±SEM (standard error of mean) of between 4 and 7 independent experiments. Data were analyzed using Two Way Analysis of Variance (ANOVA) with Tukey's

multiple comparisons test or One Way Analysis of Variance (ANOVA) with Dunnett's post-hoc comparisons test. The PRISM 6 statistical software (GraphPad Software, San Diego, CA) was used for the analyses. P< 0.05 was considered significant.

6.3 Results

6.3.1 Ovotransferrin showed no significant effects on body weight, food intake and tissue weight

A diagram of the experimental design is shown in Figure 6.1. All rats had increased body weight by the end of the experiment (Figure 6.2A). OVX rats gained significantly higher weight than that of the Sham rats. High dose ovotransferrin treatment prevented the OVX-induced weight gain. However, medium dosage and low dosage had no such influence on the body weight (Table 6.7.1). As expected, OVX caused atrophy of uterine tissue, indicating the success of the surgical procedure (Figure 6.2C and Table 6.7.2). However, treatment with ovotransferrin and drug alendronate sodium had no influence on the uterine weight. No significant differences in food intake were observed in different dietary groups (Figure 6.2B). Tissue weight was not affected by the treatments (Table 6.7.2)

6.3.2 Ovotransferrin administration prevented the decrease of bone mineral density in OVX rats

OVX rats successfully developed osteoporosis with a significant decrease in tibia trabecular bone mineral density with increasing time (Figure 6.3A), but did not affect the mineral density in tibia cortical bone (Fig. 6.3C, D). Over 12-week treatment, both PC and HOvt groups significantly retarded the decline of trabecular bone mineral density compared to negative control (NC) group

(Figure 6.3A, B), although the bone mineral density value was still less than Sham group. OVX rats supplemented with casein (NC) did not counteract the OVX-induced bone mineral density loss, suggesting that the protection role of ovotransferrin is not due to the nutritional role of food protein. OVX rats treated with medium dose and low dose of ovotransferrin did not show prominent effects on preventing BMD loss (Figure 6.3B).

6.3.3 Ovotransferrin administration preserved bone micro-architecture in OVX rats

In addition to improving bone mineral density, ovotransferrin also prevented the deterioration of bone micro-architecture in tibia trabecular bones caused by ovariectomy. Ovariectomy markedly reduced the percent bone volume, trabecular thickness, trabecular number, and increased the trabecular separation on tibia compare to the Sham rats (Figure 6.4A-D), indicating a serious damage to bone microstructure and integrity. Ovotransferrin administration significantly preserved the OVX-induced decrease of percent bone volume, trabecular thickness, and trabecular number, but not the elevation of trabecular separation (Figure 6.4A-D). Similar results were also observed in drug injection group (PC). Direct evidences were observed in the reconstructed three-dimensional trabecular images (Figure 6.4E). The plate-like structure mostly resolved into a rod-like structure, with lots of the connecting rods missing in OVX rats (NC), whereas in the ovotransferrin group, this loss of trabecular bone mass and connectivity was prevented (Figure 6.4E). Microstructure, which deteriorated substantially after OVX, was markedly improved by ovotransferrin.

6.3.4 Ovotransferrin administration regulated bone formation and bone resorption in OVX rats

To investigate the regulatory role of ovotransferrin on bone remodeling, serum and urine biomarkers, which indicate bone formation and bone resorption, were measured (Table 6.1). Serum calcium (Ca^{2+}) was not affected, while the urinary Ca^{2+} was significantly increased in OVX rats compared to the Sham group. Interestingly, ovotransferrin administration attenuated the increase of urinary Ca2+ in OVX rats without affecting serum Ca2+. Additionally, the concentrations of serum bone alkaline phosphatase (BALP) and urinary pyridinoline (PYD), the markers of bone formation and bone resorption respectively, were higher in the OVX rats (NC) than those of the Sham rats. Due to the interplay between bone resorption and formation, the exceeding bone resorption caused by OVX will also lead to a stimulation of bone formation correspondingly, but this stimulation is trivial. Both ovotransferrin administration and drug injection (PC) showed a suppression in serum BALP and urinary PYD, suggesting the inhibitory effects on OVX-induced overactive bone turnover and therefore help rehabilitate normal bone remodeling. The effect of ovotransferrin on preventing bone resorption was also evidenced by a decreased serum osteocalcin level after ovotransferrin administration. Although osteocalcin is regarded as a maker of bone formation, when bone resorbed, osteocalcin fragments are released, and the serum concentration of osteocalcin fragments reflect bone resorption (Cundy, Reid, & Grey, 2014). Generally, there is a two-fold rise in plasma osteocalcin concentration at the menopause, consistent with our results. Serum N-terminal propertide of type I collagen (PINP), a biomarker indicating collagen synthesis, was also evaluated as an indicator of bone formation. As expected, OVX showed significant inhibition in serum PINP, while ovotransferrin administration elevated the concentration. Serum PTH level in high dose ovotransferrin-treated OVX rats were

dramatically increased than other groups. PTH is the only available anabolic drug used to stimulate bone formation in osteoporosis patients. The increased serum PTH level with ovotransferrin treatment may also suggest the possible ability of ovotransferrin in promoting bone formation.

6.3.5 Ovotransferrin inhibited the production of serum cytokines in OVX rats

Bone resorption is tightly regulated by multiple cytokines, especially TNF α and IL-6, which could stimulate osteoclasts differentiation (Azuma, Kaji, Katogi, Takeshita, & Kudo, 2000; Wu, Zhou, Huang, Ji, & Kang, 2017). In our study, serum TNF α was markedly greater in the negative control group than in Sham group (Figure 6.5A). With 12-week of ovotransferrin administration, the serum TNF α level did not show significant increase compared to the Sham group, indicating that ovotransferrin is able to inhibit ovariectomy-caused high TNF α producing. Interestingly, ovariectomy did not cause a significant increase of IL-6. The serum IL-6 concentrations were similar in negative control (NC) and Sham groups; however, both high dose ovotransferrin treatment and drug intervention led to a lower serum IL-6 level (Figure 6.5B).

6.3.6 Osteoclast precursors derived from ovotransferrin-administrated OVX rats showed less differentiation ability and resorption activity

To examine the effect of ovotransferrin on the differentiation ability of osteoclast precursors, bone marrow cells extracted from different animal groups were cultured in osteoclastogenic medium and the differentiated osteoclasts were counted with TRAP staining. After 6 days of incubation, osteoclasts were successfully differentiated from bone marrow cells (Figure 6.6A). A greater number of osteoclasts were generated from bone marrow cells extracted from negative control group (NC), compared to the Sham group (Figure 6.6A, B). Importantly, bone marrow cells

extracted from the ovotransferrin administration groups showed less mature osteoclasts, especially in the high dose ovotransferrin group. Similar results were observed in drug injection group (PC). The absorption activity of differentiated osteoclasts was investigated using a calcium phosphate (CaP)-coated plate. Ovotransferrin administration not only reduced the number of differentiated osteoclasts, but also showed less absorptive activity compared to the negative group (NC) (Figure 6.6C-E). Compared with the Sham rats, OVX rats showed higher osteoclastic resorption; whereas, ovotransferrin administration attenuated OVX-induced excessive osteoclastic resorption.

To have a better understanding on the effects of ovotransferrin on bone marrow microenvironment, the expression of cathepsin K (CathK) and matrix metalloproteinase 9 (MMP9) were measured in bone marrow cells. Bone marrow CathK and MMP9 expression has been reported to replicate in the regulation of bone metastasis, suggesting the negative association between high expression of CathK, MMP9 and bone health (Ohshiba, Miyaura, Inada, & Ito, 2003; Podgorski et al., 2009). Ovariectomy significantly increased the expression of CathK and MMP9 compared to the Sham group, while both ovotransferrin administration and drug injection reduced their expression (Figure 6.7); only the HOvt group significantly decreased both CathK and MMP9 expression.

6.3.7 Ovotransferrin inhibited osteoclastogenesis by inhibiting the NF-KB pathway

In a previous *in vitro* study, ovotransferrin showed prevention on RANKL-induced osteoclastogenesis in RAW 264.7 cell by inhibiting NF- κ B pathway (Chapter 5, this thesis). This study confirmed this finding with primary osteoclast precursors extracted from bone marrow. Osteoclasts were generated and treated with different concentration of ovotransferrin (1000-10mg/mL). Similar to a previous study, ovotransferrin treatment significantly inhibited the

expression of NF- κ B p65 and I κ B α (Figure 6.8), which are two most important components involved in the NF- κ B pathway.

6.3.8 Ovotransferrin administration exerted a global regulatory effect in OVX rats

Postmenopausal leads to various health issues in addition to osteoporosis, for example obesity and inflammation. Recent studies have demonstrated an interlinked traid among postmenopausal osteoporosis, obesity and sarcopenia (Vaidya, 2014). Thus, to have a comprehensive understanding on the beneficial effects of ovotransferrin in OVX rats, the influences on adipogenesis (Figure 6.7.1), immune modulation (Figure 6.7.2, Table 6.7.3, Table 6.7.4) and short-chain fatty acid (Figure 6.7.3, Table 6.7.5) production with ovotransferrin supplementation were investigated (all results were shown on 6.7 Supplementary Data and Information)

For adipogenesis study, we investigated the expression of several major genes and proteins involved in adipogenesis. qRT-PCR results showed that ovariectomy significantly increased the expression of peroxisome proliferator-activated receptor γ (*PPAR* γ), CCAAT-enhancer-binding protein α (*C/EBP* α), *perilipin* and adipocyte protein 2 (*aP2*) gene, indicating an activation of adipogenesis in estrogen deficiency situation (Figure 6.7.1A). Ovotransferrin administration significantly inhibited the gene expression of *PPAR* γ and *C/EBP* α , but not *perilipin* and *aP2* (Figure 6.7.1A). The protein expression of PPAR γ and C/EBP α was measured by western blot (Figure 6.7.1B, C). Increased protein expression of PPAR γ and C/EBP α were observed in OVX rats, but was attenuated at the HOvt and PC groups.

For immune modulation study, we investigated the effect of ovotransferrin administration on immune alternation, including both local bone marrow cells and systemic splenocytes. Compared with the Sham group, OVX did not cause a significant change in spleenocyte phenotypes (Table 6.7.3). Meanwhile, ovotransferrin administration did not affect immune cell populations significantly either, suggesting a lack of effect on the systemic immune system (Table 6.7.3). The significant observations were made in the bone marrow compartment. Although the number of Tcells in bone marrow was relatively low, these cells are believed to play important roles in the local regulation of bone turnover. In general, ovariectomy caused relatively remarkable changes in both T-cells and B-cells in bone marrow compared to in spleen (Figure 6.7.2 and Table 6.7.4). The percentage of total T cells (CD3+) and T cytotoxic cells (CD8+) in bone marrow were significantly decreased after ovariectomy, while ovotransferrin administration attenuated this decrease (Figure 6.7.2A, B). Meanwhile, the population of T-cell subsets, including T-cell express IL-2 receptor (CD25+), T-cell express co-stimulatory factors (CD28+), T-cell express anti-integrin alpha (CD51+), dendritic cells (OX62+), and B-cells (OX12+) was increased in OVX rats, while those of CD25+, CD28+ and OX12+ cells were suppressed upon ovotransferrin administration (Figure 6.7.2C, D, H). Inspiring findings were observed on the population changes of monocytes/macrophages (CD11b/c+) and RANK+ cells, which are the precursor cells of osteoclasts. The population of CD11b/c+ and RANK+ cells were significantly increased in OVX rats, while high dose ovotransferrin treatment reduced them notably (Figure 6.7.2G, H). These results are consistent with our previous finding that fewer osteoclasts were differentiated from bone marrow cells derived from ovotransferrin treatment groups (Figure 6.6).

In this study, the concentrations of SCFAs were measured in fecal samples. Interestingly, 12-week of ovotransferrin administration significantly increased the fecal SCFAs content in OVX rats, including acetic acid, proprionic acid, isobutyric acid, butyric acid, isovaleric acid, valeric acid, and caproic acid (Figure 6.7.3 and Table 6.7.5). Although it is not known the contribution of these findings to bone outcomes, these results may help to explain the functions of ovotransferrin in an

overall perspective and provide a broader beneficial effect of ovotransferrin in postmenopausal women.

6.4 Discussion

The effect of egg white protein ovotransferrin on osteoporosis prevention of OVX rats was studied in this study. OVX rat is a well-accepted osteoporosis model that mimics the estrogen deficiencyinduced osteoporosis in postmenopausal women. In current study, adding ovotransferrin up to 1% (w/w) did not affect food intake, weight gain, and tissue weight significantly, suggesting the potential of ovotransferrin to be developed as a functional food ingredient. Our study provides direct evidence that ovotransferrin can acts as an orally active agent to preserving bone mass and micro-architecture in a growing OVX rat model (aged 3 to 6 months). Our findings indicate that ovotransferrin at a dose of 1% (w/w) could attenuate estrogen-dependent bone loss and microarchitecture deterioration in OVX rats. The mechanical properties of bone are influenced by bone mass as well as bone size, bone quality, and bone micro-architecture. In this study, ovotransferrin treatment resulted in an increase of percent bone volume consistent with the improvement in bone mineral density, suggesting ovotransferrin could prevent the loss of bone mass after ovariectomy. Ovotransferrin administration improved trabecular architecture as it increases trabecular architecture. In addition, ovotransferrin appears to partially restore trabecular connectivity by increasing trabecular numbers. Ovotransferrin showed more effective protection against trabecular loss. Our findings suggest that ovotransferrin increases both the bone mineral density and microarchitectures, therefore, has the potential to be used as a treatment of osteoporosis.

To address the mechanisms involved, several bone remodeling biomarkers including BALP, PINP, PYD and PTH were measured. After ovariectomy, bone formation markers BALP and osteocalcin were elevated with a commensurate increase of the bone resorption marker PYD. This could be interpreted as a coupling between bone resorption and formation (Han & Wang, 2017; Sims, Morris, Moore, & Durbridge, 1996; Yokose et al., 1996). Accelerated bone resorption also leads to an trivial stimulation on bone formation. Thus, the process is dominated by bone resorption and results in bone loss. Administration of ovotransferrin, on the other hand, reduced BALP, osteocalcin and PYD, which suggested the possible effects of ovotransferrin on adjusting the imbalance of bone remodeling. Moreover, ovotransferrin may function as a stimulator on bone formation as indicated by a substantial increase in collagen synthesis (PINP) after ovotransferrin administration. Furthermore, we measured two osteoclastogenic cytokines and found that ovotransferrin down regulated the production of serum $TNF\alpha$ and IL-6 under the experimental conditions. TNF α , along with IL-6, plays a critical causal role in the rapid bone loss associated with estrogen deficiency (Riggs, 2000). They promote bone resorption by affecting osteoclast differentiation and activity. Ovotransferrin was suggested as an innate immunity of laying hen eggs (Hallquist & Klasing, 1994); ovotransferrin and its cationic fragment OTAP-92 could enhance intestinal and systemic immune response (Levy, Bulet, & Ehret-Sabatier, 2004; Zhu et al., 2018). Meanwhile, ovotransferrin hydrolysates and ovotransferrin-derived peptide showed antiinflammatory activity by inhibiting the production of inflammatory cytokines, such as TNFa and IL-6 (Huang et al., 2010; Liao, Chakrabarti, Davidge, & Wu, 2016; Liu et al., 2017). Our results suggested that ovotransferrin might inhibit bone resorption as an anti-inflammatory agent, thus improving bone health.

A further study on the effect of ovotransferrin on osteoclasts formation and resorption was investigated using freshly extracted bone marrow cells. Under the same culture condition, bone marrow cells derived from OVX rats developed more osteoclasts compared to the Sham rats, indicating estrogen deficiency could activate osteoclastogenesis and promote bone loss in OVX rats. But oral administration of ovotransferrin significantly ameliorated the osteoclast formation. Furthermore, ovotransferrin treatment also prevented OVX-induced excessive osteoclastic resorption. These might be partly due to the effects of ovotransferrin on regulating osteoclast precursors in bone marrow. In our study, ovotransferrin treatment significantly suppressed the expression of CathK and MMP9 in bone marrow cells. CathK is the key enzyme responsible for osteolysis of bone, which is also the only known mammalian protease capable of degrading both helical and non-helical regions of type I collagen (Garnero et al., 2000). Overexpression of CathK has been demonstrated to result in accelerated bone turnover (Kiviranta et al., 2001; Saftig et al., 2002). Similarly, MMPs are a group of proteolytic enzymes involved in the degradation of extracellular matrix of various tissues including bone (Sundaram et al., 2007). Among all, MMP9 is highly expressed at early stages of osteoclast development and in mature osteoclasts (Reponen, Sahlberg, Munaut, Thesleff, & Tryggvason, 1994). Increased MMP9 activity has been detected in human osteoclastomas and osteoclasts in Paget's disease (Wucherpfennig, Li, Stetler-Stevenson, Rosenberg, & Stashenko, 2009). It has also been shown that upregulation of MMP9 plays an important role in the pathogenesis of dental pulp inflammation and destruction (Tsai, Chen, Huang, Su, & Chang, 2005). Cytokines such as IL-1, TNF α and granulocyte macrophage colonystimulating factor (GM-CSF) have been shown to induce the upregulation of MMP9 and enhance the activity of MMP9, therefore contributing to bone degradation (Kusano et al., 1998; Sundaram et al., 2007; Zhou, Zhang, Ardans, & Wahl, 2003). Thus, the suppression of CathK and MMP9

expression in bone marrow cells might contribute to reduce osteoclastic bone resorption by preventing osteoclast formation and activity in OVX rats.

In the previous study, ovotransferrin was able to inhibit RANKL-induced osteoclastogenesis by regulating the activation of NF- κ B pathway (Chapter 3). NF- κ B is one of the important signaling pathways activated by the binding of RANKL to RANK. This binding triggers the translocation of NF- κ B p65 into nucleus and foster the transcription of genes involved in osteoclast differentiation (Thummuri, Naidu, & Chaudhari, 2017). Using osteoclasts differentiated from bone marrow cells, ovotransferrin significantly inhibited RANKL-induced expression of NF- κ B p65 and I κ B α , indicating the involvement of NF- κ B pathway in suppressing osteoclastogenesis. The inactivation of NF- κ B has been reported interrupt osteoclastogenesis and osteoclasts function (Kim, Oh, & Lee, 2017). A2B adenosine receptor (A2BAR), a regulator of bone homeostasis, showed inhibition of NF- κ B activation and suppressed the induction of osteoclast markers, including CathK and MMP9, which contribute to the decrease in osteoclast cell-cell fusion and bone resorption activity (Kim et al., 2017).

In addition to the direct effect on bone tissue and bone cells, our study also found an inhibition effect of ovotransferrin on adipogenesis in bone marrow cells. Osteoblasts and adipocytes originate from a common precursor, MSCs; preserving bone tissue requires adequate osteoblastic differentiation while minimizing adipogenesis (Pino, Miranda, Figueroa, Rodríguez, & Rosen, 2016). Increased bone marrow fat content has been demonstrated in osteoporotic individuals (Justesen et al., 2001; Meunier, Aaron, Edouard, & Vignon, 1971). Marrow adipogenesis promoted OVX-induced osteoporosis (Moerman, Teng, Lipschitz, & Lecka-Czernik, 2004; Zayzafoon,

Gathings, & McDonald, 2004). Meanwhile, the coupling between activation of adipogenic process and suppression of osteogenesis in mesenchymal marrow stroma/stem cells has also been revealed (Moerman et al., 2004). In our study, elevated expression of PPAR γ and C/EBP α , two master transcription factors for adipogenic differentiation, was observed in OVX rats, and was downregulated after ovotransferrin administration. Although more work is needed to further validate and reveal the association between adipogenesis and osteoporosis prevention upon ovotransferrin administration, this finding suggested a new therapeutic target to develop ovotransferrin as an antiosteoporosis agent by inhibiting adipogenesis.

Over last several years, tight molecular and cellular links between immune activation and bone loss have been identified (Horowitz, Bothwell, Hesslein, Pflugh, & Schatz, 2005; Walsh et al., 2006). Evidences strongly proposed that immune system is involved in the pathogenesis of estrogen deficient osteoporosis (Safadi et al., 2000). Here, we examined the phenotypic response of immune cells in spleen and bone marrow after treated with ovotransferrin. Although the phenotype changes were observed with ovotransferrin administration, especially in bone marrow (e.g. CD3+, CD8+ and OX12+ population), it is not clear what role, if any, these changes plays in the local regulation of bone metabolism and hematopoiesis. Studies have demonstrated the involvement of CD8+ T-cells and OX12+ B-cells in regulating osteoclast differentiation by modulating the cytokines; however, the results still endure a lot of discussion and controversy (Breuil et al., 2010; John, Hock, Short, Glasebrook, & Galvin, 1996; Masuzawa et al., 1994; Onoe et al., 2010; Redoglia et al., 1990). Thus, further experiments should be conducted to explore the relationship between ovotransferrin-influenced alternations of immune system, especially in the local bone marrow microenvironment, and ovotransferrin-treatment in OVX rats.

As a food-derived compound, the role of ovotransferrin as a nutrient should be considered. Several revealing reports have highlighted the link between diet, gut microbiota and bone health (Lucas et al., 2018). High-fiber diet and its main metabolites after microbial fermentation, short-chain fatty acids (SCFAs), significantly increases bone mass and prevents postmenopausal and inflammation-induced bone loss (Lucas et al., 2018). Here we found an increase of fecal SCFA content in ovotransferrin treated OVX rats, especially acetic acid and proprionic acid. Furthermore, the concentration of isobutyric acid, isovaleric acid, valeric acid and caproic acid were significantly increased compare to the OVX rats without ovotransferrin supplementation. Although the main sources of SCFA are carbohydrates, but amino acids valine, leucine, and isoleucine obtained from protein breakdown can be converted into isobutyrate, isovalerate, and 2-methyl butyrate, known as branched-chain SCFA (Ríos-Covián et al., 2016). Thus, ovotransferrin or its derivatives after gastrointestinal digestion may benefit bone health by stimulating SCFAs production.

6.5 Conclusion

In summary, for the first time we reported the *in vivo* effects of ovotransferrin to treat osteoporosis in OVX rats. Rats supplemented with 1% w/w and 0.2% w/w of ovotransferrin significantly preserved OVX-induced loss of bone mineral density and deterioration of trabecular microarchitecture. Additionally, ovotransferrin administration suppressed the overactive bone remodeling by inhibiting osteoclastogenesis and osteoclastic bone resorption, while exerting considerable positive influence in bone formation. Ovotransferrin also inhibited marrow adipogenesis, modulated immune function (both systemic and local), and enhanced the formation of short-chain fatty acids in the gut. It is not known the contributions of these findings to bone outcomes, which warrant further studies. Above all, these results suggested a possible application of ovotransferrin on osteoporosis treatment in postmenopausal women.

6.6 Reference

- Arjmandi, B. H., Lucas, E. A., Khalil, D. A., Devareddy, L., Smith, B. J., McDonald, J., Mason,C. (2005). One year soy protein supplementation has positive effects on bone formationmarkers but not bone density in postmenopausal women. *Nutrition Journal*, 4(1), 8.
- Azuma, Y., Kaji, K., Katogi, R., Takeshita, S., & Kudo, A. (2000). Tumor necrosis factor-alpha induces differentiation of and bone resorption by osteoclasts. *The Journal of Biological Chemistry*, 275(7), 4858–4864.
- Blais, A., Malet, A., Mikogami, T., Martin-Rouas, C., & Tome, D. (2009). Oral bovine lactoferrin improves bone status of ovariectomized mice. *Am J Physiol-Endoc M*, 54, 1281– 1288.
- Breuil, V., Ticchioni, M., Testa, J., Roux, C. H., Ferrari, P., Breittmayer, J. P., Carle, G. F.
 (2010). Immune changes in post-menopausal osteoporosis: the Immunos study.
 Osteoporosis International, 21(5), 805–814.
- Conneely, O. M. (2001). Antiinflammatory activities of lactoferrin. *Journal of the American College of Nutrition*, 20(5 Suppl), 389S–395S; discussion 396S–397S.
- Cornish, J., Callon, K. E., Naot, D., Palmano, K. P., Banovic, T., Bava, U., Reid, I. R. (2004). Lactoferrin is a potent regulator of bone cell activity and increases bone formation in vivo. *Endocrinology*, 145(9), 4366–4374.
- Crockett, J., & Das, J. C. (2013). Osteoporosis & amp;ndash; a current view of pharmacological prevention and treatment. *Drug Design, Development and Therapy*, *7*, 435.
- Cundy, T., Reid, I. R., & Grey, A. (2014). Metabolic bone disease. *Clinical Biochemistry: Metabolic and Clinical Aspects*, 604–635.

de Villiers, T. J., Gass, M. L. S., Haines, C. J., Hall, J. E., Lobo, R. a, Pierroz, D. D., & Rees, M.

(2013). Global consensus statement on menopausal hormone therapy. *Climacteric*, *16*(2), 203–204.

- Drake, M. T., Clarke, B. L., & Khosla, S. (2008). Bisphosphonates: mechanism of action and role in clinical practice. *Mayo Clinic Proceedings*, *83*(9), 1032–1045.
- Fromigué, O., Modrowski, D., & Marie, P. J. (2004). Growth factors and bone formation in osteoporosis: roles for fibroblast growth factor and transforming growth factor beta. *Current Pharmaceutical Design*, 10(21), 2593–2603.
- Garnero, P., Buchs, N., Zekri, J., Rizzoli, R., Coleman, R. E., & Delmas, P. D. (2000). Markers of bone turnover for the management of patients with bone metastases from prostate cancer. *British Journal of Cancer*, 82(4), 858–864.
- Gars, M. Le, Seiler, C., Kay, A., Bayless, N., Starosvetsky, E., Moore, L., Blish, C. (2018).CD38 is a key regulator of enhanced NK cell immune responses during pregnancy through its role in immune synapse formation. *BioRxiv*, 349084.
- Giustina, A., Mazziotti, G., & Canalis, E. (2008). Growth Hormone, Insulin-Like Growth Factors, and the Skeleton. *Endocrine Reviews*, *29*(5), 535–559.
- Grey, A., Zhu, Q., Watson, M., Callon, K., & Cornish, J. (2006). Lactoferrin potently inhibits osteoblast apoptosis, via an LRP1-independent pathway. *Molecular and Cellular Endocrinology*, 251(1–2), 96–102.
- Guo, H. Y., Jiang, L., Ibrahim, S. A., Zhang, L., Zhang, H., Zhang, M., & Ren, F. Z. (2009). Orally administered lactoferrin preserves bone mass and microarchitecture in ovariectomized rats. *J Nutr*, 139(C), 958–964.
- Hallquist, N. A., & Klasing, K. C. (1994). Serotransferrin, ovotransferrin and metallothionein levels during an immune response in chickens. *Comparative Biochemistry and Physiology*.

Biochemistry and Molecular Biology, 108(3), 375–384.

- Han, J., & Wang, W. (2017). Effects of tanshinol on markers of bone turnover in ovariectomized rats and osteoblast cultures. *PloS One*, *12*(7), e0181175.
- Horowitz, M. C., Bothwell, A. L. M., Hesslein, D. G. T., Pflugh, D. L., & Schatz, D. G. (2005).
 B cells and osteoblast and osteoclast development. *Immunological Reviews*, 208(1), 141–153.
- Hou, J. M., Xue, Y., & Lin, Q.-M. (2012). Bovine lactoferrin improves bone mass and microstructure in ovariectomized rats via OPG/RANKL/RANK pathway. *Acta Pharmacologica Sinica*, 33(10), 1277–1284.
- Huang, W., Chakrabarti, S., Majumder, K., Jiang, Y., Davidge, S. T., & Wu, J. (2010). Eggderived peptide IRW inhibits TNF-α-induced inflammatory response and oxidative stress in endothelial cells. *Journal of Agricultural and Food Chemistry*, 58(20), 10840–10846.
- Ji, M. X., & Yu, Q. (2015). Primary osteoporosis in postmenopausal women. *Chronic Diseases* and Translational Medicine, 1(1), 9–13.
- John, V., Hock, J. M., Short, L. L., Glasebrook, A. L., & Galvin, R. J. (1996). A role for CD8+ T lymphocytes in osteoclast differentiation in vitro. *Endocrinology*, *137*(6), 2457–2463.
- Justesen, J., Stenderup, K., Ebbesen, E. N., Mosekilde, L., Steiniche, T., & Kassem, M. (2001). Adipocyte tissue volume in bone marrow is increased with aging and in patients with osteoporosis. - PubMed - NCBI. *Biogerontology*, 2(3), 165–171.
- Kenny, A. M., Mangano, K. M., Abourizk, R. H., Bruno, R. S., Anamani, D. E., Kleppinger, A.,
 ... Kerstetter, J. E. (2009). Soy proteins and isoflavones affect bone mineral density in older women: a randomized controlled trial. *The American Journal of Clinical Nutrition*, 90(1), 234–242.

- Kim, B. H., Oh, J. H., & Lee, N. K. (2017). The Inactivation of ERK1/2, p38 and NF-kB Is Involved in the Down-Regulation of Osteoclastogenesis and Function by A2B Adenosine Receptor Stimulation. *Molecules and Cells*, 40(10), 752–760.
- Kiviranta, R., Morko, J., Uusitalo, H., Aro, H. T., Vuorio, E., & Rantakokko, J. (2001).
 Accelerated Turnover of Metaphyseal Trabecular Bone in Mice Overexpressing Cathepsin
 K. *Journal of Bone and Mineral Research*, *16*(8), 1444–1452.
- Kobayashi, Y., Rupa, P., Kovacs-Nolan, J., Turner, P. V., Matsui, T., & Mine, Y. (2015). Oral administration of hen egg white ovotransferrin attenuates the development of colitis induced by dextran sodium sulfate in mice. *Journal of Agricultural and Food Chemistry*, 63(5), 1532–1539.
- Kusano, K., Miyaura, C., Inada, M., Tamura, T., Ito, A., Nagase, H., ... Suda, T. (1998).
 Regulation of Matrix Metalloproteinases (MMP-2, -3, -9, and -13) by Interleukin-1 and Interleukin-6 in Mouse Calvaria: Association of MMP Induction with Bone Resorption ¹. *Endocrinology*, *139*(3), 1338–1345.
- Lauderdale, D. S., Jacobsen, S. J., Furner, S. E., Levy, P. S., Brody, J. A., & Goldberg, J. (1997).
 Hip fracture incidence among elderly Asian-American populations. *American Journal of Epidemiology*, *146*(6), 502–509.
- Levy, F., Bulet, P., & Ehret-Sabatier, L. (2004). Proteomic analysis of the systemic immune response of Drosophila. *Molecular & Cellular Proteomics : MCP*, *3*(2), 156–166.
- Liao, W., Chakrabarti, S., Davidge, S. T., & Wu, J. (2016). Modulatory effects of egg white ovotransferrin-derived tripeptide IRW (Ile-Arg-Trp) on vascular smooth muscle cells against Angiotensin II stimulation. *Journal of Agricultural and Food Chemistry*, 64(39), 7342–7347.

- Liu, L., Xu, M., Tu, Y., Du, H., Zhou, Y., & Zhu, G. (2017). Immunomodulatory effect of protease hydrolysates from ovotransferrin. *Food Funct.*, 8(4), 1452–1459.
- Lucas, S., Omata, Y., Hofmann, J., Böttcher, M., Iljazovic, A., Sarter, K., ... Zaiss, M. M.
 (2018). Short-chain fatty acids regulate systemic bone mass and protect from pathological bone loss. *Nature Communications*, 9(1), 55.
- Malet, A., Bournaud, E., Lan, A., Mikogami, T., Tomé, D., & Blais, A. (2011). Bovine lactoferrin improves bone status of ovariectomized mice via immune function modulation. *Bone*, 48(5), 1028–1035.
- Masuzawa, T., Miyaura, C., Onoe, Y., Kusano, K., Ohta, H., Nozawa, S., & Suda, T. (1994). Estrogen deficiency stimulates B lymphopoiesis in mouse bone marrow. *Journal of Clinical Investigation*, 94(3), 1090–1097.
- Messina, M., & Messina, V. (2000). Soyfoods, soybean isoflavones, and bone health: a brief overview. Journal of Renal Nutrition : The Official Journal of the Council on Renal Nutrition of the National Kidney Foundation, 10(2), 63–68.
- Meunier, P., Aaron, J., Edouard, C., & Vignon, G. (1971). Osteoporosis and the replacement of cell populations of the marrow by adipose tissue. A quantitative study of 84 iliac bone biopsies. *Clinical Orthopaedics and Related Research*, 80, 147–154.
- Moerman, E. J., Teng, K., Lipschitz, D. A., & Lecka-Czernik, B. (2004). Aging activates adipogenic and suppresses osteogenic programs in mesenchymal marrow stroma/stem cells: the role of PPAR-γ2 transcription factor and TGF-β/BMP signaling pathways. *Aging Cell*, *3*(6), 379–389.
- Nieves, J. W. (2005). Osteoporosis: the role of micronutrients. *American Journal of Clinical Nutrition*, *81*(5), 1232S–1239S.

- Ohshiba, T., Miyaura, C., Inada, M., & Ito, A. (2003). Role of RANKL-induced osteoclast formation and MMP-dependent matrix degradation in bone destruction by breast cancer metastasis. *British Journal of Cancer*, *88*(8), 1318–1326.
- Onoe, Y., Miyaura, C., Ito, M., Ohta, H., Nozawa, S., & Suda, T. (2010). Comparative Effects of Estrogen and Raloxifene on B Lymphopoiesis and Bone Loss Induced by Sex Steroid Deficiency in Mice. *Journal of Bone and Mineral Research*, 15(3), 541–549.
- Pino, A. M., Miranda, M., Figueroa, C., Rodríguez, J. P., & Rosen, C. J. (2016). Qualitative Aspects of Bone Marrow Adiposity in Osteoporosis. *Frontiers in Endocrinology*, 7, 139.
- Podgorski, I., Linebaugh, B. E., Koblinski, J. E., Rudy, D. L., Herroon, M. K., Olive, M. B., & Sloane, B. F. (2009). Bone marrow-derived cathepsin K cleaves SPARC in bone metastasis. *The American Journal of Pathology*, 175(3), 1255–1269.
- Rahman, M. M., Bhattacharya, A., Banu, J., Kang, J. X., & Fernandes, G. (2009). Endogenous n-3 fatty acids protect ovariectomy induced bone loss by attenuating osteoclastogenesis. *Journal of Cellular and Molecular Medicine*, 13(8B), 1833–1844.
- Redoglia, V., Boccadoro, M., Battaglio, S., Dianzani, U., Massaia, M., & Pileri, A. (1990).
 Multiple myeloma: altered CD4/CD8 ratio in bone marrow. *Haematologica*, 75(2), 129–131.
- Reponen, P., Sahlberg, C., Munaut, C., Thesleff, I., & Tryggvason, K. (1994). High expression of 92-kD type IV collagenase (gelatinase B) in the osteoclast lineage during mouse development. *The Journal of Cell Biology*, *124*(6), 1091–1102.
- Riggs, B. L. (2000). The mechanisms of estrogen regulation of bone resorption. *The Journal of Clinical Investigation*, *106*(10), 1203–1204.

Ríos-Covián, D., Ruas-Madiedo, P., Margolles, A., Gueimonde, M., de Los Reyes-Gavilán, C.

G., & Salazar, N. (2016). Intestinal Short Chain Fatty Acids and their Link with Diet and Human Health. *Frontiers in Microbiology*, *7*, 185.

- Rosen, C. J., Donahue, L. R., & Hunter, S. J. (1994). Insulin-like growth factors and bone: the osteoporosis connection. *Proceedings of the Society for Experimental Biology and Medicine. Society for Experimental Biology and Medicine (New York, N.Y.)*, 206(2), 83–102.
- Safadi, F. F., Dissanayake, I. R., Goodman, G. G., Jago, R. A., Baker, A. E., Bowman, A. R.,
 Epstein, S. (2000). Influence of Estrogen Deficiency and Replacement on T-Cell
 Populations in Rat Lymphoid Tissues and Organs. *Endocrine*, *12*(1), 81–88.
- Saftig, P., Hunziker, E., Everts, V., Jones, S., Boyde, A., Wehmeyer, O., von Figura, K. (2002). Functions of Cathepsin K in Bone Resorption (pp. 293–303). Springer, Boston, MA.
- Shang, N., & Wu, J. (n.d.). Egg White Ovotransferrin Shows Osteogenic Activity in Osteoblast Cells.
- Silverman, S. L., & Madison, R. E. (1988). Decreased incidence of hip fracture in Hispanics, Asians, and blacks: California Hospital Discharge Data. *American Journal of Public Health*, 78(11), 1482–1483.
- Sims, N. A., Morris, H. A., Moore, R. J., & Durbridge, T. C. (1996). Increased bone resorption precedes increased bone formation in the ovariectomized rat. *Calcified Tissue International*, 59(2), 121–127.
- Smith, E., Redman, R. A., Logg, C. R., Coetzee, G. A., Kasahara, N., & Frenkel, B. (2000). Glucocorticoids inhibit developmental stage-specific osteoblast cell cycle: Dissociation of cyclin A-cyclin-dependent kinase 2 from E2F4-p130 complexes. *Journal of Biological Chemistry*, 275(26), 19992–20001.

- Sundaram, K., Nishimura, R., Senn, J., Youssef, R. F., London, S. D., & Reddy, S. V. (2007).
 RANK ligand signaling modulates the matrix metalloproteinase-9 gene expression during osteoclast differentiation. *Experimental Cell Research*, 313(1), 168–178.
- Thummuri, D., Naidu, V. G. M., & Chaudhari, P. (2017). Carnosic acid attenuates RANKLinduced oxidative stress and osteoclastogenesis via induction of Nrf2 and suppression of NF-κB and MAPK signalling. *Journal of Molecular Medicine*, *95*(10), 1065–1076.
- Tsai, C.-H., Chen, Y.-J., Huang, F.-M., Su, Y.-F., & Chang, Y.-C. (2005). The Upregulation of Matrix Metalloproteinase-9 in Inflamed Human Dental Pulps. *Journal of Endodontics*, 31(12), 860–862.
- Tu, K. N., Lie, J. D., Wan, C. K. V., Cameron, M., Austel, A. G., Nguyen, J. K., ... Hyun, D.
 (2018). Osteoporosis: A Review of Treatment Options. *P & T : A Peer-Reviewed Journal* for Formulary Management, 43(2), 92–104.
- Vaidya, R. (2014). Obesity, sarcopenia and postmenopausal osteoporosis: An interlinked triad! *Journal of Mid-Life Health*, 5(1), 1–2.
- Walsh, M. C., Kim, N., Kadono, Y., Rho, J., Lee, S. Y., Lorenzo, J., & Choi, Y. (2006).
 OSTEOIMMUNOLOGY: Interplay Between the Immune System and Bone Metabolism.
 Annual Review of Immunology, 24(1), 33–63.
- Weaver, C. M. (2015). Diet, gut microbiome, and bone health. *Current Osteoporosis Reports*, 13(2), 125–130.
- Wilsgaard, T., Emaus, N., Ahmed, L. A., Grimnes, G., Joakimsen, R. M., Omsland, T. K., & Berntsen, G. R. (2009). Lifestyle impact on lifetime bone loss in women and men. *American Journal of Epidemiology*, 169(7), 877–886.

Wu, J., & Acero-Lopez, A. (2012). Ovotransferrin: Structure, bioactivities, and preparation.

Food Research International, *46*(2), 480–487.

- Wu, Q., Zhou, X., Huang, D., Ji, Y., & Kang, F. (2017). IL-6 Enhances Osteocyte-Mediated Osteoclastogenesis by Promoting JAK2 and RANKL Activity In Vitro. *Cellular Physiology* and Biochemistry : International Journal of Experimental Cellular Physiology, Biochemistry, and Pharmacology, 41(4), 1360–1369.
- Wucherpfennig, A. L., Li, Y. P., Stetler-Stevenson, W. G., Rosenberg, A. E., & Stashenko, P. (2009). Expression of 92 kD type IV collagenase/gelatinase B in human osteoclasts. *Journal* of Bone and Mineral Research, 9(4), 549–556.
- Yagi, M., Suzuki, N., Takayama, T., Arisue, M., Kodama, T., Yoda, Y., Ito, K. (2009). Effects of lactoferrin on the differentiation of pluripotent mesenchymal cells. *Cell Biology International*, 33(3), 283–289.
- Yokose, S., Ishizuya, T., Ikeda, T., Nakamura, T., Tsurukami, H., Kawasaki, K., ... Yamaguchi,
 A. (1996). An Estrogen Deficiency Caused by Ovariectomy Increases Plasma Levels of
 Systemic Factors That Stimulate Proliferation and Differentiation of Osteoblasts in Rats*. *Endocrinology*, 137(2), 469–478.
- Yoshimaki, T., Sato, S., Tsunori, K., Shino, H., Iguchi, S., Arai, Y., Ogiso, B. (2013). Bone regeneration with systemic administration of lactoferrin in non-critical-sized rat calvarial bone defects. *Journal of Oral Science*, 55(4), 343–348.
- Zayzafoon, M., Gathings, W. E., & McDonald, J. M. (2004). Modeled Microgravity Inhibits Osteogenic Differentiation of Human Mesenchymal Stem Cells and Increases Adipogenesis. *Endocrinology*, 145(5), 2421–2432.
- Zhou, M., Zhang, Y., Ardans, J. A., & Wahl, L. M. (2003). Interferon-gamma differentially regulates monocyte matrix metalloproteinase-1 and -9 through tumor necrosis factor-alpha

and caspase 8. The Journal of Biological Chemistry, 278(46), 45406–45413.

Zhu, G., Luo, J., Du, H., Jiang, Y., Tu, Y., Yao, Y., & Xu, M. (2018). Ovotransferrin enhances intestinal immune response in cyclophosphamide-induced immunosuppressed mice. *International Journal of Biological Macromolecules*, 120(Pt A), 1–9.

Figure 6.1 The experimental design.

Rats were separated into 6 groups providing different diet for 12 weeks. Drug was injected in positive control group. NC: negative control (OVX rats without OVT supplementation); PC: positive control (OVX rats with Alendronate injection); HOvt: high ovotransferrin treatment (OVX rats supplied with 1% OVT); MOvt: medium ovotransferrin treatment (OVX rats supplied with 0.2% OVT); LOvt: low ovotransferrin treatment (OVX rats supplied with 0.04% OVT); OVX: ovariectomized; OVT: ovotransferrin; W: week.



Figure 6.2 Effects of ovotransferrin diet on body weight, food intake, and uterus weight.

(A) The change of body weight during 12-week treatment. (B) The change of food intake during 12-week treatment, (C) Uterus weight after 12 weeks. W: week. Data were expressed as Mean±SEM (n=8 per group). Means without a common letter indicated p<0.05.



С


Figure 6.3 Effects of ovotransferrin diet on tibia trabecular bone mineral density and cortical

bone mineral density.

A) The change of tibia trabecular bone mineral density during 12-week treatment, (B) The tibia trabecular bone mineral density in W0 and W12, (C) The change of tibia cortical bone mineral density during 12-weeks treatment, (D) The tibia cortical bone mineral density in W0 and W12. W: week. Data were expressed as Mean±SEM (n=8 per group). Means without a common letter indicated p<0.05.



Figure 6.4 Effects of ovotransferrin diet on tibia trabecular micro-architecture.

(A) The tibia trabecular percent bone volume in W12, (B) The tibia trabecular thickness in W12,
(C) The tibia trabecular number in W12, (D) The tibia trabecular separation in W12, (E) Threedimensional trabecular images of vertical section (a) and transversal section (b) of proximal tibia and the volume of interest of proximal tibia (c) in W12. W: week. Data were expressed as Mean±SEM (n=8 per group). Means without a common letter indicated p<0.05.

А



В





D

С



Е



Figure 6.5 Effects of ovotransferrin diet on serum cytokine production.

(A) The serum TNF α concentration in W12, (B) The serum IL-6 concentration in W12. W: week. Data were expressed as Mean±SEM (n=8 per group). Means without a common letter indicated p<0.05.

А



В



Figure 6.6 Effects of ovotransferrin diet on bone marrow cells-stimulated osteoclasts formation and resorption.

Bone marrow cells were extracted from rat left femur and culture with 10 ng/mL M-CSF and 100 ng/mL RANKL for 6 days to stimulate osteoclasts. (A) The osteoclasts after TRAP staining, (B) The number of TRAP positive cells differentiated from different animal groups, (C) The osteoclastic resorbed activity, (D) The osteoclastic resorbed pit area, (E) The images of osteoclastic resorbed area. Data were expressed as Mean±SEM (n=8 per group). Means without a common letter indicated p<0.05.





В





D



Е



Figure 6.7 Effects of ovotransferrin diet on osteoclastogenesis-related protein expression in bone marrow.

Bone marrow cells were extracted from rat left femur and whole proteins were extracted for western blot analysis. (A) The protein expression of Cathepsin K, (B) The protein expression of MMP9. Data were expressed as Mean \pm SEM (n=8 per group). Means without a common letter indicated p<0.05.



В

A



Figure 6.8 Effects of ovotransferrin treatment on osteoclast differentiation.

Bone marrow cells were extracted from left femur of Sham rats and cultured with 10 ng/mL M-CSF and 100 ng/mL and RANKL to stimulate osteoclasts. After 6 days, different concentrations of ovotransferrin were added and incubated for 24 h. Then cells were collected and protein was extracted for western blot analysis. (A) The protein expression of NF-kB p65, (B) The protein expression of IkBa. Data were expressed as Mean±SEM (n=8 per group). Means without a common letter indicated p<0.05.



В

А



Figure 6.9 Regulatory role of ovotransferrin in OVX rats.

Ovotransferrin may be investigated as a treatment for osteoporosis to improve bone quality by regulating bone remodeling. Ovotransferrin inhibits the adipogenesis, which may lead to increases in the osteoblast lineage differentiation. Ovotransferrin also alters the local immune cell phenotypes, which may lead to decrease of osteoclast lineage differentiation. Ovotransferrin also shows regulatory role to decrease the production of inflammatory cytokines and to increase the production of short-chain fatty acid.



6.7 Supplementary data and information

А

Figure 6.7.1 Effects of ovotransferrin diet on adipogenesis in bone marrow cells

Bone marrow cells were extracted from rat left femur. RNA was extracted for qRT-PCR analysis and whole proteins were extracted for western blot analysis. (A) The gene expression of adipogenesis associated gene, *PPARy*, *C/EBPa*, *Perilipin*, and *aP2*, (B) The protein expression of PPAR γ , (C) The protein expression of C/EBPa. Data were expressed as Mean±SEM (n=8 per group). Means without a common letter indicated p<0.05.









Figure 6.7.2 Phenotype analysis of freshly isolated immune cells

Values were percentage proportion of the total gate cells as determine by immunofluorescence. (A) The population of CD3+ cells, (B) The population of CD8+ cells, (C) The population of CD25+ cells, (D) The population of CD28+ cells, (E) The population of OX62+ cells, (F) The population of CD51+ cells, (H) The population of CD11b/c+ cells, (I) The population of OX12+ cells, (G) The population of RANK+ cells. Data were expressed as Mean±SEM (n=8 per group). Means without a common letter indicated p<0.05.



Figure 6.7.3 Effects of ovotransferrin diet on fecal short-chain fatty acids content

Concentration of 7 short-chain fatty acid (acetic acid, proprionic acid, isobutyric acid, butyric acid, isovaleric acid, valeric acid, caproic acid) in rat's fecal samples were analysis by GC. Data were expressed as Mean \pm SEM (n=8 per group). Means without a common letter indicated p<0.05.



	Sham	NC	HOvt	MOvt	LOvt	PC
W0	235.26±5.56	255.10±7.53	248.02±14.17	250.17±8.27	248.30±14.31	248.11±10.94
W4	320.63±18.38	401.83±21.07	329.25±21.12	382.75±26.62	390.50±23.06	376.75±32.57
W8	354.63±20.65	449.00±29.49	422.25±19.07	422.88±31.34	441.38±28.87	426.50±48.14
W12	366.00±25.63	459.00±34.13	431.25±20.95	431.00±31.96	453.63±30.97	442.50±54.71

 Table 6.7.1 The body weight (g) change in all treatment groups (n=8)

	Sham	NC	HOvt	MOvt	LOvt	РС
Heart	1.63 ± 0.08	1.73 ± 0.19	1.62 ± 0.08	1.65 ± 0.16	1.63 ± 0.08	1.63 ± 0.04
Liver	13.42 ± 1.39	15.87±3.27	12.92 ± 0.91	13.79 ± 2.94	12.21±1.76	12.10±1.22
Kidney	2.37±0.21	2.46±0.18	2.28±0.23	2.47±0.14	2.43±0.14	2.32±0.19
Spleen	0.67 ± 0.08	$0.80{\pm}0.44$	$0.69{\pm}0.05$	0.87 ± 0.38	$0.64{\pm}0.06$	$0.67{\pm}0.06$
-						
Uterus	$0.90{\pm}0.42$	$0.22{\pm}0.07$	0.27±0.03	$0.40{\pm}0.18$	$0.44{\pm}0.14$	0.43±0.17

Table 6.7.2 The tissue wei	ight (mg) in all i	treatment groups (n=8)
	a \ a/	

	Sham	NC	HOvt	MOvt	LOvt	PC
CD3+	44.8±4.9	43.7±5.0	39.8±2.1	40.7±5.4	38.6±4.0	37.0±3.8
CD4+	34.0±4.1	31.9±3.6	30.4±4.2	33.3±5.0	32.0±3.1	33.9±4.8
CD8+	27.0±3.8	26.7±2.8	24.7±4.8	22.0±1.8	23.6±2.3	22.7±2.4
CD25+	9.3±1.9	11.1±3.7	12.0±2.2	11.1±2.8	11.2±3.1	14.8±1.6
CD28+	4135±3.2	42.5±7.9	38.4±3.1	41.9±9.3	40.3±4.1	48.5±3.4
CD51+	8.7±4.9	5.9±2.7	7.3±1.7	6.0±1.7	5.8±1.6	8.9±2.3
CD71+	21.9±3.5	24.0±8.3	23.4±2.0	25.9±5.6	25.4±4.8	34.2±1.6
CD11b/c+	20.4±4.5	20.1±6.0	19.5±2.7	19.0±3.8	20.1±3.3	26.5±1.2
OX6+	37.8±6.4	37.1±3.5	40.3±1.1	43.2±2.5	40.3±3.9	42.6±2.7
OX12+	31.2±4.2	37.7±3.8	33.5±2.1	35.6±3.9	37.2±1.8	38.2±5.0
OX62+	7.8±1.9	7.7±2.6	9.4±1.9	8.9±2.4	8.9±2.1	12.6±2.9
RANK+	5.1±0.8	5.8±1.2	5.0±0.4	5.2±0.7	5.2±0.3	4.7±1.0
CD3+ CD4+	21.6±2.6	20.2±2.5	18.4±3.4	21.6±4.1	18.7±3.3	18.0±3.2

Table 6.7.3 Phenotype analysis of freshly isolated spleen cells (n=8)

CD3+ CD8+	21.5±4.3	22.0±3.0	19.9±5.0	17.4±1.8	18.4±1.7	17.6±2.4
CD4+ CD25+	2.9±0.5	3.5±1.2	3.6±0.7	3.0±0.7	3.2±0.9	4.0±0.7
CD8+ CD25+	2.7±0.6	3.1±1.2	3.5±0.6	2.7±0.8	2.8±1.0	3.6±1.2
CD4+ CD28+	6.4±0.9	7.2±1.3	6.8±1.0	6.0±1.6	6.0±1.0	6.9±0.8
CD8+ CD28+	19.4±2.3	18.2±4.9	17.2±2.7	19.3±4.5	18.5±2.4	21.8±3.0
CD4+ CD71+	13.9±3.1	14.1±5.8	14.5±2.1	15.3±4.0	15.7±3.2	20.8±2.8
CD8+ CD71+	2.9±0.8	3.5±1.7	4.3±1.0	3.7±1.1	4.1±1.2	5.7±1.1
OX6+ OX62+	11.1±2.9	11.4±4.0	12.3±2.1	12.4±3.1	11.8±2.4	17.0±1.7
OX6+ CD11b/c+	6.4±1.8	6.4±2.7	7.5±1.4	7.4±2.2	7.5±2.2	11.3±2.5
CD3+ CD4+ CD25+	17.3±4.3	21.2±6.3	24.0±2.	16.9±4.3	20.1±4.3	21.2±1.9
CD3+ CD8+ CD25+	7.2±2.4	8.9±3.4	11.2±4.3	8.9±2.9	8.5±3.0	10.7±4.0

Values were percentage proportion of the total gated cells as determined by immunofluorescence.

	Sham	NC	HOvt	MOvt	LOvt	РС
CD3+	13.0±1.3	9.4±3.3	11.7±1.1	12.2±1.90	14.1±1.3	20.7±1.4
CD4+	9.3±2.5	12.5±3.9	14.6±1.1	14.2±2.9	12.1±3.5	13.1±1.7
CD8+	10.8±1.2	7.4±2.6	9.1±0.9	10.1±1.7	11.6±1.8	17.8±2.1
CD25+	13.3±4.5	22.7±4.1	17.9±2.5	20.2±4.8	21.6±3.7	31.9±3.8
CD28+	21.5±8.3	35.0±6.3	30.2±2.1	33.3±4.5	34.5±2.8	48.6±2.6
CD51+	3.1±1.2	5.0±0.8	5.7±0.8	5.7±1.4	5.4±1.1	8.3±2.2
CD71+	27.0±3.4	24.7±8.6	26.5±2.9	26.3±5.0	20.2±5.4	25.5±4.8
CD11b/c+	17.1±3.5	22.3±4.4	15.2±3.3	18.3±4.4	18.4±3.4	21.2±4.1
OX6+	5.9±1.7	8.8±1.1	$8.8{\pm}0.8$	10.1±2.3	9.9±1.8	13.9±2.4
OX12+	15.3±3.5	18.3±4.0	16.4±3.5	17.3±5.0	18.6±2.8	18.7±2.5
OX62+	6.4±3.0	11.9±1.6	11.2±4.0	11.5±3.5	14.5±2.2	21.8±4.3
RANK+	8.3±2.7	17.8±4.4	11.0±3.2	16.4±2.8	15.2±2.9	6.4±1.4
CD3+ CD4+	2.1±0.5	3.0±0.3	3.8±0.7	3.3±0.62	3.4±0.7	4.2±0.9

 Table 6.7.4 Phenotype analysis of freshly isolated bone marrow cells (n=8)

CD3+ CD8+	2.5±0.8	2.5±0.4	2.6±0.3	2.7±0.7	2.8±0.5	3.8±0.4
CD4+ CD25+	4.3±2.0	7.3±2.5	8.4±0.7	7.9±1.5	6.5±2.5	8.1±2.1
CD8+ CD25+	2.8±1.2a	4.6±0.6	4.2±0.6	4.2±1.0	5.21±0.8	8.4±1.4
CD4+ CD28+	3.7±1.7	6.5±2.0	7.3±0.5	7.0±1.7	4.3±3.7	7.2±2.2
CD8+ CD28+	4.1±1.5	6.3±0.8	5.6±0.6	5.9±1.5	7.2±1.2	11.7±1.3
CD4+ CD71+	5.5±2.4	8.2±3.2	9.4±0.6	10.9±5.1	7.6±2.8	7.6±2.7
CD8+ CD71+	4.4±2.0	6.8±1.7	6.4±1.0	7.1±1.7	7.0±1.5	10.6±2.2
OX6+ OX62+	3.3±1.4	5.2±0.8	5.8±0.7	5.8±1.6	6.4±1.0	9.5±1.8
OX6+ CD11b/c+	0.2±0.1	0.3±0.1	0.3±0.1	0.3±0.1	0.2±0.1	0.3±0.1
CD3+ CD4+ CD25+	80.8±20.0	94.3±1.3	92.3±3.0	92.8±3.1	93.1±1.8	96.4±1.1
CD3+ CD8+ CD25+	37.6±12.0	56.1±5.9	58.6±11.9	52.5±15.8	62.7±9.6	61.7±5.2

Values were percentage proportion of the total gated cells as determined by immunofluorescence.

	Sham	NC	HOvt
Acetic acid	6.3654±1.6527	4.9498±0.0084	13.2733±0.6017
Proprionic acid	0.7940±0.2555	0.4821±0.0580	1.5274±0.0717
Isobutyric acid	0.0776±0.0694	0.0262±0.0370	0.2170±1.4442
Butyric acid	0.1786±0.0303	$0.0988 {\pm} 0.0798$	0.2967±0.1746
Isovaleric acid	0.1673±0.0703	0.0751±0.00002	0.2628±0.1389
Valeric acid	0.0288±0.0149	$0.0050 {\pm} 0.0070$	0.0620±0.0354
Caproic acid	0.1502±0.0459	0.1094±0.0020	0.2765±0.0943

Table 6.7.5 The concentration of fecal short-chain fatty acids (µmol/mL) (n=8)

Gene	Primer Sequences (5'-3')				
ΡΡΑRγ	Forward	TCCTCCTGTTGACCCAGAGCAT			
	Reverse	AGCTGATTCCGAAGTTGGTGG			
C/EBPa	Forward	TGAACAAGAACAGCAACGAG			
	Reverse	TCACTGGTCACCTCCAGCAC			
Perilipin	Forward	GACACCACCTGCATGGCT			
	Reverse	TGAAGCAGGGCCACTCTC			
aP2	Forward	GCGTGGAATTCGATGAAATCA			
	Reverse	CCCGCCATCTAGGGTTATGA			
β-actin	Forward	AGTACCCCATTGAACACGGC			
	Reverse	TTTTCACGGTTAGCCTTAGG			

Table 6.7.6 Sequences of primers used in qRT-PCR

Formula	g/kg
Casein	140.0
L-cystine	1.8
Corn starch	465.692
Maltodextrin	155.0
Sucrose	100.0
Soybean oil	40.0
Cellulose	50.0
Mineral Mix, AIN-93M-MX (64049)	35.0
Vitamin Mix, AIN-93-VX (94047)	10.0
Choline Bitartrate	2.5
TBHQ, antioxidant	0.008

Table 6.7.7 Nutritional characteristics of experimental diet

Table 6.7.8 The antibody combination for immune cells phenotype analysis

For each cell sample, 5 combinations were conducted according to the table above. Maximum 4 antibodies with different color-labeled were able to add into one well

CHAPTER 7 - General Summary and Discussion

7.1 Key Findings of the Present Research

The overall purposes of this research were to understand the regulatory roles of ovotransferrin in bone cell activity and to explore its potential application as a bioactive protein against osteoporosis. The key findings of each study are listed below:

1) The first objective was to investigate the regulatory role of ovotransferrin on osteoblastic activity and osteogenesis (Chapter 3)

Ovotransferrin, the second major protein in egg white, is an iron-binding glycoprotein belonging to transferrin family (Giansanti, Leboffe, Angelucci, & Antonini, 2015). In addition to its nutritional value, egg white ovotransferrin displays numerous biological properties, including antimicrobial, antioxidant, and immunomodulating activities (Xie, Huff, Huff, Balog, & Rath, 2002). A previous study also reported that the expression of ovotransferrin and its receptor were increased during the initial stage of bone formation in developing chick embryo, suggesting the importance of ovotransferrin in bone formation (Gentili et al., 1993; 1994). Moreover, another transferrin family member lactoferrin, derived from bovine milk, has been reported to promote bone formation by stimulating bone formation cell osteoblast activity (Cornish et al., 2004; Naot, Grey, Reid, & Cornish, 2005), and to provide benefits for bone health (Blais, Malet, Mikogami, Martin-Rouas, & Tomé, 2009; Guo et al., 2009; Hou, Xue, & Lin, 2012).

In this study, ovotransferrin directly stimulated osteoblast proliferation, differentiation and mineralization as an osteogenic agent. Osteoblasts treated with ovotransferrin exhibited increased

proliferation caused by cell-cycle arrest at the S and G2/M transition. Ovotransferrin also stimulated osteoblast differentiation by increasing the expression of type I collagen and alkaline phosphatase (ALP), and finally promoted mineralization to produce osteoid. Meanwhile, ovotransferrin also showed potent effect on inhibiting osteoclastogenesis. Ovotransferrin inhibited osteoblast-produced receptor activator of nuclear factor kappa-B (RANKL), while increased the expression of osteoprotegerin (OPG). Egg white ovotransferrin showed comparable activity to bovine lactoferrin, the well-established bioactive protein for bone formation, indicating the potential of ovotransferrin for promoting bone formation.

2) The second objective was to study the mechanisms involved in the ovotransferrin-stimulated osteogenesis (Chapter 4)

Osteoblast differentiation is a prerequisite of bone formation; thus, it is desirable to identify and develop anabolic agents to enhance bone formation by stimulating osteoblastic activity. However, regulation of osteoblast differentiation is through a complicated cooperation of various signal transductions, such as the activation of bone morphogenic proteins (BMPs) pathway (Cao & Chen, 2005), mitogen-activated protein kinases (MAPKs) pathway (Chau, Leong, & Li, 2009; Derynck & Zhang, 2003; Dijke & Hill, 2004), and phosphoinositide 3-kinases (PI3K)-protein kinase B (Akt) pathway (Chau et al., 2009; Derynck & Zhang, 2003; Dijke & Hill, 2004). Studies also reported the importance of these signaling pathways in bioactive compounds promoted osteogenesis (Grey, Zhu, Watson, Callon, & Cornish, 2006; Heo et al., 2018; Naot et al., 2011). In the previous chapter, we tested the stimulatory role of ovotransferrin on osteoblast differentiation; however, the molecular mechanism underlying the effect remains poorly understood. Therefore, the objective of this study was to explore the mechanisms involved in

ovotransferrin-stimulated osteogenesis, especially the activation of MAPK and PI3K-Akt pathways.

Ovotransferrin activated MAPK pathways by increasing phosphorylation of ERK1/2, JNK and p38, in a time-dependent manner; however, only the ERK1/2 pathway responded to ovotransferrininduced osteogenesis. Antagonist of ERK1/2 blocked the activation of ERK1/2 pathway and attenuated ovotransferrin promoted osteoblast differentiation and mineralization, while antagonist of JNK showed no effects. Adding p38 inhibitor reduced the expression of ALP and Runx2 compared with ovotransferrin alone, but remained significantly higher than control. Ovotransferrin activated ERK1/2 pathway was further validated by investigating the phosphorylation of the core component involved in ERK cascade, including c-Raf, MEK1/2, p90RSK and MSK1. Ovotransferrin caused an increase in the phosphorylation of upstream Raf and MEK1/2, as well as the downstream p90RSK and MSK1, indicating the osteoblastic promotion was due to the activation of the ERK cascade. Ovotransferrin also increased the expression of two PI3K subunits p85 and p110, leading to increased phosphorylation of Akt and osteogenesis. Akt antagonist was also used to confirm the effect of PI3K-Akt in response to the stimulation of ovotransferrin; meanwhile we found that PI3K-Akt not only regulated ovotransferrin induced osteoblast differentiation, but also contributed to the migration of osteoblast. The activation of ERK1/2 and PI3K-Akt pathways further regulated the expression of Runx2, BMP-2, and β-catenin, which play vital roles in osteoblast differentiation, and benefit for osteogenesis and bone formation. In addition to the activated pathway associated with ovotransferrin stimulation, the possible receptors that response to ovotransferrin and initiate the signal transduction were studied in this chapter. Three receptors that might be able to bind with ovotransferrin in osteoblasts, named transferrin binding protein (TFBP), transferrin receptor (TFR1), and low-density lipoprotein receptor-related protein 1 (LRP1), were significantly expressed in the presence of ovotransferrin. LRP1 antagonist pre-treatment reduced ALP and mineralization, but did not affect ovotransferrin-activated ERK1/2 or PI3K-Akt pathway, indicating LRP1 was independent to ERK1/2 and PI3K-Akt activation-associated osteoinduction.

3) The third objective was to investigate the in vitro effects of ovotransferrin on inhibiting osteoclastogenesis and bone resorptive activity (Chapter 5)

Bone metabolism relies on activities of bone formation and resorption. An excessive bone resorption will result in osteolytic bone conditions and lead to bone disorders, such as osteoporosis (Takatsuna et al., 2004; Wei et al., 2018). Osteoclast, derived from monocyte/macrophage lineage, is responsible for bone resorption (Tanaka et al., 2006). The development of osteoporosis is due to a massive recruitment of osteoclast and an excessive osteoclastic activity (Takatsuna et al., 2004; Wei et al., 2018). Thus, the inhibition agents targeting to attenuate osteoclastogenesis and osteoclastic resorption is the major therapy strategy for osteoporosis treatment (Sharif, Abdollahi, & Larijani, 2011). In previous chapters, we investigated the anabolic effect of ovotransferrin on promoting bone formation by stimulating osteoblastic activity; therefore, the objective of this study was to explore the effect of ovotransferrin on inhibiting osteoclastogenesis and resorptive activity, and its associated mechanisms.

Ovotransferrin showed significant inhibition on osteoclastogenesis and decreased the number of RANKL-induced osteoclasts. Calcium-phosphates resorption assay indicated that the resorptive activity of osteoclasts treated by ovotransferrin was significantly suppressed. Further studies found

the inhibition effect was due to the regulation of RANKL-activated NF- κ B (nuclear factor kappalight-chain-enhancer of activated B cells) and MAPK pathways. Ovotransferrin pre-treatment suppressed the phosphorylation of the major components in NF- κ B cascade, including IKK α/β , I κ B α , and NF- κ B p65. Ovotransferrin also inhibited RANKL-induced phosphorylation of MAPKs in dose-depended manner. The suppression of these two pathways resulted in a down-regulation of several osteoclastogenesis-associated proteins, such as TRAF6 (TNF receptor associated factor 6), c-Fos, NFATc1 (nuclear factor of activated T-cells cytoplasmic 1), and cathepsin K. In addition to inhibiting osteoclastogenesis, ovotransferrin also accelerated cell apoptosis in mature osteoclast via regulating the Bcl-2 family members, in which the expression of pro-apoptotic *Bim* and *Bad* were increased while the anti-apoptotic *Bcl-2* and *Bcl-x* was decreased.

4) The fourth objective was to study the in vivo efficacy of ovotransferrin on osteoporosis prevention in an ovariectomized (OVX) rat model (Chapter 6)

Ovariectomized (OVX) rat, after ovary removal, represents the clinical features of estrogen deficiency-induced bone loss, including a rapid bone loss due to exceeding bone resorption, a greater loss of cancellous bone than cortical bone, and reduced calcium absorption (Jee & Yao, 2001; Kalu, 1991; Komori, 2015). Thus, OVX rat model is the most widely used animal model used in study of postmenopausal osteoporosis and assessment of osteoporosis therapies (Jee & Yao, 2001; Kalu, 1991; Komori, 2015). Previous research demonstrated that ovotransferrin enhanced osteoblastic activity and bone formation, and inhibited osteoclastogenesis and bone resorption; therefore, the objective of this study was to study the *in vivo* efficacy of orally administrated ovotransferrin in OVX rats.

Ovotransferrin administration for 12 weeks did not affect body weight, food intake and organ weight, but prevented the OVX-induced decrease of bone mineral density, and partially preserved OVX-induced deterioration of bone microarchitecture. This improvement was due mainly to the regulation of bone turnover, in which ovotransferrin inhibited bone resorption and stimulated bone formation. Decreased bone resorption after ovotransferrin administration was evidenced 1) decreased production of serum and urinary bone resorption biomarkers; 2) decreased production of serum cytokine TNF- α and IL-6, which are two stimulators of osteoclast differentiation; and 3) interference of the differentiation of osteoclast precursors-derived from bone marrow. Stimulated bone promotion was evidenced 1) improved mineral density and trabecular bone microarchitecture; and 2) increased production of serum bone formation biomarkers. These results supported our in vitro findings that ovotransferrin could regulate bone cells activity. In addition to its regulatory roles on bone cell (osteoblast, osteoclast and their precursors) and bone remodeling, ovotransferrin also inhibited adipogenesis, modulated immune function (both systemic and local), and enhanced the formation of short-chain fatty acids in gut. It is not known the contributions of these findings to the bone outcomes, which are warranted further studies.

In conclusion, this thesis demonstrated the potential use of ovotransferrin as functional foods and/or nutraceuticals for osteoporosis prevention through the management of bone remodeling via regulating both osteoblastic and osteoclastic activity.

7.2 Significance of This Research

Egg-derived bioactive proteins and peptides have been reported to have various biological functions, such as antimicrobial, anti-oxidant, anti-inflammation, immunomodulatory and among

others. (Mine, 2007). As the second major protein in egg white, ovotransferrin has been widely studied for its bioactivities (reviewed in Chapter 2). However, to our best knowledge, this is the first study reporting *in vitro* (in cells) and *in vivo* (in rodents) bone health promoting effect of ovotransferrin. This thesis has greatly contributed to the knowledge of ovotransferrin as a regulatory molecule on bone cell activity paving a new way of value addition to the egg industry. Thus, the study exhibits significance in both egg industry and bone health management.

1) Significance to the Egg Industry

In 2017, there were 1059 egg producers in Canada, generating in total \$108 million farm cash receipts (AAFC, 2017a). Currently, the Canadian egg market is divided into table eggs and process eggs (AAFC, 2017b). The table egg market is presents approximately 73% of the Canadian market, while the processed egg market accounts for the remaining 27% (AAFC, 2017b). In addition to a small share of the market, processed egg is limited in the production of whole egg, albumen and egg yolks in frozen, dried or liquid form, which is mainly used for further manufacturing of bakery products, mayonnaise, noodles, etc. (AAFC, 2017b). Although some of the processed eggs are used for specialty items such as shampoo, pet foods and adhesives (AAFC, 2017b), the narrow applications of egg products limit the egg industry profile. Egg, as an inexpensive source of protein, can be utilized for producing bioactive proteins and/or peptides. Unfortunately, compared to other protein-rich foods such as dairy proteins, the value-added application of egg is negligible. Therefore, the potential of ovotransferrin in preventing osteoporosis and enhancing bone health will be of great interest to food, nutraceutical and pharmaceutical companies, which will eventually bring benefits to the egg industry by developing valued-add egg products.

2) Significance to Osteoporosis Treatment

According to Health Canada, osteoporosis affects ~ 1.4 million Canadians and costs over \$2.3 billion to the Canadian healthcare system for treating osteoporosis and the fractures as of 2010 (Health Canada, 2018). Some pharmaceutical drugs used in osteoporosis treatment are reported to have associated risk: increase the risk of ovarian cancer and affect the risk of cardiovascular diseases. Therefore, there is a trend to use dietary intervention or food-derived components to treat osteoporosis for long-term therapy. The findings of this study provide a new approach for osteoporosis treatment with fewer side effects and be suitable for long-term use. Meanwhile, the mechanisms and signaling pathway study provided better understanding on the regulatory role of ovotransferrin and will help with the development of new therapeutic approach to maintain bone health.

7.3 Future Research Directions

It should be noted that there are limitations in the research reported in this thesis. Firstly, we used murine cell lines for the *in vitro* studies (preosteoblast MC3T3-E1 and macrophage RAW 264.7); validation of these effects of ovotransferrin should be made with primary cells and human cell lines, such as hFOB 1.19 and Saos-2. Additionally, it is not know the metabolism of ovotransferrin after oral administration and the responsible metabolites reach to the action site. Thirdly, the animal was double-housed during the husbandry and shared the diet together. Thus, it was difficult to precisely monitor the diet intake and ovotransferrin intake of each rat. As the ovotransferrin was mixed into the diet and could be taken *ad libitum*, the ovotransferrin intake was estimated, but not accurate. Finally, in this study, ovotransferrin exhibited osteoporosis inhibitory effect in ovariectomized (OVX) rats with preserved bone mineral density and microarchitecture; however, no direct evidence indicated the stimulation effect of ovotransferrin on osteogenesis and bone

formation *in vivo*. Based on the key findings and the limitations of the research, the recommended future studies are outlined below:

1. As a protein, orally administration of ovotransferrin will go through the gastrointestinal tract and be digested into small peptides. The bioavailability of ovotransferrin and the remaining intact ovotransferrin after digestion should be studied to estimate the effective dosage of ovotransferrin to exert bioactivity. Since pig's gastrointestinal tract shows almost the same physiology as human, a piglet model might provide insights on the metabolism of ovotransferrin. Also, it remains to be determined if the digested forms are responsible for the activity after orally administration. If not, strategies are needed to protect ovotransferrin against gastrointestinal digestion.

2. In studies using cell lines, ovotransferrin contacted directly with bone cells; however, the absorption of intact high molecular protein in intestinal and transportation to action site (bone cells) is not clear. Although studies have reported that intact lactoferrin could be absorbed into circulation and exert effect in bone cells (Kitagawa et al., 2003), *in vivo* evidence for ovotransferrin is lacking. Protein labeling can be used to track the transportation and circulation of exogenous proteins both *in vitro* and *in vivo*.

3. The action of ovotransferrin involves various pathways. The involvement of MAPK and Akt pathways in regulating ovotransferrin-induced osteogenesis should be confirmed *in vivo*, for example using knockout animal models. Additionally, MAPK and Akt are two signaling pathways controlling almost all cellular activities, including proliferation, differentiation, and apoptosis. Other bone-specific regulatory pathways, such as BMP and Wnt pathways should be included into the study to further elucidate the bone health promotion role of ovotransferrin. Finally, the membrane receptors of ovotransferrin should be identified.

4. Ovotransferrin administration affected immune function, muscle synthesis, adipogenesis, and short chain fatty acid production in ovariectomized rat model. The contributions of these findings to bone health should be further explored. It remains to be established if these findings might provide additional benefits to other postmenopausal symptoms (e.g. sarcopenia, obesity).

5. In this study, ovariectomized rat model was used to study the role of ovotransferrin on osteoporosis treatment. It remains to be established if ovotransferrin shows similar benefits in other animal models (e.g. aged animal model, glucocorticoid-induced osteoporosis model, or rheumatoid arthritis model), which will broaden the application of ovotransferrin.

7.4 References

- AAFC. (2017a). Canada's poultry and egg industry profile. Retrieved January 26, 2019, from http://www.agr.gc.ca/eng/industry-markets-and-trade/canadian-agri-food-sector-intelligence/poultry-and-eggs/poultry-and-egg-market-information/industry-profile/?id=1384971854389
- AAFC. (2017b). Canada's table and processed egg industry. Retrieved January 26, 2019, from http://www.agr.gc.ca/eng/industry-markets-and-trade/canadian-agri-food-sector-intelligence/poultry-and-eggs/poultry-and-egg-market-information/table-and-processed-eggs/?id=1384971854396
- Blais, A., Malet, A., Mikogami, T., Martin-Rouas, C., & Tomé, D. (2009). Oral bovine
 lactoferrin improves bone status of ovariectomized mice. *American Journal of Physiology-Endocrinology and Metabolism*, 296(6), E1281–E1288.
 https://doi.org/10.1152/ajpendo.90938.2008
- Cao, X., & Chen, D. (2005). The BMP signaling and in vivo bone formation. *Gene*, 357(1), 1–8. https://doi.org/10.1016/j.gene.2005.06.017
- Chau, J. F. L., Leong, W. F., & Li, B. (2009). Signaling pathways governing osteoblast proliferation, differentiation and function. *Histology and Histopathology*, 24(12), 1593– 1606. https://doi.org/10.14670/HH-24.1593
- Cornish, J., Callon, K. E., Naot, D., Palmano, K. P., Banovic, T., Bava, U., ... Reid, I. R. (2004).
 Lactoferrin is a potent regulator of bone cell activity and increases bone formation in vivo.
 Endocrinology, 145(9), 4366–4374. https://doi.org/10.1210/en.2003-1307
- Derynck, R., & Zhang, Y. E. (2003). Smad-dependent and Smad-independent pathways in TGFβ family signalling. *Nature*, *425*(6958), 577–584. https://doi.org/10.1038/nature02006

- Dijke, P. ten, & Hill, C. S. (2004). New insights into TGF-β–Smad signalling. *Trends in Biochemical Sciences*, 29(5), 265–273. https://doi.org/10.1016/j.tibs.2004.03.008
- Gentili, C., Bianco, P., Neri, M., Malpeli, M., Campanile, G., Castagnola, P., Cancedda, F. D. (1993). Cell proliferation, extracellular matrix mineralization, and ovotransferrin transient expression during in vitro differentiation of chick hypertrophic chondrocytes into osteoblast-like cells. *Journal of Cell Biology*, *122*(3), 703–712. https://doi.org/10.1083/jcb.122.3.703
- Gentili, C., Doliana, R., Bet, P., Campanile, G., Colombatti, A., Cancedda, F. D., & Cancedda, R. (1994). Ovotransferrin and ovotransferrin receptor expression during chondrogenesis and endochondral bone formation in developing chick embryo. *Journal of Cell Biology*, *124*(4), 579–588.
- Giansanti, F., Leboffe, L., Angelucci, F., & Antonini, G. (2015). The nutraceutical properties of ovotransferrin and its potential utilization as a functional food. *Nutrients*, 7(11), 9105–9115. https://doi.org/10.3390/nu7115453
- Grey, A., Zhu, Q., Watson, M., Callon, K., & Cornish, J. (2006). Lactoferrin potently inhibits osteoblast apoptosis, via an LRP1-independent pathway. *Molecular and Cellular Endocrinology*, 251(1–2), 96–102. https://doi.org/10.1016/j.mce.2006.03.002
- Guo, H. Y., Jiang, L., Ibrahim, S. A., Zhang, L., Zhang, H., Zhang, M., & Ren, F. Z. (2009).
 Orally administered lactoferrin preserves bone mass and microarchitecture in ovariectomized rats. *J Nutr*, *139*(C), 958–964.
 https://doi.org/10.3945/jn.108.100586.TABLE
- Health Canada. (2018). Osteoporosis. Retrieved January 27, 2019, from https://www.canada.ca/en/public-health/services/chronic-diseases/osteoporosis.html

- Heo, S. Y., Ko, S. C., Nam, S. Y., Oh, J., Kim, Y. M., Kim, J. I., Jung, W. K. (2018). Fish bone peptide promotes osteogenic differentiation of MC3T3-E1 pre-osteoblasts through upregulation of MAPKs and Smad pathways activated BMP-2 receptor. *Cell Biochemistry and Function*, 36(3), 137–146. https://doi.org/10.1002/cbf.3325
- Hou, J. M., Xue, Y., & Lin, Q. M. (2012). Bovine lactoferrin improves bone mass and microstructure in ovariectomized rats via OPG/RANKL/RANK pathway. *Acta Pharmacologica Sinica*, 33(10), 1277–1284. https://doi.org/10.1038/aps.2012.83
- Jee, W. S., & Yao, W. (2001). Overview: animal models of osteopenia and osteoporosis. Journal of Musculoskeletal & Neuronal Interactions, 1(3), 193–207. Retrieved from http://www.ncbi.nlm.nih.gov/pubmed/15758493
- Kalu, D. N. (1991). The ovariectomized rat model of postmenopausal bone loss. *Bone and Mineral*, 15(3), 175–191. https://doi.org/10.1016/0169-6009(91)90124-I
- Kitagawa, H., Yoshizawa, Y., Yokoyama, T., Takeuchi, T., Talukder, M. J. R., Shimizu, H., Harada, E. (2003). Persorption of bovine lactoferrin from the intestinal lumen into the systemic circulation via the portal vein and the mesenteric lymphatics in growing pigs. *The Journal of Veterinary Medical Science*, 65(5), 567–572. Retrieved from http://www.ncbi.nlm.nih.gov/pubmed/12808207
- Komori, T. (2015). Animal models for osteoporosis. *European Journal of Pharmacology*, 759, 287–294. https://doi.org/10.1016/j.ejphar.2015.03.028
- Mine, Y. (2007). Egg proteins and peptides in human health--chemistry, bioactivity and production. *Current Pharmaceutical Design*, 13(9), 875–884. Retrieved from http://www.ncbi.nlm.nih.gov/pubmed/17430187

Mohan, A., Rajendran, S. R. C. K., He, Q. S., Bazinet, L., & Udenigwe, C. C. (2015).
Encapsulation of food protein hydrolysates and peptides: a review. *RSC Advances*, 5(97), 79270–79278. https://doi.org/10.1039/C5RA13419F

- Naot, D., Chhana, A., Matthews, B. G., Callon, K. E., Tong, P. C., Lin, J. M., Cornish, J. (2011).
 Molecular mechanisms involved in the mitogenic effect of lactoferrin in osteoblasts. *Bone*, 49(2), 217–224. https://doi.org/10.1016/j.bone.2011.04.002
- Naot, D., Grey, A., Reid, I. R., & Cornish, J. (2005). Lactoferrin--a novel bone growth factor. *Clinical Medicine & Research.*, 3(2), 93–101. https://doi.org/3/2/93 [pii]
- Salari Sharif, P., Abdollahi, M., & Larijani, B. (2011). Current, new and future treatments of osteoporosis. *Rheumatology International*, 31(3), 289–300. https://doi.org/10.1007/s00296-010-1586-z
- Takatsuna, H., Asagiri, M., Kubota, T., Oka, K., Osada, T., Sugiyama, C., Umezawa, K. (2004). Inhibition of RANKL-Induced Osteoclastogenesis by (–)-DHMEQ, a Novel NF-κB Inhibitor, Through Downregulation of NFATc1. *Journal of Bone and Mineral Research*, 20(4), 653–662. https://doi.org/10.1359/JBMR.041213
- Tanaka, S., Miyazaki, T., Fukuda, A., Akiyama, T., Kadono, Y., Wakeyama, H., Nakamura, K.
 (2006). Molecular Mechanism of the Life and Death of the Osteoclast. *Annals of the New York Academy of Sciences*, *1068*(1), 180–186. https://doi.org/10.1196/annals.1346.020
- Wei, C. M., Liu, Q., Song, F. M., Lin, X. X., Su, Y. J., Xu, J., Zhao, J. M. (2018). Artesunate inhibits RANKL-induced osteoclastogenesis and bone resorption in vitro and prevents LPSinduced bone loss in vivo. *Journal of Cellular Physiology*, 233(1), 476–485. https://doi.org/10.1002/jcp.25907
- Xie, H., Huff, G. R., Huff, W. E., Balog, J. M., & Rath, N. C. (2002). Effects of ovotransferrin on chicken macrophages and heterophil-granulocytes. *Developmental & Comparative*

Immunology, 26(9), 805-815. https://doi.org/10.1016/S0145-305X(02)00028-9

BIBILIOGRAPHY

AAFC. (2015). Trends and Market Opportunities for the Functional Foods and Natural Health Products Sector. Retrieved January 23, 2019, from http://www.agr.gc.ca/eng/industry-marketsand-trade/canadian-agri-food-sector-intelligence/functional-foods-and-natural-healthproducts/trends-and-market-opportunities-for-the-functional-foods-and-natural-health-productssector/?id=1170940080209

AAFC. (2017a). Canada's poultry and egg industry profile. Retrieved January 26, 2019, from http://www.agr.gc.ca/eng/industry-markets-and-trade/canadian-agri-food-sector-intelligence/poultry-and-eggs/poultry-and-egg-market-information/industry-profile/?id=1384971854389

AAFC. (2017b). Canada's table and processed egg industry. Retrieved January 26, 2019, from http://www.agr.gc.ca/eng/industry-markets-and-trade/canadian-agri-food-sector-intelligence/poultry-and-eggs/poultry-and-egg-market-information/table-and-processed-eggs/?id=1384971854396

Abeyrathne, E. D. N. S., Lee, H. Y., & Ahn, D. U. (2013). Egg white proteins and their potential use in food processing or as nutraceutical and pharmaceutical agents - a review. *Poultry Science*, *92*, 3292–3299.

Abu-Amer, Y. (2013). NF-κB signaling and bone resorption. Osteoporosis International : A Journal Established as Result of Cooperation between the European Foundation for Osteoporosis and the National Osteoporosis Foundation of the USA, 24(9), 2377–2386.

Aguilera, O., Quiros, L. M., & Fierro, J. F. (2003). Transferrins selectively cause ion efflux through bacterial and artificial membranes. *FEBS Letters*, *548*(1–3), 5–10.

Akiyama, T., Bouillet, P., Miyazaki, T., Kadono, Y., Chikuda, H., Chung, U. I., Tanaka, S. (2003). Regulation of osteoclast apoptosis by ubiquitylation of proapoptotic BH3-only Bcl-2 family member Bim. *The EMBO Journal*, *22*(24), 6653–6664.

Akiyama, Y., Oshima, K., Kuhara, T., Shin, K., Abe, F., Iwatsuki, K., Matsuda, T. (2013). A lactoferrin-receptor, intelectin 1, affects uptake, sub-cellular localization and release of

immunochemically detectable lactoferrin by intestinal epithelial Caco-2 cells. *Journal of Biochemistry*, *154*(5), 437–448.

Alderton, G., Ward, W. H., & Fevold, H. L. (1946). Identification of the bacteria-inhibiting ironbinding protein of egg white as conalbumin. *Archives of Biochemistry*, *11*, 9–13.

Alekel, D. L., Germain, A. S., Peterson, C. T., Hanson, K. B., Stewart, J. W., & Toda, T. (2000). Isoflavone-rich soy protein isolate attenuates bone loss in the lumbar spine of perimenopausal women. *The American Journal of Clinical Nutrition*, 72(3), 844–852.

Alekel, D. L., Weaver, C. M., Ronis, M. J. J., & Ward, W. E. (2013). Nutritional Influences on Bone Health and Overview of Methods. *Bioactive Food as Dietary Interventions for the Aging Population*, 357–370.

Andersen, C. J. (2015). Bioactive egg components and inflammation. *Nutrients*, 7(9), 7889–7913.

Anderson, J. J., Ambrose, W. W., & Garner, S. C. (1998). Biphasic effects of genistein on bone tissue in the ovariectomized, lactating rat model. *Proceedings of the Society for Experimental Biology and Medicine. Society for Experimental Biology and Medicine (New York, N.Y.)*, 217(3), 345–350.

Aoe, S., Koyama, T., Toba, Y., Itabashi, A., & Takada, Y. (2005). A controlled trial of the effect of milk basic protein (MBP) supplementation on bone metabolism in healthy menopausal women. *Osteoporosis International*, *16*(12), 2123–2128.

Aoe, S., Toba, Y., Yamamura, J., Kawakami, H., Yahiro, M., Kumegawa, M., Takada, Y.
(2001). Controlled Trial of the Effects of Milk Basic Protein (MBP) Supplementation on Bone
Metabolism in Healthy Adult Women. *Bioscience, Biotechnology, and Biochemistry*, 65(4), 913–918.

Aoyagi, Y., Park, H., Park, S., Yoshiuchi, K., Kikuchi, H., Kawakami, H., Shephard, R. J. (2010). Interactive effects of milk basic protein supplements and habitual physical activity on bone health in older women: A 1-year randomized controlled trial. *International Dairy Journal*, *20*(10), 724–730. https://doi.org/10.1016/J.IDAIRYJ.2010.03.007

Arboleya, L., & Castañeda, S. (2013). Osteoimmunology: the study of the relationship between the immune system and bone tissue. *Reumatología Clínica*, *9*(5), 303–315.

Arjmandi, B. H., Birnbaum, R., Goyal, N. V, Getlinger, M. J., Juma, S., Alekel, L., Kukreja, S.
C. (1998). Bone-sparing effect of soy protein in ovarian hormone-deficient rats is related to its isoflavone content. *The American Journal of Clinical Nutrition*, 68(6), 1364S–1368S.

Arjmandi, B. H., Lucas, E. A., Khalil, D. A., Devareddy, L., Smith, B. J., McDonald, J., Mason, C. (2005). One year soy protein supplementation has positive effects on bone formation markers but not bone density in postmenopausal women. *Nutrition Journal*, *4*(1), 8.

Azuma, Y., Kaji, K., Katogi, R., Takeshita, S., & Kudo, A. (2000). Tumor necrosis factor-alpha induces differentiation of and bone resorption by osteoclasts. *The Journal of Biological Chemistry*, *275*(7), 4858–4864.

Bae, S. J., Kim, H. J., Won, H. Y., Min, Y. K., & Hwang, E. S. (2017). Acceleration of osteoblast differentiation by a novel osteogenic compound, DMP-PYT, through activation of both the BMP and Wnt pathways. *Scientific Reports*, *7*(1), 8455.

Bain, G., Müller, T., Wang, X., & Papkoff, J. (2003). Activated beta-catenin induces osteoblast differentiation of C3H10T1/2 cells and participates in BMP2 mediated signal transduction. *Biochemical and Biophysical Research Communications*, *301*(1), 84–91.

Balan, G., Bauman, J., Bhattacharya, S., Castrodad, M., Healy, D. R., Herr, M., Paralkar, V. M. (2009). The discovery of novel calcium sensing receptor negative allosteric modulators. *Bioorganic & Medicinal Chemistry Letters*, *19*(12), 3328–3332.

Baron, F., Jan, S., Gonnet, F., Pasco, M., Jardin, J., Giudici, B., Nau, F. (2014). Ovotransferrin Plays a Major Role in the Strong Bactericidal Effect of Egg White against the Bacillus cereus Group. *Journal of Food Protection*, 77(6), 955–962.

Baron, R., & Rawadi, G. (2007). Targeting the Wnt/β-Catenin Pathway to Regulate Bone Formation in the Adult Skeleton. *Endocrinology*, *148*(6), 2635–2643.

Bates, P. (2007). Metabolic bone disease. Orthopaedics and Trauma, 31(5), 306-311.

Bharadwaj, S., Naidu, A. G. T., Betageri, G. V., Prasadarao, N. V., & Naidu, A. S. (2009). Milk ribonuclease-enriched lactoferrin induces positive effects on bone turnover markers in postmenopausal women. *Osteoporosis International*, *20*(9), 1603–1611.

Bhargavan, B., Gautam, A. K., Singh, D., Kumar, A., Chaurasia, S., Tyagi, A. M., Chattopadhyay, N. (2009). Methoxylated isoflavones, cajanin and isoformononetin, have nonestrogenic bone forming effect via differential mitogen activated protein kinase (MAPK) signaling. *Journal of Cellular Biochemistry*, *108*(2), 388–399.

Bi, B. Y., Lefebvre, A. M., Duś, D., Spik, G., & Mazurier, J. (1997). Effect of lactoferrin on proliferation and differentiation of the Jurkat human lymphoblastic T cell line. *Archivum Immunologiae et Therapiae Experimentalis*, *45*(4), 315–320.

Bischoff-Ferrari, H. A., Dawson-Hughes, B., Orav, E. J., Staehelin, H. B., Meyer, O. W., Theiler, R., Egli, A. (2016). Monthly High-Dose Vitamin D Treatment for the Prevention of Functional Decline. *JAMA Internal Medicine*, *176*(2), 175.

Blais, A., Malet, A., Mikogami, T., Martin-Rouas, C., & Tome, D. (2009). Oral bovine lactoferrin improves bone status of ovariectomized mice. *Am J Physiol-Endoc M*, *54*, 1281–1288.

Blais, A., Malet, A., Mikogami, T., Martin-Rouas, C., & Tomé, D. (2009). Oral bovine lactoferrin improves bone status of ovariectomized mice. *American Journal of Physiology-Endocrinology and Metabolism*, 296(6), E1281–E1288.

Blake, G. M., Lewiecki, E. M., Kendler, D. L., & Fogelman, I. (2007). A Review of Strontium Ranelate and Its Effect on DXA Scans. *Journal of Clinical Densitometry*, *10*(2), 113–119.

Bolander, M. E., Youngf, M. F., Fishert, L. W., Yamada, Y., & Terminet, J. D. (1988). Osteonectin cDNA sequence reveals potential binding regions for calcium and hydroxyapatite and shows homologies with both a basement membrane protein (SPARC) and a serine proteinase inhibitor (ovomucoid) (bone/calcium-binding proteins). *Biochemistry*, *85*, 2919–2923.

Bonjour, J. P. (2005). Dietary protein: An essential nutrient for bone health. *Journal of the American College of Nutrition*, 24(6), 526S–536S.

Boyle, W. J., Simonet, W. S., & Lacey, D. L. (2003). Osteoclast differentiation and activation. *Nature*, *423*(6937), 337–342.

Brand, H. S., Lerner, U. H., Grubb, A., Beertsen, W., Nieuw Amerongen, A. V., & Everts, V. (2004). Family 2 cystatins inhibit osteoclast-mediated bone resorption in calvarial bone explants. *Bone*, *35*(3), 689–696.

Brandwood, C. P., Hoyland, J. A., Hillarby, M. C., Berry, J. L., Davies, M., Selby, P. L., & Mee, A. P. (2003). Apoptotic gene expression in Paget's disease: a possible role for Bcl-2. *The Journal of Pathology*, *201*(3), 504–512.

Breuil, V., Ticchioni, M., Testa, J., Roux, C. H., Ferrari, P., Breittmayer, J. P., Carle, G. F. (2010). Immune changes in post-menopausal osteoporosis: the Immunos study. *Osteoporosis International*, *21*(5), 805–814.

Brommage, R., & Neuman, W. F. (1979). Passive accumulation of magnesium, sodium, and potassium by chick calvaria. *Calcified Tissue International*, *28*(1), 57–63.

Brown, K. (2009). Is tamoxifen a genotoxic carcinogen in women? *Mutagenesis*, 24(5), 391–404.

Bruinink, A., Sidler, C., & Birchler, F. (1996). Neurotrophic effects of transferrin on embryonic chick brain and neural retinal cell cultures. *International Journal of Developmental Neuroscience : The Official Journal of the International Society for Developmental Neuroscience*, 14(6), 785–795.

Buccigrossi, V., De Marco, G., Bruzzese, E., Ombrato, L., Bracale, I., Polito, G., & Guarino, A. (2007). Lactoferrin induces concentration-dependent functional modulation of intestinal proliferation and differentiation. *Pediatric Research*, *61*(4), 410–414.

Buckley, L., Guyatt, G., Fink, H. A., Cannon, M., Grossman, J., Hansen, K. E., McAlindon, T. (2017). 2017 American College of Rheumatology Guideline for the Prevention and Treatment of Glucocorticoid-Induced Osteoporosis. *Arthritis & Rheumatology*, *69*(8), 1521–1537.

Bullen, J. J., Rogers, H. J., & Griffiths, E. (1978). Role of iron in bacterial infection. *Current Topics in Microbiology and Immunology*, 80, 1–35.

Caballero-Alias, A. M., Loverldge, N., Pitsillides, A., Parker, M., Kaptoge, S., Lyon, A., Reeve, J. (2005). Osteocytic expression of constitutive NO synthase isoforms in the femoral neck cortex: a case-control study of intracapsular hip fracture. J. Bone Miner. Res., 20(2), 268-273.

Camacho, P. M., Petak, S. M., Binkley, N., Clarke, B. L., Harris, S. T., Hurley, D. L., Watts, N.
B. (2016). American Association of Clinical Endocrinologists and American Collage of
Endocrinology Clinical Practice Guidelines for the Diagnosis and Treatment of Postmenopausal
Osteoporosis - 2016. *Endocrine Practice*, 22(9), 1111–1118.

Cancedda, R., Castagnola, P., Cancedda, F. D., Dozin, B., & Quarto, R. (2000). Developmental control of chondrogenesis and osteogenesis. *The International Journal of Developmental Biology*, *44*(6), 707–714.

Cao, H., Ke, Y., Zhang, Y., Zhang, C. J., Qian, W., & Zhang, G. L. (2012). Icariin stimulates MC3T3-E1 cell proliferation and differentiation through up-regulation of bone morphogenetic protein-2. *International Journal of Molecular Medicine*, *29*(3), 435–439.

Cao, X., & Chen, D. (2005). The BMP signaling and in vivo bone formation. Gene, 357(1), 1–8.

Cargnello, M., & Roux, P. P. (2011). Activation and function of the MAPKs and their substrates, the MAPK-activated protein kinases. *Microbiology and Molecular Biology Reviews : MMBR*, 75(1), 50–83.

Carpena, N. T., Abueva, C. D. G., Padalhin, A. R., & Lee, B. T. (2017). Evaluation of egg white ovomucin-based porous scaffold as an implantable biomaterial for tissue engineering. *Journal of Biomedical Materials Research Part B: Applied Biomaterials*, *105*(7), 2107–2117.

Celil, A. B., & Campbell, P. G. (2005). BMP-2 and Insulin-like Growth Factor-I Mediate Osterix (Osx) Expression in Human Mesenchymal Stem Cells via the MAPK and Protein Kinase D Signaling Pathways. *Journal of Biological Chemistry*, *280*(36), 31353–31359.

Chae, H. J., Jeong, B. J., Ha, M. S., Lee, J. K., Byun, J. O., Jung, W. Y., Kim, H. R. (2002). Erk Map Kinase is required in 1,25(OH)2D3-induced differentiatio in human osteobalsts. *Immunopharmacology and Immunotoxicology*, *24*(1), 31–41.

Chamanza, R., Toussaint, M. J. M., van Ederen, A. M., van Veen, L., Hulskamp-Koch, C., & Fabri, T. H. F. (1999). Serum amyloid a and transferrin in chicken. A preliminary investigation of using acute - phase variables to assess diseases in chickens. *Veterinary Quarterly*, *21*(4), 158–162.

Chau, J. F. L., Leong, W. F., & Li, B. (2009). Signaling pathways governing osteoblast proliferation, differentiation and function. *Histology and Histopathology*, *24*(12), 1593–1606.

Chen, C., Koh, A. J., Datta, N. S., Zhang, J., Keller, E. T., Xiao, G., McCauley, L. K. (2004). Impact of the Mitogen-activated Protein Kinase Pathway on Parathyroid Hormone-related Protein Actions in Osteoblasts. *Journal of Biological Chemistry*, *279*(28), 29121–29129.

Chen, G., Deng, C., & Li, Y. P. (2012). TGF-β and BMP signaling in osteoblast differentiation and bone formation. *International Journal of Biological Sciences*, 8(2), 272–288.

Chen, J. S., & Sambrook, P. N. (2012). Antiresorptive therapies for osteoporosis: a clinical overview. *Nature Reviews Endocrinology*, *8*(2), 81–91.

Cheng, B., Kato, Y., Zhao, S., Luo, J., Sprague, E., Bonewald, L. F., Jiang, J. X. (2001). PGE(2) is essential for gap junction-mediated intercellular communication between osteocyte-like MLO-Y4 cells in response to mechanical strain. *Endocrinology*, 142(8), 3464-3473.

Chung, M., Lee, J., Terasawa, T., Lau, J., & Trikalinos, T. A. (2011). Vitamin D With or Without Calcium Supplementation for Prevention of Cancer and Fractures: An Updated Metaanalysis for the U.S. Preventive Services Task Force. *Annals of Internal Medicine*, *155*(12), 827.

Clement-Lacroix, P., Ai, M., Morvan, F., Roman-Roman, S., Vayssiere, B., Belleville, C., Rawadi, G. (2005). Lrp5-independent activation of Wnt signaling by lithium chloride increases bone formation and bone mass in mice. *Proceedings of the National Academy of Sciences*, *102*(48), 17406–17411. Coheley, L. M., Kindler, J. M., Laing, E. M., Oshri, A., Hill Gallant, K. M., Warden, S. J., Lewis, R. D. (2018). Whole egg consumption and cortical bone in healthy children. *Osteoporosis International*, 1–9.

Conneely, O. M. (2001). Antiinflammatory activities of lactoferrin. *Journal of the American College of Nutrition*, *20*(5 Suppl), 389S–395S; discussion 396S–397S.

Cornish, J., Callon, K. E., Lin, C. Q., Xiao, C. L., Gamble, G. D., Cooper, G. J. S., & Reid, I. R. (1999). Comparison of the Effects of Calcitonin Gene-Related Peptide and Amylin on Osteoblasts. *Journal of Bone and Mineral Research*, *14*(8), 1302–1309.

Cornish, J., Callon, K. E., Naot, D., Palmano, K. P., Banovic, T., Bava, U., Reid, I. R. (2004). Lactoferrin is a potent regulator of bone cell activity and increases bone formation in vivo. *Endocrinology*, *145*(9), 4366–4374. https://doi.org/10.1210/en.2003-1307

Cosman, F., de Beur, S. J., LeBoff, M. S., Lewiecki, E. M., Tanner, B., Randall, S., National Osteoporosis Foundation. (2014). Clinician's Guide to Prevention and Treatment of Osteoporosis. Osteoporosis International : A Journal Established as Result of Cooperation between the European Foundation for Osteoporosis and the National Osteoporosis Foundation of the USA, 25(10), 2359–2381.

Crockett, J. C., Mellis, D. J., Scott, D. I., & Helfrich, M. H. (2011). New knowledge on critical osteoclast formation and activation pathways from study of rare genetic diseases of osteoclasts: focus on the RANK/RANKL axis. *Osteoporosis International*, *22*(1), 1–20.

Crockett, J. C., Rogers, M. J., Coxon, F. P., Hocking, L. J., & Helfrich, M. H. (2011). Bone remodelling at a glance. *Journal of Cell Science*, *124*(Pt 7), 991–998.

Crockett, J., & Das, J. C. (2013). Osteoporosis: a current view of pharmacological prevention and treatment. *Drug Design, Development and Therapy*, *7*, 435.

Cundy, T., Reid, I. R., & Grey, A. (2014). Metabolic bone disease. *Clinical Biochemistry: Metabolic and Clinical Aspects*, 604–635. David, M., Wannecq, E., Descotes, F., Jansen, S., Deux, B., Ribeiro, J., Peyruchaud, O. (2010). Cancer Cell Expression of Autotaxin Controls Bone Metastasis Formation in Mouse through Lysophosphatidic Acid-Dependent Activation of Osteoclasts. *PLoS ONE*, *5*(3), e9741.

Day, T. F., Guo, X., Garrett-Beal, L., & Yang, Y. (2005). Wnt/β-Catenin Signaling in Mesenchymal Progenitors Controls Osteoblast and Chondrocyte Differentiation during Vertebrate Skeletogenesis. *Developmental Cell*, 8(5), 739–750.

de Villiers, T. J., Gass, M. L. S., Haines, C. J., Hall, J. E., Lobo, R. a, Pierroz, D. D., & Rees, M. (2013). Global consensus statement on menopausal hormone therapy. *Climacteric*, *16*(2), 203–204.

Del Giacco, G. S., Leone, A. L., & Ferlazzo, A. (1985). Total IgE in newborns treated prophylactically with conalbumin. *International Journal of Tissue Reactions*, 7(6), 535–537.

Derynck, R., & Zhang, Y. E. (2003). Smad-dependent and Smad-independent pathways in TGFβ family signalling. *Nature*, *425*(6958), 577–584.

Dijke, P. ten, & Hill, C. S. (2004). New insights into TGF-β–Smad signalling. *Trends in Biochemical Sciences*, *29*(5), 265–273.

Donida, B. M., Mrak, E., Gravaghi, C., Villa, I., Cosentino, S., Zacchi, E., Ferraretto, A. (2009). Casein phosphopeptides promote calcium uptake and modulate the differentiation pathway in human primary osteoblast-like cells. *Peptides*, *30*(12), 2233–2241.

Drake, M. T., Clarke, B. L., & Khosla, S. (2008). Bisphosphonates: mechanism of action and role in clinical practice. *Mayo Clinic Proceedings*, *83*(9), 1032–1045.

Droogenbroeck, C. Van, & Vanrompay, D. (2013). Use of ovotransferrin on a turkey farm to reduce respiratory disease. *Veterinary Record*, *172*(3), 71.1-71.4

Du, M., Wang, K., Wu, C., & Zhang, L. (2009). Effects of bovine colostrum acid protein on bone loss and hemobiochemistry indexes in rats. *Dairy Science and Technology*, *89*(5), 449–461.

Du, M., Xu, W., Yi, H., Han, X., Wang, C., & Zhang, L. (2011). Protective effects of bovine colostrum acid proteins on bone loss of ovariectomized rats and the ingredients identification. *Molecular Nutrition & Food Research*, *55*(2), 220–228. https://doi.org/10.1002/mnfr.200900593

Easton, R. M., Cho, H., Roovers, K., Shineman, D. W., Mizrahi, M., Forman, M. S., Birnbaum, M. J. (2005). Role for Akt3/Protein Kinase B in Attainment of Normal Brain Size. *Molecular and Cellular Biology*, *25*(5), 1869–1878.

Eaton-Evans, J. (1994). Osteoporosis and the role of diet. *British Journal of Biomedical Science*, *51*(4), 358–370.

Eferl, R., Hoebertz, A., Schilling, A. F., Rath, M., Karreth, F., Kenner, L., Wagner, E. F. (2004). The Fos-related antigen Fra-1 is an activator of bone matrix formation. *The EMBO Journal*, *23*(14), 2789–2799.

Ehara, A., Ogata, K., Imazato, S., Ebisu, S., Nakano, T., & Umakoshi, Y. (2003). Effects of α -TCP and TetCP on MC3T3-E1 proliferation, differentiation and mineralization. *Biomaterials*, 24(5), 831–836.

Ehret-Sabatier, L., Loew, D., Goyffon, M., Fehlbaum, P., Hoffmann, J. A., van Dorsselaer, A., & Bulet, P. (1996). Characterization of novel cysteine-rich antimicrobial peptides from scorpion blood. *The Journal of Biological Chemistry*, *271*(47), 29537–29544.

Elias, R. J., Kellerby, S. S., & Decker, E. A. (2008). Antioxidant Activity of Proteins and Peptides. *Critical Reviews in Food Science and Nutrition*, *48*(5), 430–441.

Elmore, S. (2007). Apoptosis: A Review of Programmed Cell Death. *Toxicologic Pathology*, *35*(4), 495–516.

Evans, E. M., Racette, S. B., Van Pelt, R. E., Peterson, L. R., & Villareal, D. T. (2007). Effects of soy protein isolate and moderate exercise on bone turnover and bone mineral density in postmenopausal women. *Menopause*, *14*(3), 481–488.

Everts, V., Delaissé, J. M., Korper, W., Jansen, D. C., Tigchelaar-Gutter, W., Saftig, P., & Beertsen, W. (2002). The Bone Lining Cell: Its Role in Cleaning Howship's Lacunae and Initiating Bone Formation. *Journal of Bone and Mineral Research*, *17*(1), 77–90.

Fanti, P., Monier-Faugere, M. C., Geng, Z., Schmidt, J., Morris, P. E., Cohen, D., & Malluche,
H. H. (1998). The Phytoestrogen Genistein Reduces Bone Loss in Short-Term Ovariectomized
Rats. *Osteoporosis International*, 8(3), 274–281.

Farrar, G., Barone, J., & Morgan, A. (2010). Ovalbumin-based porous scaffolds for bone tissue regeneration. *Journal of Tissue Engineering*, *2010*, 209860.

Feng, X. (2009). Chemical and Biochemical Basis of Cell-Bone Matrix Interaction in Health and Disease. *Current Chemical Biology*, *3*(2), 189–196.

FitzGerald, R. J. (1998). Potential Uses of Caseinophosphopeptides. *International Dairy Journal*, 8(5–6), 451–457.

Fitzpatrick, L. A., Dabrowski, C. E., Cicconetti, G., Gordon, D. N., Fuerst, T., Engelke, K., & Genant, H. K. (2012). Ronacaleret, a calcium-sensing receptor antagonist, increases trabecular but not cortical bone in postmenopausal women. *Journal of Bone and Mineral Research*, *27*(2), 255–262.

Fogelman, I., & Blake, G. M. (2005). Strontium ranelate for the treatment of osteoporosis. *BMJ* (*Clinical Research Ed.*), 330(7505), 1400–1401.

Fritsche, K. L., Setchell, K. D. R., Conneely, O. M., Beck, M. A., Prasad, K. N., Keen, C. L., & Klurfeld, D. M. (2001). Anti-inflammatory activities of lactoferrin. *Journal of the American College of Nutrition*, *20*(5 SUPPL.), 396S–397S.

Fromigué, O., Modrowski, D., & Marie, P. J. (2004). Growth factors and bone formation in osteoporosis: roles for fibroblast growth factor and transforming growth factor beta. *Current Pharmaceutical Design*, *10*(21), 2593–2603.

Fujita, T., Azuma, Y., Fukuyama, R., Hattori, Y., Yoshida, C., Koida, M., Komori, T. (2004). Runx2 induces osteoblast and chondrocyte differentiation and enhances their migration by coupling with PI3K-Akt signaling. *The Journal of Cell Biology*, *166*(1), 85–95.

Gabay, C., & Kushner, I. (1999). Acute-Phase Proteins and Other Systemic Responses to Inflammation. *New England Journal of Medicine*, *340*(6), 448–454.

Galvez, A., Gimenez-Gallego, G., Reuben, J. P., Roy-Contancin, L., Feigenbaum, P., Kaczorowski, G. J., & Garcia, M. L. (1990). Purification and characterization of a unique, potent, peptidyl probe for the high conductance calcium-activated potassium channel from venom of the scorpion Buthus tamulus. *The Journal of Biological Chemistry*, *265*(19), 11083– 11090.

Garcia, A. J., Tom, C., Guemes, M., Polanco, G., Mayorga, M. E., Wend, K., Krum, S. A. (2013). ERα signaling regulates MMP3 expression to induce FasL cleavage and osteoclast apoptosis. *Journal of Bone and Mineral Research*, *28*(2), 283–290.

Garnero, P., Buchs, N., Zekri, J., Rizzoli, R., Coleman, R. E., & Delmas, P. D. (2000). Markers of bone turnover for the management of patients with bone metastases from prostate cancer. *British Journal of Cancer*, *82*(4), 858–864.

Garofalo, R. S., Orena, S. J., Rafidi, K., Torchia, A. J., Stock, J. L., Hildebrandt, A. L., Coleman, K. G. (2003). Severe diabetes, age-dependent loss of adipose tissue, and mild growth deficiency in mice lacking Akt2/PKB beta. *The Journal of Clinical Investigation*, *112*(2), 197–208.

Gars, M. Le, Seiler, C., Kay, A., Bayless, N., Starosvetsky, E., Moore, L., Blish, C. (2018). CD38 is a key regulator of enhanced NK cell immune responses during pregnancy through its role in immune synapse formation. *BioRxiv*, 349084.

Ge, C., Yang, Q., Zhao, G., Yu, H., Kirkwood, K. L., & Franceschi, R. T. (2012). Interactions between extracellular signal-regulated kinase 1/2 and P38 Map kinase pathways in the control of RUNX2 phosphorylation and transcriptional activity. *Journal of Bone and Mineral Research*, *27*(3), 538–551.

Gentili, C., Bianco, P., Neri, M., Malpeli, M., Campanile, G., Castagnola, P., Cancedda, F. D. (1993). Cell proliferation, extracellular matrix mineralization, and ovotransferrin transient expression during in vitro differentiation of chick hypertrophic chondrocytes into osteoblast-like cells. *Journal of Cell Biology*, *122*(3), 703–712.

Gentili, C., Doliana, R., Bet, P., Campanile, G., Colombatti, A., Cancedda, F. D., & Cancedda, R. (1994). Ovotransferrin and ovotransferrin receptor expression during chondrogenesis and endochondral bone formation in developing chick embryo. *The Journal of Cell Biology*, *124*(4), 579–588.

Giansanti, F., Giardi, M. F., Massucci, M. T., Botti, D., & Antonini, G. (2007). Ovotransferrin expression and release by chicken cell lines infected with Marek's disease virus. *Biochemistry and Cell Biology*, *85*(1), 150–155.

Giansanti, F., Leboffe, L., Angelucci, F., & Antonini, G. (2015). The nutraceutical properties of ovotransferrin and its potential utilization as a functional food. *Nutrients*, 7(11), 9105–9115. https://doi.org/10.3390/nu7115453

Giansanti, F., Leboffe, L., Pitari, G., Ippoliti, R., & Antonini, G. (2012). Physiological roles of ovotransferrin. *Biochimica et Biophysica Acta (BBA) - General Subjects*, *1820*(3), 218–225.

Giansanti, F., Massucci, M. T., Giardi, M. F., Nozza, F., Pulsinelli, E., Nicolini, C., Antonini, G. (2005). Antiviral activity of ovotransferrin derived peptides. *Biochemical and Biophysical Research Communications*, *331*(1), 69–73.

Giansanti, F., Rossi, P., Massucci, M. T., Botti, D., Antonini, G., Valenti, P., & Seganti, L. (2002). Antiviral activity of ovotransferrin discloses an evolutionary strategy for the defensive activities of lactoferrin. *Biochemistry and Cell Biology*, *80*(1), 125–130.

Giustina, A., Mazziotti, G., & Canalis, E. (2008). Growth Hormone, Insulin-Like Growth Factors, and the Skeleton. *Endocrine Reviews*, *29*(5), 535–559.

Glantschnig, H., Fisher, J. E., Wesolowski, G., Rodan, G. A., & Reszka, A. A. (2003). M-CSF, TNFα and RANK ligand promote osteoclast survival by signaling through mTOR/S6 kinase. *Cell Death & Differentiation*, *10*(10), 1165–1177. Goldring, S. R. (2015). The osteocyte: key player in reguating bone turnover. RMD Open, 15, 1.

Grand View Research. (2016). Functional foods market analysis by product (carotenoids, dietary fibers, fitty acids, minerals, prebiotics, probiotics, vitamins), by application, by End-Use (Sports nutrition, weight managment, immunity, digestive health) and segment forecasts, 2018 to 2024.

Grand view research. (2017). Osteoporosis drug market analysis by product (branded, bishosphonates, pharathyroid hormone therapy, calcitonin, selective estrogen inhibitors modulator (SERM), Rank ligand inhibitos, generics), and segment forecasts, 2018-2025.

Greenblatt, M. B., Shim, J. H., Zou, W., Sitara, D., Schweitzer, M., Hu, D., Glimcher, L. H. (2010). The p38 MAPK pathway is essential for skeletogenesis and bone homeostasis in mice. *The Journal of Clinical Investigation*, *120*(7), 2457–2473.

Greendale, G. A., FitzGerald, G., Huang, M. H., Sternfeld, B., Gold, E., Seeman, T., Sowers, M. (2002). Dietary Soy Isoflavones and Bone Mineral Density: Results from the Study of Women's Health Across the Nation. *American Journal of Epidemiology*, *155*(8), 746–754.

Grey, A., Banovic, T., Zhu, Q., Watson, M., Callon, K., Palmano, K., Cornish, J. (2004). The low-density lipoprotein receptor-related protein 1 is a mitogenic receptor for lactoferrin in osteoblastic cells. *Molecular Endocrinology (Baltimore, Md.)*, *18*(9), 2268–2278.

Groves, J. T., & Kuriyan, J. (2010). Molecular mechanisms in signal transduction at the membrane. *Nature Structural & Molecular Biology*, *17*(6), 659–665.

Guo, H. Y., Jiang, L., Ibrahim, S. A., Zhang, L., Zhang, H., Zhang, M., & Ren, F. Z. (2009). Orally administered lactoferrin preserves bone mass and microarchitecture in ovariectomized rats. *J Nutr*, *139*(C), 958–964.

Hadji, P., Coleman, R., & Gnant, M. (2013). Bone effects of mammalian target of rapamycin (mTOR) inhibition with everolimus. *Critical Reviews in Oncology/Hematology*, 87(2), 101–111.

Hall, B. K., & Miyake, T. (2000). All for one and one for all: Condensations and the initiation of skeletal development. *BioEssays*, *22*(2), 138–147.

Hallquist, N. A., & Klasing, K. C. (1994). Serotransferrin, ovotransferrin and metallothionein levels during an immune response in chickens. *Comparative Biochemistry and Physiology*. *Biochemistry and Molecular Biology*, *108*(3), 375–384.

Hamdy, N. A. T. (2008). Denosumab: RANKL inhibition in the management of bone loss. *Drugs* of Today, 44(1), 7.

Han, J., & Wang, W. (2017). Effects of tanshinol on markers of bone turnover in ovariectomized rats and osteoblast cultures. *PloS One*, *12*(7), e0181175.

Hartmann, R., & Meisel, H. (2007). Food-derived peptides with biological activity: from research to food applications. *Current Opinion in Biotechnology*, *18*(2), 163–169.

Hatta, H., Kapoor, M. P., & Juneja, L. R. (2007). Bioactive Components in Egg Yolk. In *Egg Bioscience and Biotechnology* (pp. 185–237). Hoboken, NJ, USA: John Wiley & Sons, Inc.

Hayashi, M., Nakashima, T., Taniguchi, M., Kodama, T., Kumanogoh, A., & Takayanagi, H. (2012). Osteoprotection by semaphorin 3A. *Nature*, *485*(7396), 69–74.

He, X., Semenov, M., Tamai, K., & Zeng, X. (2004). LDL receptor-related proteins 5 and 6 in Wnt/-catenin signaling: Arrows point the way. *Development*, *131*(8), 1663–1677.

Health Canada. (2018). Osteoporosis. Retrieved January 27, 2019, from https://www.canada.ca/en/public-health/services/chronic-diseases/osteoporosis.html

Heaney, R. P. (2009). Dairy and bone health. *Journal of the American College of Nutrition*, 28 *Suppl 1*, 82S–90S.

Heino, T. J., Hentunen, T. A., Vaananen, H. K. (2002). Osteocytes inhibit osteoclastic bone resorption through transforming growth factor-beta: enhancement by estrogen. *J. Cell Biochem.*, 85(1), 185-197.

Heo, S. Y., Ko, S. C., Nam, S. Y., Oh, J., Kim, Y. M., Kim, J. I., Jung, W. K. (2018). Fish bone peptide promotes osteogenic differentiation of MC3T3-E1 pre-osteoblasts through upregulation

of MAPKs and Smad pathways activated BMP-2 receptor. *Cell Biochemistry and Function*, *36*(3), 137–146. https://doi.org/10.1002/cbf.3325

Herz, J., & Strickland, D. K. (2001). LRP: a multifunctional scavenger and signaling receptor. *Journal of Clinical Investigation*, *108*(6), 779–784.

Horowitz, M. C., Bothwell, A. L. M., Hesslein, D. G. T., Pflugh, D. L., & Schatz, D. G. (2005). B cells and osteoblast and osteoclast development. *Immunological Reviews*, *208*(1), 141–153.

Hou, J. M., Xue, Y., & Lin, Q. M. (2012). Bovine lactoferrin improves bone mass and microstructure in ovariectomized rats via OPG/RANKL/RANK pathway. *Acta Pharmacologica Sinica*, *33*(10), 1277–1284. https://doi.org/10.1038/aps.2012.83

Houde, N., Chamoux, E., Bisson, M., & Roux, S. (2009). Transforming growth factor-beta1 (TGF-beta1) induces human osteoclast apoptosis by up-regulating Bim. *The Journal of Biological Chemistry*, 284(35), 23397–23404.

Hu, Y., Chan, E., Wang, S. X., & Li, B. (2003). Activation of p38 Mitogen-Activated Protein Kinase Is Required for Osteoblast Differentiation. *Endocrinology*, *144*(5), 2068–2074.

Huang, D., Zhao, L., Zong, Y., Yang, S., Zhang, L., Song, Q., Lu, C. (2006, May 26). Ovotransferrin iron supplement agent, and its preparing method and use.

Huang, W. Y., Majumder, K., & Wu, J. (2010). Oxygen radical absorbance capacity of peptides from egg white protein ovotransferrin and their interaction with phytochemicals. *Food Chemistry*, *123*(3), 635–641.

Hughes, D. E., & Boyce, B. F. (1997). Apoptosis in bone physiology and disease. *Molecular Pathology* : *MP*, *50*(3), 132–137.

Hughes, D. E., Dai, A., Tiffee, J. C., Li, H. H., Mundy, G. R., & Boyce, B. F. (1996). Estrogen promotes apoptosis of murine osteoclasts mediated by TGF-beta. *Nature Medicine*, *2*(10), 1132–1136.

Humphrey, M. B., Lanier, L. L., & Nakamura, M. C. (2005). Role of ITAM-containing adapter proteins and their receptors in the immune system and bone. *Immunological Reviews*, 208(1), 50–65.

Hung, R. W., & Chow, A. W. (1997). Apoptosis: Molecular mechanisms, regulation and role in pathogenesis. *The Canadian Journal of Infectious Diseases = Journal Canadian Des Maladies Infectieuses*, 8(2), 103–109.

Ibrahim, H. R. (1996). Insights into the structure-function relationships of ovalbumin, ovotransferrin, and lysozyme. In *Hen Eggs: Basic and Applied Science*.

Ibrahim, H. R., & Kiyono, T. (2009). Novel Anticancer Activity of the Autocleaved Ovotransferrin against Human Colon and Breast Cancer Cells. *Journal of Agricultural and Food Chemistry*, *57*(23), 11383–11390.

Ibrahim, H. R., Haraguchi, T., & Aoki, T. (2006). Ovotransferrin is a redox-dependent autoprocessing protein incorporating four consensus self-cleaving motifs flanking the two kringles. *Biochimica et Biophysica Acta (BBA) - General Subjects*, *1760*(3), 347–355.

Ibrahim, H. R., Hoq, M. I., & Aoki, T. (2007). Ovotransferrin possesses SOD-like superoxide anion scavenging activity that is promoted by copper and manganese binding. *International Journal of Biological Macromolecules*, *41*(5), 631–640.

Ibrahim, H. R., Iwamori, E., Sugimoto, Y., & Aoki, T. (1998). Identification of a distinct antibacterial domain within the N-lobe of ovotransferrin. *Biochimica et Biophysica Acta*, *1401*(3), 289–303.

Ibrahim, H. R., Sugimoto, Y., & Aoki, T. (2000). Ovotransferrin antimicrobial peptide (OTAP-92) kills bacteria through a membrane damage mechanism. *Biochimica et Biophysica Acta*, *1523*(2–3), 196–205.

Ibrahim, H. R., Tatsumoto, S., Ono, H., Van Immerseel, F., Raspoet, R., & Miyata, T. (2015). A novel antibiotic-delivery system by using ovotransferrin as targeting molecule. *European Journal of Pharmaceutical Sciences*, *66*, 59–69.

Itabashi, A. (2006). [Prevention of osteoporosis by foods and dietary supplements. Milk basic protein (MBP) increases bone mineral density in young adult women and perimenopausal women]. *Clinical Calcium*, *16*(10), 1632–1638.

Iwasawa, M., Miyazaki, T., Nagase, Y., Akiyama, T., Kadono, Y., Nakamura, M., Tanaka, S. (2009). The antiapoptotic protein Bcl-xL negatively regulates the bone-resorbing activity of osteoclasts in mice. *The Journal of Clinical Investigation*, *119*(10), 3149–3159.

Jahandideh, F., Chakrabarti, S., Davidge, S. T., & Wu, J. (2016). Antioxidant Peptides Identified from Ovotransferrin by the ORAC Method Did Not Show Anti-Inflammatory and Antioxidant Activities in Endothelial Cells. *Journal of Agricultural and Food Chemistry*, *64*(1), 113–119.

Jaiswal, R. K., Jaiswal, N., Bruder, S. P., Mbalaviele, G., Marshak, D. R., & Pittenger, M. F. (2000). Adult human mesenchymal stem cell differentiation to the osteogenic or adipogenic lineage is regulated by mitogen-activated protein kinase. *The Journal of Biological Chemistry*, *275*(13), 9645–9652.

Javed, A., Chen, H., & Ghori, F. Y. (2010). Genetic and Transcriptional Control of Bone Formation. *Oral and Maxillofacial Surgery Clinics of North America*, *22*(3), 283–293.

Jee, W. S., & Yao, W. (2001). Overview: animal models of osteopenia and osteoporosis. *Journal of Musculoskeletal & Neuronal Interactions*, 1(3), 193–207. Retrieved from http://www.ncbi.nlm.nih.gov/pubmed/15758493

Jha, S., Wang, Z., Laucis, N., & Bhattacharyya, T. (2015). Trends in Media Reports, Oral Bisphosphonate Prescriptions, and Hip Fractures 1996-2012: An Ecological Analysis. *Journal of Bone and Mineral Research*, *30*(12), 2179–2187.

Ji, M, Leem, K. H., Kim, M., & Kim, H. K. (2007). Egg Yolk Soluble Protein Stimulates the Proliferation and Differentiation of Osteoblastic MC3T3-E1 Cells. *Bioscience, Biotechnology, and Biochemistry*, *71*(5), 1327–1329. https://doi.org/10.1271/bbb.60217

Ji, M. X., & Yu, Q. (2015). Primary osteoporosis in postmenopausal women. *Chronic Diseases* and *Translational Medicine*, *1*(1), 9–13.

Ji, M., Leem, K. H., Kim, M., & Klim H. K. (2007). Egg Yolk Soluble Protein Stimulates the Proliferation and Differentiation of Osteoblastic MC3T3-E1 Cells. *Bioscience, Biotechnology, and Biochemistry*, *71*(5), 1327–1329.

John, V., Hock, J. M., Short, L. L., Glasebrook, A. L., & Galvin, R. J. (1996). A role for CD8+ T lymphocytes in osteoclast differentiation in vitro. *Endocrinology*, *137*(6), 2457–2463.

Justesen, J., Stenderup, K., Ebbesen, E. N., Mosekilde, L., Steiniche, T., & Kassem, M. (2001). Adipocyte tissue volume in bone marrow is increased with aging and in patients with osteoporosis. - PubMed - NCBI. *Biogerontology*, *2*(3), 165–171.

Kalu, D. N. (1991). The ovariectomized rat model of postmenopausal bone loss. *Bone and Mineral*, *15*(3), 175–191. https://doi.org/10.1016/0169-6009(91)90124-I

Kanno, T., Takahashi, T., Tsujisawa, T., Ariyoshi, W., & Nishihara, T. (2007). Mechanical stress-mediated Runx2 activation is dependent on Ras/ERK1/2 MAPK signaling in osteoblasts. *Journal of Cellular Biochemistry*, *101*(5), 1266–1277.

Kapinas, K., & Delany, A. M. (2011). MicroRNA biogenesis and regulation of bone remodeling. *Arthritis Research & Therapy*, *13*(3), 220.

Karin, M. (2005). Inflammation-activated Protein Kinases as Targets for Drug Development. *Proceedings of the American Thoracic Society*, *2*(4), 386–390.

Katz, M., Amit, I., & Yarden, Y. (2007). Regulation of MAPKs by growth factors and receptor tyrosine kinases. *Biochimica et Biophysica Acta*, *1773*(8), 1161–1176.

Kawabata, H. (2018). Transferrin and transferrin receptors update. *Free Radical Biology and Medicine*.

Kawamura, N., Kugimiya, F., Oshima, Y., Ohba, S., Ikeda, T., Saito, T., & Shinoda, Y. (2007). Akt1 in Osteoblasts and Osteoclasts Controls Bone Remodeling, (10). Kenner, L., Hoebertz, A., Beil, F. T., Keon, N., Karreth, F., Eferl, R., Wagner, E. F. (2004). Mice lacking JunB are osteopenic due to cell-autonomous osteoblast and osteoclast defects. *The Journal of Cell Biology*, *164*(4), 613–623.

Kenny, A. M., Mangano, K. M., Abourizk, R. H., Bruno, R. S., Anamani, D. E., Kleppinger, A., Kerstetter, J. E. (2009). Soy proteins and isoflavones affect bone mineral density in older women: a randomized controlled trial. *The American Journal of Clinical Nutrition*, 90(1), 234–242.

Kerstetter, J. E., & Insogna, K. L. (2003). Dietary protein, calcium metabolism, and skeletal homeostasis. *The American Journal of Clinical Nutrition*, 78, 584–592.

Kerstetter, J. E., Caseria, D. M., Mitnick, M. E., Ellison, A. F., Gay, L. F., Liskov, T. A. P., Insogna, K. L. (1997). Increased circulating concentrations of parathyroid hormonein healthy, youngwomen consuminga, 1188–1196.

Khosla, S., & Shane, E. (2016). A Crisis in the Treatment of Osteoporosis. *Journal of Bone and Mineral Research*, *31*(8), 1485–1487.

Khosla, S., Burr, D., Cauley, J., Dempster, D. W., Ebeling, P. R., Felsenberg, D., American Society for Bone and Mineral Research. (2007). Bisphosphonate-Associated Osteonecrosis of the Jaw: Report of a Task Force of the American Society for Bone and Mineral Research. *Journal of Bone and Mineral Research*, 22(10), 1479–1491.

Kim, B. H., Oh, J. H., & Lee, N. K. (2017). The Inactivation of ERK1/2, p38 and NF-kB Is Involved in the Down-Regulation of Osteoclastogenesis and Function by A2B Adenosine Receptor Stimulation. *Molecules and Cells*, *40*(10), 752–760.

Kim, H. K., Kim, M. G., & Leem, K. H. (2008). Inhibitory Effects of Egg Yolk Soluble Protein on Bone Resorption. *Bioscience, Biotechnology, and Biochemistry*, *72*(7), 1929–1931.

Kim, H. K., Lee, S., & Leem, K. H. (2011). Protective effect of egg yolk peptide on bone metabolism. *Menopause*, *18*(3), 307–313.

Kim, H., Kim, M. G., Leem, K. H., Kim, H. K., Kim, M. G., & Leem, K. H. (2013). Osteogenic Activity of Collagen Peptide via ERK/MAPK Pathway Mediated Boosting of Collagen Synthesis and Its Therapeutic Efficacy in Osteoporotic Bone by Back-Scattered Electron Imaging and Microarchitecture Analysis. *Molecules*, *18*(12), 15474–15489.

Kim, J. H., & Kim, N. (2016). Signaling Pathways in Osteoclast Differentiation. *Chonnam Medical Journal*, *52*(1), 12–17.

Kim, J. H., Kim, M. S., Oh, H. G., Lee, H. Y., Park, J. W., Lee, B. G., Chae, H. J. (2013). Treatment of eggshell with casein phosphopeptide reduces the severity of ovariectomy-induced bone loss. *Laboratory Animal Research*, *29*(2), 70.

Kim, M. C., Hong, M. H., Lee, B. H., Choi, H. J., Ko, Y. M., & Lee, Y. K. (2015). Bone Tissue Engineering by Using Calcium Phosphate Glass Scaffolds and the Avidin–Biotin Binding System. *Annals of Biomedical Engineering*, *43*(12), 3004–3014.

Kini, Usha; Nandeesh, B. N. (2012). *Physiology of Bone Formation, Remodeling, and Metabolism. Radionuclide and Hybrid Bone Imaging* (Vol. 9783642024).

Kitagawa, H., Yoshizawa, Y., Yokoyama, T., Takeuchi, T., Talukder, M. J. R., Shimizu, H., Harada, E. (2003). Persorption of bovine lactoferrin from the intestinal lumen into the systemic circulation via the portal vein and the mesenteric lymphatics in growing pigs. *The Journal of Veterinary Medical Science*, *65*(5), 567–572.

Kiviranta, R., Morko, J., Uusitalo, H., Aro, H. T., Vuorio, E., & Rantakokko, J. (2001). Accelerated Turnover of Metaphyseal Trabecular Bone in Mice Overexpressing Cathepsin K. *Journal of Bone and Mineral Research*, *16*(8), 1444–1452.

Ko, K. Y., Mendoncam, A. F., Ismail, H., & Ahn, D. U. (2009). Ethylenediaminetetraacetate and lysozyme improves antimicrobial activities of ovotransferrin against Escherichia coli O157:H7. *Poultry Science*, *88*(2), 406–414.

Kobayashi, T., Koie, H., Watanabe, A., Ino, A., Watabe, K., Kim, M., Otsuji, K. (2015). Effects of food enriched with egg yolk hydrolysate (bone peptide) on bone metabolism in orchidectomized dogs. *J. Vet. Med. Sci*, 77(4), 503–506.

Kobayashi, Y., Rupa, P., Kovacs-Nolan, J., Turner, P. V., Matsui, T., & Mine, Y. (2015). Oral administration of hen egg white ovotransferrin attenuates the development of colitis induced by dextran sodium sulfate in mice. *Journal of Agricultural and Food Chemistry*, 63(5), 1532–1539.

Komori, T. (2015). Animal models for osteoporosis. *European Journal of Pharmacology*, 759, 287–294.

Kramer, I., Halleux, C., Keller, H., Pegurri, M., Gooi, J. H., Weber, P. B., Feng, J. Q., Bonewald,L. F., Knessel M. (2010). Osteocyte Wnt/beta-catenin signaling is required for normal bonehomeostasis. *Mol. Cell Biol.*, 30, 3071-3085

Kreijkamp-Kaspers, S., Kok, L., Grobbee, D. E., de Haan, E. H. F., Aleman, A., Lampe, J. W., & van der Schouw, Y. T. (2004). Effect of Soy Protein Containing Isoflavones on Cognitive Function, Bone Mineral Density, and Plasma Lipids in Postmenopausal Women. *JAMA*, 292(1), 65–74.

Kruger, M. C., Plimmer, G. G., Schollum, L. M., Haggarty, N., Ram, S., & Palmano, K. (2017). The effect of whey acidic protein fractions on bone loss in the ovariectomised rat. *British Journal of Nutrition*, *93*, 244–252.

Krum, S. A., Miranda-Carboni, G. A., Hauschka, P. V, Carroll, J. S., Lane, T. F., Freedman, L. P., & Brown, M. (2008). Estrogen protects bone by inducing Fas ligand in osteoblasts to regulate osteoclast survival. *The EMBO Journal*, *27*(3), 535–545.

Kuang, H., Yang, F., Zhang, Y., Wang, T., & Chen, G. (2018). The Impact of Egg Nutrient Composition and Its Consumption on Cholesterol Homeostasis. *Cholesterol*, 2018, 6303810.

Kugimiya, F., Kawaguchi, H., Ohba, S., Kawamura, N., Hirata, M., Chikuda, H., Chung, U. (2007). GSK-3β Controls Osteogenesis through Regulating Runx2 Activity. *PLoS ONE*, *2*(9), e837.

Kulkarni, N. H., Onyia, J. E., Zeng, Q., Tian, X., Liu, M., Halladay, D. L., Ma, Y. L. (2006). Orally Bioavailable GSK-3α/β Dual Inhibitor Increases Markers of Cellular Differentiation In Vitro and Bone Mass In Vivo. *Journal of Bone and Mineral Research*, *21*(6), 910–920. Kusano, K., Miyaura, C., Inada, M., Tamura, T., Ito, A., Nagase, H., ... Suda, T. (1998). Regulation of Matrix Metalloproteinases (MMP-2, -3, -9, and -13) by Interleukin-1 and Interleukin-6 in Mouse Calvaria: Association of MMP Induction with Bone Resorption 1. *Endocrinology*, *139*(3), 1338–1345.

Kusu, N., Laurikkala, J., Imanishi, M., Usui, H., Konishi, M., Miyake, A., Itoh, N. (2003). Sclerostin Is a Novel Secreted Osteoclast-derived Bone Morphogenetic Protein Antagonist with Unique Ligand Specificity. *Journal of Biological Chemistry*, *278*(26), 24113–24117.

Lai, C. F., Chaudhary, L., Fausto, A., Halstead, L. R., Ory, D. S., Avioli, L. V, & Cheng, S. L. (2001). Erk is essential for growth, differentiation, integrin expression, and cell function in human osteoblastic cells. *The Journal of Biological Chemistry*, *276*(17), 14443–14450.

Lapierre, D. M., Tanabe, N., Pereverzev, A., Spencer, M., Shugg, R. P. P., Dixon, S. J., & Sims, S. M. (2010). Lysophosphatidic Acid Signals through Multiple Receptors in Osteoclasts to Elevate Cytosolic Calcium Concentration, Evoke Retraction, and Promote Cell Survival. *Journal of Biological Chemistry*, *285*(33), 25792–25801.

Laskey, J., Webb, I., Schulman, H. M., & Ponka, P. (1988). Evidence that transferrin supports cell proliferation by supplying iron for DNA synthesis. *Experimental Cell Research*, *176*(1), 87–95.

Lauderdale, D. S., Jacobsen, S. J., Furner, S. E., Levy, P. S., Brody, J. A., & Goldberg, J. (1997). Hip fracture incidence among elderly Asian-American populations. *American Journal of Epidemiology*, *146*(6), 502–509.

Lee, A. W. M., Oates, P. S., & Trinder, D. (2003). Effects of cell proliferation on the uptake of transferrin-bound iron by human hepatoma cells. *Hepatology*, *38*(4), 967–977.

Lee, D. C., McKnight, G. S., & Palmiter, R. D. (1980). The chicken transferrin gene. Restriction endonuclease analysis of gene sequences in liver and oviduct DNA. *The Journal of Biological Chemistry*, *255*(4), 1442–1450.

Lee, J. H., Ahn, D. U., & Paik, H. D. (2018). In vitro Immune-Enhancing Activity of Ovotransferrin from Egg White via MAPK Signaling Pathways in RAW 264.7 Macrophages. *Korean Journal for Food Science of Animal Resources*.

Lee, K. S., Hong, S. H., & Bae, S. C. (2002). Both the Smad and p38 MAPK pathways play a crucial role in Runx2 expression following induction by transforming growth factor- β and bone morphogenetic protein. *Oncogene*, *21*(47), 7156–7163.

Lee, K., Chung, Y. H., Ahn, H., Kim, H., Rho, J., & Jeong, D. (2016). Selective Regulation of MAPK Signaling Mediates RANKL-dependent Osteoclast Differentiation. *International Journal of Biological Sciences*, *12*(2), 235–245.

Lee, N., Cheng, J., enomoto, T., & Nakano, Y. (2006). One peptide derived from hen ovotransferrin as pro-drug to inhibit angiotensin converting enzyme. *Journal of Food and Drug Analysis*, *14*(1), 31–35.

Lee, S. E., Chung, W. J., Kwak, H. B., Chung, C.-H., Kwack, K., Lee, Z. H., & Kim, H. H. (2001). Tumor Necrosis Factor-α Supports the Survival of Osteoclasts through the Activation of Akt and ERK. *Journal of Biological Chemistry*, *276*(52), 49343–49349.

Lee, Y. K., Song, J., Lee, S. B., Kim, K. M., Choi, S. H., Kim, C. K., Kim, K. N. (2004). Proliferation, differentiation, and calcification of preosteoblast-like MC3T3-E1 cells cultured onto noncrystalline calcium phosphate glass. *Journal of Biomedical Materials Research - Part A*, 69(1).

Leem, K. H., Kim, M. G., Kim, H. M., Kim, M., Lee, Y. J., & Kim, H. K. (2004). Effects of Egg Yolk Proteins on the Longitudinal Bone Growth of Adolescent Male Rats. *Bioscience, Biotechnology, and Biochemistry*, *68*(11), 2388–2390.

Leibbrandt, A., & Penninger, J. M. (2008). RANK/RANKL: Regulators of Immune Responses and Bone Physiology. *Annals of the New York Academy of Sciences*, *1143*(1), 123–150.

Levy, F., Bulet, P., & Ehret-Sabatier, L. (2004). Proteomic analysis of the systemic immune response of Drosophila. *Molecular & Cellular Proteomics : MCP*, *3*(2), 156–166.

Lewiecki, E. M., Miller, P. D., McClung, M. R., Cohen, S. B., Bolognese, M. A., Liu, Y., AMG 162 Bone Loss Study Group. (2007). Two-Year Treatment With Denosumab (AMG 162) in a Randomized Phase 2 Study of Postmenopausal Women With Low BMD. *Journal of Bone and Mineral Research*, *22*(12), 1832–1841.

Li, C., Geng, F., Huang, X., Ma, M., & Zhang, X. (2014). Phosvitin phosphorus is involved in chicken embryo bone formation through dephosphorylation. *Poultry Science*, *93*(12), 3065–3072.

Li, D., Liu, J., Guo, B., Liang, C., Dang, L., Lu, C., Zhang, G. (2016). Osteoclast-derived exosomal miR-214-3p inhibits osteoblastic bone formation. *Nature Communications*, *7*(1), 10872.

Li, X., Qin, L., Bergenstock, M., Bevelock, L. M., Novack, D. V., & Partridge, N. C. (2007). Parathyroid Hormone Stimulates Osteoblastic Expression of MCP-1 to Recruit and Increase the Fusion of Pre/Osteoclasts. *Journal of Biological Chemistry*, *282*(45), 33098–33106.

Li, Y., Ge, C., Long, J. P., Begun, D. L., Rodriguez, J. A., Goldstein, S. A., & Franceschi, R. T. (2012). Biomechanical stimulation of osteoblast gene expression requires phosphorylation of the RUNX2 transcription factor. *Journal of Bone and Mineral Research*, *27*(6), 1263–1274.

Lian, J. B., Stein, G. S., van Wijnen, A. J., Stein, J. L., Hassan, M. Q., Gaur, T., & Zhang, Y.
(2012). MicroRNA control of bone formation and homeostasis. *Nature Reviews Endocrinology*, 8(4), 212–227.

Liao, W., Chakrabarti, S., Davidge, S. T., & Wu, J. (2016). Modulatory effects of egg white ovotransferrin-derived tripeptide IRW (Ile-Arg-Trp) on vascular smooth muscle cells against Angiotensin II stimulation. *Journal of Agricultural and Food Chemistry*, *64*(39), 7342–7347.

Lin, G. L., & Hankenson, K. D. (2011). Integration of BMP, Wnt, and notch signaling pathways in osteoblast differentiation. *Journal of Cellular Biochemistry*, *112*(12), 3491–3501.

Liu, L., Xu, M., Tu, Y., Du, H., Zhou, Y., & Zhu, G. (2017). Immunomodulatory effect of protease hydrolysates from ovotransferrin. *Food Funct.*, *8*(4), 1452–1459.

Liu, M., Fan, F., Shi, P., Tu, M., Yu, C., Yu, C., & Du, M. (2018). Lactoferrin promotes MC3T3-E1 osteoblast cells proliferation via MAPK signaling pathways. *International Journal of Biological Macromolecules*, *107*(Pt A), 137–143.

Liu, Q., Li, C., Geng, F., Huang, X., & Ma, M. (2017). Hen egg yolk phosvitin stimulates osteoblast differentiation in the absence of ascorbic acid. *J Sci Food Agric*, *94*, 4532–4538.

Lorget, F., Clough, J., Oliveira, M., Daury, M. C., Sabokbar, A., & Offord, E. (2002). Lactoferrin reduces in vitro osteoclast differentiation and resorbing activity. *Biochemical and Biophysical Research Communications*, *296*(2), 261–266.

Lucas, S., Omata, Y., Hofmann, J., Böttcher, M., Iljazovic, A., Sarter, K., Zaiss, M. M. (2018). Short-chain fatty acids regulate systemic bone mass and protect from pathological bone loss. *Nature Communications*, *9*(1), 55.

Luckman, S. P., Hughes, D. E., Coxon, F. P., Russell, R. G. G., Rogers, M. J., & Rogers, M. J. (1998). Nitrogen-Containing Bisphosphonates Inhibit the Mevalonate Pathway and Prevent Post-Translational Prenylation of GTP-Binding Proteins, Including Ras. *Journal of Bone and Mineral Research*, *13*(4), 581–589.

Luna-Ulloa, L. B., Hernández-Maqueda, J. G., Castañeda-Patlán, M. C., & Robles-Flores, M. (2011). Protein kinase C in Wnt signaling: Implications in cancer initiation and progression. *IUBMB Life*, *63*(10), 915–921.

Luo, J., Yang, Z., Ma, Y., Yue, Z., Lin, H., Qu, G., Liu, M. (2016). LGR4 is a receptor for RANKL and negatively regulates osteoclast differentiation and bone resorption. *Nature Medicine*, *22*(5), 539–546.

Macedo, M. F., De Sousa, M., Ned, R. M., Mascarenhas, C., Andrews, N. C., & Correia-Neves,M. (2004). Transferrin is required for early T-cell differentiation. *Immunology*, *112*(4), 543–549.

MacGillivray, R. T. A., Moore, S. A., Chen, J., Anderson, B. F., Baker, H., Luo, Y., Baker, E. N. (1998). Two High-Resolution Crystal Structures of the Recombinant N-Lobe of Human Transferrin Reveal a Structural Change Implicated in Iron Release † , ‡. *Biochemistry*, *37*(22), 7919–7928.

MacNabb, C., Patton, D., & Hayes, J. S. (2016). Sclerostin Antibody Therapy for the Treatment of Osteoporosis: Clinical Prospects and Challenges. *Journal of Osteoporosis*, 2016, 1–22.

Majumder, K., Chakrabarti, S., Davidge, S. T., & Wu, J. (2013). Structure and Activity Study of Egg Protein Ovotransferrin Derived Peptides (IRW and IQW) on Endothelial Inflammatory Response and Oxidative Stress. *Journal of Agricultural and Food Chemistry*, *61*(9), 2120–2129.

Majumder, K., Liang, G., Chen, Y., Guan, L., Davidge, S. T., & Wu, J. (2015). Egg ovotransferrin-derived ACE inhibitory peptide IRW increases ACE2 but decreases proinflammatory genes expression in mesenteric artery of spontaneously hypertensive rats. *Molecular Nutrition & Food Research*, *59*(9), 1735–1744.

Malet, A., Bournaud, E., Lan, A., Mikogami, T., Tomé, D., & Blais, A. (2011). Bovine lactoferrin improves bone status of ovariectomized mice via immune function modulation. *Bone*, *48*(5), 1028–1035.

Marshall, K. (2004). Therapeutic applications of whey protein. *Alternative Medicine Review : A Journal of Clinical Therapeutic*, 9(2), 136–156.

Martins, A. M., Pham, Q. P., Malafaya, P. B., Raphael, R. M., Kasper, F. K., Reis, R. L., & Mikos, A. G. (2009). Natural Stimulus Responsive Scaffolds/Cells for Bone Tissue Engineering: Influence of Lysozyme upon Scaffold Degradation and Osteogenic Differentiation of Cultured Marrow Stromal Cells Induced by CaP Coatings. *Tissue Engineering Part A*, *15*(8), 1953–1963.

Masuda, H., Hirose, J., Omata, Y., Tokuyama, N., Yasui, T., Kadono, Y., Tanaka, S. (2014). Anti-apoptotic Bcl-2 family member Mcl-1 regulates cell viability and bone-resorbing activity of osteoclasts. *Bone*, *58*, 1–10.

Masuzawa, T., Miyaura, C., Onoe, Y., Kusano, K., Ohta, H., Nozawa, S., & Suda, T. (1994). Estrogen deficiency stimulates B lymphopoiesis in mouse bone marrow. *Journal of Clinical Investigation*, *94*(3), 1090–1097.

Matsuguchi, T., Chiba, N., Bandow, K., Kakimoto, K., Masuda, A., & Ohnishi, T. (2009a). JNK Activity Is Essential for *Atf4* Expression and Late-Stage Osteoblast Differentiation. *Journal of Bone and Mineral Research*, *24*(3), 398–410. Matsumoto, M., Sudo, T., Saito, T., Osada, H., & Tsujimoto, M. (2000). Involvement of p38 Mitogen-activated Protein Kinase Signaling Pathway in Osteoclastogenesis Mediated by Receptor Activator of NF-κB Ligand (RANKL). *Journal of Biological Chemistry*, *275*(40), 31155–31161.

Matsuo, K., & Irie, N. (2008). Osteoclast–osteoblast communication. *Archives of Biochemistry and Biophysics*, 473(2), 201–209.

Matsushita, T., Chan, Y. Y., Kawanami, A., Balmes, G., Landreth, G. E., & Murakami, S. (2009). Extracellular signal-regulated kinase 1 (ERK1) and ERK2 play essential roles in osteoblast differentiation and in supporting osteoclastogenesis. *Molecular and Cellular Biology*, *29*(21), 5843–5857.

Matthews, D. M., & Laster, L. (1965). Absorption of protein digestion products: a review. *Gut*, 6(5), 411–426.

Mazurier, J., Metz-Boutigue, M. H., Jollès, J., Spik, G., Montreuil, J., & Jollès, P. (1983). Human lactotransferrin: molecular, functional and evolutionary comparisons with human serum transferrin and hen ovotransferrin. *Experientia*, *39*(2), 135–141.

McConnell, D. (1964). Bone Mineral. Science, 145(3638), 1336–1336.

McConnell, D., Foreman, D. W., Drew, I., Perkins, D., & Daly, P. (1971). Texture and composition of bone. *Science (New York, N.Y.)*, *172*(3986), 971–973.

McGonnell, I. M., Grigoriadis, A. E., W-F Lam, E., Price, J. S., Sunters, A., & Gartland, A. (2012). A specific role for phosphoinositide 3-kinase and AKT in osteoblasts?, *719*(19), 28–1.

McMichael, B. K., Meyer, S. M., & Lee, B. S. (2010). c-Src-mediated Phosphorylation of Thyroid Hormone Receptor-interacting Protein 6 (TRIP6) Promotes Osteoclast Sealing Zone Formation. *Journal of Biological Chemistry*, *285*(34), 26641–26651.

Medscapte. (2017). FDA rejects Amgen's osteoporosis hope romosozumab. Retrieved September 19, 2018, from www.medscapte.com/view-artical/882966

Mee, A. P. (1999). Paramyxoviruses and Paget's disease: the affirmative view. *Bone*, 24(5 Suppl), 19S–21S.

Messina, M., & Messina, V. (2000). Soyfoods, soybean isoflavones, and bone health: a brief overview. *Journal of Renal Nutrition : The Official Journal of the Council on Renal Nutrition of the National Kidney Foundation*, 10(2), 63–68.

Meunier, P., Aaron, J., Edouard, C., & Vignon, G. (1971). Osteoporosis and the replacement of cell populations of the marrow by adipose tissue. A quantitative study of 84 iliac bone biopsies. *Clinical Orthopaedics and Related Research*, *80*, 147–154.

Miljuš, G., Malenković, V., Đukanović, B., Kolundžić, N., & Nedić, O. (2015). IGFBP-3/transferrin/transferrin receptor 1 complexes as principal mediators of IGFBP-3 delivery to colon cells in non-cancer and cancer tissues. *Experimental and Molecular Pathology*, *98*(3), 431–438.

Millán, J. L., & Whyte, M. P. (2016). Alkaline Phosphatase and Hypophosphatasia. *Calcified Tissue International*, *98*(4), 398–416.

Mine, Y. (2007). Egg proteins and peptides in human health--chemistry, bioactivity and production. *Current Pharmaceutical Design*, *13*(9), 875–884. Retrieved from http://www.ncbi.nlm.nih.gov/pubmed/17430187

Mine, Y., Li-Chan. E., & Jiang, B. (2010). *Bioactive Proteins and Peptides as Functional Foods and Nutraceuticals : Yoshinori Mine : 9780813813110*. (Y. Mine, Li-Chan. E., & B. Jiang, Eds.). Arnes, AI: Iowa State University Press.

Miranda, J. M., Anton, X., Redondo-Valbuena, C., Roca-Saavedra, P., Rodriguez, J. A., Lamas, A., Cepeda, A. (2015). Egg and egg-derived foods: effects on human health and use as functional foods. *Nutrients*, *7*(1), 706–729.

Mizunoya, W., Tashima, A., Sato, Y., Tatsumi, R., & Ikeuchi, Y. (2015). The growth-promoting activity of egg white proteins in the C2C12 myoblast cell line. *Animal Science Journal*, *86*(2), 194–199.

Moe, S. M., & Chen, N. X. (2008). Mechanisms of vascular calcification in chronic kidney disease. *Journal of the American Society of Nephrology : JASN*, *19*(2), 213–216.

Moerman, E. J., Teng, K., Lipschitz, D. A., & Lecka-Czernik, B. (2004). Aging activates adipogenic and suppresses osteogenic programs in mesenchymal marrow stroma/stem cells: the role of PPAR- γ 2 transcription factor and TGF- β /BMP signaling pathways. *Aging Cell*, *3*(6), 379–389.

Mohan, A., Rajendran, S. R. C. K., He, Q. S., Bazinet, L., & Udenigwe, C. C. (2015). Encapsulation of food protein hydrolysates and peptides: a review. *RSC Advances*, 5(97), 79270–79278. https://doi.org/10.1039/C5RA13419F

Mollazadeh, S., Fazly Bazzaz, B., & Kerachian, M. (2015). Role of apoptosis in pathogenesis and treatment of bone-related diseases. *Journal of Orthopaedic Surgery and Research*, *10*(1), 15.

Morgan, E. F., Barnes, G. L., & Einhorn, T. A. (2008). The bone organ system: form and function. In R. Marcus, D. Feldman, D. W. Dempster, M. Luckey, & J. A. Cauley (Eds.), *Osteoporosis* (3rd ed., pp. 3–26). San Diego: Elsevier Academic Press.

Morley, P., Whitfield, J. F., & Willick, G. E. (2001). Parathyroid hormone: an anabolic treatment for osteoporosis. *Current Pharmaceutical Design*, *7*(8), 671–687.

Mullard, A. (2016). Merck & amp; Co. drops osteoporosis drug odanacatib. *Nature Reviews Drug Discovery*, *15*(10), 669–669.

Murray, B., & FitzGerald, R. (2007). Angiotensin Converting Enzyme Inhibitory Peptides Derived from Food Proteins: Biochemistry, Bioactivity and Production. *Current Pharmaceutical Design*, *13*(8), 773–791.

Murray, M. J., Meyer, W. R., Lessey, B. A., Oi, R. H., DeWire, R. E., & Fritz, M. A. (2003). Soy protein isolate with isoflavones does not prevent estradiol-induced endometrial hyperplasia in postmenopausal women: a pilot trial. *Menopause*, *10*(5), 456–464.

Naor, Z., Benard, O., & Seger, R. (2000). Activation of MAPK cascades by G-protein-coupled receptors: the case of gonadotropin-releasing hormone receptor. *Trends in Endocrinology and Metabolism: TEM*, *11*(3), 91–99.

Naot, D., Chhana, A., Matthews, B. G., Callon, K. E., Tong, P. C., Lin, J. M., Cornish, J. (2011). Molecular mechanisms involved in the mitogenic effect of lactoferrin in osteoblasts. *Bone*, *49*(2), 217–224.

Naot, D., Grey, A., Reid, I. R., & Cornish, J. (2005). Lactoferrin-a novel bone growth factor. *Clinical Medicine & Research.*, *3*(2), 93–101. https://doi.org/3/2/93 [pii]

Negishi-Koga, T., Shinohara, M., Komatsu, N., Bito, H., Kodama, T., Friedel, R. H., & Takayanagi, H. (2011). Suppression of bone formation by osteoclastic expression of semaphorin 4D. *Nature Medicine*, *17*(11), 1473–1480.

Nieves, J. W. (2005). Osteoporosis: the role of micronutrients. *American Journal of Clinical Nutrition*, *81*(5), 12328–12398.

NIH Consensus Development Panel on Osteoporosis Prevention, Diagnosis, and Therapy. (2001). Osteoporosis prevention, diagnosis, and therapy. *JAMA*, *285*(6), 785–795.

Ohshiba, T., Miyaura, C., Inada, M., & Ito, A. (2003). Role of RANKL-induced osteoclast formation and MMP-dependent matrix degradation in bone destruction by breast cancer metastasis. *British Journal of Cancer*, *88*(8), 1318–1326.

Omi, N., Aoi, S., Murata, K., & Ezawa, I. (1994). Evaluation of the effect of soybean milk and soybean milk peptide on bone metabolism in the rat model with ovariectomized osteoporosis. *Journal of Nutritional Science and Vitaminology*, *40*(2), 201–211.

Onoe, Y., Miyaura, C., Ito, M., Ohta, H., Nozawa, S., & Suda, T. (2010). Comparative Effects of Estrogen and Raloxifene on B Lymphopoiesis and Bone Loss Induced by Sex Steroid Deficiency in Mice. *Journal of Bone and Mineral Research*, *15*(3), 541–549.

Otani, H., & Odashima, M. (1997). Inhibition of proliferative responses of mouse spleen lymphocytes by lacto - and ovotransferrins. *Food and Agricultural Immunology*, 9(3), 193–201. Ou-Yang, H., Paschalis, E. P., Mayo, W. E., Boskey, A. L., & Mendelsohn, R. (2001). Infrared Microscopic Imaging of Bone: Spatial Distribution of CO32–. *Journal of Bone and Mineral Research*, *16*(5), 893–900.

Owen, T. A., Aronow, M., Shalhoub, V., Barone, L. M., Wilming, L., Tassinari, M. S., Stein, G. S. (1990). Progressive development of the rat osteoblast phenotype in vitro - reciprocal relationships in expression of genes associated with osteoblast proliferation and differentiation during formation of the bone extracellular-matrix. *Journal of Cellular Physiology*, *143*(3), 420–430.

Paez, P. M., García, C. I., Campagnoni, A. T., Soto, E. F., & Pasquini, J. M. (2005). Overexpression of human transferrin in two oligodendroglial cell lines enhances their differentiation. *Glia*, 52(1), 1–15.

Pang, T., Gong, M., Han, J., & Liu, D. (2018). Relationship between osteoporosis and expression of Bcl-2 and CXCL12. *Experimental and Therapeutic Medicine*, *15*(2), 1293–1297.

Panupinthu, N., Rogers, J. T., Zhao, L., Solano-Flores, L. P., Possmayer, F., Sims, S. M., & Dixon, S. J. (2008). P2X7 receptors on osteoblasts couple to production of lysophosphatidic acid: a signaling axis promoting osteogenesis. *The Journal of Cell Biology*, *181*(5), 859–871.

Parfitt, A. M. (2008). Osteoporosis. In R. Marcus, D. Feldman, D. A. Nelson, & C. J. Rosen (Eds.), *Osteoporosis* (pp. 71–92). San Diego: Academic Press.

Peng, X. D., Xu, P. Z., Chen, M. L., Hahn-Windgassen, A., Skeen, J., Jacobs, J., Hay, N. (2003). Dwarfism, impaired skin development, skeletal muscle atrophy, delayed bone development, and impeded adipogenesis in mice lacking Akt1 and Akt2. *Genes & Development*, *17*(11), 1352–1365.

Perazella, M. A., & Markowitz, G. S. (2008). Bisphosphonate nephrotoxicity. *Kidney International*, *74*(11), 1385–1393.

Picherit, C., Bennetau-Pelissero, C., Chanteranne, B., Lebecque, P., Davicco, M. J., Barlet, J. P.,& Coxam, V. (2001). Soybean Isoflavones Dose-Dependently Reduce Bone Turnover but Do

Not Reverse Established Osteopenia in Adult Ovariectomized Rats. *The Journal of Nutrition*, 131(3), 723–728.

Picherit, C., Chanteranne, B., Bennetau-Pelissero, C., Davicco, M. J., Lebecque, P., Barlet, J. P.,
& Coxam, V. (2001). Dose-dependent bone-sparing effects of dietary isoflavones in the ovariectomised rat. *The British Journal of Nutrition*, *85*(3), 307–316.

Pino, A. M., Miranda, M., Figueroa, C., Rodríguez, J. P., & Rosen, C. J. (2016). Qualitative Aspects of Bone Marrow Adiposity in Osteoporosis. *Frontiers in Endocrinology*, *7*, 139.

Podgorski, I., Linebaugh, B. E., Koblinski, J. E., Rudy, D. L., Herroon, M. K., Olive, M. B., & Sloane, B. F. (2009). Bone marrow-derived cathepsin K cleaves SPARC in bone metastasis. *The American Journal of Pathology*, *175*(3), 1255–1269.

Ponnapakkam, T., Katikaneni, R., Sakon, J., Stratford, R., & Gensure, R. C. (2014). Treating osteoporosis by targeting parathyroid hormone to bone. *Drug Discovery Today*, *19*(3), 204–208.

Qi, H., Aguiar, D. J., Williams, S. M., Pean, A. La, Pan, W., & Verfaillie, C. M. (n.d.). *Identification of genes responsible for osteoblast differentiation from human mesodermal progenitor cells.*

Qiu, W., Chen, L., & Kassem, M. (2011). Activation of non-canonical Wnt/JNK pathway by Wnt3a is associated with differentiation fate determination of human bone marrow stromal (mesenchymal) stem cells. *Biochemical and Biophysical Research Communications*, *413*(1), 98–104.

Rahman, M. M., Bhattacharya, A., Banu, J., Kang, J. X., & Fernandes, G. (2009). Endogenous n-3 fatty acids protect ovariectomy induced bone loss by attenuating osteoclastogenesis. *Journal of Cellular and Molecular Medicine*, *13*(8B), 1833–1844.

Ramos, J. W. (2008). The regulation of extracellular signal-regulated kinase (ERK) in mammalian cells. *The International Journal of Biochemistry & Cell Biology*, 40(12), 2707–2719.

Rath, N. C., Anthony, N. B., Kannan, L., Huff, W. E., Huff, G. R., Chapman, H. D., Wakenell, P. (2009). Serum ovotransferrin as a biomarker of inflammatory diseases in chickens. *Poultry Science*, *88*(10), 2069–2074.

Redlich, K., & Smolen, J. S. (2012). Inflammatory bone loss: pathogenesis and therapeutic intervention. *Nature Reviews Drug Discovery*, *11*(3), 234–250.

Redoglia, V., Boccadoro, M., Battaglio, S., Dianzani, U., Massaia, M., & Pileri, A. (1990). Multiple myeloma: altered CD4/CD8 ratio in bone marrow. *Haematologica*, *75*(2), 129–131.

Reginster, J. Y., Sarlet, N., Lejeune, E., & Leonori, L. (2005). Strontium ranelate: a new treatment for postmenopausal osteoporosis with a dual mode of action. *Current Osteoporosis Reports*, *3*(1), 30–34.

Reinholz, G. G., Getz, B., Pederson, L., Sanders, E. S., Subramaniam, M., Ingle, J. N., & Spelsberg, T. C. (2000). Bisphosphonates directly regulate cell proliferation, differentiation, and gene expression in human osteoblasts. *Cancer Research*, *60*(21), 6001–6007.

Reponen, P., Sahlberg, C., Munaut, C., Thesleff, I., & Tryggvason, K. (1994). High expression of 92-kD type IV collagenase (gelatinase B) in the osteoclast lineage during mouse development. *The Journal of Cell Biology*, *124*(6), 1091–1102.

Rey, A., Manen, D., Rizzoli, R., Ferrari, S. L., & Caverzasio, J. (2007). Evidences for a role of p38 MAP kinase in the stimulation of alkaline phosphatase and matrix mineralization induced by parathyroid hormone in osteoblastic cells. *Bone*, *41*(1), 59–67.

Richie, E. R., Hilliard, J. K., Gilmore, R., & Gillespie, D. J. (1987). Human milk-derived lactoferrin inhibits mitogen and alloantigen induced human lymphocyte proliferation. *Journal of Reproductive Immunology*, *12*(2), 137–148.

Richon, V. M., Lyle, R. E., & McGehee, R. E. (1997). Regulation and expression of retinoblastoma proteins p107 and p130 during 3T3-L1 adipocyte differentiation. *Journal of Biological Chemistry*, *272*(15), 10117–10124.
Riggs, B. L. (2000). The mechanisms of estrogen regulation of bone resorption. *The Journal of Clinical Investigation*, *106*(10), 1203–1204.

Ríos-Covián, D., Ruas-Madiedo, P., Margolles, A., Gueimonde, M., de Los Reyes-Gavilán, C.G., & Salazar, N. (2016). Intestinal Short Chain Fatty Acids and their Link with Diet and Human Health. *Frontiers in Microbiology*, *7*, 185.

Rodrigues, T. A., Freire, A. O., Bonfim, B. F., Cartágenes, M. S. S., Garcia, J. B. S., Rodrigues, T. A., Garcia, J. B. S. (2018). Strontium ranelate as a possible disease-modifying osteoarthritis drug: a systematic review. *Brazilian Journal of Medical and Biological Research*, *51*(8).

Rodríguez-Carballo, E., Gámez, B., Sedó-Cabezón, L., Sánchez-Feutrie, M., Zorzano, A., Manzanares-Céspedes, C., Ventura, F. (2014). The p38α MAPK Function in Osteoprecursors Is Required for Bone Formation and Bone Homeostasis in Adult Mice. *PLoS ONE*, *9*(7), e102032.

Rosen, C. J. (2004). What's new with PTH in osteoporosis: where are we and where are we headed? *Trends in Endocrinology & Metabolism*, *15*(5), 229–233.

Rosen, C. J. (2013). Primer on the metabolic bone diseases and disorders of mineral metabolism. (C. J. Rosen, R. Bouillon, J. Compston, & V. Rosen, Eds.) (8th ed.). Ames, Iowa: Wiley-Blackwell.

Rosen, C. J., Donahue, L. R., & Hunter, S. J. (1994). Insulin-like growth factors and bone: the osteoporosis connection. *Proceedings of the Society for Experimental Biology and Medicine*. *Society for Experimental Biology and Medicine (New York, N.Y.)*, 206(2), 83–102.

Roux, S., Lambert-Comeau, P., Saint-Pierre, C., Lépine, M., Sawan, B., & Parent, J. L. (2005).
Death receptors, Fas and TRAIL receptors, are involved in human osteoclast apoptosis. *Biochemical and Biophysical Research Communications*, 333(1), 42–50.

Rucci, N. (2008). Molecular biology of bone remodelling. *Clinical Cases in Mineral and Bone Metabolism : The Official Journal of the Italian Society of Osteoporosis, Mineral Metabolism, and Skeletal Diseases, 5,* 49–56. Rutkovskiy, A., Stensløkken, K.-O., & Vaage, I. J. (2016). Osteoblast Differentiation at a Glance. *Medical Science Monitor Basic Research*, *22*, 95–106.

Safadi, F. F., Dissanayake, I. R., Goodman, G. G., Jago, R. A., Baker, A. E., Bowman, A. R., Epstein, S. (2000). Influence of Estrogen Deficiency and Replacement on T-Cell Populations in Rat Lymphoid Tissues and Organs. *Endocrine*, *12*(1), 81–88.

Saftig, P., Hunziker, E., Everts, V., Jones, S., Boyde, A., Wehmeyer, O., von Figura, K. (2002). Functions of Cathepsin K in Bone Resorption (pp. 293–303). Springer, Boston, MA.

Sakamoto, S., Sakamoto, M., Goldhaber, P., & Glimcher, M. J. (1974). The inhibition of mouse bone collagenase by lysozyme. *Calcified Tissue Research*, *14*(1), 291–299.

Salari Sharif, P., Abdollahi, M., & Larijani, B. (2011). Current, new and future treatments of osteoporosis. *Rheumatology International*, *31*(3), 289–300.

Sambandam, Y., Blanchard, J. J., Daughtridge, G., Kolb, R. J., Shanmugarajan, S., Pandruvada,
S. N. M., Reddy, S. V. (2010). Microarray Profile of Gene Expression During Osteoclast
Differentiation in Modelled Microgravity. *Journal of Cellular Biochemistry*, *111*(5), 1179–1187.

San Martin, I. A., Varela, N., Gaete, M., Villegas, K., Osorio, M., Tapia, J. C., Galindo, M. (2009). Impaired cell cycle regulation of the osteoblast-related heterodimeric transcription factor Runx2-Cbfβ in osteosarcoma cells. *Journal of Cellular Physiology*, *221*(3), 560–571.

Seol, K., Lim. D., Jang, A., Jo, C., & Lee, M. (2009). Antimicrobial effect of j-carrageenanbased edible film containing ovotransferrin in fresh chicken breast stored at 5 C. *Meat Science*, *83*, 479–483.

Schaffler, M. B., Cheung, W. Y., Majeska, R., Kennedy, O. (2014). Osteocytes: master orchestrators of bone. Calcif. Tissue. Int., 94(1): 5-24.

Shah, R. B., & Khan, M. A. (2004). Protection of salmon calcitonin breakdown with serine proteases by various ovomucoid species for oral drug delivery. *Journal of Pharmaceutical Sciences*, *93*(2), 392–406.

Shahi, M., Peymani, A., & Sahmani, M. (2017). Regulation of Bone Metabolism. *Reports of Biochemistry & Molecular Biology*, 5(2), 73–82.

Shams-White, M. M., Chung, M., Du, M., Fu, Z., Insogna, K. L., Karlsen, M. C., Weaver, C. M. (2017). Dietary protein and bone health: a systematic review and meta-analysis from the National Osteoporosis Foundation. *The American Journal of Clinical Nutrition*, *105*(6), ajcn145110.

Shane, E., Burr, D., Abrahamsen, B., Adler, R. A., Brown, T. D., Cheung, A. M., Whyte, M. P. (2014). Atypical Subtrochanteric and Diaphyseal Femoral Fractures: Second Report of a Task Force of the American Society for Bone and Mineral Research. *Journal of Bone and Mineral Research*, *29*(1), 1–23.

Shang, N., & Wu, J. (2018). Egg White Ovotransferrin Shows Osteogenic Activity in Osteoblast Cells. *Journal of Agricutrual Food and Chemistry*, *66*, 2775-2782.

Sharff, K. A., Song, W. X., Luo, X., Tang, N., Luo, J., Chen, J., He, T. C. (2009). Heyl Basic Helix-Loop-Helix Protein Plays an Important Role in Mediating BMP9-induced Osteogenic Differentiation of Mesenchymal Progenitor Cells. *Journal of Biological Chemistry*, 284(1), 649– 659.

Shen, S., Chahal, B., Majumder, K., You, S. J., & Wu, J. (2010). Identification of Novel Antioxidative Peptides Derived from a Thermolytic Hydrolysate of Ovotransferrin by LC-MS/MS. *Journal of Agricultural and Food Chemistry*, *58*(13), 7664–7672.

Sheng, M. H., Lau, K. H., Baylink, D. J. (2014). Role of osteocyte-derived insulin-like growth factor I in developmental growth, modeling, remodeling, and regeneration of the bone. J. Bone Metab. 21(1), 41-54.

Shimo-Oka, T., Hagiwara, Y., & Ozawa, E. (1986). Class specificity of transferrin as a muscle trophic factor. *Journal of Cellular Physiology*, *126*(3), 341–351.

Shimo, T., Matsumura, S., Ibaragi, S., Isowa, S., Kishimoto, K., Mese, H., Sasaki, A. (2007). Specific inhibitor of MEK-mediated cross-talk between ERK and p38 MAPK during differentiation of human osteosarcoma cells. *Journal of Cell Communication and Signaling*, *1*(2), 103–111.

Shin, K., Wakabayashi, H., Yamauchi, K., Yaeshima, T., & Iwatsuki, K. (2008). Recombinant human intelectin binds bovine lactoferrin and its peptides. *Biological & Pharmaceutical Bulletin*, *31*(8), 1605–1608. Retrieved from

Siciliano, R., Rega, B., Marchetti, M., Seganti, L., Antonini, G., & Valenti, P. (1999). Bovine Lactoferrin Peptidic Fragments Involved in Inhibition of Herpes Simplex Virus Type 1 Infection. *Biochemical and Biophysical Research Communications*, *264*(1), 19–23.

Silverman, S. L., & Madison, R. E. (1988). Decreased incidence of hip fracture in Hispanics, Asians, and blacks: California Hospital Discharge Data. *American Journal of Public Health*, 78(11), 1482–1483.

Silvestroff, L., Franco, P. G., & Pasquini, J. M. (2013). Neural and Oligodendrocyte Progenitor Cells: Transferrin Effects on Cell Proliferation. *ASN Neuro*, *5*(1), AN20120075.

Sims, N. A., Morris, H. A., Moore, R. J., & Durbridge, T. C. (1996). Increased bone resorption precedes increased bone formation in the ovariectomized rat. *Calcified Tissue International*, *59*(2), 121–127.

Sims, S. M., Panupinthu, N., Lapierre, D. M., Pereverzev, A., & Dixon, S. J. (2013).
Lysophosphatidic acid: A potential mediator of osteoblast–osteoclast signaling in bone. *Biochimica et Biophysica Acta (BBA) - Molecular and Cell Biology of Lipids*, 1831(1), 109–116.

Smith, E., Coetzee, G. A., & Frenkel, B. (2002). Glucocorticoids inhibit cell cycle progression in differentiating osteoblasts via glycogen synthase kinase-3β. *Journal of Biological Chemistry*, *277*(20), 18191–18197.

Smith, E., Redman, R. A., Logg, C. R., Coetzee, G. A., Kasahara, N., & Frenkel, B. (2000).
Glucocorticoids inhibit developmental stage-specific osteoblast cell cycle: Dissociation of cyclin
A-cyclin-dependent kinase 2 from E2F4-p130 complexes. *Journal of Biological Chemistry*, 275(26), 19992–20001.

Stevenson, M., Davis, S., Lloyd-Jones, M., & Beverley, C. (2007). The clinical effectiveness and cost-effectiveness of strontium ranelate for the prevention of osteoporotic fragility fractures in postmenopausal women. *Health Technology Assessment (Winchester, England)*, *11*(4), 1–134.

Strahilevitz, J., Mor, A., Nicolas, P., & Shai, Y. (1994). Spectrum of antimicrobial activity and assembly of dermaseptin-b and its precursor form in phospholipid membranes. *Biochemistry*, *33*(36), 10951–10960.

Sundaram, K., Nishimura, R., Senn, J., Youssef, R. F., London, S. D., & Reddy, S. V. (2007). RANK ligand signaling modulates the matrix metalloproteinase-9 gene expression during osteoclast differentiation. *Experimental Cell Research*, *313*(1), 168–178.

Superti, F., Ammendolia, M. G., Berlutti, F., & Valenti, P. (2007). Chapter 7 Ovotransferrin. In R. Huopalahti, R. López-Fandiño, M. Anton, & R. Schade (Eds.), *Bioactive Egg Compounds* (pp. 43–48). Berlin Heidelberg: Springer.

Suzuki, A., Palmer, G., Bonjour, J. P., & Caverzasio, J. (1999). Regulation of Alkaline Phosphatase Activity by p38 MAP Kinase in Response to Activation of Gi Protein-Coupled Receptors by Epinephrine in Osteoblast-Like Cells. *Endocrinology*, *140*(7), 3177–3182.

Suzuki, E., Ochiai-Shino, H., Aoki, H., Onodera, S., Saito, A., Saito, A., & Azuma, T. (2014). Akt Activation is Required for TGF-β1-Induced Osteoblast Differentiation of MC3T3-E1 Pre-Osteoblasts. *PLoS ONE*, *9*(12), e112566.

Suzuki, Y. A., Lopez, V., & Lönnerdal, B. (2005). Mammalian lactoferrin receptors: structure and function. *Cellular and Molecular Life Sciences*, *62*(22), 2560–2575.

Swarthout, J. T., Doggett, T. A., Lemker, J. L., & Partridge, N. C. (2001). Stimulation of extracellular signal-regulated kinases and proliferation in rat osteoblastic cells by parathyroid hormone is protein kinase C-dependent. *The Journal of Biological Chemistry*, *276*(10), 7586–7592.

Takada, I., Kouzmenko, A. P., & Kato, S. (2009). Wnt and PPARγ signaling in osteoblastogenesis and adipogenesis. *Nature Reviews Rheumatology*, *5*(8), 442–447.

Takatsuna, H., Asagiri, M., Kubota, T., Oka, K., Osada, T., Sugiyama, C., Umezawa, K. (2004). Inhibition of RANKL-Induced Osteoclastogenesis by (–)-DHMEQ, a Novel NF-κB Inhibitor, Through Downregulation of NFATc1. *Journal of Bone and Mineral Research*, *20*(4), 653–662.

Takeuchi, T., Kitagawa, H., & Harada, E. (2004). Evidence of lactoferrin transportation into blood circulation from intestine via lymphatic pathway in adult rats. *Experimental Physiology*, *89*(3), 263–270.

Tanaka, S., Miyazaki, T., Fukuda, A., Akiyama, T., Kadono, Y., Wakeyama, H., Nakamura, K. (2006). Molecular Mechanism of the Life and Death of the Osteoclast. *Annals of the New York Academy of Sciences*, *1068*(1), 180–186.

Taylor, C. L., Yaktine, A. L., & Del Balle, H. B. (2011). *Dietary Reference Intakes for Calcium and Vitamin D*. Washington, D.C.: National Academies Press.

Teitelbaum, S. L. (2000). Bone resorption by osteoclasts. *Science (New York, N.Y.)*, 289(5484), 1504–1508.

Teitelbaum, S. L. (2004). RANKing c-Jun in osteoclast development. *The Journal of Clinical Investigation*, *114*(4), 463–465.

Teitelbaum, S. L., & Ross, F. P. (2003). Genetic regulation of osteoclast development and function. *Nature Reviews Genetics*, *4*(8), 638–649.

Teplyuk, N. M., Galindo, M., Teplyuk, V. I., Pratap, J., Young, D. W., Lapointe, D., van Wijnen, A. J. (2008). Runx2 regulates G protein-coupled signaling pathways to control growth of osteoblast progenitors. *The Journal of Biological Chemistry*, *283*(41), 27585–27597.

Thibodeau, S. N., Lee, D. C., & Palmiter, R. D. (1978). Identical precursors for serum transferrin and egg white conalbumin. *The Journal of Biological Chemistry*, *253*(11), 3771–3774.

Thouverey, C., & Caverzasio, J. (2012). The p38α MAPK positively regulates osteoblast function and postnatal bone acquisition. *Cellular and Molecular Life Sciences*, *69*(18), 3115–3125.

Thummuri, D., Jeengar, M. K., Shrivastava, S., Nemani, H., Ramavat, R. N., Chaudhari, P., & Naidu, V. G. M. (2015). Thymoquinone prevents RANKL-induced osteoclastogenesis activation and osteolysis in an in vivo model of inflammation by suppressing NF-KB and MAPK Signalling. *Pharmacological Research*, *99*, 63–73.

Thummuri, D., Naidu, V. G. M., & Chaudhari, P. (2017). Carnosic acid attenuates RANKLinduced oxidative stress and osteoclastogenesis via induction of Nrf2 and suppression of NF-κB and MAPK signalling. *Journal of Molecular Medicine*, *95*(10), 1065–1076.

Toba, Y., Takada, Y., Yamamura, J., Tanaka, M., Matsuoka, Y., Kawakami, H., Kumegawa, M. (2000). Milk basic protein: a novel protective function of milk against osteoporosis. *Bone*, *27*(3), 403–408.

Tohjo, H., Miyoshi, F., Uchida, E., Niyama, M., Syuto, B., Moritsu, Y., Takeuchi, M. (1995). Polyacrylamide Gel Electrophoretic Patterns of Chicken Serum in Acute Inflammation Induced by Intramuscular Injection of Turpentine. *Poultry Science*, *74*(4), 648–655.

Tosteson, A. N. A., Burge, R. T., Marshall, D. A., & Lindsay, R. (2008). Therapies for treatment of osteoporosis in US women: cost-effectiveness and budget impact considerations. *The American Journal of Managed Care*, *14*(9), 605–615.

Tranter, H. S., & Board, R. G. (1982). The antimicrobial defense of avian eggs: biological perspective and chemical basis [Poultry]. *Journal of Applied Biochemistry*.

Tsai, C. H., Chen, Y. J., Huang, F. M., Su, Y. F., & Chang, Y. C. (2005). The Upregulation of Matrix Metalloproteinase-9 in Inflamed Human Dental Pulps. *Journal of Endodontics*, *31*(12), 860–862.

Tsunoi, M., Hakeda, Y., Kurihara, N., Maeda, N., Utsum, N., & Kumegawa, M. (1984). Effect of transferrin on alkaline phosphatase activity and collagen synthesis in osteoblastic cell derived from newborn mouse calvaria. *Experimental Cell Research*, *153*, 240–244.

Tu, K. N., Lie, J. D., Wan, C. K. V., Cameron, M., Austel, A. G., Nguyen, J. K., Hyun, D. (2018). Osteoporosis: A Review of Treatment Options. *P & T : A Peer-Reviewed Journal for Formulary Management*, *43*(2), 92–104.

Tulipano, G., Bulgari, O., Chessa, S., Nardone, A., Cocchi, D., & Caroli, A. (2010). Direct effects of casein phosphopeptides on growth and differentiation of in vitro cultured osteoblastic cells (MC3T3-E1). *Regulatory Peptides*, *160*(1–3), 168–174.

Uenishi, K., Ishida, H., Toba, Y., Aoe, S., Itabashi, A., & Takada, Y. (2007). Milk basic protein increases bone mineral density and improves bone metabolism in healthy young women. *Osteoporosis International*, *18*(3), 385–390.

US Department of Health and Human Services. (2004). Bone health and osteoporosis: a report of the Surgeon General. In *US Health and Human Services*.

Vaidya, R. (2014). Obesity, sarcopenia and postmenopausal osteoporosis: An interlinked triad! *Journal of Mid-Life Health*, *5*(1), 1–2.

Valenti, P., Antonini, G., Fanelli, M. R., Orsi, N., & Antonini, E. (1982). Antibacterial activity of matrix-bound ovotransferrin. *Antimicrobial Agents and Chemotherapy*, *21*(5), 840–841.

Valenti, P., Antonini, G., Von Hunolstein, C., Visca, P., Orsi, N., & Antonini, E. (1983). Studies of the antimicrobial activity of ovotransferrin. *International Journal of Tissue Reactions*, *5*(1), 97–105.

Valenti, P., Stasio, A., Mastromerino, P., Seganti, L., Sinibaldi, L., & Orsi, N. (1981). Influence of bicarbonate and citrate on the bacteriostatic action of ovotransferrin towards staphylococci. *FEMS Microbiology Letters*, *10*(1), 77–79.

Valenti, P., Visca, P., Antonini, G., Mycopathologia, N. O.-, & 1985, undefined. (1985). Antifungal activity of ovotransferrin towards genus Candida. *Mycopathologia*, 89(3), 169–175.

Valenti, P., Visca, P., Antonini, G., Orsi, N., & Antonini, E. (1987). The effect of saturation with Zn2+ and other metal ions on the antibacterial activity of ovotransferrin. *Medical Microbiology and Immunology*, *176*(3), 123–130.

Van Droogenbroeck, C., Dossche, L., Wauman, T., Van Lent, S., Phan, T. T. T., Beeckman, D.S. A., & Vanrompay, D. (2011). Use of ovotransferrin as an antimicrobial in turkeys naturally

infected with Chlamydia psittaci, avian metapneumovirus and Ornithobacterium rhinotracheale. *Veterinary Microbiology*, *153*(3–4), 257–263.

Vliegenthart, J. F. G., Dorland, L., Haverkamp, J., Spik, G., Fournet, B., & Montreuil, J. (1979). Investigation by 360-MHz 1H-nuclear-magnetic-resonance spectroscopy and methylation analysis of the single glycan chain of chicken ovotransferrin. *European Journal of Biochemistry*, *100*(2), 569–574.

Vupadhyayula, P. M., Gallagher, J. C., Templin, T., Logsdon, S. M., & Smith, L. M. (2009). Effects of soy protein isolate on bone mineral density and physical performance indices in postmenopausal women-a 2-year randomized, double-blind, placebo-controlled trial. *Menopause*, *16*(2), 320–328.

Waalen, J. (2010). Current and emerging therapies for the treatment of osteoporosis. *Journal of Experimental Pharmacology*, 121.

Walsh, M. C., Kim, N., Kadono, Y., Rho, J., Lee, S. Y., Lorenzo, J., & Choi, Y. (2006). Osteoimmunology: interplay between the immune system and bone metabolism. *Annual Review of Immunology*, *24*(1), 33–63.

Wang, L., Liu, S., Zhao, Y., Liu, D., Liu, Y., Chen, C., Jin, Y. (2015). Osteoblast-induced osteoclast apoptosis by fas ligand/FAS pathway is required for maintenance of bone mass. *Cell Death & Differentiation*, 22(10), 1654–1664.

Wang, X., Zhao, Y., Yao, Y., Xu, M., Du, H., Zhang, M., & Tu, Y. (2017). Anti-inflammatory activity of di-peptides derived from ovotransferrin by simulated peptide-cut in TNF-α-induced Caco-2 cells. *Journal of Functional Foods*, *37*, 424–432.

Watts, N. B., Adler, R. A., Bilezikian, J. P., Drake, M. T., Eastell, R., Orwoll, E. S., Endocrine Society. (2012). Osteoporosis in Men: An Endocrine Society Clinical Practice Guideline. *The Journal of Clinical Endocrinology & Metabolism*, 97(6), 1802–1822.

Weaver, C. M. (2015). Diet, gut microbiome, and bone health. *Current Osteoporosis Reports*, 13(2), 125–130.

Wei, C. M., Liu, Q., Song, F. M., Lin, X. X., Su, Y. J., Xu, J., Zhao, J. M. (2018). Artesunate inhibits RANKL-induced osteoclastogenesis and bone resorption in vitro and prevents LPS-induced bone loss in vivo. *Journal of Cellular Physiology*, *233*(1), 476–485.

Weinzimer, S. A., Gibson, T. B., Collett-Solberg, P. F., Khare, A., Liu, B., & Cohen, P. (2001). Transferrin Is an Insulin-Like Growth Factor-Binding Protein-3 Binding Protein1. *The Journal of Clinical Endocrinology & Metabolism*, *86*(4), 1806–1813.

Wellman-Labadie, O., Picman, J., & Hincke, M. T. (2008). Comparative antibacterial activity of avian egg white protein extracts. *British Poultry Science*, *49*(2), 125–132.

WHO. (2003). Prevention and management of osteoporosis. World Health Organization technical report series (Vol. 921). Retrieved from http://www.ncbi.nlm.nih.gov/pubmed/15293701

Widler, L. (2011). Calcilytics: antagonists of the calcium-sensing receptor for the treatment of osteoporosis. *Future Medicinal Chemistry*, *3*(5), 535–547.

Wiktor-Jedrzejczak, W., Bartocci, A., Ferrante, A. W., Ahmed-Ansari, A., Sell, K. W., Pollard, J. W., & Stanley, E. R. (1990). Total absence of colony-stimulating factor 1 in the macrophagedeficient osteopetrotic (op/op) mouse. *Proceedings of the National Academy of Sciences of the United States of America*, 87(12), 4828–4832.

Williams, J. (1982). The evolution of transferrin. *Trends in Biochemical Sciences*, 7(11), 394–397.

Williams. J. (1968). A comparison of glycopeptides from the ovotransferrin and serum transferrin of the hen. *Biochemical Journal*, *108*(1), 57–67.

Wilsgaard, T., Emaus, N., Ahmed, L. A., Grimnes, G., Joakimsen, R. M., Omsland, T. K., & Berntsen, G. R. (2009). Lifestyle impact on lifetime bone loss in women and men. *American Journal of Epidemiology*, *169*(7), 877–886.

Wilting, I., de Vries, F., Thio, B. M. K. S., Cooper, C., Heerdink, E. R., Leufkens, H. G. M., van Staa, T. P. (2007). Lithium use and the risk of fractures. *Bone*, *40*(5), 1252–1258.

Winkler, D. G., Sutherland, M. K., Geoghegan, J. C., Yu, C., Hayes, T., Skonier, J. E., Latham, J. A. (2003). Osteocyte control of bone formation via sclerostin, a novel BMP antagonist. *The EMBO Journal*, *22*(23), 6267–6276.

Woo, K. H., Kim, H. M., & Ko, J. S. (2002). Macrophage colony-stimulating factor promortes the survival of osteoclast precursors by upregulating Bcl-X(L). *Experimental & Molecular Medicine*, *34*(5), 340–346.

Wortzel, I., & Seger, R. (2011). The ERK Cascade: Distinct Functions within Various Subcellular Organelles. *Genes & Cancer*, 2(3), 195–209.

Wozney, J. M., Rosen, V., Celeste, A. J., Mitsock, L. M., Whitters, M. J., Kriz, R. W., ... Wang, E. A. (1988). Novel regulators of bone formation: molecular clones and activities. *Science (New York, N.Y.), 242*(4885), 1528–1534.

Wu, J., & Acero-Lopez, A. (2012). Ovotransferrin: Structure, bioactivities, and preparation. *Food Research International*, *46*(2), 480–487.

Wu, Q., Zhou, X., Huang, D., Ji, Y., & Kang, F. (2017). IL-6 Enhances Osteocyte-Mediated Osteoclastogenesis by Promoting JAK2 and RANKL Activity In Vitro. *Cellular Physiology and Biochemistry : International Journal of Experimental Cellular Physiology, Biochemistry, and Pharmacology, 41*(4), 1360–1369.

Wucherpfennig, A. L., Li, Y. P., Stetler-Stevenson, W. G., Rosenberg, A. E., & Stashenko, P. (2009). Expression of 92 kD type IV collagenase/gelatinase B in human osteoclasts. *Journal of Bone and Mineral Research*, *9*(4), 549–556.

Xi, J. C., Zang, H. Y., Guo, L. X., Xue, H. B., Liu, X. D., Bai, Y. B., & Ma, Y. Z. (2015). The PI3K/AKT cell signaling pathway is involved in regulation of osteoporosis. *Journal of Receptors and Signal Transduction*, *35*(6), 640–645.

Xiang, M. X., Su, H. W., Hu, J., & Yan, Y. J. (2011). Stimulative effects of Polygonum amplexicaule var. sinense on osteoblastic MC3T3-E1 cells. *Pharmaceutical Biology*, *49*(10), 1091–1096.

Xiao, G., Jiang, D., Gopalakrishnan, R., & Franceschi, R. T. (2002). Fibroblast Growth Factor 2 Induction of the Osteocalcin Gene Requires MAPK Activity and Phosphorylation of the Osteoblast Transcription Factor, Cbfa1/Runx2. *Journal of Biological Chemistry*, 277(39), 36181–36187.

Xiao, G., Jiang, D., Thomas, P., Benson, M. D., Guan, K., Karsenty, G., & Franceschi, R. T. (2000). MAPK pathways activate and phosphorylate the osteoblast-specific transcription factor, Cbfa1. *The Journal of Biological Chemistry*, *275*(6), 4453–4459.

Xie, H., Huff, G. R., Huff, W. E., Balog, J. M., & Rath, N. C. (2002). Effects of ovotransferrin on chicken macrophages and heterophil-granulocytes. *Developmental and Comparative Immunology*, *26*(9), 805–815.

Xie, H., Newberry, L., Clark, F. D., Huff, W. E., Huff, G. R., Balog, J. M., & Rath, N. C. (2002). Changes in Serum Ovotransferrin Levels in Chickens with Experimentally Induced Inflammation and Diseases. *Avian Diseases*, *46*(1), 122–132.

Xing, L., & Boyce, B. F. (2005). Regulation of apoptosis in osteoclasts and osteoblastic cells. *Biochemical and Biophysical Research Communications*, *328*(3), 709–720.

Yagi, M., Suzuki, N., Takayama, T., Arisue, M., Kodama, T., Yoda, Y., Ito, K. (2009). Effects of lactoferrin on the differentiation of pluripotent mesenchymal cells. *Cell Biology International*, *33*(3), 283–289.

Yamaguchi, A., Komori, T., & Suda, T. (2000). Regulation of Osteoblast Differentiation
Mediated by Bone Morphogenetic Proteins, Hedgehogs, and Cbfa1. *Endocrine Reviews*, 21(4),
393–411.

Yamaguchi, M., Tezuka, M., Yoshihiro TUKADA, S., & Kish, M. (1998). Casein Phosphopeptides in Dietary Calcium Tofu Ovariectomized Rats: Prevention of Bone Loss Enhance Calcium Availability in. *Food Sci. Technol. Int.*, *4*(3), 209–212.

Yamamura, J., Aoe, S., Toba, Y., Motouri, M., Kawakami, H., Kumegawa, M., Takada, Y. (2002). Milk Basic Protein (MBP) Increases Radial Bone Mineral Density in Healthy Adult Women. *Bioscience, Biotechnology, and Biochemistry*, *66*(3), 702–704.

Yamashita, M., Ying, S.-X., Zhang, G., Li, C., Cheng, S. Y., Deng, C., & Zhang, Y. E. (2005). Ubiquitin Ligase Smurf1 Controls Osteoblast Activity and Bone Homeostasis by Targeting MEKK2 for Degradation. *Cell*, *121*(1), 101–113.

Yang, S., Xu, H., Yu, S., Cao, H., Fan, J., Ge, C., Xiao, G. (2011). Foxo1 Mediates Insulin-like Growth Factor 1 (IGF1)/Insulin Regulation of Osteocalcin Expression by Antagonizing Runx2 in Osteoblasts. *Journal of Biological Chemistry*, *286*(21), 19149–19158.

Yang, X., Matsuda, K., Bialek, P., Jacquot, S., Masuoka, H. C., Schinke, T., Karsenty, G. (2004). ATF4 is a substrate of RSK2 and an essential regulator of osteoblast biology; implication for Coffin-Lowry Syndrome. *Cell*, *117*(3), 387–398.

Yavropoulou, M. P., & Yovos, J. G. (n.d.). The role of the Wnt signaling pathway in osteoblast commitment and differentiation. *Hormones (Athens, Greece)*, *6*(4), 279–294.

Yeh, W. C., Bierert, B. E., & Mcknightt, S. L. (1995). Rapamycin inhibits clonal expansion and adipogenic differentiation of 3T3-L1 cells. *Biochemistry*, *92*(November), 11086–11090.

Ying, X., Cheng, S., Wang, W., Lin, Z., Chen, Q., Zhang, W., Zhu Lu, C. (2012). Effect of lactoferrin on osteogenic differentiation of human adipose stem cells. *International Orthopaedics*, *36*(3), 647–653.

Yokose, S., Ishizuya, T., Ikeda, T., Nakamura, T., Tsurukami, H., Kawasaki, K., ... Yamaguchi, A. (1996). An Estrogen Deficiency Caused by Ovariectomy Increases Plasma Levels of Systemic Factors That Stimulate Proliferation and Differentiation of Osteoblasts in Rats*. *Endocrinology*, *137*(2), 469–478.

Yoshimaki, T., Sato, S., Tsunori, K., Shino, H., Iguchi, S., Arai, Y., Ogiso, B. (2013). Bone regeneration with systemic administration of lactoferrin in non-critical-sized rat calvarial bone defects. *Journal of Oral Science*, *55*(4), 343–348.

You, J., Luo, Y., & Wu, J. (2014). Conjugation of ovotransferrin with catechin shows improved antioxidant activity. *Journal of Agricultural and Food Chemistry*, 62(12), 2581–2587.

Zayzafoon, M., Gathings, W. E., & McDonald, J. M. (2004). Modeled Microgravity Inhibits Osteogenic Differentiation of Human Mesenchymal Stem Cells and Increases Adipogenesis. *Endocrinology*, *145*(5), 2421–2432.

Zeng, F., Fan, F., Xue, W., Xie, H., Wu, B., Tu, S., Chen, Y. (2013). The association of red meat, poultry, and egg consumption with risk of hip fractures in elderly Chinese: a case-control study. *Bone*, *56*(2), 242–248.

Zerega, B., Camardella, L., Cermelli, S., Sala, R., Cancedda, R., & Cancedda, F. D. (2001). Avidin expression during chick chondrocyte and myoblast development in vitro and in vivo: regulation of cell proliferation. *Journal of Cell Science*, *114*, 1473–1482.

Zhang, H., Pan, Y., Zheng, L., Choe, C., Lindgren, B., Jensen, E. D., ... Huang, H. (2011).
FOXO1 Inhibits Runx2 Transcriptional Activity and Prostate Cancer Cell Migration and Invasion. *Cancer Research*, 71(9), 3257–3267.

Zhang, Q., Badell, I., Schwarz, E. M., Boyce, B. F., & Xing, L. (2004). TNF Protects Osteoclasts from Alendronate - Induced Apoptosis by Stimulating Bcl - xL Expression through the Transcription Factor, Ets 2. *Journal of Bone and Mineral Research*, *19*(S1), S171.

Zhang, W., Guo, H., Jing, H., Li, Y., Wang, X., Zhang, H., Ren, F. (2014). Lactoferrin stimulates osteoblast differentiation through PKA and p38 pathways independent of Lactoferrin's receptor LRP1. *Journal of Bone and Mineral Research*, *29*(5), 1232–1243.

Zhang, Y., Lima, C. F., & Rodrigues, L. R. (2015). In vitro evaluation of bovine lactoferrin potential as an anticancer agent. *International Dairy Journal*, 40, 6–15.

Zhao, C., Irie, N., Takada, Y., Shimoda, K., Miyamoto, T., Nishiwaki, T., Matsuo, K. (2006). Bidirectional ephrinB2-EphB4 signaling controls bone homeostasis. *Cell Metabolism*, *4*(2), 111–121.

Zheng, X., Lee, S.-K., & Chun, O. K. (2016). Soy Isoflavones and Osteoporotic Bone Loss: A Review with an Emphasis on Modulation of Bone Remodeling. *Journal of Medicinal Food*, *19*(1), 1–14.

Zhou, M., Zhang, Y., Ardans, J. A., & Wahl, L. M. (2003). Interferon-gamma differentially regulates monocyte matrix metalloproteinase-1 and -9 through tumor necrosis factor-alpha and caspase 8. *The Journal of Biological Chemistry*, 278(46), 45406–45413.

Zhou, Z. R., & Smith, D. L. (1990). Assignment of disulfide bonds in proteins by partial acid hydrolysis and mass spectrometry. *Journal of Protein Chemistry*, *9*(5), 523–532.

Zhu, G., Luo, J., Du, H., Jiang, Y., Tu, Y., Yao, Y., & Xu, M. (2018). Ovotransferrin enhances intestinal immune response in cyclophosphamide-induced immunosuppressed mice. *International Journal of Biological Macromolecules*, *120*(Pt A), 1–9.